



DISSERTATION

Analysis of kinetic and diffusive multi-species systems

Ausgeführt zum Zwecke der Erlangung des akademischen Grades einer Doktorin der Naturwissenschaften unter der Leitung von

Univ.-Prof. Dr. rer. nat. Ansgar Jüngel

E101

Institut für Analysis und Scientific Computing

eingereicht an der Technischen Universität Wien Fakultät für Mathematik und Geoinformation

von

Dipl.-Ing. Esther Sarah Daus

Matrikelnummer: 0508401 Wiedner Hauptstrasse 8-10, A-1040 Wien Es ist nicht das Wissen, sondern das Lernen, nicht das Besitzen, sondern das Erwerben, nicht das Dasein, sondern das Hinkommen, was den größten Genuß gewährt.

It is not the knowledge, but the act of learning, not the possession of but the act of getting there, which grants the greatest enjoyment.

Carl Friedrich Gauss (1777–1855)

Kurzfassung

Ziel der vorliegenden Arbeit ist das mathematische Studium der Analysis diverser kinetischer und diffusiver Mehrkomponentensysteme, die in der Physik, der Biologie und der Chemie ihre Anwendung finden, und gewisse Kreuzeffekte zwischen den Spezies aufweisen. Die Arbeit ist in zwei Hauptteile gegliedert, und zwar in einen kinetischen Teil, der in Kapitel 4 und Kapitel 5 behandelt wird, und in einen diffusiven Teil, der in Kapitel 6 und Kapitel 7 ausgearbeitet wird.

Der erste Teil ist physikalischen Mehrkomponentensystemen kinetischer Gleichungen gewidmet, und untersucht das Langzeitverhalten von Gasgemischen, welche aus monoatomischen, nichtreaktiven Gaspartikeln mehrerer Spezies bestehen. Unser Ziel ist es, unter physikalisch möglichst plausiblen Voraussetzungen mit expliziten Methoden zu zeigen, dass diese Gemische für hinreichend reguläre Anfangsdaten mit exponentieller Abklingrate zum globalen Gleichgewichtszustand konvergieren. Mathematisch verwenden wir hierfür die mehrkomponentige Boltzmann-Gleichung auf dem dreidimensionalen Torus, welche aus einem degeneriert-dissipativen Kollisionsoperator und einem konservativen Transportoperator besteht. Im Gegensatz zur Theorie der klassischen einkomponentigen Boltzmann-Gleichung treten hier während des Kollisionsvorgangs zusätzliche Kreuzeffekten auf, die dadurch entstehen, dass im Gasgemisch auch Teilchen unterschiedlicher Spezies miteinander kollidieren können. Dieses Phänomen stellt somit die Hauptschwierigkeit unseres Problems dar. Unsere Studie führen wir in zwei Teilen durch.

In Kapitel 4 arbeiten wir mit der linearisierten Gleichung und unter der vereinfachenden Voraussetzung, dass alle Teilchen die gleiche Masse besitzen. Es gelingt uns, eine bisher nur für den einkomponentige Boltzmannschen Kollisionsoperator gezeigte degenerierte Koerzivitätsungleichung auf den Mehrkomponentenfall zu verallgemeinern. Diese Ungleichung wird in der englischsprachigen Fachliteratur neben dem Namen "coercivity estimate" auch unter dem Namen "spectral-gap estimate" angeführt, da sie nicht nur das degeneriert dissipative Verhalten des Kollisionsoperators beweist, sondern weiters auch eine für das Abklingverhalten wichtige Eigenschaft des Spektrums des Operators zeigt, nämlich dass er einen "spectral gap" besitzt. Mit Hilfe der Hypokoerzivitätsmethode von C. Mouhot und L. Neumann [113] zusammen mit der von uns verallgemeinerten Koerzivitätsungleichung zeigen wir schlussendlich exponentielle Konvergenz zum globalen Gleichgewicht für Lösungen der mehrkomponentigen linearisierten Boltzmann-Gleichung.

In Kapitel 5 verfeinern wir unsere Studie in mehrfacher Hinsicht. Einerseits schwächen wir die physikalischen Voraussetzungen an unser Modell insofern ab, dass wir nun auch Gasgemische zulassen, die aus Spezies mit unterschiedlichen Massen bestehen. Andererseits betrachten wir nicht mehr nur die um das globale Gleichgewicht herum linearisierte Gleichung, sondern erlauben kleine Störungen. Dies ist mathematisch äquivalent dazu, die volle nichtlineare mehrkomponentige Boltzmann-Gleichung nahe des globalen Gleichgewichts zu

betrachten. Der Schwerpunkt unserer Untersuchungen liegt nun darin, in aus physikalischer Sicht bestmöglich passenden Funktionenräumen eine rigorose Cauchytheorie (Existenz, Eindeutigkeit, exponenzielle Konvergenz zum Gleichgewicht) für diese Gleichung zu beweisen, indem wir neueste Techniken für das mehrkomponentige Problem verallgemeinern. Dabei zeigen wir auch eine neue Carleman-Darstellung und eine Art Povzner Ungleichung für Gasgemische mit unterschiedlichen Massen. Ausgangspunkt ist wieder die linearisierte Theorie, für die sich heraustellt, dass sich auch im Falle ungleicher Massen eine Koerzivitätsungleichung herleiten lässt. Nach Kombination mehrerer (zum Teil hypokoerziver) Methoden, wobei die Stärke und Neuartigkeit dieser Kombination darin liegt, dass sie ohne die sonst übliche Verwendung von Sobolevräumen höherer Ordnung auskommt, erhalten wir schließlich Existenz, Eindeutigkeit, und exponentielle Konvergenz zum globalen Gleichgewicht der Lösung im Raum $L_v^1 L_x^{\infty}(m)$, wobei $m \sim (1+|v|^k)$ ein polynomielles Gewicht der Ordnung $k > k_0$, und k_0 ein physikalischer Schwellenwert ist. Im Spezialfall von Spezies mit gleichen Massen und unter der "hard-spheres" Voraussetzung liefert dies den Schwellenwert $k_0 = 2$, und somit (zumindest fast) den physikalisch "optimalen" Raum aller Dichtefunktionen mit endlicher Masse und Energie.

Der zweite Teil dieser Arbeit ist diffusiven Mehrkomponentensystemen gewidmet, die in der Populationsdynamik der Biologie (Kapitel 6) und in der Reaktionskinetik der Chemie (Kapitel 7) eine wichtige Rolle spielen.

In Kapitel 6 untersuchen wir das nach Shigesada, Kawasaki und Teramoto [129] benannte SKT-Modell der Populationsdynamik, welches ein Kreuzdiffusionsmodell ist, und das Phänomen der räumlichen Abkapselung von miteinander sich im Konkurrenzkampf (z.B. um eine Futterquelle) befindenden Spezies beschreibt. Unser Ziel ist es, globale Existenz schwacher Lösungen des SKT Models zu beweisen. Neu an unseren Ergebnissen ist, dass wir (unter algebraischen Bedingungen an die Diffusionskoeffizienten) Existenz von Lösungen für eine beliebige Anzahl an Spezies zeigen können. Bisherige Ergebnisse beinhalteten (bis auf Spezialfälle in [51]) nur den Fall zweier Populationen. Dabei verwenden und erweitern wir eine von A. Jüngel [94] entwickelte Entropiemethode, die uns die für den Existenzbeweis wichtigen Gradientenabschätzungen liefert. Die Bedingungen an die Diffusionskoeffizienten, die wir erhalten, stellen sicher, dass die Entropie monoton fallend ist, und lassen sich salopp folgendermaßen charakterisieren: entweder wir benötigen eine aus der Theorie von Markovketten her bekannte Bedingung namens "detailed balance". oder die Kreuzdiffusionseffekte müssen schwach im Vergleich zu den Selbstdiffusionseffekten sein. Mehrere Gegenbeispiele illustrieren, dass die Entropie (anfangs) monoton steigt statt zu fallen, falls keine dieser beiden Bedingungen erfüllt ist. Schlussendlich leiten wir eine graphentheoretische Bedingung her, unter der "detailed balance" erfüllt ist.

In Kapitel 7 beweisen wir den rigorosen Limes von verallgemeinerten Reaktions- Diffusionssystemen aus der Reaktionskinetik zu neuartigen Kreuzdiffusionssystemen mit Reaktionstermen, indem wir annehmen, dass einige der chemischen Reaktionen im Vergleich zu allen anderen extrem schnell ablaufen. Dieser Limes besitzt nun die interessante Eigenschaft, dass er Entropie erhaltend ist, und somit Entropien zu neuartigen Klassen von Kreuzdiffusionssystemen mit Reaktionstermen generieren kann. Für den rigorosen Konvergenzbeweis verwenden wir neben Entropieabschätzungen auch die von M. Pierre und D. Schmitt entwickelte Dualitätsmethode [121].

Abstract

The objective of this thesis is the analysis of kinetic and diffusive multi-species systems with certain cross effects between the species, which are very important in many applications in physics, biology and chemistry.

In the first part of this thesis, we study the multi-species Boltzmann equation for hard potentials or Maxwellian molecules under Grad's angular cut-off condition on the torus, which describes the evolution of a dilute gaseous mixture. First, we work on the linearized level with same molar masses, where we prove a multi-species spectral-gap estimate of the collision operator, which leads to exponential trend to global equilibrium using the hypocoercive properties of the linearized Boltzmann equation. Next, we study the full Cauchy theory of the nonlinear multi-species Boltzmann equation close to global equilibrium for different molar masses in $L_v^1 L_x^{\infty}(m)$, where $m \sim (1+|v|^k)$ is a polynomial weight of order $k > k_0$, recovering the optimal physical threshold of finite energy $k_0 = 2$ in the particular case of a multi-species hard spheres mixture with same molar masses.

The second part of this thesis is devotet to cross-diffusion systems. First, we prove global existence of weak solutions for a generalized SKT cross-diffusion population dynamics model with an arbitrary number of species under detailed balance or weak cross-diffusion condition using entropy methods. Finally, we study a rigorous fast-reaction limit from reaction-diffusion systems to cross-diffusion systems using entropy estimates and additional duality estimates. Since the reaction-diffusion system exhibits an entropy structure, performing the fast-reaction limit leads to a limiting entropy of the limiting cross-diffusion system. In this way, we obtain new entropies for new classes of cross-diffusion systems.

Contents

I.	Int	roduct	ion	1	
1.	General introduction				
	1.1.	Overv	iew	3	
	1.2.	The m	ain results	3	
	1.3.	Presen	tation of the mathematical problems	4	
	1.4.	Outlin	e of the thesis	10	
2.	Intro	oductio	n to the kinetic multi-species models	11	
	2.1.	The li	nearized Boltzmann system with same molar masses	11	
		2.1.1.	The model for same molar masses	11	
		2.1.2.	Linearized Boltzmann collision operator	13	
		2.1.3.	State of the art	14	
		2.1.4.	Assumptions on the collision kernels	14	
		2.1.5.	Main results for same molar masses	15	
		2.1.6.	Comments and outlook	19	
	2.2.	The B	oltzmann system with different molar masses close to equilibrium	19	
		2.2.1.	The model for different molar masses	20	
		2.2.2.	State of the art for the Cauchy problem	22	
		2.2.3.	Assumptions on the collision kernels	22	
		2.2.4.	Main results for different molar masses close to equilibrium	23	
		2.2.5.	Comments and outlook	27	
3.	Intro	oductio	n to the diffusive multi-species models	29	
	3.1.	Cross-	diffusion population systems for multiple species	29	
		3.1.1.	The model	29	
		3.1.2.	State of the art	30	
		3.1.3.	Key ideas	31	
		3.1.4.	Main results	33	
	3.2.	From	reaction diffusion to cross diffusion in the fast-reaction limit	35	
		3.2.1.	The strategy of our work	35	
		3.2.2.	Main results	37	
		3.2.3.	State of the art	40	

II.	Th	e kine	tic models	43			
4.			zed multi-species Boltzmann system	45			
	4.1.		odel				
		4.1.1.	1				
			Discussion of the assumptions				
			Notation and definitions				
	4.0		Main results				
		_	rties of the kinetic model				
			etric properties of the spectrum				
	4.4.		it spectral-gap estimate				
			Decomposition				
			Spectral-gap estimate for the mono-species part				
		4.4.3.	Absorption of the orthogonal parts				
			Estimate for the remaining part				
			Estimate for the momentum and energy differences				
	4.5.	Conve	rgence to equilibrium	. 67			
5.	The	The nonlinear multi-species Boltzmann system close to global equilibrium 7					
	5.1.		odel				
		5.1.1.	Different molar masses				
		5.1.2.	The perturbative regime and its motivation	. 72			
		5.1.3.	Notations and assumptions on the collision kernel	. 74			
			Main results				
			Organisation of this chapter				
	5.2.	Spectr	al gap with different masses	. 77			
		5.2.1.	First properties of the linear multi-species Boltzmann operator $$. $$. 77			
		5.2.2.	Explicit spectral gap	. 80			
	5.3.	L^2 -the	eory for the linear part with Maxwellian weight	. 88			
		5.3.1.	A priori control of the fluid part by the microscopic part	. 89			
		5.3.2.	Generation of a semigroup	. 103			
	5.4.	L^{∞} -th	eory for the linear part with Maxwellian weight	. 104			
		5.4.1.	Pointwise estimate on the compact part	. 104			
		5.4.2.	Semigroup generated by the linear part	. 111			
	5.5.	The fu	all nonlinear equation in a perturbative regime	. 119			
		5.5.1.	Existence of a solution that decays exponentially	. 119			
		5.5.2.	Uniqueness of solutions in the perturbative regime	. 134			
		5.5.3.	Positivity of solutions	. 139			
111	Th	e diffu	sive models	141			
•••		c umu	Sive models	171			
6.			sion population systems for multiple species	143			
	6.1.		odel	. 143			
		6 I T	Main regulte	1/15			

6.2.	Positiv	ve definiteness of the mobility matrix	. 146
	6.2.1.	Sublinear and linear transition rates	. 148
	6.2.2.	Superlinear transition rates	. 150
	6.2.3.	Approximate matrices	. 152
6.3.	Linear	transition rates: proof of Theorem 6.1	
6.4.		near transition rates: proof of Theorem 6.2	
6.5.		onal and auxiliary results	
		Detailed balance condition	
	6.5.2.	Nonlinear Aubin-Lions lemmas	
	6.5.3.	Increasing entropies	
7. Fro	m react	ion diffusion to cross diffusion in the fast-reaction limit	169
7.1.	Motiva	ation	. 169
	7.1.1.	Outline	. 169
	7.1.2.	Notation	. 169
7.2.	Rigoro	ous limit from three species with one fast reaction	. 169
	7.2.1.	Model for three species with one fast reaction	. 169
	7.2.2.	Existence of solutions	. 170
	7.2.3.	Fast-reaction limit to a cross-diffusion system	. 174
	7.2.4.	Transformed system	. 184
	7.2.5.	Robustness of the method	. 185
7.3.	Rigoro	ous limit from four species with one fast and one slow reaction	. 188
	7.3.1.	Model for four species with one fast and one slow reaction	. 188
	7.3.2.	Existence of solutions	. 189
	7.3.3.	Fast-reaction limit to a cross-diffusion system	. 189
	7.3.4.	A priori estimates	. 189
	7.3.5.	Strong compactness	. 194
	7.3.6.	Passing to the limit	. 196
	7.3.7.	Global strong solutions to the reaction-diffusion system	. 198
	7.3.8.	Transformed system	. 200
7.4.	Auxili	ary result: regularizing effect of the heat kernel	. 201
Bibliog	raphy		203
Acknov	wledgen	nents	213
	_		
Currici	ılum vit	ae	215

Part I. Introduction

1. General introduction

This thesis is devoted to the analysis of kinetic and diffusive multi-species systems in physics, biology and chemistry, where the interactions between the different species play an important role. On both the kinetic and the macroscopic level of description, we will study how these interactions create certain cross effects between the species, which are able to describe cross-diffusion phenomena.

1.1. Overview

Kinetic theory [32, 33, 34, 138] is a part of non-equilibrium statistical physics, and started in the 19th century with the founding works of J. Clerk Maxwell [107, 108] and L. Boltzmann [59] on the evolution of a dilute gas, described by the Boltzmann equation. The basic idea is to use a statistical description to model the evolution of a system consisting of many particles, like a gas, plasma or galaxy, by assuming that it can be regarded as a continuum, instead of describing the trajectory of each particle individually by Newton's laws of classical mechanics. However, the Boltzmann equation is not a "first principle" of physics, but represents a mesoscopic level, since it can be formally derived from the microscopic level of molecular dynamics of a gas (see [33][34]), and it was rigorously derived from molecular gas dynamics in the Boltzmann-Grad limit (see for short times Landford's theorem [102], or more recently [70, 125]).

Cross diffusion in multi-species systems represents the effect when the gradient of the concentration of one of the species induces a flux on the other species. Cross-diffusion equations are *macroscopic* models, and they can be derived from kinetic equations in the diffusive limit [21, 91, 18]. Cross diffusion was first conjectured by Onsager and Fuoss [117] in 1932, and experimentally justified by Gosting and Dunlop [65] in 1955. One of the most prominent experiments was performed by Duncan and Toor [64], showing that cross effects between the species can completely change the diffusive fluxes. Nowadays, the field of cross-diffusion in multi-species systems is a very active research area, see [95] for a mathematical viewpoint, and [136] and the references therein for many experimental results on cross-diffusion phenomena.

1.2. The main results

The results presented in this thesis can be summarized as follows. On the kinetic level, assuming hard potentials or Maxwellian molecules under Grad's cutoff assumption, we prove an explicit spectral-gap estimate for the multi-species linearized Boltzmann collision operator for same molar masses, and show exponential convergence to global equilibrium

for the inhomogeneous linearized equation using hypocoercive techniques. Afterwards, we study the Cauchy theory for the perturbed multi-species Boltzmann equation with different molar masses in physically relevant function spaces, without using higher order Sobolev regularity. The asymmetry due to the different molar masses leads to new multi-species results, including a spectral-gap estimate for different molar masses, a multi-species Carleman representation, and a new multi-species Povzner-type inequality.

On the macroscopic level, we investigate two different issues. First, we study the existence of global weak solutions of a generalized Shigesada-Kawasaki-Teramoto type cross-diffusion system for an arbitrary number of species under a detailed balance of weak cross-diffusion condition, by using entropy methods. This has been proven (up to some very specific cases) only for two species so far. Finally, by starting already from a macroscopic equation of reaction-diffusion type, we perform a rigorous fast-reaction limit in order to obtain more complicated cross-diffusion systems with known entropy structure in the limit. This is done with the help of entropy estimates and additional duality estimates.

1.3. Presentation of the mathematical problems

We start our observations by considering a distribution function $F_i(t, x, v)$ for $1 \le i \le n$, which describes the probability density of the *i*-th species of a gaseous mixture of $n \ge 2$ different species with same molar masses on the phase space $(x, v) \in \mathbb{T}^3 \times \mathbb{R}^3$. The quantity $F_i(t, x, v) \, dx dv$ measures the number of particles of the species *i* in the mixture at time *t* in an elementary volume of the phase space of the size dx dv centered at the point (x, v). The particles are assumed to be mono-atomic and non-reactive, moreover, we restrict our observations to classical mechanics, ignoring any kind of relativistic or quantum effects. Then, the evolution of F_i can be described by the multi-species Boltzmann equation

$$\partial_t F_i + v \cdot \nabla_x F_i = Q_i(F), \quad 1 \le i \le n, \quad t > 0, \tag{1.3.1}$$

with initial conditions and periodic boundary conditions. The term $v \cdot \nabla_x F_i$ models the transport of the particles, whereas $Q(F) = (Q_1(F), \dots, Q_n(F))$ is the quadratic multispecies Boltzmann collision operator, local in time t and in position x, with

$$Q_i(F)(v) = \sum_{j=1}^{n} Q_{ij}(F_i, F_j)(v), \quad 1 \le i \le n.$$

It models the collisions between particles of the same (i = j) or of different $(i \neq j)$ species, where

$$Q_{ij}(F_i, F_j)(v) = \sum_{j=1}^n \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij}(|v - v_*|, \cos(\theta))(F_i(v')F_j(v'^*) - F_i(v)F_j(v^*))dv^*d\sigma.$$

The most important property of the Boltzmann equation is the H-theorem, first proved by L. Boltzmann for a single-species gas, and later for reactive multi-species kinetic models in [122, 23, 52]. It states that the entropy is nonincreasing in time along the flow of (1.3.1)

towards its global equilibrium called global Maxwellian $M(v) = (M_1(v), \dots, M_n(v))$, which is a Gaussian function. This coincides with the Second Law of Thermodynamics, which says that the physical entropy (which is the negative mathematical entropy) of an isolated system is nondecreasing in time. Besides the existence of a unique global equilibrium, there exist also local equilibria, which are in equilibrium with respect to the velocity v, but not with respect to the position x. It turns out that the entropy stops decaying whenever it reaches a local equilibrium, and only the interplay between the conservative transport part and the dissipative collision part of the kinetic equation leads to convergence to global equilibrium. This property of the Boltzmann equation and also of several other kinetic models is called hypocoercivity [139]. It was first investigated for the single-species kinetic Fokker-Planck equation in [54, 86], and then for the confined nonlinear single-species Boltzmann equation assuming regularity in [55]. Hypocoercivity of a general class of linear single-species kinetic models was investigated in [113] and in [58]. In both papers, a linear energy method is established, which combines the coercivity of the collision operator in the velocity space (spectral-gap estimate) together with transport effects in order to overcome the lack of coercivity on the set of local equilibria. The abstract strategy presented in [113] includes the linearized Boltzmann equation for hard spheres on the torus, yielding exponential convergence in H^1 . The work [58] studies hypocoercivity for a large class of single-species linear kinetic models on the whole space with confinement potential in L^2 , but for only one-dimensional collision kernels. Recently, an abstract space-extension technique was established in [78], which is capable of enlarging the functional space of the explicit convergence result of a semigroup, by using a high-order quantitative factorization argument on the resolvent of the semigroup. In this way, a first constructive proof of exponential convergence for the full nonlinear Boltzmann equation for hard spheres was proved in $L_v^1 L_x^{\infty} (1+|v|^k)$ for k>2.

In Chapter 4, we show that also the multi-species linearized Boltzmann equation

$$\partial_t f_i + v \cdot \nabla_x f_i = L_i(f), \quad 1 \le i \le n, \quad t > 0,$$

$$L_i(f)(v) = \sum_{j=1}^{n} L_{ij}(f_i, f_j)(v),$$

with

$$L_{ij}(f_i, f_j) = M_i^{-1/2} (Q_{ij}(M_i, M_j^{1/2} f_j) + Q_{ij}(M_i^{1/2} f_i, M_j))$$

exhibits a hypocoercive property by generalizing the abstract hypocoercivity procedure of C. Mouhot and L. Neumann [113] to the multi-species case. The core idea in [113] is to construct a modified entropy by adding an additional mixed term of the form (2.1.15) to the H^1 -norm with an appropriate weight, whose dissipation functional is coercive. This yields exponential convergence to global equilibrium with an explicitly computable exponential convergence rate in $H^1(\mathbb{T}^3 \times \mathbb{R}^3)$. The crucial part for our multi-species problem will be to show that an explicit spectral-gap estimate like in [111] holds true also in the multi-species case, by carefully exploiting the multi-species conservation laws. These conservation laws

turn out to be significantly different from the single-species ones due to certain cross effects between the species, namely, that the mass of each species is conserved individually, but only the sum of the momenta and the sum of the energies is conserved, which can be derived from the multi-species H-theorem [52]. Thus, the idea is to study the differences of the momenta and the differences of the energies, in order to handle the cross effects arising due to the collisions between different species, see section 2.1.5, and in particular the estimate on the differences of the energies and differences of the momenta presented in (2.1.11) for an explicitly computable constant C > 0:

$$-(\Pi^m(f), L^b(\Pi^m(f)))_{L_v^2} \ge C \sum_{i,j=1}^n (|u_i - u_j|^2 + |e_i - e_j|^2),$$

where we decomposed the linearized operator $L = L^m + L^b$ into a mono-species part L^m and a bi-species part L^b , and thus $\Pi^m(f)$ denotes the orthogonal projection of f onto the kernel of L^m in L^2 , and u_i are the momenta and e_i are the energies of f, defined in (4.4.4). In this way, we obtain the explicit multi-species spectral-gap estimate

$$-(f, L(f))_{L_v^2} \ge \lambda \|f - \Pi^L(f)\|_{L_v^2(\nu)}^2,$$

where $\lambda > 0$ is an explicitly computable constant, $L^2(\nu)$ is a weighted (vector-valued) L^2 space weighted by the collision frequency ν , and $H^L(f)$ denotes the orthogonal projection
of $f = (f_1, \ldots, f_n)$ onto the null space of the linearized multi-species collision operator $L = (L_1, \ldots, L_n)$ with respect to the L^2 scalar product. This part of the thesis is based on
a research collaboration with A. Jüngel (TU Wien), C. Mouhot (University of Cambridge)
and Nicola Zamponi (TU Wien), and was published under the title Hypocoercivity for a
linearized multispecies Boltzmann system in the journal SIAM Journal on Mathematical
Analysis [46].

In Chapter 5, we extend the previous results, by establishing a full Cauchy theory in $L_v^1 L_x^{\infty}(m)$ for the nonlinear multi-species Boltzmann equation with different molar masses on the torus, where $m \sim (1+|v|^k)$ is a polynomial weight of order $k \geq k_0$ with an explicit threshold k_0 , under the assumption that f_i is close to global equilibrium, which is equivalent of solving the perturbed multi-species Boltzmann equation. In the particular case of a multi-species hard spheres mixture with same molar masses, this result recovers the optimal physical threshold of finite energy $k_0 = 2$, which was obtained for the single-species equation in [78]. The strategy of our proof is to combine and adapt several very recent methods, combined with new hypocoercivity estimates, in order to develop a new constructive approach that allows to deal with polynomial weights without requiring any spatial Sobolev regularity. Moreover, dealing with different masses induces a loss of symmetry in the Boltzmann operator, which prevents the direct adaptation of standard mono-species methods. Thus, our results include a refinement of the multi-species spectral-gap estimate [46] (see Theorem 5.6), a multi-species Carleman representation (section 5.4.1), which takes into account the concentric spheres arising due to the different masses already pointed out by Pettersson in [119, p. 364], and a new multi-species Povzner-type inequality (see Proposition 5.22). The different molar masses do not disturb too much the procedure for proving the multi-species spectral-gap estimate which we developed in [46] and presented in Chapter 4. The Carleman representation for different molar masses has the interesting feature of producing admissible Carleman sets of the form of a hyperplane that passes through $V_E(v,v')$ defined in (5.4.7) and is orthogonal to v-v' (similar to the single-species case), or of a sphere defined in (5.4.9), depending on which change of variables we perform. For the Povzner-type inequality, we used a method similar to [9, Lemma 1 and Corollary 3], and our main idea is to consider kinetic energies $m_i |v'|^2$ and $m_j |v'_*|^2$ in order to exhibit the problematic term arising from $m_i - m_j$, which can be non-zero in the case of different masses.

Before, global existence of L^1 -renormalized solutions was proved in [123] for reactive multi-species Boltzmann systems with initial data only satisfying the natural condition that total mass, energy and entropy are finite. In the recent work [27], additional L^{∞} stability estimates for the multi-species Boltzmann equation are derived. Other methods for proving the compactness of the compact part of the multi-species linearized collision operator can be found in [19, 22], and for previous results on a multi-species Povzner-type inequality for a coupling of two Boltzmann-like equations modeling the evolution of dust particles in a rarefied atmosphere, we refer to [36]. The work presented in this thesis is based on a research collaboration with M. Briant (Paris 6), which will appear under the title The Boltzmann equation for a multi-species mixture close to global equilibrium in the journal Archive for Rational Mechanics and Analysis [28].

Starting from a diffusive scaling of the multi-species Boltzmann equation (1.3.1) under the assumption of volume-filling and with $\varepsilon > 0$ representing the mean-free path

$$\varepsilon \, \partial_t f_i^{\varepsilon} + v \cdot \nabla_x f_i^{\varepsilon} = \frac{1}{\varepsilon} \, Q_i^{\varepsilon}(f^{\varepsilon}), \quad 1 \le i \le n,$$

a class of cross-diffusion systems called Maxwell-Stefan equations can be derived in the diffusive limit $\varepsilon \to 0$, see [21, 91, 18]. Existence of solutions for this system was investigated e.g. in [72, 12, 20, 96], including a global existence result of weak solutions under quite general assumptions in [96] by using entropy methods. A general strategy, which generalizes the approach in [96] of proving global existence of weak solutions for strongly coupled cross-diffusion systems of the form

$$\partial_t u - \operatorname{div}(A(u)\nabla_x u) = f(u), \quad u = (u_1, \dots, u_n), \tag{1.3.2}$$

where the diffusion matrix A(u) is neither symmetric nor positive definite and f(u) represents the reaction part, was established in [94] (see also [95]) by the Boundedness-by-entropy method. The core assumption for this method is the existence of an entropy, which is a Lyapunov functional along the flow of the cross-diffusion system, such that this system can be rewritten in (formal) gradient-flow form. In particular, we assume that equation (1.3.2) can be written as

$$\partial_t u - \operatorname{div}\left(B\nabla \frac{\delta \mathcal{H}}{\delta u}\right) = f(u),$$
 (1.3.3)

where B is a positive semi-definite matrix and $\frac{\delta \mathcal{H}}{\delta u}$ is the variational derivative of the entropy functional $\mathcal{H}[u] = \int_{\Omega} h(u) dx$ with an entropy density $h: (0, \infty)^n \to [0, \infty)$. If we identify

 $\frac{\delta \mathcal{H}}{\delta u}$ by its Riesz representative h'(u) (the derivative of h) and introduce the entropy variable w = h'(u), we obtain the equation

$$\partial_t u - \operatorname{div} (B(w)\nabla w) = f(u). \tag{1.3.4}$$

The simple calculation $\nabla w = \nabla h'(u) = h''(u)\nabla u$ yields B(w)h''(u) = A(u), which leads to

$$B(w) = A(u) (h''(u))^{-1}. (1.3.5)$$

Thus we can also study equation (1.3.4) and by supposing that $h':(0,\infty)^n\to\mathbb{R}^n$ is invertible, transform back from the w- variable to the u- variable.

The important consequence of the gradient-flow formulation is that by just assuming the matrix B to be positive-semidefinite, we can show (omitting the reaction term) that $\mathcal{H}[u(t)]$ is a Lyapunov functional along the solutions u(t) of the reaction diffusion system (1.3.2), *i.e.*

$$\frac{d}{dt}\mathcal{H}\left[u(t)\right] = \frac{d}{dt}\int_{\Omega}h(u)dx = -\int_{\Omega}\nabla w:B(w)\nabla wdx = -\int_{\Omega}\nabla u:h''(u)A(u)\nabla udx \leq 0.$$

The resulting entropy-dissipation inequality provides the necessary a priori estimates for the gradients in order to apply a nonlinear Aubin-Lions lemma.

In **Chapter 6**, we extend the Boundedness-by-entropy method [94, 95] to a population dynamics cross-diffusion system, which was first investigated by Shigesada, Kawasaki and Teramoto in 1979 in order to study the segregation between two competing species due to repulsive effects [129]. First global existence results for this so called SKT model in the case of two species by using entropy methods were shown in [37, 38]. Later, these techniques were refined for generalized SKT models by combining entropy estimates with duality estimates [50, 51]. In [94], global existence of weak solutions for a generalized SKT model was shown in the case of only two species. We extend this result to an arbitrary number of species. The model has the form (1.3.2), with diffusion matrix

$$A_{ij}(u) = \delta_{ij}p_i(u) + u_i \frac{\partial p_i}{\partial u_j}(u), \quad p_i(u) = a_{i0} + \sum_{k=1}^n a_{ik}u_k^s, \quad a_{ij} \ge 0, \quad s > 0,$$

and Lotka-Volterra competition term

$$f_i(u) = u_i \left(b_{i0} - \sum_{j=1}^n b_{ij} u_j^{\sigma} \right), \quad 0 \le \sigma < 2s - 1 + 2/d,$$

on a d-dimensional open bounded domain $\Omega \subseteq \mathbb{R}^d$. For s=1, n=2 and $\sigma=1$, this yields the classical SKT model of Shigesada, Kawasaki and Teramoto [129]. The existence proof is carried out under a detailed balance or weak cross-diffusion condition. The detailed balance condition is related to the symmetry of the mobility matrix, mirroring Onsager's principle in thermodynamics. Under detailed balance (and without reaction term), the entropy is non-increasing in time, but counter-examples show that the entropy may increase initially

if the detailed balance condition does not hold. The crucial gradient estimate in the linear case s = 1 is derived from the following entropy-dissipation inequality:

$$\frac{d}{dt}\mathcal{H}[u] + 4\int_{\Omega} \sum_{i=1}^{n} \pi_i a_{i0} |\nabla \sqrt{u_i}|^2 dx + 2\int_{\Omega} \sum_{i=1}^{n} \pi_i a_{ii} |\nabla u_i|^2 dx \le 0,$$

where $\pi_i > 0$ are positive constants inside the entropy functional

$$\mathcal{H}[u] = \sum_{i=1}^{n} \int_{\Omega} \pi_i(u_i \log u_i - u_i + 1) dx,$$

which satisfy the detailed balance condition

$$\pi_i a_{ij} = \pi_j a_{ji}, \quad 1 \le i, j \le n.$$

These results are based on a research collaboration with X. Chen (Beijing University of Posts and Telecommunications) and A. Jüngel (TU Wien), which is submitted under the title Global existence analysis of cross-diffusion population systems for multiple species [39].

In **Chapter 7**, we study the rigorous derivation of a class of (reaction-) cross diffusion systems in the fast-reaction limit, by already starting from a macroscopic equation of reaction-diffusion type. We consider two concrete examples and use a strategy similar to [16]. In the first model, we study the fast-reaction limit $\varepsilon \to 0$ of a system of three reaction-diffusion equations of the form

$$\begin{cases}
\partial_t u_1^{\varepsilon} - \Delta_x f_1(u_1^{\varepsilon}) = -\frac{1}{\varepsilon} \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right), \\
\partial_t u_2^{\varepsilon} - \Delta_x f_2(u_2^{\varepsilon}) = +\frac{1}{\varepsilon} \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right), \\
\partial_t u_3^{\varepsilon} - \Delta_x f_3(u_3^{\varepsilon}) = +\frac{1}{\varepsilon} \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right),
\end{cases} (1.3.6)$$

for $\varepsilon>0$ on a bounded domain $\Omega\subseteq\mathbb{R}^d$ together with no-flux boundary conditions and nonnegative initial conditions. The functions $u_i^\varepsilon:=u_i^\varepsilon(t,x)\geq 0$ for i=1,2,3 represent the concentrations of the i-th reacting species, and $f_i(u_i^\varepsilon)$ and $q_i(u_i^\varepsilon)$ are general smooth functions with some restrictions on their growth. The advantage of this rather general system is, that its structure is still simple enough that an entropy naturally appears as a generalization of the physical entropy of systems arising from reversible reaction chemistry. Then, by using entropy estimates and duality estimates, we show that this system converges rigorously in the fast-reaction limit $\varepsilon\to 0$ towards

$$q_1(u_1) - q_2(u_2)q_3(u_3) = 0,$$
 (1.3.7)

$$\partial_t (u_1 + u_2) - \Delta_x (f_1(u_1) + f_2(u_2)) = 0, \tag{1.3.8}$$

$$\partial_t (u_1 + u_3) - \Delta_x (f_1(u_1) + f_3(u_3)) = 0.$$
 (1.3.9)

By using the algebraic condition (1.3.7) and performing the change of variables $w_2 := u_1 + u_2$ and $w_3 := u_1 + u_3$ in (1.3.8) and (1.3.9), this limiting system turns out to be a system of

two cross-diffusion equations for which an entropy exists: it is the limiting entropy of the well-known physical entropy of the system (1.3.6) for $\varepsilon \to 0$. In this way, we obtain a new large class of cross-diffusion systems with known entropy structure. Our second model is one example of a more complex situation, in which a system of four equations of reaction-diffusion type converges in the fast-reaction limit towards a "hybrid" system of reaction-cross diffusion consisting of three equations, still presenting an entropy structure. In both rigorous convergence proofs, entropy estimates as well as duality estimates play an important role. Entropy estimates are obtained from the entropy-dissipation inequality of the Lyapunov functional of these systems:

$$\sup_{t \in [0,T]} H[u^{\varepsilon}(t)] + \int_0^T D_{\varepsilon}[u^{\varepsilon}(s)] ds \leq H[u^{in}],$$

where $H[u^{\varepsilon}(t)]$ is the entropy (Lyapunov function) along the flow of the system, and $D_{\varepsilon}[u^{\varepsilon}(t)] = -\frac{d}{dt}H[u^{\varepsilon}(t)]$ is the entropy dissipation. The use of duality estimates for reaction-diffusion systems goes back to M. Pierre and D. Schmitt [121] and has been refined in many variants since then, see e.g [49, 29]. Recently, they have been introduced also for cross-diffusion systems, see for instance [50, 51]. We present two of the variants explicitly in Lemma 7.5 and Lemma 7.21, which can be found in [49] and [29] respectively. The core idea of these duality estimates is to gain regularity for a system of the type

$$\partial_t u - \Delta_x(Mu) = 0$$
, $M = M(t, x) \ge 0$, $u = u(t, x) \ge 0$, $u = (u_1, \dots, u_n)$,

with no-flux boundary conditions and regular initial conditions, roughly speaking, by multiplying the system by a positive solution of its adjoint problem backward in time, *i.e.* its dual problem. This part of the thesis is based on a research collaboration with L. Desvillettes (Paris Diderot) and A. Jüngel (TU Wien) under the title *Cross-diffusion systems and fast-reaction limit*, and will be submitted soon.

1.4. Outline of the thesis

In **Chapter 1**, we introduce the mathematical problems investigated in this thesis, discuss how they fit into the context of the existing literature, and briefly sketch some ideas of the proofs. A more detailed introduction into the models presented in this thesis can be found in **Chapter 2** and **Chapter 3**, which are based on my own works [46, 28] and [39] respectively. We refer to these two chapters for more details on the state of the art and the methods used for the proofs. In **Chapter 4**, **5**, **6**, **7** we present the detailed proofs. The proofs of the first three chapters (Chapter 4, 5, 6) can be found in [46, 28, 39].

2. Introduction to the kinetic multi-species models

2.1. The linearized Boltzmann system with same molar masses

In Chaper 4, our goal is to prove an explicit spectral-gap estimate of the linearized Boltzmann operator for gas mixtures in the case of hard and Maxwellian potentials as well as the exponential decay of solutions to a multi-species Boltzmann system. Spectral-gap estimates and the large-time behavior of the mono-species Boltzmann equations were intensively studied in the literature, but are unknown for multi-species systems. Our aim is to extend the spectral-gap analysis to the case of the linearized multi-species Boltzmann system modeling an ideal gas mixture. This is achieved by generalizing the coercivity method of [113], including quantitative estimates on the spectral gap for the multi-species collision operator. A crucial step of our analysis is the observation that the multi-species version of the H-theorem implies conservation of mass for each species but conservation of momentum and energy only for the sum of all species. As a consequence, we need to study carefully the "cross-effects" of the collisions, i.e., how collisions between different species act on distribution functions which are elements of the nullspace of the mono-species collision operator. The crucial step is to relate these "cross-effects" to the differences of momentum and energy.

2.1.1. The model for same molar masses

The evolution of a dilute ideal gas composed of $n \geq 2$ different species of chemically non-interacting mono-atomic particles (see [52] for chemically reacting gases) with the same particle mass (see Chapter 5 for different molar masses) can be modeled by the following system of Boltzmann equations, stated on the three-dimensional torus \mathbb{T}^3 ,

$$\partial_t F_i + v \cdot \nabla_x F_i = Q_i(F), \quad t > 0, \quad F_i(x, v, 0) = F_{I,i}(x, v), \quad (x, v) \in \mathbb{T}^3 \times \mathbb{R}^3, \quad (2.1.1)$$

where $1 \leq i \leq n$. The vector $F = (F_1, \ldots, F_n)$ is the distribution function of the system, with F_i describing the *i*th species. The variables are the position $x \in \mathbb{T}^3$, the velocity $v \in \mathbb{R}^3$, and the time $t \geq 0$. The right-hand side of the kinetic equation in (2.1.1) is the *i*th component of the nonlinear collision operator, defined by

$$Q_i(F) = \sum_{j=1}^{n} Q_{ij}(F_i, F_j), \quad 1 \le i \le n,$$

where Q_{ij} models interactions between particles of the same (i = j) or of different species $(i \neq j)$,

$$Q_{ij}(F_i, F_j)(v) = \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij}(|v - v^*|, \cos \vartheta) (F_i' F_j'^* - F_i F_j^*) dv^* d\sigma,$$

with the abbreviations $F'_i = F_i(v')$, $F_i^* = F_i(v^*)$, $F_i'^* = F_i(v'^*)$, the three-dimensional unit sphere \mathbb{S}^2 , and

$$v' = \frac{v + v^*}{2} + \frac{|v - v^*|}{2}\sigma, \quad v'^* = \frac{v + v^*}{2} - \frac{|v - v^*|}{2}\sigma \tag{2.1.2}$$

are the pre-collisional velocities depending on the post-collisional velocities (v, v^*) . These expressions follow from the fact that we assume the collisions to be elastic, i.e., the momentum and kinetic energy are conserved on the microscopic level:

$$v' + v'^* = v + v^*, \quad \frac{1}{2}|v'|^2 + \frac{1}{2}|v'^*|^2 = \frac{1}{2}|v|^2 + \frac{1}{2}|v^*|^2.$$
 (2.1.3)

The collision kernels B_{ij} are nonnegative functions of the modulus $|v-v^*|$ and the cosine of the deviation angle $\theta \in [0, \pi]$, defined by $\cos \theta = \sigma \cdot (v-v^*)/|v-v^*|$.

Let us first recall the main properties of the nonlinear operator Q_i . Using the techniques from [34, pp. 36-42], it is not difficult to see that $Q := (Q_1, \ldots, Q_n)$ conserves the mass of each species but only total momentum and energy, i.e.

$$\int_{\mathbb{R}^3} \sum_{i,j=1}^n Q_{ij}(F_i, F_j) \psi_i(v) dv = 0$$

if and only if $\psi(v) \in \text{span}\{e^{(1)}, \dots, e^{(n)}, v_1 \mathbf{1}, v_2 \mathbf{1}, v_3 \mathbf{1}, |v|^2 \mathbf{1}\}$, where $e^{(i)}$ is the *i*th unit vector in \mathbb{R}^n and $\mathbf{1} = (1, \dots, 1) \in \mathbb{R}^n$. It is shown in [52] that Q satisfies a multi-species version of the H-theorem which implies that any local equilibrium, i.e. any function F being the maximum of the Boltzmann entropy, has the form of a local Maxwellian $M_{\text{loc}} = (M_{\text{loc},1}, \dots, M_{\text{loc},n})$ with

$$F_i(x,v,t) = M_{\text{loc},i}(x,v,t) = \frac{\rho_{\text{loc},i}(x,t)}{(2\pi\theta_{\text{loc}}(x,t))^{3/2}} \exp\left(-\frac{|v-u_{\text{loc}}(x,t)|^2}{2\theta_{\text{loc}}(x,t)}\right),$$

where, introducing the total local density $\rho_{\text{loc}} = \sum_{i=1}^{n} \rho_{\text{loc},i}$,

$$\rho_{\text{loc},i} = \int_{\mathbb{R}^3} F_i dv, \quad u_{\text{loc}} = \frac{1}{\rho_{\text{loc}}} \sum_{i=1}^n \int_{\mathbb{R}^3} F_i v dv, \quad \theta_{\text{loc}} = \frac{1}{3\rho_{\text{loc}}} \sum_{i=1}^n \int_{\mathbb{R}^3} F_i |v - u|^2 dv$$

are the (local) masses of the species, the total momentum and total energy, respectively.

On the other hand, the global equilibrium, which is the unique stationary solution F to (2.1.1), is given by $M = (M_1, \ldots, M_n)$ with

$$F_i(x,v) = M_i(v) = \frac{\rho_{\infty,i}}{(2\pi\theta_{\infty})^{3/2}} \exp\left(-\frac{|v - u_{\infty}|^2}{2\theta_{\infty}}\right),$$

where now, setting $\rho_{\infty} = \sum_{i=1}^{n} \rho_{\infty,i}$,

$$\rho_{\infty,i} = \int_{\mathbb{T}^3 \times \mathbb{R}^3} F_i dx dv, \quad u_{\infty} = \frac{1}{\rho_{\infty}} \int_{\mathbb{T}^3 \times \mathbb{R}^3} F_i v dx dv, \quad \theta_{\infty} = \frac{1}{3\rho_{\infty}} \int_{\mathbb{T}^3 \times \mathbb{R}^3} F_i |v - u|^2 dx dv$$

do not depend on (x,t). By translating and scaling the coordinate system, we may assume that $u_{\infty} = 0$ and $\theta_{\infty} = 1$ such that the global equilibrium becomes

$$M_i(v) = \frac{\rho_{\infty,i}}{(2\pi)^{3/2}} e^{-|v|^2/2}, \quad 1 \le i \le n.$$
(2.1.4)

2.1.2. Linearized Boltzmann collision operator

We assume that the distribution function F_i is close to the global equilibrium such that we can write $F_i = M_i + M_i^{1/2} f_i$ for some small perturbation f_i , where M_i is given by (2.1.4). Then, dropping the small nonlinear remaining term, f_i satisfies the linearized equation

$$\partial_t f_i + v \cdot \nabla_x f_i = L_i(f), \quad t > 0, \quad f_i(x, v, 0) = f_{I,i}(x, v), \quad (x, v) \in \mathbb{T}^3 \times \mathbb{R}^3, \quad (2.1.5)$$

for $1 \leq i \leq n$, where $f = (f_1, \ldots, f_n)$ and the *i*th component of the linearized collision operator $L = (L_1, \ldots, L_n)$ is given by

$$L_i(f) = \sum_{j=1}^{n} L_{ij}(f_i, f_j), \quad 1 \le i \le n,$$

with

$$L_{ij}(f_i, f_j) = M_i^{-1/2} \left(Q_{ij}(M_i, M_j^{1/2} f_j) + Q_{ij}(M_i^{1/2} f_i, f_j) \right)$$

$$= \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} M_i^{1/2} M_j^* (h_i' + h_j'^* - h_i - h_j^*) dv^* d\sigma, \quad h_i := M_i^{-1/2} f_i. \quad (2.1.6)$$

Here, we have used $M_i^{'*}M_j^{'}=M_i^*M_j$ for any i,j, which follows from (2.1.3). Notice that we have chosen the linearization considered in, e.g., [113, 135], which allows to work in a pure $L_{x,v}^2$ framework without weight function. Another linearization is given by $F_i=M_i+g_i$, which we use in Chapter 5, making it necessary to work in a weighted L^2 space, where the weight function is a Maxwellian of the form $1/M_i$. Note that this choice gives the same results as with the linearization (2.1.6) since both linearizations correspond to the same space of solutions. It turned out that in Chapter 4 the computations are easier using (2.1.6), whereas in Chapter 5 they are easier using the other linearization.

The linearized Boltzmann system satisfies an H-theorem with the linearized entropy $H(f) = \frac{1}{2} \sum_{i=1}^{n} \int_{\mathbb{R}^3} f_i^2 dv$,

$$-\frac{dH}{dt} = -\sum_{i=1}^{n} \int_{\mathbb{R}^3} f_i L_i(f) dv =: -(f, L(f))_{L_v^2} \ge 0,$$

where $(\cdot,\cdot)_{L_v^2}$ is the scalar product on $L_v^2 := L^2(\mathbb{R}^3;\mathbb{R}^n)$. We will prove in Lemma 4.4 that $(f,L(f))_{L_v^2} = 0$ if and only if $M_i^{-1/2}f_i$ lies in span $\{e^{(1)},\ldots,e^{(n)},v_1\mathbf{1},v_2\mathbf{1},v_3\mathbf{1},|v|^2\mathbf{1}\}$, which is the null space $\operatorname{Ker}(L)$ of the linear operator L.

Our main goal is to show that, under suitable assumptions on the collision kernels, there exists a constant $\lambda > 0$, which can be computed explicitly, such that for all suitable functions f, $-(f, L(f))_{L_v^2} \ge \lambda ||f - \Pi^L(f)||_{\mathcal{H}}^2$, where Π^L is the projection onto $\operatorname{Ker}(L)$ and \mathcal{H} is a subset of L_v^2 (see Theorem 2.3 for the precise statement). This spectral-gap estimate, together with hypocoercivity techniques, allows us to conclude that exponential decay of the solutions f(t) towards the global equilibrium holds (see Theorem 2.4).

2.1.3. State of the art

The study of the linearized mono-species collision operator, in the spatially homogeneous and hard-potential case, goes back to Hilbert [87]. For this operator, Carleman [30] proved the existence of a spectral gap. The results were extended by Grad [74] for hard potentials with cut-off. Baranger and Mouhot [4] derived constructive estimates in the hard-sphere case. For Maxwell molecules, Fourier transform methods were employed in [35] to achieve explicit spectral properties. A spectral-gap estimate for the linearized Boltzmann operator, consisting of the sum of the linearized collision operator and the transport operator, was first shown by Ukai [134]. Improved estimates (in smaller spaces of Sobolev type), still for hard potentials, were established in [113]. In [114], spectral-gap estimates for moderately soft potentials (without angular cut-off) were proved, improving and extending previous results by Pao [118]. Hypoelliptic estimates for the linearized operator without cut-off can be found in [1] and references therein. A spectral analysis with relaxed tail decay and regularity conditions on the solutions was performed recently in an abstract framework [78], which we will use in Chapter 5. Dolbeault et al. [58] derived exponential decay rates in weighted L^2 spaces, which improves previous Sobolev estimates. For further references, we refer to [114, Section 1.5].

Spectral properties of the linearized Boltzmann operator were already investigated by Grad [75]. Based on these results, Schechter [128] located the essential spectrum of the classical collision operator in L^2 . The spectrum of the Boltzmann operator for hard spheres was also analyzed in L^p for $p \neq 2$; see [100]. We refer to the recent work [63] for further results in L^p for $1 \leq p \leq \infty$ and more references. A detailed analysis of the resolvent and spectrum of the linearized Boltzmann operator can be found in [135, Section 2.2]. A complete analysis for the essential and discrete spectra for the linearized collision operator with hard potentials was performed in [112].

Before we state our main results, let us introduce the general assumptions.

2.1.4. Assumptions on the collision kernels

We impose the following assumptions on the collision kernels B_{ij} arising in (2.1.6). For a discussion of these assumptions, see section 4.1.2.

(A1) The collision kernels satisfy

$$B_{ij}(|v-v^*|,\cos\vartheta) = B_{ii}(|v-v^*|,\cos\vartheta)$$
 for $1 \le i, j \le n$.

(A2) The collision kernels decompose in the kinetic part $\Phi_{ij} \geq 0$ and the angular part $b_{ij} \geq 0$ according to

$$B_{ij}(|v-v^*|,\cos\vartheta) = \Phi_{ij}(|v-v^*|)b_{ij}(\cos\vartheta), \quad 1 \le i, j \le n.$$

(A3) For the kinetic part, there exist constants C_1 , $C_2 > 0$, $\gamma \in [0, 1]$, and $\delta \in (0, 1)$ such that for all $1 \le i, j \le n$ and r > 0,

$$C_1 r^{\gamma} \le \Phi_{ij}(r) \le C_2 (r + r^{-\delta}).$$

(A4) For the angular part, there exist constants C_3 , $C_4 > 0$ such that for all $1 \le i, j \le n$ and $\theta \in [0, \pi]$,

$$0 < b_{ij}(\cos \vartheta) \le C_3 |\sin \vartheta| |\cos \vartheta|, \quad b'_{ij}(\cos \vartheta) \le C_4.$$

Furthermore,

$$C^b := \min_{1 \le i \le n} \inf_{\sigma_1, \sigma_2 \in \mathbb{S}^2} \int_{\mathbb{S}^2} \min \left\{ b_{ii}(\sigma_1 \cdot \sigma_3), b_{ii}(\sigma_2 \cdot \sigma_3) \right\} d\sigma_3 > 0.$$

- (A5) For all $1 \leq i, j \leq n$, b_{ij} is even in [-1, 1] and the mapping $v \mapsto \Phi'_{ij}(|v|)$ on \mathbb{R}^3 is locally integrable on \mathbb{R}^3 and bounded as $|v| \to \infty$.
- (A6) There exists $\beta > 0$ such that for all $1 \leq i, j \leq n, s > 0$, and $\sigma \in [-1, 1]$, we have $B_{ij}(s, \sigma) \leq \beta B_{ii}(s, \sigma)$.

Following [113], since the functions b_{ij} are integrable, we define

$$\ell^b := \min_{1, \le i, j \le n} \int_0^\pi b_{ij}(\cos \theta) \sin \theta d\theta > 0. \tag{2.1.7}$$

2.1.5. Main results for same molar masses

We present the main results and discuss them briefly. The first result is a geometric property of the essential spectrum of the linearized collision operator L and the linearized Boltzmann operator G = L - T, where T denotes the transport part $T = v \cdot \nabla_x$.

Theorem 2.1 (Essential spectrum of L and L-T). Let the collision kernels B_{ij} satisfy assumptions (A1)-(A4) and set

$$J = \bigcup_{i=1}^{n} \operatorname{Im}(\nu_i) \subset [\nu_0, \infty),$$

where $\nu_0 = \min_{i=1,\dots,n} \sup_{v \in \mathbb{R}^3} \nu_i(v) > 0$, where ν_i is the i-th component of the collision frequency ν_i , given by

$$\nu_i(v) = \sum_{j=1}^n \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij}(|v^* - v|, \cos \theta) M_j^* dv^* d\sigma, \quad i = 1, \dots, n.$$
 (2.1.8)

Then

$$\sigma_{ess}(L) = -J, \quad \sigma_{ess}(L-T) = \{\lambda \in \mathbb{C} : \Re(\lambda) \in -J\},$$

where $\sigma_{ess}(L)$ denotes the essential spectrum of the multi-species operator L.

Remark 2.2. We observe that if $\lim_{|v|\to\infty}\nu_i(v)=\infty$ for $i=1,\ldots,n$ then

$$\sigma_{\text{ess}}(L) = (-\infty, -\nu_0], \quad \sigma_{\text{ess}}(L-T) = \{\lambda \in \mathbb{C} : \Re(\lambda) < -\nu_0\}.$$

Indeed, under the assumption $\nu_i(v) \to \infty$ as $|v| \to \infty$, the continuity of ν_i , and the Weierstraß theorem show that $J = [\nu_0, \infty)$. Thus, the essential spectrum of the linearized multi-species collision operator is very similar to the mono-species operator, where ν_0 corresponds to the infimum in \mathbb{R}^3 of the single collision frequency; see [111, Section 3] and [112, Prop. 3.1].

The proof of Theorem 2.1 is based on perturbation theory [97, Chap. IV] and is similar to the proof for the mono-species collision operator [135], but we show new explicit spectral-gap estimates related to the particular structure of the kernel in the multi-species case. More precisely, we write $L = K - \Lambda$, where it turns out that $K = L + \Lambda$ is compact on L_v^2 (see section 4.3 for details). Weyl's theorem [84, Theorem S] states that the essential spectrum of $L = K - \Lambda$ coincides with that of $-\Lambda$. Thus it remains to show that $\sigma_{\text{ess}}(\Lambda) = J$. This is done by using Weyl's singular sequences, which allow for a sufficient and necessary condition for $\lambda \in \mathbb{C}$ being an element of the essential spectrum of the selfadjoint operator Λ .

The proof of the second statement in Theorem 2.1 is more involved since K is not compact on $L_{x,v}^2$ and hence, Weyl's theorem cannot be applied directly. The idea is to employ an extended Weyl theorem, which states that the essential spectrum is conserved under a relatively compact perturbation [97, Section IV.5.6, Theorem 5.35]. Indeed, if K is relatively compact with respect to $\Lambda + T$ then $\sigma_{\text{ess}}(L - T) = \sigma_{\text{ess}}(K - (\Lambda + T)) = -\sigma_{\text{ess}}(\Lambda + T)$, and it remains to compute the essential spectrum of $\Lambda + T$.

The next theorem concerns an explicit spectral-gap estimate, which is our main result here.

Theorem 2.3 (Explicit spectral-gap estimate). Let the collision kernels B_{ij} satisfy assumptions (A1)-(A4). Then there exists a constant $\lambda > 0$ such that

$$-(f, L(f))_{L_n^2} \ge \lambda \|f - \Pi^L(f)\|_{\mathcal{H}}^2 \quad \text{for all } f \in \mathcal{D},$$
 (2.1.9)

where Π^L is the projection onto the null space $\operatorname{Ker}(L)$, \mathcal{D} denotes the domain of L, and \mathcal{H} is a weighted $L^2(\nu)$ space weighted by the collision frequency ν , see (4.1.6) for the precise definitions. If additionally hypothesis (A6) holds, the constant λ can be computed explicitly:

$$\lambda = \frac{\eta D^b}{8C_k}, \quad \eta = \min\left\{1, \frac{4C^m C_k}{16C_k + D^b}\right\},$$

where C^m , D^b , and C_k are defined in (4.4.8), (4.4.13), and (4.4.15), respectively.

We present two proofs of this theorem. The first proof is non-constructive and relies on an abstract functional theoretical argument, based on the decomposition $L = K - \Lambda$ and Weyl's perturbation theorem. This abstract spectral-gap estimate is proved in Lemma 4.9. The second proof provides a constructive spectral-gap estimate, generalizing the result in [111] (also see [110, Theorem 6.1]) from the mono-species to the multi-species case. For this, we split the operator $L = L^m + L^b$ in the mono-species part $L^m = (L_1^m, \ldots, L_n^m)$ and the bi-species part $L^b = (L_1^b, \ldots, L_n^b)$,

$$L_i^m(f_i) = L_{ii}(f_i, f_i), \quad L_i^b(f) = \sum_{j \neq i} L_{ij}(f_i, f_j).$$

The proof consists of four main steps.

Step 1: Coercivity of the mono-species operator L^m . The bi-species part of L satisfies $-(f, L^b(f))_{L^2_v} \geq 0$ for all $f \in \mathcal{D}$. Furthermore, the results of [110, Theorem 6.1] show that for the mono-species part,

$$-(f, L^{m}(f))_{L_{v}^{2}} \ge C^{m} \|f - \Pi^{m}(f)\|_{\mathcal{H}}^{2} \quad \text{for } f \in \text{Dom}(L^{m}), \tag{2.1.10}$$

where the constant $C^m > 0$ can be computed explicitly and Π^m is the projection onto $\operatorname{Ker}(L^m)$ (see Lemma 4.10). Inequality (2.1.10) may be interpreted as a coercivity estimate for L^m on $\operatorname{Ker}(L^m)^{\perp}$. It is related to the "microscopic coercivity" in [58, Section 1.3] for the mono-species setting. Hence, we obtain the "naive" spectral-gap estimate

$$-(f, L(f))_{L^2_v} \ge C^m \|f - \Pi^m(f)\|_{\mathcal{H}}^2 \quad \text{for } f \in \mathcal{D}.$$

This estimate is not sharp enough for the multi-species case since we need an inequality for all $f - \Pi^L(f) \in \operatorname{Ker}(L)^{\perp}$ and not only for $f - \Pi^m(f) \in \operatorname{Ker}(L^m)^{\perp} \subset \operatorname{Ker}(L)^{\perp}$. By projecting onto $\operatorname{Ker}(L^m)^{\perp}$ only, we neglect the "cross-effects" coming from the bi-species part of the collision operator. Thus, we need a better estimate for $-(f, L(f))_{L_v^2}$, which is achieved as follows.

Step 2: Absorption of the orthogonal parts. The contribution $f^{\perp} := f - \Pi^m(f)$ in $-(f, L^b(f))_{L^2_v}$ can be absorbed by the \mathcal{H} norm of f^{\perp} (see Lemma 4.11), giving for a certain $\eta > 0$,

$$-(f, L(f))_{L_v^2} \ge (C^m - 4\eta) \|f^{\perp}\|_{\mathcal{H}}^2 - \frac{\eta}{2} (\Pi^m(f), L^b(\Pi^m(f)))_{L_v^2}.$$

Step 3: Coercivity of the bi-species operator L^b . The projection $\Pi^m(f)$ depends on the velocities u_i and energies e_i of the *i*th species, and thus, the cross terms can be bounded by the differences of momentum and differences of energies,

$$- (\Pi^{m}(f), L^{b}(\Pi^{m}(f)))_{L_{v}^{2}} \ge C \sum_{i,j=1}^{n} (|u_{i} - u_{j}|^{2} + |e_{i} - e_{j}|^{2}),$$
 (2.1.11)

for some constant C>0. This is the key step of the proof. The inequality may be considered as a coercivity estimate for the bi-species operator. A key observation is that the differences of momenta and energies converge to zero as f approaches the global equilibrium. Whereas (2.1.10) acts on $\operatorname{Ker}(L^m)^{\perp}$, (2.1.11) gives an estimate on the orthogonal complement $\operatorname{Ker}(L^m)$.

Step 4: Lower bound for the differences of momenta and energy. The last step consists in estimating the differences $|u_i - u_j|$ and $|e_i - e_j|$ from below by the error made by projecting onto $\text{Ker}(L^m)^{\perp}$ instead of $\text{Ker}(L)^{\perp}$:

$$\sum_{i,j=1}^{n} (|u_i - u_j|^2 + |e_i - e_j|^2) \ge C \|f - \Pi^L(f)\|_{\mathcal{H}}^2 - 2C \|f - \Pi^m(f)\|_{\mathcal{H}}^2.$$

Putting together the above inequalities, Theorem 2.3 follows; we refer to section 4.4 for details.

As a consequence of the spectral-gap estimate, we are able to prove the exponential decay of the solution f(t) to (2.1.5) to the global equilibrium with an explicit decay rate.

Theorem 2.4 (Convergence to equilibrium). Let the collision kernels B_{ij} satisfy assumptions (A1)-(A5) and let $f_I \in H^1_{x,v}$. Then the linearized Boltzmann operator G = L - T with $T = v \cdot \nabla_x$ generates a strongly continuous semigroup $S_G(t)$ on $H^1_{x,v}$, which satisfies

$$||S_G(t)(I - \Pi^G)||_{H^1_{x,y}} \le Ce^{-\tau t}, \quad t \ge 0,$$
 (2.1.12)

for some constants $C, \tau > 0$. In particular, the solution $f(t) = S_G(t)f_I$ to (2.1.5) satisfies

$$||f(t) - f_{\infty}||_{H^{1}_{x,y}} \le Ce^{-\tau t} ||f_{I} - f_{\infty}||_{H^{1}_{x,y}}, \quad t \ge 0,$$
 (2.1.13)

where $f_{\infty} := \Pi^G(f_I)$ is the global equilibrium of (2.1.5). Moreover, under the additional assumption (A6) and lower bound in (A4), the constants C and τ depend only on the constants appearing in hypotheses (M1)-(M3) in section 4.5 and in particular on λ defined in Theorem 2.3.

The idea of the proof is to employ the hypocoercivity of the linearized Boltzmann operator L-T, using the interplay between the degenerate-dissipative properties of L and the conservative properties of T. The aim is to find a functional $\widetilde{H}[f]$ which is equivalent to the square of the norm of a Banach space (here, $H_{x,v}^1$),

$$\kappa_1 \|f\|_{H^1_{x,v}}^2 \le \widetilde{H}[f] \le \kappa_2 \|f\|_{H^1_{x,v}}^2 \quad \text{for } f \in H^1_{x,v},$$

leading to

$$\frac{d}{dt}\widetilde{H}[f(t)] \le -\kappa \|f(t)\|_{H_{x,v}^1}^2, \quad t > 0, \tag{2.1.14}$$

where κ_1 , κ_2 , $\kappa > 0$ and $f(t) = S_G(t)f_I$. These two estimates yield exponential convergence of f(t) in $H^1_{x,v}$. It turns out that the obvious choice $\widetilde{H}[f] = c_1 ||f||_{L^2_{x,v}} + c_2 ||\nabla_x f||_{L^2_{x,v}} + c_3 ||\nabla_v f||_{L^2_{x,v}}$ does not lead to a closed estimate. The key idea, inspired from [139] and worked out in [113], is to add the "mixed term" of the form

$$c_4(\nabla_x f, \nabla_v f)_{L^2_x} \tag{2.1.15}$$

to the definition of $\widetilde{H}[f]$. Then

$$\frac{d}{dt}(\nabla_x f, \nabla_v f)_{L^2_{x,v}} = -\|\nabla_x f\|_{L^2_{x,v}}^2 + 2(\nabla_x L(f), \nabla_v f)_{L^2_{x,v}},$$

and the last term can be estimated in terms of expressions arising from the time derivative of the other norms in $\widetilde{H}[f(t)]$. Thus, choosing $c_i > 0$ in a suitable way, one may conclude that (2.1.14) holds.

In [113], the calculation of (2.1.14) is reduced to the validity of certain abstract conditions on the operators K and Λ (see section 4.5). These conditions state that Λ is coercive in a certain sense, K has a regularizing effect, and $L = K - \Lambda$ has a local spectral gap. The last condition is proved in Theorem 2.3, while the other conditions follow from direct calculations, since the operators K and Λ are given explicitly. As a consequence, the proof of Theorem 2.4 essentially consists in verifying the abstract conditions stated in [113]. In contrast to the estimate of Theorem 2.3, where the multi-species character plays a role in the spectral-gap estimate, there are no "cross-effects" here and the same modified functional $\widetilde{H}[f]$ as above, including the mixed term, can be used. However, the decay rate τ changes, since the constant in hypothesis (M3) (see section 4.5) differs in the mono- and multi-species case and τ depends also on that constant.

2.1.6. Comments and outlook

First, it seems to be not trivial to extend the results to the whole-space case. The problem is that one loses the compactness in the x-space. One possibility is to assume some confinement potential which, under some appropriate weighted Poincaré inequality, can yield compactness of the resolvent and hence a spectral gap. For instance, Duan [62] used non-constructive techniques to prove decay rates for the mono-species linearized Boltzmann equation. Still in the mono-species case, with one-dimensional collisional invariants and using constructive methods, the decay is investigated by, e.g., Hérau and Nier [86] and Villani [139], working in the space $H_{x,v}^1$, and by Hérau [85] and Dolbeault, Mouhot, Schmeiser [58], working in the space $L_{x,v}^2$. The tasks in the multi-species case are first to extend the non-constructive methods, which probably does not contain new difficulties, and second to devise a constructive method, which is more involved and work in progress; see [57].

Second, the convergence result based on hypocoercivity requires some regularity on the initial data, namely $f_I \in H^1_{x,v}$. The extension of the exponential decay to initial data from $L^2_{x,v}$ is done in Chapter 5, where a Cauchy theory for the full nonlinear multi-species Boltzmann equation in a perturbative framework is presented. In [58], exponential convergence was proved for solutions to scalar linear kinetic equations in the whole space with a confining potential, where just L^2 regularity for the initial data was needed, but only for one-dimensional collision kernels which is not the case for the Boltzmann equation.

Third, the technique of proving a multi-species explicit spectral-gap estimate seems to be quite robust also to other multi-species models, since it has been recently successfully adapted to the case of a multi-species Landau equation with soft potentials [77].

2.2. The Boltzmann system with different molar masses close to equilibrium

In Chapter 5 our goal will be to extend the previous results obtained in Chapter 4 to the more general case of a gaseous mixture with different molar masses. Furthermore, our goal will be not only to study the trend to global equilibrium, but to perform a full Cauchy theory for this equation, including also existence, uniqueness, positivity besides exponential trend to equilibrium in appropriate functional spaces.

The physically most relevant space for such a Cauchy theory for this nonlinear equation close to equilibrium is the space of density functions that only have finite mass and energy, which are the first and second moments in the velocity variable. We will obtain this result in the space $L_v^1 L_x^{\infty}(1+|v|^k)$ for any $k > k_0$, where k_0 is an explicit threshold depending heavily on the differences of the masses, recovering the physically optimal threshold $k_0 = 2$ when all the masses of the mixture are the same and the particles are approximated to be hard spheres.

Moreover, by combining very recent strategies in combination with new hypocoercivity estimates, we develop a new constructive approach that deals with polynomial weights without the need of any spatial Sobolev regularity. This is new even in the mono-species case even though the final result we obtain has recently been proved for the mono-species hard sphere model [78]) (which we therefore also extend to more general hard and Maxwellian

potential kernels.).

Also, as a by-product, we prove explicitly that the linear operator $\mathbf{L} - v \cdot \nabla_x$ generates a strongly continuous semigroup with exponential decay both in $L_{x,v}^2\left(\boldsymbol{\mu}^{-1/2}\right)$ and in $L_{x,v}^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)$; such constructive and direct results on the torus are new to our knowledge, even for the single-species Boltzmann equation.

At last, we derive new estimates in order to deal with different masses and the multispecies cross-interaction operators, and we also extend recent mono-species estimates to more general collision kernels. Note that the asymmetry of the elastic collisions requires to derive a new description of Carleman's representation of the Boltzmann operator as well as new Povzner-type inequalities suitable for this lack of symmetry.

2.2.1. The model for different molar masses

Since we will use a slightly different notation in Chapter 5 compared to Chapter 4, let us briefly introduce the model. For a detailed presentation of the properties of this model, we refer to subsection 5.1.1 and 5.1.2.

We are now interested in the evolution of a dilute gas on the torus \mathbb{T}^3 composed of N different species of chemically non-reacting mono-atomic particles with different molar masses, which can be modeled by the following system of Boltzmann equations, stated on $\mathbb{R}^+ \times \mathbb{T}^3 \times \mathbb{R}^3$,

$$\forall 1 \le i \le N, \quad \partial_t F_i(t, x, v) + v \cdot \nabla_x F_i(t, x, v) = Q_i(\mathbf{F})(t, x, v) \tag{2.2.1}$$

with initial data

$$\forall 1 \le i \le N, \ \forall (x, v) \in \mathbb{T}^3 \times \mathbb{R}^3, \quad F_i(0, x, v) = F_{0,i}(x, v).$$

The distribution function of the system is given by the vector $\mathbf{F} = (F_1, \dots, F_N)$, where F_i describes the i^{th} species at time t, position x and velocity v.

The Boltzmann operator $\mathbf{Q}(\mathbf{F}) = (Q_1(\mathbf{F}), \dots, Q_N(\mathbf{F}))$ is given for all i by

$$Q_i(\mathbf{F}) = \sum_{j=1}^{N} Q_{ij}(F_i, F_j),$$

where Q_{ij} describes interactions between particles of either the same (i = j) or of different $(i \neq j)$ species and are local in time and space.

$$Q_{ij}(F_i, F_j)(v) = \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} (|v - v_*|, \cos \theta) \left[F_i' F_j'^* - F_i F_j^* \right] dv_* d\sigma,$$

where we used the shorthands $F'_i = F_i(v')$, $F_i = F_i(v)$, $F'_j = F_j(v'_*)$ and $F^*_j = F_j(v_*)$. Due to the fact that now we allow also different molar masses for each species, the microscopic collision rules have the following form:

$$\begin{cases} v' = \frac{1}{m_i + m_j} \left(m_i v + m_j v_* + m_j | v - v_* | \sigma \right) \\ v'_* = \frac{1}{m_i + m_j} \left(m_i v + m_j v_* - m_i | v - v_* | \sigma \right) \end{cases}, \text{ and } \cos \theta = \left\langle \frac{v - v_*}{|v - v_*|}, \sigma \right\rangle.$$

which comes from the conservation of momentum and kinetic energy on the microscopic level due to elastic collisions:

$$m_{i}v + m_{j}v_{*} = m_{i}v' + m_{j}v'_{*},$$

$$\frac{1}{2}m_{i}|v|^{2} + \frac{1}{2}m_{j}|v_{*}|^{2} = \frac{1}{2}m_{i}|v'|^{2} + \frac{1}{2}m_{j}|v'_{*}|^{2}.$$
(2.2.2)

We want to mention the lack of symmetry between v' and v'_* compared to v for the transformation $(v, v_*) \mapsto (v_*, v)$ due to different masses, which creates additional technical difficulties. By translating and rescaling the coordinate system, we can always assume that the only global equilibrium is the normalized Maxwellian

$$\mu = (\mu_i)_{1 \le i \le N} \quad \text{with} \quad \mu_i(v) = c_{\infty,i} \left(\frac{m_i}{2\pi}\right)^{3/2} e^{-m_i \frac{|v|^2}{2}}.$$
(2.2.3)

We mention that we have the following macroscopic conservation laws: conservation of the total number density $c_{\infty,i}$ of each species, of the total momentum of the gas $\rho_{\infty}u_{\infty}$ and its total energy $3\rho_{\infty}\theta_{\infty}/2$:

$$\forall t \geq 0, \quad c_{\infty,i} = \int_{\mathbb{T}^3 \times \mathbb{R}^3} F_i(t, x, v) \, dx dv \quad (1 \leq i \leq N)$$

$$u_{\infty} = \frac{1}{\rho_{\infty}} \sum_{i=1}^{N} \int_{\mathbb{T}^3 \times \mathbb{R}^3} m_i v F_i(t, x, v) \, dx dv$$

$$\theta_{\infty} = \frac{1}{3\rho_{\infty}} \sum_{i=1}^{N} \int_{\mathbb{T}^3 \times \mathbb{R}^3} m_i |v - u_{\infty}|^2 F_i(t, x, v) \, dx dv,$$

$$(2.2.4)$$

where $\rho_{\infty} = \sum_{i=1}^{N} m_i c_{\infty,i}$ is the global density of the gas. For more details on the modified conservation rules and other differences due to these different molar masses, we refer to subsection 5.1.2.

The aim is to construct a Cauchy theory for the multi-species Boltzmann equation (2.2.1) around the global equilibrium μ . In other terms we study the existence, uniqueness and exponential decay of solutions of the form $F_i(t, x, v) = \mu_i(v) + f_i(t, x, v)$ for all i.

Under this perturbative regime, the Cauchy problem amounts to solving the perturbed multi-species Boltzmann system of equations

$$\partial_t \mathbf{f} + v \cdot \nabla_x \mathbf{f} = \mathbf{L}(\mathbf{f}) + \mathbf{Q}(\mathbf{f}),$$
 (2.2.5)

or equivalently in the non-vectorial form

$$\forall 1 \leq i \leq N, \quad \partial_t f_i + v \cdot \nabla_x f_i = L_i(\mathbf{f}) + Q_i(\mathbf{f}),$$

where $\mathbf{f} = (f_1, \dots, f_N)$ and the operator $\mathbf{L} = (L_1, \dots, L_N)$ is the linear Boltzmann operator given for all $1 \le i \le N$ by

$$L_i(\mathbf{f}) = \sum_{j=1}^{N} L_{ij}(f_i, f_j),$$

with

$$L_{ij}(f_i, f_j) = Q_{ij}(\mu_i, f_j) + Q_{ij}(f_i, \mu_j).$$

As mentioned previously, we prove the existence, uniqueness, positivity and exponential trend to equilibrium for the full nonlinear multi-species Boltzmann equation (2.2.1) in $L_v^1 L_x^{\infty} ((1+|v|)^k)$ with the explicit threshold $k > k_0$ defined in Lemma 5.20, when the initial data $\mathbf{F_0}$ is close enough to the global equilibrium $\boldsymbol{\mu}$. This is equivalent to solving the perturbed equation (2.2.5) for small $\mathbf{f_0}$.

2.2.2. State of the art for the Cauchy problem

The perturbed single-species Boltzmann equation around a global Maxwellian has been extensively studied over the past fifty years (see [135] for an exhaustive review). Starting with Grad [76], the Cauchy problem has been tackled in $L_v^2 H_x^s \left(\mu^{-1/2}\right)$ spaces [134], in $H_{x,v}^s \left(\mu^{-1/2}(1+|v|)^k\right)$ [81][143] was then extended to $H_{x,v}^s \left(\mu^{-1/2}\right)$ where an exponential trend to equilibrium has also been obtained [113][82]. Recently, [78] proved existence and uniqueness for single-species Boltzmann equation in more the general spaces $\left(W_v^{\alpha,1} \cap W_v^{\alpha,q}\right) W_x^{\beta,p} \left((1+|v|)^k\right)$ for $\alpha \leq \beta$ and β and k large enough with explicit thresholds. The latter paper thus includes $L_v^1 L_x^\infty \left((1+|v|)^k\right)$. All the results presented above hold in the case of the torus for hard and Maxwellian potentials. We refer the reader interested in the Cauchy problem to the review [135].

All the works mentioned above involve to working in spaces with derivatives in the space variable x (we shall discuss some of the reasons later) with exponential weight. The recent breakthrough [78] gets rid of both the Sobolev regularity and the exponential weight but uses a new extension method which still requires to have a well-established linear theory in $H_{x,v}^s(\mu^{-1/2})$.

2.2.3. Assumptions on the collision kernels

We will use the following assumptions on the collision kernels B_{ij} .

(H1) The following symmetry holds

$$B_{ii}(|v - v_*|, \cos \theta) = B_{ii}(|v - v_*|, \cos \theta)$$
 for $1 \le i, j \le N$.

(H2) The collision kernels decompose into the product

$$B_{ij}(|v - v_*|, \cos \theta) = \Phi_{ij}(|v - v_*|)b_{ij}(\cos \theta), \quad 1 \le i, j \le N,$$

where the functions $\Phi_{ij} \geq 0$ are called kinetic part and $b_{ij} \geq 0$ angular part.

(H3) The kinetic part has the form of hard or Maxwellian ($\gamma = 0$) potentials, i.e.

$$\Phi_{ij}(|v-v_*|) = C_{ij}^{\Phi}|v-v_*|^{\gamma}, \quad C_{ij}^{\Phi} > 0, \quad \gamma \in [0,1], \quad \forall \ 1 \le i, j \le N.$$

(H4) For the angular part, we assume a strong form of Grad's angular cutoff (first introduced in [74]), that is: there exist constants C_{b1} , $C_{b2} > 0$ such that for all $1 \le i, j \le N$ and $\theta \in [0, \pi]$,

$$0 < b_{ij}(\cos \theta) \le C_{b1} |\sin \theta| |\cos \theta|, \quad b'_{ij}(\cos \theta) \le C_{b2}.$$

Furthermore,

$$C^b := \min_{1 \le i \le N} \inf_{\sigma_1, \sigma_2 \in \mathbb{S}^2} \int_{\mathbb{S}^2} \min \left\{ b_{ii}(\sigma_1 \cdot \sigma_3), b_{ii}(\sigma_2 \cdot \sigma_3) \right\} d\sigma_3 > 0.$$

Moreover, we shall use the (standard) shorthand notations

$$b_{ij}^{\infty} = \|b_{ij}\|_{L_{[-1,1]}^{\infty}} \quad \text{and} \quad l_{b_{ij}} = \|b \circ \cos\|_{L_{\mathbb{S}^2}^1}.$$
 (2.2.6)

and

$$\langle v \rangle = \sqrt{1 + |v|^2}.\tag{2.2.7}$$

2.2.4. Main results for different molar masses close to equilibrium

As already explained, the ultimate goal in this chapter will be to perform a full perturbative Cauchy theory for the multi-species Boltzmann equation (2.2.1). Along the way, we shall also prove the following important results about the linear perturbed operator $\mathbf{L} - v \cdot \nabla_x$.

Theorem 2.5. Let the collision kernels B_{ij} satisfy assumptions (H1) - (H4). Then the following holds.

(i) The operator **L** is a closed self-adjoint operator in $L_v^2(\mu^{-1/2})$ and there exists $\lambda_L > 0$ such that

$$\forall \mathbf{f} \in L_v^2\left(\boldsymbol{\mu}^{-1/2}\right), \quad \langle \mathbf{f}, \mathbf{L}\left(\mathbf{f}\right) \rangle_{L_v^2\left(\boldsymbol{\mu}^{-1/2}\right)} \leq -\lambda_L \left\|\mathbf{f} - \pi_{\mathbf{L}}\left(\mathbf{f}\right)\right\|_{L_v^2\left(\langle v \rangle^{\gamma/2}\boldsymbol{\mu}^{-1/2}\right)}^2;$$

(ii) Let $E = L_{x,v}^2(\boldsymbol{\mu}^{-1/2})$ or $E = L_{x,v}^{\infty}(\langle v \rangle^{\beta}\boldsymbol{\mu}^{-1/2})$ with $\beta > 3/2$. The linear perturbed operator $\mathbf{G} = \mathbf{L} - v \cdot \nabla_x$ generates a strongly continuous semigroup $S_{\mathbf{G}}(t)$ on E and there exist C_E , $\lambda_E > 0$ such that

$$\forall t \geq 0, \quad \|S_{\mathbf{G}}(t) \left(Id - \Pi_{\mathbf{G}}\right)\|_{E} \leq C_{E}e^{-\lambda_{E}t},$$

where $\pi_{\mathbf{L}}$ is the orthogonal projection onto $Ker(\mathbf{L})$ in $L_v^2(\boldsymbol{\mu}^{-1/2})$ and $\Pi_{\mathbf{G}}$ is the orthogonal projection onto $Ker(\mathbf{G})$ in $L_{x,v}^2(\boldsymbol{\mu}^{-1/2})$.

The constants λ_L , C_E and λ_E are explicit and depend on N, E, the different masses m_i and the collision kernels.

Spectral gap for the linear operator in $L_v^2(\mu^{-1/2})$. It has been known for long that the single-species linear Boltzmann operator L is a self-adjoint non positive linear operator in the space $L_v^2(\mu^{-1/2})$. Moreover it has a spectral gap λ_0 . This has been proved in [30][74][75] with non constructive methods for hard potential with cutoff and in [6][7] in the Maxwellian case. These results were made constructive in [4][111] for more general collision operators. One can easily extend this spectral gap to Sobolev spaces $H_v^s(\mu^{-1/2})$ (see for instance [78] Section 4.1).

In Chapter 4 based on [46], we proved the existence of an explicit spectral gap for the operator **L** for multi-species mixtures where all the masses are the same $(m_i = m_j)$. Our constructive spectral gap estimate in $L_v^2(\mu^{-1/2})$ proved in this chapter for different molar masses closely follows the methods presented in Chapter 4 that consist in proving that the cross-interactions between different species do not perturb too much the spectral gap that is known to exist for the diagonal operator L_{ii} (single-species operators). We emphasize here that not only we adapt the methods of [46] to fit the different masses framework but we also derive estimates on the collision frequencies that allow us to get rid of their strong requirement on the collision kernels: $B_{ij} \leq \beta B_{ii}$ for all i, j. The latter assumption is indeed physically irrelevant in our framework.

 $L_{x,v}^2\left(\boldsymbol{\mu}^{-1/2}\right)$ theory for the full perturbed linear operator. The next step is to prove that the existence of a spectral gap for \mathbf{L} in the sole velocity variable can be transposed to $L_{x,v}^2\left(\boldsymbol{\mu}^{-1/2}\right)$ when one adds the skew-symmetric transport operator $-v\cdot\nabla_x$. In other words, we prove that $\mathbf{G}=\mathbf{L}-v\cdot\nabla_x$ generates a strongly continuous semigroup in $L_{x,v}^2\left(\boldsymbol{\mu}^{-1/2}\right)$ with exponential decay.

One thus wants to derive an exponential decay for solutions to the linear perturbed Boltzmann equation

$$\partial_t \mathbf{f} + v \cdot \nabla_x \mathbf{f} = L(\mathbf{f})$$
.

A straightforward use of the spectral gap λ_L of **L** shows for such a solution

$$\frac{d}{dt} \left\| \mathbf{f} \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} \leq -2\lambda_{L} \left\| \mathbf{f} - \pi_{\mathbf{L}} \left(\mathbf{f} \right) \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2},$$

where $\pi_{\mathbf{L}}$ stands for the orthogonal projection in $L_v^2\left(\boldsymbol{\mu}^{-1/2}\right)$ onto the kernel of the operator \mathbf{L} . This inequality exhibits the hypocoercivity of \mathbf{L} . Roughly speaking, the exponential decay in $L_{x,v}^2\left(\boldsymbol{\mu}^{-1/2}\right)$ would follow for solutions \mathbf{f} if the microscopic part $\pi_{\mathbf{L}}^{\perp}(\mathbf{f}) = \mathbf{f} - \pi_{\mathbf{L}}(\mathbf{f})$ controls the fluid part which has the following form

$$\forall 1 \le i \le N, \quad \pi_{\mathbf{L}}(\mathbf{f})_i(t, x, v) = \left[a_i(t, x) + b(t, x) \cdot v + c(t, x) \frac{|v|^2 - 3m_i^{-1}}{2} \right] m_i \mu_i(v),$$

where $a_i(t,x), c(t,x) \in \mathbb{R}$ and $b(t,x) \in \mathbb{R}^3$ are the coordinates of an orthogonal basis.

The standard strategies in the case of the single-species Boltzmann equation are based on higher Sobolev regularity either from hypocoercivity methods [113] or elliptic regularity of the coefficients a, b and c [80][82]. Roughly speaking one has [80][82]

$$\Delta \pi_L(f) \sim \partial^2 \pi_L^{\perp} f + \text{higher order terms},$$
 (2.2.8)

which can be combined with elliptic estimates to control the fluid part by the microscopic part in Sobolev spaces H^s . Our main contribution to avoid involving high regularity is based on an adaptation of the recent work [66] (dealing with the single-species Boltzmann equation with diffusive boundary conditions). The key idea consists in integrating against test functions that contains a weak version of the elliptic regularity of $\mathbf{a}(t,x)$, b(t,x) and c(t,x). Basically, the elliptic regularity of $\pi_{\mathbf{L}}(\mathbf{f})$ will be recovered thanks to the transport part applied to these test functions while on the other side \mathbf{L} encodes the control by $\pi_{\mathbf{L}}^{\perp}(\mathbf{f})$.

It has to be emphasized that thanks to boundary conditions, [66] only needed the conservation of mass whereas in our case this "weak version" of estimates (2.2.8) strongly relies on all the conservation laws. The choice of test functions thus has to take into account the delicate interaction between each species and the total mixture we already pointed out. This leads to intricate technicalities since for each species we need to deal with different reference rates of decay m_i . Finally, our proof also involves elliptic regularity in negative Sobolev spaces to deal with $\partial_t \mathbf{a}$, $\partial_t b$ and $\partial_t c$.

 $L^{\infty}_{x,v}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)$ theory for the full nonlinear equation. Thanks to the first two steps we have a satisfactory L^2 semigroup theory for the full linear operator. Unfortunately, as it is already the case for the single-species Boltzmann equation (see [33][34] or [138] for instance), the underlying $L^2_{x,v}$ -norm is not an algebraic norm for the nonlinear operator \mathbf{Q} whereas the $L^{\infty}_{x,v}$ -norm is.

The key idea of proving a semigroup property in L^{∞} is thanks to an $L^2 - L^{\infty}$ theory "à la Guo" [83], where the L^{∞} -norm will be controlled by the L^2 -norm along the characteristics. As we shall see, each component L_i can be decomposed into $L_i = K_i - \nu_i$ where $\nu_i(f) = \nu_i(v)f_i$ is a multiplicative operator. If we denote by $\mathbf{S}_{\mathbf{G}}(t)$ the semigroup generated by $\mathbf{G} = \mathbf{L} - v \cdot \nabla_x$, we have the following implicit Duhamel representation of its i^{th} component along the characteristics

$$S_{\mathbf{G}}(t)_i = e^{-\nu_i(v)t} + \int_0^t e^{-\nu_i(v)(t-s)} K_i \left[\mathbf{S}_{\mathbf{G}}(s) \right] ds.$$

Following the idea of Vidav [137] and later used in [83], an iteration of the above should yield a certain compactness property. Hiding here all the cross-interactions, we end up with

$$\mathbf{S}_{\mathbf{G}}(t) = e^{-\boldsymbol{\nu}(v)t} + \int_{0}^{t} e^{-\boldsymbol{\nu}(v)(t-s)} \mathbf{K} e^{-\boldsymbol{\nu}(v)s} ds + \int_{0}^{t} \int_{0}^{s} e^{-\boldsymbol{\nu}(v)(t-s)} \mathbf{K} e^{-\boldsymbol{\nu}(v)(s-s_{1})} \mathbf{K} \left[\mathbf{S}_{\mathbf{G}}(s_{1}) \right] ds_{1} ds.$$

We shall prove that \mathbf{K} is compact and is a kernel operator. The first two terms will be easily estimated and the last term will be roughly of the form

$$\int_0^t \int_0^s \int_{v_1, v_2 \text{ bounded}} |\mathbf{S}_{\mathbf{G}}(s_1, x - (t - s)v - (s - s_1)v_1, v_2| \ dv_2 dv_1 ds_1 ds.$$

The double integration implies that v_1 and v_2 are independent and we can thus perform a change of variables which changes the integral in v_1 into an integral over \mathbb{T}^3 that we

can bound thanks to the previous L^2 theory. For integrability reasons, this third step actually proves that **G** generates a strongly continuous semigroup with exponential decay in L^{∞} ($\langle v \rangle^{\beta} \mu^{-1/2}$) for $\beta > 3/2$.

Our work provides two key contributions to prove the latter result. First, to prove the desired pointwise estimate for the kernel of \mathbf{K} , we need to give a new representation of the operator in terms of the parameters (v', v'_*) instead of (v_*, σ) . In the single-species case, such a representation is the well-known Carleman representation [30] and requires integration onto the so-called Carleman hyperplanes $\langle v' - v, v'_* - v \rangle = 0$. However, when particles have different masses, the lack of symmetry between v' and v'_* compared to v obliges us to derive new Carleman admissible sets (some become spheres). Second, the decay of the exponential weight differs from one species to the other. To obtain estimates that are similar to the case of single-species we exhibit the property that \mathbf{K} mixes the exponential rate of decay among the cross-interaction between species. This enables us to close the L^{∞} estimate for the first two terms of the iterated Duhamel representation.

We now state the results we obtain for the full nonlinear equation.

Theorem 2.6. Let the collision kernels B_{ij} satisfy assumptions (H1) - (H4) and let $E = L_v^1 L_x^{\infty} (\langle v \rangle^k)$ with $k > k_0$, where k_0 is the minimal integer such that

$$C_{k} = \frac{2}{k+2} \frac{1 - \left[\max_{i,j} \frac{|m_{i} - m_{j}|}{m_{i} + m_{j}} \right]^{\frac{k+2}{2}} + \left[1 - \left(\max_{i,j} \frac{|m_{i} - m_{j}|}{m_{i} + m_{j}} \right) \right]^{\frac{k+2}{2}}}{1 - \max_{i,j} \frac{|m_{i} - m_{j}|}{m_{i} + m_{j}}} \max_{i,j} \frac{4\pi b_{ij}^{\infty}}{l_{b_{ij}}} < 1. \quad (2.2.9)$$

where $l_{b_{ij}}$ and b_{ij}^{∞} are angular kernel constants (2.2.6).

Then there exist η_E , C_E and $\lambda_E > 0$ such that for any $\mathbf{F_0} = \boldsymbol{\mu} + \mathbf{f_0} \geq 0$ satisfying the conservation of mass, momentum and energy (2.2.4) with $u_{\infty} = 0$ and $\theta_{\infty} = 1$, if

$$\|\mathbf{F_0} - \boldsymbol{\mu}\| \le \eta_E$$

then there exists a unique solution $\mathbf{F} = \boldsymbol{\mu} + \mathbf{f}$ in E to the multi-species Boltzmann equation (2.2.1) with initial data $\mathbf{f_0}$. Moreover, \mathbf{F} is non-negative, satisfies the conservation laws and

$$\forall t \geq 0, \quad \|\mathbf{F} - \boldsymbol{\mu}\|_E \leq C_E e^{-\lambda_E t} \|\mathbf{F_0} - \boldsymbol{\mu}\|_E.$$

The constants are explicit and only depend on N, k, the different masses m_i and the collision kernels.

The extension to polynomial weights and $L_v^1 L_x^{\infty}$ space is done by developing an analytic and nonlinear version of the recent work [78], also recently adapted in a nonlinear setting [26]. The main strategy is to find a decomposition of the full linear operator \mathbf{G} into $\mathbf{G_1} + \mathbf{A}$. We shall prove that $\mathbf{G_1}$ acts like a small perturbation of the operator $\mathbf{G}_{\nu} = -v \cdot \nabla_x - \boldsymbol{\nu}(v)$ and is thus hypodissipative, and that \mathbf{A} has a regularizing effect. The regularizing property

of the operator \mathbf{A} allows us to decompose the perturbative equation (2.2.5) into a system of differential equations

$$\partial_t \mathbf{f_1} = \mathbf{G_1}(\mathbf{f_1}) + \mathbf{Q}(\mathbf{f_1} + \mathbf{f_2}, \mathbf{f_1} + \mathbf{f_2})$$
$$\partial_t \mathbf{f_2} + v \cdot \nabla_x \mathbf{f_2} = \mathbf{L}(\mathbf{f_2}) + \mathbf{A}(\mathbf{f_1})$$

The first equation is solved in $L_{x,v}^{\infty}(m)$ or $L_v^1 L_x^{\infty}(m)$ with the initial data $\mathbf{f_0}$ thanks to the hypodissipativity of $\mathbf{G_1}$. The regularity of $\mathbf{A}(\mathbf{f_1})$ allows us to use Step 3 and thus solve the second equation with null initial data in $L_{x,v}^{\infty}(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2}))$. First, the existence of a solution to the system having exponential decay is obtained thanks to an iterative scheme combined with new estimates on the multi-species operators $\mathbf{G_1}$ and \mathbf{A} . Then uniqueness follows a new stability estimate in an equivalent norm (proposed in [78]), that fits the dissipativity of the semigroup generated by \mathbf{G} . Finally, positivity of the unique solution comes from a different iterative scheme.

In the case of the single-species Boltzmann equation, the less regular weight m(v) one can achieve with this method is determined by the hypodissipative property of \mathbf{G}_1 and gives $m = \langle v \rangle^k$ with k > 2, which is indeed obtained also in the multi-species framework of same masses in the case of hard spheres. In the general case of different masses, the threshold k_0 is more intricate (see Theorem 2.6), since it also depends on the different masses m_i .

2.2.5. Comments and outlook

We make a few comments about the theorem above.

- (1) As already mentioned, μ can be replaced by any global equilibrium $\mathbf{M}(c_{i,\infty}, u_{\infty}, \theta_{\infty})$.
- (2) The natural weight for this theory is the one associated to the conservation of individual masses and total energy: $(1+m_i^{k/2}|v|^k)_{1\leq i\leq N}$, which is equivalent to $\langle v\rangle^k$ and we keep the latter weight to work without vector-valued masses outside Subsection 5.5.1.
- (3) The uniqueness has to be understood in a perturbative regime, that is among the solutions that can be written under the form $\mathbf{F} = \boldsymbol{\mu} + \mathbf{f}$. We do not give a global uniqueness in $L_v^1 L_x^{\infty} \left(\langle v \rangle^k \right)$ (as proved in [78] for the single-species Boltzmann equation).
- (4) As a by-product of the proof of uniqueness, we prove that the spectral-gap estimate of Theorem 2.5 also holds for $E = L_v^1 L_x^{\infty} (\langle v \rangle^k)$ with $k > k_0$.
- (5) In the case of identical masses and hard sphere collision kernels (b=1) we recover $C_k = 4/(k+2)$ and thus $k_0 = 2$ which has recently been obtained in the mono-species case [78].
- (6) We want to mention, that a stability analysis of the global equilibrium in L^{∞} settings has been recently obtained in [27].

3. Introduction to the diffusive multi-species models

3.1. Cross-diffusion population systems for multiple species

Shigesada, Kawasaki, and Teramoto suggested in their seminal paper [129] a diffusive Lotka-Volterra system for two competing species, which is able to describe the segregation of the population and to show pattern formation when time increases. Starting from an on-lattice random-walk model, this system was extended to an arbitrary number of species in [144, Appendix]. While the existence analysis of global weak solutions to the two-species model is well understood by now [37, 38], only very few results for the n-species model under very restrictive conditions exist (see the discussion below). Here, we provide for the first time a global existence analysis for an arbitrary number of population species using the entropy method of [94], and we reveal an astonishing relation between the monotonicity of the entropy and the detailed balance condition of an associated Markov chain.

3.1.1. The model

We consider the reaction-cross-diffusion equations

$$\partial_t u_i - \operatorname{div}\left(\sum_{j=1}^n A_{ij}(u)\nabla u_j\right) = f_i(u) \quad \text{in } \Omega, \ t > 0, \quad i = 1, \dots, n,$$
(3.1.1)

with no-flux boundary and initial conditions

$$\sum_{i=1}^{n} A_{ij}(u) \nabla u_j \cdot \nu = 0 \quad \text{on } \partial \Omega, \ t > 0, \quad u_i(\cdot, 0) = u_i^0 \quad \text{in } \Omega.$$
 (3.1.2)

Here, u_i models the density of the *i*th species, $u=(u_1,\ldots,u_n),\ \Omega\subset\mathbb{R}^d\ (d\geq 1)$ is a bounded domain with Lipschitz boundary, and ν is the exterior unit normal vector to $\partial\Omega$. The diffusion coefficients are given by

$$A_{ij}(u) = \delta_{ij}p_i(u) + u_i \frac{\partial p_i}{\partial u_j}(u), \quad p_i(u) = a_{i0} + \sum_{k=1}^n a_{ik}u_k^s, \quad i, j = 1, \dots, n,$$
 (3.1.3)

where a_{i0} , $a_{ij} \ge 0$ and s > 0. The functions p_i are the transition rates of the underlying random-walk model [95, 144]. The source terms f_i are of Lotka-Volterra type,

$$f_i(u) = u_i \left(b_{i0} - \sum_{j=1}^n b_{ij} u_j \right), \quad i = 1, \dots, n,$$
 (3.1.4)

and we suppose that b_{i0} , $b_{ij} \ge 0$ (competition case). Note that (3.1.1) can be written more compactly as

$$\partial_t u - \operatorname{div}(A(u)\nabla u) = f(u), \quad f(u) = (f_1(u), \dots, f_n(u)).$$

3.1.2. State of the art

From a mathematical viewpoint, the analysis of (3.1.1)-(3.1.2) is highly nontrivial since the diffusion matrix A(u) is neither symmetric nor generally positive definite. Although the maximum principle may be applied to prove the nonnegativity of the densities, it is generally not possible to show upper bounds. Moreover, there is no general regularity theory for diffusion systems, which makes the analysis very delicate. Equations (3.1.1) can be written in the form

$$\partial_t u_i - \Delta(u_i p_i(u)) = f_i(u), \tag{3.1.5}$$

which allows for the proof of an L^{2+s} estimate by the duality method [48, 120], but we will not exploit this method here, only in Chapter 7.

The case of n = 2 species and linear transition rates s = 1 corresponds to the original population model of Shigesada, Kawasaki, and Teramota [129],

$$\partial_t u_1 - \Delta \left(u_1 (a_{10} + a_{11} u_1 + a_{12} u_2) \right) = f_1(u),
\partial_t u_2 - \Delta \left(u_2 (a_{20} + a_{21} u_1 + a_{22} u_2) \right) = f_2(u).$$
(3.1.6)

The numbers a_{i0} are the diffusion coefficients, a_{ii} are the self-diffusion coefficients, and a_{ij} for $i \neq j$ are called the cross-diffusion coefficients. This model attracted a lot of attention in the mathematical literature. The first global existence result is due to Kim [99] who studied the equations in one space dimension, neglected self-diffusion, and assumed equal coefficients $(a_{ij} = 1)$. His result was extended to higher space dimensions in [56]. Most of the papers made restrictive structural assumptions, for instance supposing that the diffusion matrix is triangular $(a_{21} = 0)$, since this allows for the maximum principle in the second equation [2, 103, 106]. Another restriction is to suppose that the cross-diffusion coefficients are small, since in this situation the diffusion matrix becomes positive definite [56, 141].

Significant progress was made by Amann [2] who showed that a priori estimates in the $W^{1,p}$ norm with p > d are sufficient for the solutions to general quasilinear parabolic systems to exist globally in time, and he applied his result to the triangular case. The first global existence result without any restriction on the diffusion coefficients (except positivity) was achieved in [69] in one space dimension and in [37, 38] in several space dimensions. The results were extended to the whole space in [60]. The existence of global classical solutions was proved in, e.g., [104], under suitable conditions on the coefficients.

Nonlinear transition rates, but still for two species, were analyzed by Desvillettes and co-workers, assuming sublinear (0 < s < 1) [50] or superlinear rates (s > 1) and the weak cross-diffusion condition $((s-1)/(s+1))^2 a_{12} a_{21} \le a_{11} a_{22}$ [51]. Similar results, but under a slightly stronger weak cross-diffusion hypothesis, were proved in [94].

As already mentioned, there are very few results for more than two species. The existence of positive stationary solutions and the stability of the constant equilibrium was investigated

in [3, 127]. The existence of global weak solutions in one space dimension assuming a positive definite diffusion matrix was proved in [140], based on Amann's results. Using an entropy approach, the global existence of solutions was shown in [51] for three species under the condition $0 < s < 1/\sqrt{3}$ (which guarantees that $\det(A(u)) > 0$). To our knowledge, a global existence theorem under more general conditions seems to be not available in the literature. Here, we prove such a result and relate a structural condition on the coefficients a_{ij} with Onsager's principle of thermodynamics.

3.1.3. Key ideas

Before we state the main results, let us explain our strategy. The idea is to find a priori estimates by employing a Lyapunov functional approach with

$$\mathcal{H}[u] = \int_{\Omega} h(u)dx = \int_{\Omega} \sum_{i=1}^{n} \pi_i h_s(u_i) dx, \qquad (3.1.7)$$

where $\pi_i > 0$ are some numbers and

$$h_s(z) = \begin{cases} z(\log z - 1) + 1 & \text{for } s = 1, \\ \frac{z^s - sz}{s - 1} + 1 & \text{for } s \neq 1. \end{cases}$$
 (3.1.8)

Because of the connection of our method to nonequilibrium thermodynamics [95, Section 4.3], we refer to $\mathcal{H}[u]$ as an entropy and to h(u) as an entropy density. Introducing the so-called entropy variable $w = (w_1, \ldots, w_n)$ (called chemical potential in thermodynamics) by

$$w_i = \frac{\partial h}{\partial u_i}(u) = \begin{cases} \pi_i \log u_i & \text{for } s = 1, \\ \frac{s\pi_i}{s-1}(u_i^{s-1} - 1) & \text{for } s \neq 1, \end{cases}$$

equations (3.1.1) can be written as

$$\partial_t u(w) - \text{div}(B(w)\nabla w) = f(u(w)), \quad B(w) = A(u)H(u)^{-1},$$
 (3.1.9)

where $u(w) := (h')^{-1}(w)$ is the inverse transformation and H(u) = h''(u) is the Hessian of the entropy density. We claim that if f = 0 and B(w) or, equivalently, H(u)A(u) is positive semi-definite (we say that an arbitrary matrix $M \in \mathbb{R}^{n \times n}$ is positive (semi-) definite if $z^{\top}Mz > (\geq) 0$ for all $z \in \mathbb{R}^n$, $z \neq 0$.), $\mathcal{H}[u]$ is a Lyapunov functional along solutions to (3.1.1). Indeed, a (formal) computation shows that

$$\frac{d}{dt}\mathcal{H}[u] = -\int_{\Omega} \nabla w : B(w)\nabla w dx \le 0,$$

which implies that $t \mapsto \mathcal{H}[u(t)]$ is nonincreasing. The entropy method provides more than just the monotonicity of $\mathcal{H}[u]$. If, for instance, $z^{\top}H(u)A(u)z \geq \sum_{i=1}^{n} c_i u_i^{\alpha-2} z_i^2$ for some constants $\alpha > 0$, $c_i > 0$, it follows that

$$\frac{d}{dt}\mathcal{H}[u] + \frac{4}{\alpha^2} \int_{\Omega} \sum_{i=1}^n c_i |\nabla u_i^{\alpha/2}|^2 dx \le 0,$$

which yields gradient estimates for $u_i^{\alpha/2}$. This strategy was employed in many papers on cross-diffusion systems; see, e.g., [37, 38, 50, 60, 69, 94, 144]. In this work, we introduce two new ideas which we explain for the case s = 1 ($s \neq 1$ is studied below).

It is known that the entropy (3.1.7) with $\pi_i = 1$ is a Lyapunov functional for the twospecies model (3.1.6) with $f_1 = f_2 = 0$. This property is generally not satisfied for the corresponding *n*-species system. Our *first idea* is to introduce the numbers $\pi = (\pi_1, \ldots, \pi_n)$ in the entropy (3.1.7). It turns out that (3.1.7) is a Lyapunov functional and H(u)A(u) is symmetric and positive definite if

$$\pi_i a_{ij} = \pi_j a_{ji}$$
 for all $i, j = 1, \dots, n$. (3.1.10)

More precisely, this property is equivalent to the symmetry of H(u)A(u) (see Proposition 6.19). We recognize (3.1.10) as the detailed balance condition for the Markov chain associated to (a_{ij}) . The equivalence of the symmetry and the detailed balance condition is new but not surprising. In fact, the latter condition means that π is a reversible measure, and time-reversibility of a thermodynamic system is equivalent to the symmetry of the so-called Onsager matrix B(w), so symmetry and reversibility are related both from a mathematical and physical viewpoint. We detail these relations in Section 6.5.1. In Section 6.2.1, we derive a refined estimate for H(u)A(u) leading to

$$\frac{d}{dt}\mathcal{H}[u] + 4\int_{\Omega} \sum_{i=1}^{n} \pi_i a_{i0} |\nabla \sqrt{u_i}|^2 dx + 2\int_{\Omega} \sum_{i=1}^{n} \pi_i a_{ii} |\nabla u_i|^2 dx \le 0, \tag{3.1.11}$$

and thus giving an H^1 estimate for $\sqrt{u_i}$ (if $a_{i0} > 0$) and u_i (if $a_{ii} > 0$). This is the key estimate for the global existence result. (Below we also take into account the reaction terms (3.1.4).)

One may ask whether the detailed balance condition is necessary for the monotonicity of the entropy. It is not. We show that if self-diffusion dominates cross-diffusion in the sense

$$\eta_0 := \min_{i=1,\dots,n} \left(a_{ii} - \frac{s}{2(s+1)} \sum_{j=1}^n \left(\sqrt{a_{ij}} - \sqrt{a_{ji}} \right)^2 \right) > 0, \tag{3.1.12}$$

and detailed balance may be not satisfied, then the estimate leading to (3.1.11) still holds (with different constants), and global existence follows. (Throughout this work, we set $\pi_i = 1$ when detailed balance does not hold.) However, if conditions (3.1.10) or (3.1.12) are both not satisfied, there exist coefficients a_{ij} and initial data u^0 such that $t \mapsto \mathcal{H}[u(t)]$ is increasing on $[0, t_0]$ for some $t_0 > 0$; see Section 6.5.3. Numerical experiments (not shown) indicate that after the initial increase, the entropy decays and, in fact, it stays bounded for all time. We conjecture that the entropy is bounded for all time for all nonnegative coefficients and nonnegative initial data and that global existence of weak solutions holds for any (positive) coefficients a_{ij} .

Our results can be extended to nonlinear transition rates of type (3.1.3). One may choose more general terms $a_{ij}u_j^{s_j}$ with different exponents s_j but the results are easier to formulate if all exponents are equal. Coefficients with exponents $s \neq 1$ were also considered in [50, 51, 94] but in the two-species case only. We generalize these results to the multispecies case for any $n \geq 2$. The entropy method has to be adapted since the inverse of

 $h'_s(z) = (s/(s-1))(z^{s-1}-1)$ cannot be defined on \mathbb{R} and thus, $u(w) = (h')^{-1}(w)$ is not defined for all $w \in \mathbb{R}^n$. This issue can be overcome by regularization as in [50, 94]. In fact, we introduce

$$h_{\varepsilon}(u) = h(u) + \varepsilon \sum_{i=1}^{n} (u_i(\log u_i - 1) + 1).$$

Then $h'_{\varepsilon}: (0,\infty)^n \to \mathbb{R}^n$ can be inverted and $(h'_{\varepsilon})^{-1}: \mathbb{R}^n \to (0,\infty)^n$ is defined on \mathbb{R}^n . As a consequence, $u_i = (h'_{\varepsilon})^{-1}(w)_i$ is positive for any $w \in \mathbb{R}^n$ and even strongly positive if w varies in a compact subset of \mathbb{R}^n .

Unfortunately, the product $H_{\varepsilon}(u)A(u)$, where $H_{\varepsilon}(u)=h_{\varepsilon}''(u)$, is generally not positive definite and we need to approximate A(u). In contrast to the approximations suggested in [50, 94], we employ a non-diagonal matrix; see (6.2.5) below. More specifically, we introduce $A_{\varepsilon}(u)=A(u)+\varepsilon A^0(u)+\varepsilon^{\eta}A^1(u)$ with non-diagonal $A^0(u)$, diagonal $A^1(u)$, and $\eta \leq 1/2$ such that

$$z^{\top} H_{\varepsilon}(u) A_{\varepsilon}(u) z \ge z^{\top} H(u) A(u) z$$
 for all $z \in \mathbb{R}^n$.

The choice of the non-diagonal approximation satisfying this inequality is nontrivial, and this construction is our *second idea*.

3.1.4. Main results

First, we show that global existence of weak solutions holds for linear transition rates (s = 1). In the following, we set $Q_T = \Omega \times (0, T)$.

Theorem 3.1 (Global existence for linear transition rates). Let T > 0, s = 1 and $u^0 = (u_1^0, \ldots, u_n^0)$ be such that $u_i^0 \ge 0$ for $i = 1, \ldots, n$ and $\int_{\Omega} h(u^0) dx < \infty$. Let either detailed balance and $a_{ii} > 0$ for $i = 1, \ldots, n$; or (3.1.12) hold. Then there exists a weak solution $u = (u_1, \ldots, u_n)$ to (3.1.1)-(3.1.2) satisfying $u_i \ge 0$ in Ω , t > 0, and

$$u_i \in L^2(0, T; H^1(\Omega)), \quad u_i \in L^{\infty}(0, T; L^1(\Omega)),$$

 $u_i \in L^{2+2/d}(Q_T), \quad \partial_t u_i \in L^{q'}(0, T; W^{1,q}(\Omega)'), \quad i = 1, \dots, n,$

where q = 2(d+1) and q' = (2d+2)/(2d+1). The solution u solves (3.1.1) in the weak sense

$$\int_0^T \langle \partial_t u, \phi \rangle dt + \int_0^T \int_{\Omega} \nabla \phi : A(u) \nabla u dx dt = \int_0^T \int_{\Omega} f(u) \cdot \phi dx dt$$
 (3.1.13)

for all test functions $\phi \in L^q(0,T;W^{1,q}(\Omega))$, and the initial condition in (3.1.2) is satisfied in the sense of $W^{1,q}(\Omega)'$.

The theorem can be generalized to the case of vanishing self-diffusion, i.e. $a_{ii} = 0$ if detailed balance, $a_{i0} > 0$, and $b_{ii} > 0$ hold; see Remark 6.12.

Our second result is concerned with nonlinear transition rates $(s \neq 1)$. The entropy inequality yields the regularity $u_i \in L^{2s+2/d}(Q_T)$ which may not include L^2 for "small" exponents s < 1 and large dimensions d. For this reason, we need to suppose, in the

sublinear case, the lower bound s > 1 - 2/d and a weaker growth of the Lotka-Volterra terms:

$$f_i(u) = u_i \left(b_{i0} - \sum_{j=1}^n b_{ij} u_j^{\sigma} \right), \quad i = 1, \dots, n, \quad 0 \le \sigma < 2s - 1 + 2/d.$$
 (3.1.14)

The superlinear case (s > 1) is somehow easier than the sublinear one since the entropy inequality gives the higher regularity $u_i \in L^p(Q_T)$ with p > 2. On the other hand, we need a weak cross-diffusion constraint. More precisely, if detailed balance holds, we require that

$$\eta_1 := \min_{i=1,\dots,n} \left(a_{ii} - \frac{s-1}{s+1} \sum_{j=1, j \neq i}^n a_{ij} \right) > 0,$$
(3.1.15)

and if detailed balance does not hold, we suppose that

$$\eta_2 := \min_{i=1,\dots,n} \left(a_{ii} - \frac{1}{2(s+1)} \sum_{j=1, j \neq i} \left(s(a_{ij} + a_{ji}) - 2\sqrt{a_{ij}a_{ji}} \right) \right) > 0.$$
 (3.1.16)

For $m \geq 2$ and $1 \leq q \leq \infty$ we introduce the space

$$W_{\nu}^{m,q}(\Omega) = \{ \phi \in W^{m,q}(\Omega) : \nabla \phi \cdot \nu = 0 \text{ on } \partial \Omega \}.$$
 (3.1.17)

Theorem 3.2 (Global existence for nonlinear transition rates). Assume that T > 0, $s > \max\{0, 1 - 2/d\}$, and let the initial data u^0 be such that $u_i^0 \ge 0$ for i = 1, ..., n and $\int_{\Omega} h(u^0) dx < \infty$. If s < 1, we suppose that (3.1.14) and either detailed balance and $a_{ii} > 0$ for i = 1, ..., n; or (3.1.12) hold. If s > 1, we suppose that (3.1.4) and either detailed balance and (3.1.15) or (3.1.16) hold. Then there exist a number $2 \le q < \infty$ and a weak solution $u = (u_1, ..., u_n)$ to (3.1.1)-(3.1.2) satisfying $u_i \ge 0$ in Ω , t > 0, and

$$u_i^s \in L^2(0, T; H^1(\Omega)), \quad u_i \in L^{\infty}(0, T; L^{\max\{1, s\}}(\Omega)),$$

 $u_i \in L^{p(s)}(Q_T), \quad \partial_t u_i \in L^{q'}(0, T; W_{\nu}^{m, q}(\Omega)'), \quad i = 1, \dots, n,$

where $p(s) = 2s + (2/d) \max\{1, s\}$, 1/q + 1/q' = 1, and $m > \max\{1, d/2\}$. The solution u solves (6.1.1) in the "very weak" sense

$$\int_0^T \langle \partial_t u, \phi \rangle dt - \int_0^T \int_{\Omega} \sum_{i=1}^n u_i p_i(u) \Delta \phi_i dx dt = \int_0^T \int_{\Omega} f(u) \cdot \phi dx dt$$
 (3.1.18)

for all $\phi = (\phi_1, \dots, \phi_n) \in L^q(0, T; W^{m,q}_{\nu}(\Omega))$, and the initial condition holds in the sense of $W^{m,q}_{\nu}(\Omega)'$.

In the superlinear case, it can be shown that the solution satisfies (3.1.1) in the weak sense (3.1.13); see Remark 6.16. Moreover, for any $s > \max\{0, 1 - 2/d\}$, it is sufficient to consider test functions from $L^{\beta}(0,T;W^{2,\beta}_{\nu}(\Omega))$ with $1/\beta + 1/p(s) = 1$, and the initial condition holds in the sense of $W^{2,\beta}_{\nu}(\Omega)'$. We can generalize the theorem to the case of vanishing self-diffusion if either $s > \max\{1, d/2\}$; or 0 < s < 1, d = 1, and $\sigma < s + 1$ hold; see Remark 6.17.

The lower bound s > 1 - 2/d can be avoided if the regularity $u_i \in L^{2+s}(Q_T)$ holds, which is expected to follow from the duality method [48, 120]. Unfortunately, this method is not compatible with our approximation scheme (see (6.3.1) below). This issue can possibly be overcome by employing the scheme proposed in [51] which is specialized to diffusion systems like (3.1.5). In this work, however, we prefer to employ scheme (6.3.1).

3.2. From reaction diffusion to cross diffusion in the fast-reaction limit

The motivation of this work can be summarized in the following question: For which nonnegative functions $p_2(w), p_3(w) \ge 0$ can we find an entropy for the cross-diffusion system

$$\begin{cases} \partial_t w_2 - \Delta_x (p_2(w)w_2) = 0, \\ \partial_t w_3 - \Delta_x (p_3(w)w_3) = 0, \\ n(x) \cdot \nabla_x w_i = 0 \text{ on } \partial\Omega, \quad i = 2, 3 \\ w_i(0, \cdot) = w_{i0}(\cdot) \text{ in } \Omega, \quad i = 2, 3, \end{cases}$$
(3.2.1)

where $w(t,x) = (w_2(t,x), w_3(t,x))$ with x on a bounded domain $\Omega \subseteq \mathbb{R}^N$? This means that we are looking for a Lyapunov functional $H[w_2(t), w_3(t)]$ which is decreasing along the flow of the system (3.2.1)

$$\frac{d}{dt}H[w_2(t), w_3(t)] \le 0 \quad \forall t \ge 0.$$

The idea will be to start from a reaction-diffusion model with known entropy and by performing a fast-reaction limit to obtain the entropy for the cross-diffusion system (3.2.1).

3.2.1. The strategy of our work

We present the motivation of our work on a toy problem, which was also studied in [16]. Given the (non-standard) cross-diffusion system

$$\begin{cases}
\partial_t w_2 - \Delta_x \left(p_2(w) w_2 \right) = 0, & p_2(w) = \frac{d_1 \psi_3(w) + d_2}{\psi_3(w) + 1}, & d_i > 0, \\
\partial_t w_3 - \Delta_x \left(p_3(w) w_3 \right) = 0, & p_3(w) = \frac{d_1 \psi_2(w) + d_3}{\psi_2(w) + 1}, & d_i > 0,
\end{cases}$$
(3.2.2)

with

$$\psi_2(w) = \frac{\sqrt{(w_3 - w_2 + 1)^2 + 4w_2}}{2} - \frac{w_3 - w_2 + 1}{2}, \quad \psi_3(w) = \frac{w_3}{1 + \psi_2(w)}, \tag{3.2.3}$$

the question is whether we can find an entropy for this system. Indeed we can, namely we will show that

$$H[w(t)] = \int_{\Omega} (\psi_2(w)\psi_3(w)(\log \psi_2(w)\psi_3(w) - 1) + 1) dx +$$

$$+ \int_{\Omega} (\psi_2(w)(\log \psi_2(w) - 1) + 1) dx + \int_{\Omega} (\psi_3(w)(\log \psi_3(w) - 1) + 1) dx$$
 (3.2.4)

is an entropy for (3.2.2), (3.2.3). Our strategy will be to start from a simpler reaction-diffusion system with known entropy and to perform the fast reaction limit.

Let us start from a reaction-diffusion system for species A_i , i = 1, ..., 3, of the form

$$\begin{cases}
\partial_t u_1^{\varepsilon} - d_1 \, \Delta_x u_1^{\varepsilon} &= \frac{1}{\varepsilon} \left(u_2^{\varepsilon} u_3^{\varepsilon} - u_1^{\varepsilon} \right) \\
\partial_t u_2^{\varepsilon} - d_2 \, \Delta_x u_2^{\varepsilon} &= -\frac{1}{\varepsilon} \left(u_2^{\varepsilon} u_3^{\varepsilon} - u_1^{\varepsilon} \right) \\
\partial_t u_3^{\varepsilon} - d_3 \, \Delta_x u_3^{\varepsilon} &= -\frac{1}{\varepsilon} \left(u_2^{\varepsilon} u_3^{\varepsilon} - u_1^{\varepsilon} \right),
\end{cases} \tag{3.2.5}$$

with nonnegative initial conditions and no-flux boundary conditions, which describes the reversible chemical reaction

$$A_1 \rightleftharpoons A_2 + A_3$$

with fast reaction rate $1/\varepsilon > 0$, where the functions $u_i^{\varepsilon} := u_i^{\varepsilon}(t, x) \geq 0$ represent the concentrations of species A_i with diffusivities $d_i > 0$ on a bounded smooth open domain $\Omega \subseteq \mathbb{R}^N$. Since this system is coming from reversible reaction chemistry, it is not surprising that it has an entropy of the form

$$H[u^{\varepsilon}(t)] = \sum_{i=1}^{3} \int_{\Omega} \left(u_i^{\varepsilon} (\log u_i^{\varepsilon} - 1) + 1 \right) dx.$$
 (3.2.6)

Performing the fast reaction limit $\varepsilon \to 0$ leads to the system

$$\begin{cases} u_2 u_3 - u_1 = 0, \\ \partial_t (u_1 + u_2) - \Delta_x (d_1 u_1 + d_2 u_2) = 0, \\ \partial_t (u_1 + u_3) - \Delta_x (d_1 u_1 + d_3 u_3) = 0, \end{cases}$$
(3.2.7)

which can be rewritten by using the algebraic condition $u_1 = u_2u_3$ in the following way:

$$\begin{cases} \partial_t(u_2(1+u_3)) + \Delta_x \left[\frac{d_1u_3 + d_2}{u_3 + 1} (u_2(1+u_3)) \right] = 0, \\ \partial_t(u_3(1+u_2)) + \Delta_x \left[\frac{d_1u_2 + d_3}{u_2 + 1} (u_3(1+u_2)) \right] = 0. \end{cases}$$

Inverting the following change of variables

$$w_2 := u_1 + u_2$$
 and $w_3 := u_1 + u_3$ with $w = (w_2, w_3)$, (3.2.8)

yields the relation

$$u_2 = \psi_2(w_2, w_3), \quad u_3 = \psi_3(w_2, w_3),$$
 (3.2.9)

from which we obtain the desired cross-diffusion system (3.2.2)

$$\begin{cases} \partial_t w_2 - \Delta_x \left[\frac{d_1 \psi_3(w) + d_2}{\psi_3(w) + 1} w_2 \right] = 0, \\ \partial_t w_3 - \Delta_x \left[\frac{d_1 \psi_2(w) + d_3}{\psi_2(w) + 1} w_3 \right] = 0. \end{cases}$$

The key step is the inversion in (3.2.9), where $\psi = (\psi_2, \psi_3) := \varphi^{-1}$ is the inverse of the homeomorphism

$$\varphi: \begin{cases} \mathbb{R}_+^2 \to \mathbb{R}_+^2 \\ (u_2, u_3) \mapsto (\underbrace{u_2(1+u_3)}_{=:w_2}, \underbrace{u_3(1+u_2)}_{:=w_3}). \end{cases}$$

In this toy problem, this leads to the quadratic equation $u_2^2 + (w_3 - w_2 + 1)u_2 - w_2 = 0$, which has the unique nonnegative solution

$$u_2 = \psi_2(w) = \frac{\sqrt{(w_3 - w_2 + 1)^2 + 4w_2}}{2} - \frac{w_3 - w_2 + 1}{2}, \quad u_3 = \psi_3(w) = \frac{w_3}{1 + u_2}.$$

By performing the same limit $\varepsilon \to 0$ and the same variable transformation (3.2.8),(3.2.9) again, but now for the entropy (3.2.6), we obtain the limiting entropy (3.2.4).

3.2.2. Main results

Our goal will be to use the above sketched strategy on two generalized starting systems, namely (3.2.10) and (3.2.15), in order to obtain new entropies for new large classes of cross-diffusion systems. We briefly introduce both models considered and present the main results.

Model 1

The first model is obtained by starting from the following reaction-diffusion system:

$$\begin{cases}
\partial_t u_1^{\varepsilon} - \Delta_x f_1(u_1^{\varepsilon}) = -\frac{1}{\varepsilon} \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right), \\
\partial_t u_2^{\varepsilon} - \Delta_x f_2(u_2^{\varepsilon}) = +\frac{1}{\varepsilon} \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right), \\
\partial_t u_3^{\varepsilon} - \Delta_x f_3(u_3^{\varepsilon}) = +\frac{1}{\varepsilon} \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right),
\end{cases} (3.2.10)$$

where the entropy of this system reads

$$H[u^{\varepsilon}(t)] = \sum_{i=1}^{3} \int_{\Omega} h_i(u_i^{\varepsilon}) dx, \text{ with } h_i(u_i^{\varepsilon}) = \int_{c}^{u_i^{\varepsilon}} \log(q_i(z)) dz.$$
 (3.2.11)

The limiting cross-diffusion system for $\varepsilon \to 0$ has the form

$$\begin{cases} q_1(u_1) - q_2(u_2)q_3(u_3) = 0, \\ \partial_t (u_1 + u_2) - \Delta_x (f_1(u_1) + f_2(u_2)) = 0, \\ \partial_t (u_1 + u_3) - \Delta_x (f_1(u_1) + f_3(u_3)) = 0, \end{cases}$$
(3.2.12)

which can be transformed by inverting the following homeomorphism

$$\varphi: \begin{cases} \mathbb{R}_{+}^{2} \to \mathbb{R}_{+}^{2}, \\ (u_{2}, u_{3}) \mapsto (w_{2}, w_{3}), \end{cases}$$
 (3.2.13)

where $w_2 := u_2 + q_1^{-1}(q_2(u_2)q_3(u_3))$ and $w_3 := u_3 + q_1^{-1}(q_2(u_2)q_3(u_3))$, and its inverse

$$\psi := \varphi^{-1}, \quad \psi(w) = (\psi_2(w), \psi_3(w)).$$
 (3.2.14)

This leads to the following cross-diffusion system

$$\begin{cases} \partial_t w_2 - \Delta_x \left[\left(\frac{f_1(q_1^{-1}(q_2(\psi_2(w))q_3(\psi_3(w)))) + f_2(\psi_2(w))}{w_2} \right) w_2 \right] = 0, \\ \partial_t w_3 - \Delta_x \left[\left(\frac{f_1(q_1^{-1}(q_2(\psi_2(w))q_3(\psi_3(w)))) + f_3(\psi_3(w))}{w_3} \right) w_3 \right] = 0. \end{cases}$$

The entropy can be expressed in the new variables $w = (w_2, w_3)$ in the following way

$$H[w(t)] = \int_{\Omega} \int_{c}^{q_{1}^{-1}(q_{2}(\psi_{2}(w))q_{3}(\psi_{3}(w)))} \log(q_{1}(z)) dzdx + \int_{\Omega} \int_{c}^{\psi_{2}(w)} \log(q_{2}(z)) dzdx + \int_{\Omega} \int_{c}^{\psi_{3}(w)} \log(q_{3}(z)) dzdx.$$

The assumptions for the rigorous fast-reaction limit are the following:

- (B1) There exists a constant $C_1 > 0$, such that (for large x) it holds that $0 < f_i(x) < C_1 x^2 \log(x)$ for all x > 0, i = 1, 2, 3.
- (B2) There exist $\alpha > 0$ and $C_2 > 0$, such that

$$q_i(x) \ge C_2 x^{\alpha}$$
 for all $x \ge 0$, $i = 1, 2, 3$.

(B3) For some s > 1 and $1 \le p, p' \le \infty$ with 1/p + 1/p' = 1, there exist positive constants $C_3, C_4, C_5 > 0$ such that (for large x)

$$q_1^s(x) \le C_3 x f_1(x)$$
 for all $x \ge 0$,
 $q_2^{sp}(x) \le C_4 x f_2(x)$ for all $x \ge 0$,
 $q_2^{sp'}(x) \le C_5 x f_3(x)$ for all $x \ge 0$.

(B4) The functions $f_i: \mathbb{R}_+ \to \mathbb{R}_+$ and $q_i: \mathbb{R}_+^* \to \mathbb{R}_+$ are in C^1 with $f_i'(x) > 0$ and $q_i'(x) > 0$ for all $x \ge 0$, i = 1, 2, 3.

(B5) There exists a constant $C_6 > 0$, such that

$$q_i(x) \le C_6 x \left(f_i f_i' q_i' \right)(x)$$
 for all $x \ge 0$, $i = 1, 2, 3$.

(B6) For $a: \mathbb{R}^2_+ \to \mathbb{R}^2_+$ with $a(u_2, u_3) = (a_2(u_2, u_3), a_3(u_2, u_3))$ it holds that

$$a_i: \begin{cases} \mathbb{R}_+^2 \to \mathbb{R}_+ \\ (u_2, u_3) \mapsto u_i + q_1^{-1} (q_2(u_2)q_3(q_3)) / u_i \end{cases}$$

are continuous for i = 2, 3.

The main theorem for this model reads as follows:

Theorem 3.3. Let assumptions (B1)-(B6) for $f_i : \mathbb{R}_+ \to \mathbb{R}_+$ and $q_i : \mathbb{R}_+ \to \mathbb{R}_+$ hold. Let $\Omega \subseteq \mathbb{R}^N$ be a bounded regular open set of \mathbb{R}^N , and let for any $\varepsilon > 0$, $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}$ denote a weak solution of the reaction-diffusion system (3.2.10) with initial data $u_i^{\varepsilon}(0,x) = u_i^{in}(x) \in L^{\infty}$ for all $x \in \Omega$, i = 1, 2, 3.

Then the following holds: If $\varepsilon \to 0$, there exists a subsequence of $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}$ (which we still denote by $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}$), which converges to u_1, u_2, u_3 in $L^1_{loc}([0, \infty); L^1(\Omega))$.

Moreover, this limit is a weak solution of the cross-diffusion system (3.2.12) belonging to $L^1_{loc}([0,\infty);L^1(\Omega))$.

Model 2

The second model is a 'hybrid" reaction-cross diffusion system in four species, which is derived from the follwing reaction-diffusion system

$$\begin{cases} \partial_{t}u_{1}^{\varepsilon} - d_{1} \, \Delta_{x}u_{1}^{\varepsilon} &= \frac{1}{\varepsilon} \left(u_{2}^{\varepsilon} \, u_{3}^{\varepsilon} - u_{1}^{\varepsilon} \right) + \left(u_{4}^{\varepsilon} - u_{1}^{\varepsilon} \, u_{3}^{\varepsilon} \right) \\ \partial_{t}u_{2}^{\varepsilon} - d_{2} \, \Delta_{x}u_{2}^{\varepsilon} &= -\frac{1}{\varepsilon} \left(u_{2}^{\varepsilon} \, u_{3}^{\varepsilon} - u_{1}^{\varepsilon} \right) \\ \partial_{t}u_{3}^{\varepsilon} - d_{3} \, \Delta_{x}u_{3}^{\varepsilon} &= -\frac{1}{\varepsilon} \left(u_{2}^{\varepsilon} \, u_{3}^{\varepsilon} - u_{1}^{\varepsilon} \right) + \left(u_{4}^{\varepsilon} - u_{1}^{\varepsilon} \, u_{3}^{\varepsilon} \right) \\ \partial_{t}u_{4}^{\varepsilon} - d_{4} \, \Delta_{x}u_{4}^{\varepsilon} &= -\left(u_{4}^{\varepsilon} - u_{1}^{\varepsilon} \, u_{3}^{\varepsilon} \right) \end{cases}$$

$$(3.2.15)$$

with the entropy

$$H[u^{\varepsilon}(t)] = \sum_{i=1}^{4} \int_{\Omega} \left(u_i^{\varepsilon} (\log u_i^{\varepsilon} - 1) + 1 \right) dx.$$
 (3.2.16)

The limiting system for (w_2, w_3, u_4) with $w_2 = u_2(1 + u_3), w_3 = u_3(1 + u_2)$ reads

$$\begin{cases}
\partial_t w_2 - \Delta_x \left(\frac{d_1 \psi_3(w) + d_2}{\psi_3(w) + 1} w_2 \right) = \left(u_4 - \psi_2^2 \psi_3(w) \right), \\
\partial_t w_3 - \Delta_x \left(\frac{d_1 \psi_2(w) + d_3}{\psi_2(w) + 1} w_3 \right) = 2\left(u_4 - \psi_2^2 \psi_3(w) \right), \\
\partial_t u_4 - d_4 \Delta_x u_4 = -\left(u_4 - \psi_2^2 \psi_3(w) \right).
\end{cases}$$
(3.2.17)

and has the limiting entropy

$$H[w_2(t),_3(t), u_4(t)] = \int_{\Omega} (\psi_2(w)\psi_3(w)(\log(\psi_2(w)\psi_3(w)) - 1) + 1) dx$$
$$+ \sum_{i=2,3} \int_{\Omega} (\psi_i(w)(\log(\psi_i(w)) - 1) + 1) dx + \int_{\Omega} (u_4(\log(u_4) - 1) + 1) dx.$$

The main theorem concerning this model states the following.

Theorem 3.4. Let $\Omega \subseteq \mathbb{R}^N$ be a bounded regular open set of \mathbb{R}^N , and let for any $\varepsilon > 0$, $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}, u_4^{\varepsilon}$ denote a weak solution of the reaction-diffusion system (3.2.15) with initial data $u^{in} \log(u^{in}) \in L^2(\Omega)$.

Then the following holds: If $\varepsilon \to 0$, there exists a subsequence of u_1^{ε} , u_2^{ε} , u_3^{ε} , u_4^{ε} (which we still denote by u_1^{ε} , u_2^{ε} , u_3^{ε} , u_4^{ε}), which converges to u_1, u_2, u_3, u_4 in $L^1_{loc}([0, \infty); L^1(\Omega))$.

Moreover, this limit is a weak solution of the reaction-cross-diffusion system (3.2.17) belonging to $L^2_{loc}([0,\infty);L^2(\Omega))$.

The proof of this theorem is based on entropy and duality estimates.

3.2.3. State of the art

Our work generalizes the strategy initiated in [16, 79, 17] to perform a fast-reaction limit from reaction-diffusion systems coming from reversible reaction chemistry to cross-diffusion systems. The strategy presented on a toy problem in subsection 3.2.1 was already investigated in [79], and also generalized for more general starting systems including a model with fast and slow reactions similar to our second model (3.2.15). However, the results concerning the rigorous fast-rection limit of our first model (3.2.10) are (up to our knowledge) completely new, only existence of solutions for fixed $\varepsilon > 0$ has been already studied in [17, Theorem 3, p.21] for models including our first model. Moreover, also our results about how the limiting entropy looks like for the limiting cross-diffusion system are completely new and cannot be found in any of these works, neither the robustness analysis for the entropy performed in subsection 7.2.5.

Fast-reaction limits from reaction-diffusion systems have been well-known in the engineering literature for a long time, but the rigorous mathematical treatment started only very recently. For the presentation of the underlying mass action law of the reaction-diffusion model, see [72]. In [89], a fast reaction limit from a reaction-diffusion system modeling the chemical reaction $A + B \rightarrow C$ to a free boundary value problem was investigated. In [11], limits of ODE systems were studied with several fast-reaction and

additional slow-reaction processes. This was extended in [13] to the case of a simple fast reaction of the type $A \rightleftharpoons B$ from a reaction diffusion to a nonlinear diffusion equation using invariant sets. In [10], limits of reaction-diffusion systems with equal diffusion rates were investigated using also invariant sets. In [5] and [14], Quasi-Steady-State Approximations (QSSA) of the form $A + B \rightleftharpoons C \rightleftharpoons D + E$ with highly reactive intermediate C were investigated. For fast-reaction limits in irreversible chemistry, see [15] for a limit to a free boundary value problem, and for QSSA in irreversible chemistry, see [90] and [43]. We also want to mention the recent work [45], which focusses mainly on the modeling aspects of singular limits from reaction-diffusion equations. Moreover, we want to mention the works [92, 93, 88, 44, 53, 115] for a passage from reaction-diffusion equations to SKT-type cross-diffusion models of population dynamics. We also want to mention, that a rigorous fast-reaction limit starting from a kinetic system to a reaction-diffusion system was investigated in [31, 116].

 ${\it 3. \ Introduction \ to \ the \ diffusive \ multi-species \ models}$

Part II. The kinetic models

4. The linearized multi-species Boltzmann system

4.1. The model

In this chapter, we prove a new coercivity estimate on the spectral gap of the linearized Boltzmann collision operator for multiple species under assumptions on the collision kernels including hard and Maxwellian potentials under Grad's angular cut-off condition. We present two proofs: a non-constructive one, based on the decomposition of the collision operator into a compact and a coercive part, and a constructive one, which exploits the "cross-effects" coming from collisions between different species and which yields explicit constants. Furthermore, the essential spectra of the linearized collision operator and the linearized Boltzmann operator are calculated. Based on the spectral-gap estimate, the exponential convergence towards global equilibrium with explicit rate is shown for solutions to the linearized multi-species Boltzmann system on the torus. The convergence is achieved by hypocoercive effects between degenerately dissipative collision term and conservative transport term, proved by using the hypocoercivity method of Mouhot and Neumann.

For a presentation of the model, see subsection 2.1.1 and subsection 2.1.2. Here, we just repeat the main definitions. We are interested in the linearized multi-species Boltzmann system

$$\partial_t f_i + v \cdot \nabla_x f_i = L_i(f), \quad t > 0, \quad f_i(x, v, 0) = f_{I,i}(x, v), \quad (x, v) \in \mathbb{T}^3 \times \mathbb{R}^3,$$
 (4.1.1)

for $1 \leq i \leq n$, where $f = (f_1, \ldots, f_n)$ and the *i*th component of the linearized collision operator $L = (L_1, \ldots, L_n)$ is given by

$$L_i(f) = \sum_{j=1}^{n} L_{ij}(f_i, f_j), \quad 1 \le i \le n,$$

with

$$L_{ij}(f_i, f_j) = M_i^{-1/2} \left(Q_{ij}(M_i, M_j^{1/2} f_j) + Q_{ij}(M_i^{1/2} f_i, f_j) \right)$$

$$= \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} M_i^{1/2} M_j^* (h_i' + h_j'^* - h_i - h_j^*) dv^* d\sigma, \quad h_i := M_i^{-1/2} f_i.$$
 (4.1.2)

The global equilibrium has the form

$$M_i(v) = \frac{\rho_{\infty,i}}{(2\pi)^{3/2}} e^{-|v|^2/2}, \quad 1 \le i \le n,$$
 (4.1.3)

and the collision frequency reads

$$\nu_i(v) = \sum_{i=1}^n \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij}(|v^* - v|, \cos \theta) M_j^* dv^* d\sigma, \quad i = 1, \dots, n.$$
 (4.1.4)

4.1.1. Assumptions on the collision kernels

We repeat the assumptions on the collision kernels B_{ij} arising in (4.1.2), already introduced in subsection 2.1.4, and discuss them briefly.

(A1) The collision kernels satisfy

$$B_{ij}(|v-v^*|,\cos\vartheta) = B_{ji}(|v-v^*|,\cos\vartheta)$$
 for $1 \le i, j \le n$.

(A2) The collision kernels decompose in the kinetic part $\Phi_{ij} \geq 0$ and the angular part $b_{ij} \geq 0$ according to

$$B_{ij}(|v-v^*|,\cos\vartheta) = \Phi_{ij}(|v-v^*|)b_{ij}(\cos\vartheta), \quad 1 \le i, j \le n.$$

(A3) For the kinetic part, there exist constants C_1 , $C_2 > 0$, $\gamma \in [0, 1]$, and $\delta \in (0, 1)$ such that for all $1 \le i, j \le n$ and r > 0,

$$C_1 r^{\gamma} \le \Phi_{ij}(r) \le C_2(r + r^{-\delta}).$$

(A4) For the angular part, there exist constants C_3 , $C_4 > 0$ such that for all $1 \le i, j \le n$ and $\theta \in [0, \pi]$,

$$0 < b_{ij}(\cos \vartheta) \le C_3 |\sin \vartheta| |\cos \vartheta|, \quad b'_{ij}(\cos \vartheta) \le C_4.$$

Furthermore,

$$C^b := \min_{1 \le i \le n} \inf_{\sigma_1, \sigma_2 \in \mathbb{S}^2} \int_{\mathbb{S}^2} \min \left\{ b_{ii}(\sigma_1 \cdot \sigma_3), b_{ii}(\sigma_2 \cdot \sigma_3) \right\} d\sigma_3 > 0.$$

- (A5) For all $1 \leq i, j \leq n$, b_{ij} is even in [-1,1] and the mapping $v \mapsto \Phi'_{ij}(|v|)$ on \mathbb{R}^3 is locally integrable on \mathbb{R}^3 and bounded as $|v| \to \infty$.
- (A6) There exists $\beta > 0$ such that for all $1 \leq i, j \leq n, s > 0$, and $\sigma \in [-1, 1]$, we have $B_{ij}(s, \sigma) \leq \beta B_{ii}(s, \sigma)$.

Following [113], since the functions b_{ij} are integrable, we define

$$\ell^b := \min_{1, \le i, j \le n} \int_0^{\pi} b_{ij}(\cos \theta) \sin \theta d\theta > 0. \tag{4.1.5}$$

4.1.2. Discussion of the assumptions

The first hypothesis (A1) means that the collisions are micro-reversible. Assumption (A2) is satisfied, for instance, for collision kernels derived from interaction potentials behaving like inverse-power laws. The lower bound in hypothesis (A3) includes power-law functions $\Phi_{ij}(r) = r^{\gamma}$ with $\gamma > 0$ (hard potential) and $\gamma = 0$ (Maxwellian molecules). The assumption $\gamma \geq 0$ is crucial since the linearized collision operator in the mono-species case for soft potentials ($\gamma < 0$) with angular cut-off has no spectral gap [4]; however, degenerate spectral-gap estimates are possible [73, 110]. The upper bound in (A3) means that the kinetic part is of restricted growth for both small and large values of $|v-v^*|$. In hypothesis (A4), the upper bound for b_{ij} implies Grad's cut-off assumption. The positivity of C^b in Assumption (A4) is used in the constructive proof of the multi-species spectral-gap estimate (Theorem 4.2) via the mono-species spectral-gap estimate which depends on C^b ; see also the proofs of Theorem 1.1 in [4] and Theorem 6.1 in [110]. The positivity of C^b is satisfied for the main physical case of a collision kernel satisfying Grad's cut-off, i.e. for hard spheres with $B_{ij}(|v-v^*|,\cos\vartheta) = |v-v^*|$. Conditions (A1)-(A4) are also imposed in [4, 110, 111] for the linearized mono-species Boltzmann operator.

Assumption (A5) imposes technical conditions needed to verify the abstract hypotheses in [113]. More precisely, the evenness of b_{ij} is employed to show hypothesis (M2) (see section 4.5) and the properties on Φ'_{ij} are used to verify (4.5.2) in hypothesis (M1). The conditions on Φ'_{ij} are satisfied for hard and Maxwellian power-law potentials $\Phi_{ij}(r) = r^{\gamma}$ with exponent $\gamma \in [0, 1]$, for instance. Finally, condition (A6) states that the ratio of the off-diagonal and diagonal collision kernels can be bounded uniformly from above by a constant $\beta > 0$. This hypothesis will be needed for the explicit computation of the constants in Theorems 4.2 and 4.3. More precisely, (A6) allows us to estimate the mono-species part of the collision operator using the computation of [110]; see Lemma 4.10.

4.1.3. Notation and definitions

We call Dom(F) the domain of an operator F and Im(f) the image of a mapping f. We introduce the spaces $L_v^2 = L^2(\mathbb{R}^3; \mathbb{R}^n)$, $L_{x,v}^2 = L^2(\mathbb{T}^3 \times \mathbb{R}^3; \mathbb{R}^n)$, $H_{x,v}^1 = H^1(\mathbb{T}^3 \times \mathbb{R}^3; \mathbb{R}^n)$, and

$$\mathcal{H} = \left\{ f \in L_v^2 : \|f\|_{\mathcal{H}}^2 = \sum_{i=1}^n \int_{\mathbb{R}^3} f_i^2 \nu_i dv < \infty \right\},$$

$$\mathcal{D} = \left\{ f \in L_v^2 : \|f\|_{\mathcal{H}}^2 = \sum_{i=1}^n \int_{\mathbb{R}^3} f_i^2 \nu_i^2 dv < \infty \right\}.$$
(4.1.6)

Here, ν_i is the collision frequency, defined in (4.1.4). For Maxwellian modelcules $\Phi_{ij}(r) = \text{const.}$, the collision frequency is constant but for strictly hard potentials $\Phi_{ij}(r) = r^{\gamma}$ with $0 < \gamma \le 1$, ν_i is unbounded. In fact, it satisfies $\nu_0(1+|v|)^{\gamma} \le \nu_i(v) \le \nu_1(1+|v|)^{\gamma}$ for some constants $\nu_1 \ge \nu_0 > 0$ [113, p. 991]. In the physically most relevant case of hard spheres $(\gamma = 1, b_{ij} = 1)$, the collision frequency can be computed explicitly, see formula (2.13) in [34, Section 7.2]. For more properties of the collision frequencies, we refer to [32, Section III.3]. If the collision frequencies are bounded, $\mathcal{H} = L_v^2$. Generally, ν_i is unbounded and so, \mathcal{H} is a proper subset of L_v^2 . The norm on L_v^2 (and similarly for the other spaces) is defined

by

$$||f||_{L_v^2}^2 = \sum_{i=1}^n \int_{\mathbb{R}^3} f_i^2 dv$$
 for $f = (f_1, \dots, f_n) \in L_v^2$.

We distinguish the following linear operators. We define the operator $\Lambda = (\Lambda_1, \dots, \Lambda_n)$: $\mathrm{Dom}(\Lambda) \to L^2_v$ by

$$\Lambda_i(f) = \nu_i f_i, \quad i = 1, \dots, n,$$

where $\text{Dom}(\Lambda) = \{f \in L_v^2 : \Lambda f \in L_v^2\} = \mathcal{D}$. It is closed, densely defined, selfadjoint and, by Lemma 4.6 below, coercive. The linearized collision operator $L : \text{Dom}(L) \to L_v^2$, defined in (4.1.2), can be written as $L = K - \Lambda$, where $K := L + \Lambda$, or, more explicitly,

$$K_i(f) = \sum_{j=1}^n \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} M_i^{1/2} M_j^* (h_i' + h_j'^* - h_j^*) dv^* d\sigma, \quad i = 1, \dots, n.$$

It was shown in [19] that K is a compact operator in L_v^2 . Thus, $Dom(L) = Dom(\Lambda) = \mathcal{D}$ and L is closed and densely defined. Furthermore, L is nonpositive and selfadjoint on L_v^2 . We define the transport operator

$$T = v \cdot \nabla_x : \mathrm{Dom}(T) \to L_v^2$$

where $\text{Dom}(T) = \{ f \in L^2_{x,v} : v \cdot \nabla_x f \in L^2_{x,v} \}$. Finally, we consider the linearized Boltzmann operator

$$G = L - T : Dom(G) \rightarrow L_v^2$$

which is unbounded, closed, and densely defined with $Dom(G) = Dom(L) \cap Dom(T)$.

We denote by $\operatorname{Ker}(A)$ and $\operatorname{Ran}(A)$ the kernel and range of a linear operator A, respectively. Its resolvent set is denoted by $\rho(A)$ and its spectrum by $\sigma(A) = \mathbb{C} \setminus \rho(A)$. For a linear unbounded operator A with $\sigma(A) \subset (-\infty, 0]$, we say that A has a spectral gap when the distance between 0 and $\sigma(A) \setminus \{0\}$ is positive. Finally, the essential spectrum of A is defined as the set of all complex numbers $\lambda \in \mathbb{C}$ such that $A - \lambda I$ is not Fredholm, where I is the identity operator. We refer to section 4.3 for details regarding this definition.

4.1.4. Main results

In this subsection, we list the main results of this part of the thesis. For a discussion of these results, see section 2.1.5.

Theorem 4.1 (Essential spectrum of L and L-T). Let the collision kernels B_{ij} satisfy assumptions (A1)-(A4) and set

$$J = \bigcup_{i=1}^n \mathrm{Im}(\nu_i) \subset [\nu_0, \infty),$$

where $\nu_0 = \min_{i=1,\dots,n} \sup_{v \in \mathbb{R}^3} \nu_i(v) > 0$ (see Lemma 4.6). Then

$$\sigma_{ess}(L) = -J, \quad \sigma_{ess}(L-T) = \{\lambda \in \mathbb{C} : \Re(\lambda) \in -J\}.$$

The next theorem is the main result.

Theorem 4.2 (Explicit spectral-gap estimate). Let the collision kernels B_{ij} satisfy assumptions (A1)-(A4). Then there exists a constant $\lambda > 0$ such that

$$-(f, L(f))_{L_{v}^{2}} \ge \lambda \|f - \Pi^{L}(f)\|_{\mathcal{H}}^{2} \quad for \ all \ f \in \mathcal{D}, \tag{4.1.7}$$

where Π^L is the projection onto the null space Ker(L). If additionally hypothesis (A6) holds, the constant λ can be computed explicitly:

$$\lambda = \frac{\eta D^b}{8C_k}, \quad \eta = \min\left\{1, \frac{4C^mC_k}{16C_k + D^b}\right\},$$

where C^m , D^b , and C_k are defined in (4.4.8), (4.4.13), and (4.4.15), respectively.

Note that the constant C^m depends on the mono-species spectral-gap constant C^b via (4.4.8) below.

As a consequence of the spectral-gap estimate, we are able to prove the exponential decay of the solution f(t) to (4.1.1) to the global equilibrium with an explicit decay rate.

Theorem 4.3 (Convergence to equilibrium). Let the collision kernels B_{ij} satisfy assumptions (A1)-(A5) and let $f_I \in H^1_{x,v}$. Then the linearized Boltzmann operator G = L - T generates a strongly continuous semigroup $S_G(t)$ on $H^1_{x,v}$, which satisfies

$$||S_G(t)(I - \Pi^G)||_{H^1_{x,\eta}} \le Ce^{-\tau t}, \quad t \ge 0,$$
 (4.1.8)

for some constants C, $\tau > 0$. In particular, the solution $f(t) = S_G(t)f_I$ to (4.1.1) satisfies

$$||f(t) - f_{\infty}||_{H_{x,v}^{1}} \le Ce^{-\tau t} ||f_{I} - f_{\infty}||_{H_{x,v}^{1}}, \quad t \ge 0,$$
 (4.1.9)

where $f_{\infty} := \Pi^G(f_I)$ is the global equilibrium of (4.1.1). Moreover, under the additional assumption (A6) and lower bound in (A4), the constants C and τ depend only on the constants appearing in hypotheses (M1)-(M3) in section 4.5 and in particular on λ defined in Theorem 4.2.

4.2. Properties of the kinetic model

We show some properties of the linearized collision operator (4.1.2) and the collision frequencies (4.1.4). Let assumptions (A1)-(A4) hold. First we prove an H-theorem for (4.1.2).

Lemma 4.4 (*H*-theorem for the linearized collision operator). It holds that $(f, L(f))_{L_v^2} \leq 0$ for all $f \in \mathcal{D}$ and $(f, L(f))_{L_v^2} = 0$ if and only if $f \in \text{Ker}(L)$, where

$$\operatorname{Ker}(L) = \left\{ f \in L_v^2 : \exists \alpha_1, \dots, \alpha_n, \ e \in \mathbb{R}, \ u \in \mathbb{R}^3, \ \forall 1 \le i \le n, \right.$$
$$f_i = M_i^{1/2} (\alpha_i + u \cdot v + e|v|^2) \right\},$$

and M_i is given by (4.1.3).

The proof is similar to the mono-species case except that the elements of the null space of L depend on the total mean velocity u and total energy e instead of the individual velocities and energies. Therefore, we give a complete proof. We note that an H-theorem for the nonlinear Boltzmann operator for a mixture of reactive gases was proved in [52].

Proof. By the change of variables $(v, v^*) \mapsto (v^*, v)$ and $(v, v^*) \mapsto (v', v'^*)$ and the symmetry of B_{ij} (assumption (A1)), we can write for $f \in L_v^2$,

$$(f, L(f))_{L_v^2} = -\frac{1}{4} \sum_{i,j=1}^n \int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} M_i M_j^* (h_i' + h_j'^* - h_i - h_j^*)^2 dv^* dv d\sigma,$$

where we recall that $h_i = M_i^{-1/2} f_i$. This shows that $(f, L(f))_{L_v^2} \leq 0$ for all $f \in \mathcal{D}$. Moreover, $(f, L(f))_{L_v^2} = 0$ if and only if

$$h'_i + h'^*_i - h_i - h^*_i = 0$$
 for all $(v, v^*) \in \mathbb{R}^3 \times \mathbb{R}^3$, $1 \le i, j \le n$. (4.2.1)

It is shown in [34, pp. 36-42] that (4.2.1) for i = j implies that h_i has the form $h_i(v) = \alpha_i + u_i \cdot v + e_i |v|^2$ for suitable constants α_i , $e_i \in \mathbb{R}$ and $u_i \in \mathbb{R}^3$. Inserting this expression into (4.2.1) leads to

$$u_i \cdot (v' - v) + u_j \cdot (v'^* - v^*) + e_i(|v'|^2 - |v|^2) + e_i(|v'^*|^2 - |v^*|^2) = 0$$

$$(4.2.2)$$

for $1 \le i, j \le n$. We consider the particular type of collisions with $v' = v^*$, $v'^* = v$, and |v| = |v'|. For such collisions, $\sigma = (v^* - v)/|v^* - v|$. Then the above equation becomes

$$(u_i - u_j) \cdot (v' - v) = 0, \quad 1 \le i, j \le n.$$

By rotating the velocities v, v' in all possible ways, we deduce that $(u_i - u_j) \cdot w = 0$ for all $w \in \mathbb{R}^3$ and thus, $u_i = u_j$ for all $1 \le i, j \le n$. We set $u := u_1$. This fact, together with the conservation of momentum $v' - v + v'^* - v^* = 0$, implies that (4.2.2) becomes

$$e_i(|v'|^2 - |v|^2) + e_j(|v'^*|^2 - |v^*|^2) = 0, \quad 1 \le i, j \le n.$$

Taking into account the conservation of energy $|v'|^2 - |v|^2 + |v'^*|^2 - |v^*|^2 = 0$, we infer that $(e_i - e_j)(|v'|^2 - |v|^2) = 0$ and consequently, $e_i = e_j$ for all $1 \le i, j \le n$. Set $e := e_1$. We have shown that $(f, L(f))_{L_v^2} = 0$ if and only if there exist $\alpha_1, \ldots, \alpha_n$, $e \in \mathbb{R}$ and $u \in \mathbb{R}^3$ such that for $1 \le i \le n$, $f_i(v) = M_i^{1/2}(\alpha_i + u \cdot v + e|v|^2)$. These functions clearly belong to $\operatorname{Ker}(L)$, which finishes the proof.

The next result is concerned with the stationary solutions of (4.1.1).

Lemma 4.5. The global equilibrium $f_{\infty} = (f_{\infty,1}, \dots, f_{\infty,n})$ of (4.1.1), i.e. the unique stationary solution, is given by

$$f_{\infty,i}(v) = M_i^{1/2}(\alpha_i + u \cdot v + e|v|^2), \quad 1 \le i \le n,$$

where α_i , $e \in \mathbb{R}$ and $u \in \mathbb{R}^3$ are uniquely determined by the global conservation laws of mass, momentum, and energy, i.e. by the equations

$$\int_{\mathbb{D}^3} M_i^{1/2} (f_{\infty,i} - f_{I,i}) \psi(v) dv = 0, \quad 1 \le i \le n,$$

for $\psi(v) = 1, v_1, v_2, v_3, |v|^2$, where $f_{I,i}$ are the initial data.

Proof. First, we claim that $\operatorname{Ker}(G) = \operatorname{Ker}(L) \cap \operatorname{Ker}(T)$, where G = L - T and $T = v \cdot \nabla_x$ are considered on $\mathbb{T}^3 \times \mathbb{R}^3$. The inclusion $\operatorname{Ker}(L) \cap \operatorname{Ker}(T) \subset \operatorname{Ker}(G)$ being trivial, let $f \in \operatorname{Ker}(G)$. Then, using the skew-symmetry of T,

$$0 = (f, G(f))_{L^2_{x,v}} = (f, L(f))_{L^2_{x,v}} - (f, T(f))_{L^2_{x,v}} = (f, L(f))_{L^2_{x,v}}.$$

Lemma 4.4 shows that $f \in \operatorname{Ker}(L)$. But this implies that T(f) = L(f) - G(f) = 0 and hence $f \in \operatorname{Ker}(T)$. This shows the claim. Let f_{∞} be a stationary solution. Then $f_{\infty} \in \operatorname{Ker}(G)$ and by our claim, $f \in \operatorname{Ker}(L) \cap \operatorname{Ker}(T)$. Since $\operatorname{Ker}(T) = \{f \in L^2_{x,v} : \nabla_x f = 0\}$ [25, Lemma B.2], f_{∞} does not depend on x. Because of $f_{\infty} \in \operatorname{Ker}(L)$, Lemma 4.4 shows the result. \square

Finally, we prove that the collision frequencies (4.1.4) are strictly positive with bounded derivative.

Lemma 4.6. Let Assumptions (A2)-(A4) hold. The collision frequencies (4.1.4) satisfy

$$\min_{1 \le i \le n} \inf_{v \in \mathbb{R}^3} \nu_i(v) \ge \nu_0 := 2^{3\gamma/2} \frac{C_1 \ell^b \rho_\infty}{\sqrt{\pi}} \Gamma\left(\frac{\gamma+3}{2}\right) > 0, \tag{4.2.3}$$

where $C_1 > 0$ is given by assumption (A3), $\ell^b > 0$ is defined in (4.1.5), $\rho_{\infty} := \sum_{j=1}^n \rho_{j,\infty}$ (see (2.1.4)), and Γ is the Gamma function. Furthermore, if additionally (A5) holds, then $\nabla_v \nu_i \in L_v^{\infty}(\mathbb{R}^3)$, implying that $|\nu_i(v)| \leq C_{\nu}(1+|v|)$ for some $C_{\nu} > 0$ and for all $v \in \mathbb{R}^3$, $i = 1, \ldots, n$.

Proof. The decomposition of B_{ij} , according to assumption (A2), implies that

$$\nu_i(v) = \sum_{i=1}^n \int_{\mathbb{R}^3} \Phi_{ij}(|v - v^*|) M_j^* dv^* \int_{\mathbb{S}^2} b_{ij}(\cos \vartheta) d\sigma.$$

The integral

$$c_{ij} := \int_{\mathbb{S}^2} b_{ij}(\cos \theta) d\sigma = 2\pi \int_0^{\pi} b_{ij}(\cos \theta) \sin \theta d\theta$$

does not depend on v or v^* . We conclude from (A2)-(A4) that

$$\nu_{i}(v) = (2\pi)^{-3/2} \sum_{j=1}^{n} c_{ij} \rho_{\infty,j} \int_{\mathbb{R}^{3}} \Phi_{ij}(|v-v^{*}|) e^{-|v^{*}|^{2}/2} dv^{*}$$

$$\geq \frac{C_{1} \ell^{b} \rho_{\infty}}{(2\pi)^{3/2}} \int_{\mathbb{R}^{3}} |v-v^{*}|^{\gamma} e^{-|v^{*}|^{2}/2} dv^{*}.$$
(4.2.4)

Observe that the function

$$G(v) := \int_{\mathbb{R}^3} |v - v^*|^{\gamma} e^{-|v^*|^2/2} dv^*$$

is uniformly positive since the transformation $v^* \mapsto -v^*$ and the elementary inequality

$$|v - v^*|^{\gamma} + |v + v^*|^{\gamma} \ge |(v - v^*) + (v + v^*)|^{\gamma} = 2^{\gamma} |v^*|^{\gamma}$$

for $\gamma \in [0,1]$ lead to

$$G(v) = \frac{1}{2} \int_{\mathbb{R}^3} (|v - v^*|^{\gamma} + |v + v^*|^{\gamma}) e^{-|v^*|^2/2} dv^* \ge 2^{\gamma - 1} \int_{\mathbb{R}^3} |v^*|^{\gamma} e^{-|v^*|^2/2} dv^* = 2^{\gamma - 1} G(0).$$

Actually, using spherical coordinates and the change of unknowns $s = r^2/2$,

$$G(0) = 4\pi \int_0^\infty r^{\gamma+2} e^{-r^2/2} dr = 2^{(\gamma+5)/2} \pi \int_0^\infty s^{(\gamma+1)/2} e^{-s} ds = 2^{(\gamma+5)/2} \pi \Gamma\left(\frac{\gamma+3}{2}\right).$$

Inserting the above estimate on G(v) into (4.2.4) shows (4.2.3).

It remains to prove that $\nabla_v \nu_i \in L_v^{\infty}(\mathbb{R}^3)$. To this end, we compute

$$|\nabla \nu_{i}(v)| = (2\pi)^{-3/2} \left| \sum_{j=1}^{n} c_{ij} \rho_{\infty,j} \int_{\mathbb{R}^{3}} \Phi'_{ij}(|v-v^{*}|) \cdot \frac{v-v^{*}}{|v-v^{*}|} e^{-|v^{*}|^{2}/2} dv^{*} \right|$$

$$\leq (2\pi)^{-3/2} \sum_{j=1}^{n} c_{ij} \rho_{\infty,j} \int_{\mathbb{R}^{3}} |\Phi'_{ij}(|v-v^{*}|)| e^{-|v^{*}|^{2}/2} dv^{*}. \tag{4.2.5}$$

For given R > 0, we decompose

$$|\Phi'_{ij}(|v|)| = |\Phi'_{ij}(|v|)|\chi_{\{|v| < R\}}(v) + |\Phi'_{ij}(|v|)|\chi_{\{|v| \ge R\}}(v).$$

Assumption (A5) means that there exists R > 0 such that

$$|\Phi'_{ij}(|\cdot|)|\chi_{\{|\cdot|< R\}} \in L^1_v(\mathbb{R}^3)$$
 and $|\Phi'_{ij}(|\cdot|)|\chi_{\{|\cdot|\geq R\}} \in L^\infty_v(\mathbb{R}^3)$.

Thus, the right-hand side of (4.2.5) is bounded since it can be written as the sum of two terms, each of which is the convolution of an L^1 and an L^{∞} function. This shows that $\nabla_v \nu_i \in L_v^{\infty}(\mathbb{R}^3)$.

Remark 4.7. We observe that ν_i is generally not bounded since the kinetic part $\Phi_{ij}(r)$ may grow like r as $r \to \infty$. It is possible to show that ν_i is bounded if Φ_{ij} is bounded. The unboundedness of ν_i implies that the spaces L^2_v and \mathcal{H} are not isomorphic.

4.3. Geometric properties of the spectrum

In this section, we prove Theorem 4.1 and give a non-constructive proof of the spectral-gap estimate (4.1.7) in Theorem 4.2 by using arguments from functional analysis.

First, we study the essential spectrum of L and L-T. There exist several definitions of the essential spectrum of a linear operator. Given a linear, closed and densely defined operator $A: \text{Dom}(A) \subset X \to X$ on a Banach space X, we define

$$\sigma_{\text{ess}}(A) = \{ \lambda \in \mathbb{C} : A - \lambda I \text{ is not Fredholm} \}.$$

We recall that a linear, closed, and densely defined operator A is Fredholm if its range Ran(A) is closed and both its kernel and cokernel are finite-dimensional. For other definitions of the essential spectrum, we refer to [84]. The essential spectrum is closed and

conserved under compact perturbations, i.e., the bounded operators A and B have the same essential spectrum if A - B is compact (Weyl's theorem; see [84, Theorem S].

If X is a Hilbert space and A is selfadjoint, it holds $\sigma_{\mathrm{ess}}(A) \subset \mathbb{R}$ and for given $\lambda \in \mathbb{R}$, we have $\lambda \in \sigma_{\mathrm{ess}}(A)$ if and only if $A - \lambda I$ is not closed or the kernel of $A - \lambda I$ is infinite dimensional. (This follows from the fact that $\mathrm{Ran}(A - \lambda I)^{\perp} = \mathrm{Ker}(A - \lambda I)$ for closed, selfadjoint operators A [97, Chap. V.3.1].) Moreover, Weyl's criterion holds [133, Lemma 5.17]: $\lambda \in \sigma_{\mathrm{ess}}(A)$ if and only if $A - \lambda I$ admits a singular sequence, i.e. a sequence $(f_k) \subset \mathrm{Dom}(A)$ such that (i) $||f_k||_X = 1$ for all $k \in \mathbb{N}$; (ii) $||(A - \lambda I)f_k||_X \to 0$ as $k \to \infty$; and (iii) (f_k) has no convergent subsequences in X.

We decompose L as $L = K - \Lambda$, where $K = (K_1, \ldots, K_n)$, $\Lambda = (\Lambda_1, \ldots, \Lambda_n)$, and

$$K_{i}(f) = \sum_{j=1}^{n} \int_{\mathbb{R}^{3} \times \mathbb{S}^{2}} B_{ij} M_{i}^{1/2} M_{j}^{*} (h_{i}' + h_{j}'^{*} - h_{j}^{*}) dv^{*} d\sigma,$$

$$\Lambda_{i}(f) = \nu_{i} f_{i}, \quad 1 \leq i \leq n,$$
(4.3.1)

and the collision frequencies ν_i are defined in (4.1.4). We recall from Lemma 4.6 that they satisfy $\nu_i(v) \ge \nu_0 > 0$ and $|\nu_i(v)| \le C_{\nu}(1+|v|)$ for all i = 1, ..., n and $v \in \mathbb{R}^3$.

Proof of Theorem 4.1. Since K is compact on $X=L_v^2$ [19, Prop. 2], it follows that $\sigma_{\mathrm{ess}}(L)=\sigma_{\mathrm{ess}}(-\Lambda)=-\sigma_{\mathrm{ess}}(\Lambda)$. Thus, we will first study the essential spectrum of Λ . The proof is divided into several steps. Recall that $J=\cup_{i=1}^n\mathrm{Im}(\nu_i)\subset [\nu_0,\infty)$.

Step 1: $\overline{J} \subset \sigma_{ess}(\Lambda)$. Let $\lambda \in J$. There exists $j \in \{1, ..., n\}$ and $\widehat{v} \in \mathbb{R}^3$ such that $\lambda = \nu_j(\widehat{v})$. We define the sequence $(f_k) \subset \mathcal{D}$ by

$$f_{k,i}(v) = (2\pi\sigma_k)^{-3/4} \exp\left(-\frac{|v-\hat{v}|^2}{4\sigma_k}\right)$$
 if $i = j$, $f_{k,i}(v) = 0$ if $i \neq j$,

where $\sigma_k = 1/k$, $k \in \mathbb{N}$. Clearly, condition (i) for the singular sequence is satisfied. Furthermore,

$$\|(\Lambda - \lambda I)f_k\|_{L_v^2}^2 = \sum_{i=1}^n \int_{\mathbb{R}^3} (\nu_i(v) - \lambda)^2 f_{k,i}(v)^2 dv$$

= $(2\pi\sigma_k)^{-3/2} \int_{\mathbb{R}^3} (\nu_i(v) - \nu_j(\widehat{v}))^2 \exp\left(-\frac{|v - \widehat{v}|^2}{2\sigma_k}\right) dv.$

The limit of a sequence of Gaussians with variance tending to zero converges to the delta distribution $\delta_{\widehat{v}}$ (in the sense of distributions), which means that

$$(2\pi\sigma_k)^{-3/2} \int_{\mathbb{R}^3} u(v) \exp\left(-\frac{|v-\widehat{v}|^2}{2\sigma_k}\right) dv \to u(\widehat{v}) \quad \text{as } k \to \infty$$

for all functions $u \in C^0(\mathbb{R}^3)$ with polynomial growth at infinity. Since $|\nu_i(v)| \leq C_{\nu}(1+|v|)$, this condition is satisfied and we conclude that $\|(\Lambda - \lambda I)f_k\|_{L^2_v} \to 0$ as $k \to \infty$, showing that condition (ii) holds.

Let us assume by contradiction that condition (iii) does not hold. Then there exists a subsequence $(f_{k_{\ell}})$ of (f_k) that converges in L_v^2 to some function $f \in L_v^2$. As a consequence,

 $|f_{k_{\ell}}|^2 \to |f|^2$ in L_v^1 as $\ell \to \infty$. In particular, $f \in L_v^2$. However, the distributional limit $|f_{k_{\ell}}|^2 \to \delta_{\widehat{v}}$ and the uniqueness of the limit imply that $\delta_{\widehat{v}} = |f|^2 \in L_v^1$, which is absurd. Thus, condition (iii) holds, and we infer that $\lambda \in \sigma_{\mathrm{ess}}(\Lambda)$. Then, since $\sigma_{\mathrm{ess}}(\Lambda)$ is closed, $\overline{J} \subset \sigma_{\mathrm{ess}}(\Lambda)$.

Step 2: $\sigma_{ess}(\Lambda) \subset \overline{J}$. Let $\lambda \in \mathbb{R} \setminus \overline{J}$. Then there exists a constant c > 0 such that for all $v \in \mathbb{R}^3$ and $i = 1, \ldots, n, |\nu_i(v) - \lambda| \ge c$. If $(f_k) \subset \mathcal{D}$ with $||f_k||_{L^2_x} = 1$ for all $k \in \mathbb{N}$, we have

$$\|(\Lambda - \lambda I)f_k\|_{L_v^2}^2 = \sum_{i=1}^n \int_{\mathbb{R}^3} (\nu_i(v) - \lambda)^2 f_{k,i}(v)^2 dv \ge c^2 \sum_{i=1}^n \int_{\mathbb{R}^3} f_{k,i}(v)^2 dv = c^2 > 0$$

for all $k \in \mathbb{N}$. Thus, condition (ii) cannot hold which implies that $\lambda \notin \sigma_{\text{ess}}(\Lambda)$.

Steps 1 and 2 imply that $\sigma_{\text{ess}}(\Lambda) = \overline{J}$.

Step 3: $\{\lambda \in \mathbb{C} : \Re(\lambda) \in J\} \subset \sigma_{ess}(\Lambda + T)$. Let $\lambda \in \mathbb{C}$ be such that $\Re(\lambda) \in J$. It follows from Step 1 that $\Re(\lambda) \in \sigma_{ess}(\Lambda)$. Since Λ is selfadjoint on the Hilbert space L_v^2 , $\Lambda - \Re(\lambda)I$ is not closed or the kernel of $\Lambda - \Re(\lambda)I$ is infinite dimensional. As the operator $\Lambda - \Re(\lambda)I$ is closed, its kernel must be infinite dimensional. Therefore, there exists a sequence $(f_k) \subset L_v^2$ such that $\Lambda(f_k) - \Re(\lambda)f_k = 0$ and $(f_k, f_\ell)_{L_v^2} = \delta_{k\ell}$ for $k, \ell \in \mathbb{N}$. Let us define $\phi(x,v) = \exp(i\Im(\lambda)x \cdot v/|v|^2)$ and $g_k = \phi f_k \in L_{x,v}^2$. Since $|\phi| = 1$, we have

$$(g_k, g_\ell)_{L^2_{x,v}} = (f_k, f_\ell)_{L^2_v} = \delta_{k\ell} \text{ for } k, \ell \in \mathbb{N}.$$
 (4.3.2)

Furthermore, $\phi \in \text{Dom}(T)$ and $T(\phi) = i\Im(\lambda)\phi$ for $v \neq 0$, and thus,

$$(\Lambda + T - \lambda I)g_k = \phi(\Lambda - \Re(\lambda)I)f_k + f_k(T - i\Im(\lambda)I)\phi = 0,$$

which shows that $g_k \in \text{Ker}(\Lambda + T - \lambda I)$ for $k \in \mathbb{N}$. This fact, together with relation (4.3.2), implies that $\text{Ker}(\Lambda + T - \lambda I)$ is infinite dimensional. As a consequence, $\Lambda + T - \lambda I$ is not Fredholm and $\lambda \in \sigma_{\text{ess}}(\Lambda + T)$, which proves the claim.

Step 4: $\{\lambda \in \mathbb{C} : \Re(\lambda) \notin \overline{J}\} \subset \rho(\Lambda + T)$. Clearly, this gives

$$\sigma_{\mathrm{ess}}(\Lambda + T) \subset \sigma(\Lambda + T) \subset \{\lambda \in \mathbb{C} : \Re(\lambda) \in \overline{J}\}.$$

Let $\lambda \in \mathbb{C}$ be such that $\Re(\lambda) \in \mathbb{R} \setminus \overline{J}$. We show first that $\operatorname{Ker}(\Lambda + T - \lambda I) = \{0\}$. We assume by contradiction that there exists $f \in \operatorname{Dom}(\Lambda + T)$ satisfying $||f||_{L^2_{x,v}} > 0$ and $(\Lambda + T - \lambda I)f = 0$. In particular, there is an index $\ell \in \{1, \ldots, n\}$ such that $\int_{\mathbb{R}^3} \int_{\mathbb{T}^3} f_\ell^2 dx dv > 0$. Then, multiplying $\nu_\ell f_\ell + T(f_\ell) = \lambda f_\ell$ by \overline{f}_ℓ (the complex conjugate of f_ℓ) and integrating in $\mathbb{T}^3 \times \mathbb{R}^3$, we obtain

$$\int_{\mathbb{R}^3} \int_{\mathbb{T}^3} \nu_{\ell} |f_{\ell}|^2 dx dv + \int_{\mathbb{R}^3} \int_{\mathbb{T}^3} \overline{f}_{\ell} v \cdot \nabla_x f_{\ell} dx dv = \lambda \int_{\mathbb{R}^3} \int_{\mathbb{T}^3} |f_{\ell}|^2 dx dv. \tag{4.3.3}$$

By the divergence theorem, the real part of the second integral vanishes,

$$2\Re \int_{\mathbb{R}^3} \int_{\mathbb{T}^3} \overline{f}_{\ell} v \cdot \nabla_x f_{\ell} dx dv = \int_{\mathbb{R}^3} \int_{\mathbb{T}^3} \left(\overline{f}_{\ell} v \cdot \nabla_x f_{\ell} + f_{\ell} v \cdot \nabla_x \overline{f}_{\ell} \right) dx dv$$

$$= \int_{\mathbb{R}^3} \int_{\mathbb{T}^3} v \cdot \nabla_x |f_{\ell}|^2 dx dv = 0.$$

$$(4.3.4)$$

Then, taking the real part of (4.3.3), we infer that

$$\Re(\lambda) = \frac{\int_{\mathbb{R}^3} \int_{\mathbb{T}^3} \nu_{\ell} |f_{\ell}|^2 dx dv}{\int_{\mathbb{R}^3} \int_{\mathbb{T}^3} |f_{\ell}|^2 dx dv}.$$

Consequently, $\inf_{\mathbb{R}^3} \nu_{\ell} \leq \Re(\lambda) \leq \sup_{\mathbb{R}^3} \nu_{\ell}$ and, thanks to the continuity of ν_{ℓ} , $\Re(\lambda) \in \overline{\operatorname{Im}(\nu_{\ell})} \subset \overline{J}$, which is a contradiction. Thus, $\operatorname{Ker}(\Lambda + T - \lambda I) = \{0\}$. Similarly, we can show that $\operatorname{Ker}((\Lambda + T - \lambda I)^*) = \operatorname{Ker}(\Lambda + T^* - \overline{\lambda}I) = \{0\}$ as well.

The operator L-T is closed [135, Theorems 2.2.1 and 2.2.2]. Thus, the boundedness of the compact operator K and the stability of closedness under bounded perturbations [97, Chap. III, Problem 5.6] imply that $\Lambda + T = K - (L - T)$ is closed (and also densely defined). Hence, $\operatorname{Ran}(\Lambda + T - \lambda I)^{\perp} = \operatorname{Ker}((\Lambda + T - \lambda I)^*) = \{0\}$, meaning that $\Lambda + T - \lambda I$ is invertible. If $f \in L^2_{x,v}$ is given, there exists $u \in \operatorname{Dom}(\Lambda + T)$ such that $(\Lambda + T - \lambda I)u = f$, which translates into

$$(\nu_j - \Re(\lambda))u_j + (T - i\Im(\lambda))u_j = f, \quad j = 1, \dots, n.$$
 (4.3.5)

We point out that, since ν_j is continuous, $\operatorname{Im}(\nu_j)$ is an interval (or a point, in case that ν_j is constant). This fact and the assumption $\Re(\lambda) \notin \overline{J}$ imply that either $\nu_j - \Re(\lambda) > 0$ in \mathbb{R}^3 or $\nu_j - \Re(\lambda) < 0$ in \mathbb{R}^3 . This means that the sign s_j of $\nu_j - \Re(\lambda)$ is constant in \mathbb{R}^3 , for $j = 1, \ldots, n$. By multiplying (4.3.5) by $s_j \overline{u}_j$, integrating over $\mathbb{T}^3 \times \mathbb{R}^3$, taking the real part, and summing over $j = 1, \ldots, n$, we find that

$$\sum_{j=1}^{n} \int_{\mathbb{R}^3} \int_{\mathbb{T}^3} |\nu_j - \Re(\lambda)| |u_j|^2 dx dv = \frac{1}{2} \sum_{j=1}^{n} s_j \int_{\mathbb{R}^3} \int_{\mathbb{T}^3} \left(\overline{u}_j f_j + u_j \overline{f}_j \right) dx dv.$$

The (real part of the) second term in (4.3.5) vanishes after integration; see (4.3.4). Since $\lambda \in \mathbb{R} \setminus \overline{J}$, by definition of J, there exists $c_{\lambda} > 0$ such that $|\nu_{j} - \Re(\lambda)| \geq c_{\lambda}$ in \mathbb{R}^{3} for all $j = 1, \ldots, n$. Then the Cauchy-Schwarz inequality shows that $||u||_{L_{x,v}^{2}} \leq c_{\lambda}^{-1}||f||_{L_{x,v}^{2}}$. This means that $(\Lambda + T - \lambda I)^{-1}$ is bounded, so $\lambda \in \rho(\Lambda + T)$.

Steps 3 and 4 show that $\sigma_{\text{ess}}(\Lambda + T) = \{\lambda \in \mathbb{C} : \Re(\lambda) \in \overline{J}\}.$

Step 5: $\sigma_{ess}(-\Lambda - T) = \sigma_{ess}(K - \Lambda - T)$. The operator K is compact on L^2_v but not on $L^2_{x,v}$, so the claim does not follow from the original form of Weyl's theorem. Instead we will employ the fact that the essential spectrum is conserved under a relatively compact perturbation [97, Section IV.5.6, Theorem 5.35]. More precisely, we prove that K is relatively compact with respect to $\Lambda + T$, i.e., $G_z := K(\Lambda + T - zI)^{-1}$ is compact on $L^2_{x,v}$ for some $z \in \mathbb{C}$ with $\Re(z) \in \mathbb{R} \setminus \overline{J}$. (Notice that by Step 4, $z \in \rho(\Lambda + T)$.) Then

$$\sigma_{\rm ess}(K - \Lambda - T) = \sigma_{\rm ess}(-\Lambda - T) = \{\lambda \in \mathbb{C} : \Re(\lambda) \in -J\}. \tag{4.3.6}$$

The second identity is a consequence of Steps 3 and 4.

To prove the compactness of G_z , we introduce the space $W:=\ell^2(\mathbb{Z}^3;L_v^2)$ of sequences $f=(f_m)\subset L_v^2$ with the canonical norm $\|f\|_W=(\sum_{m\in\mathbb{Z}^3}\|f_m\|_{L_v^2}^2)^{1/2}$. Clearly, W is a Hilbert space with the scalar product $(f,g)_W=\sum_{m\in\mathbb{Z}^3}(f_m,g_m)_{L_v^2}$. Furthermore, we introduce the Fourier mapping $F:L_{x,v}^2(\mathbb{T}^3\times\mathbb{R}^3)\to W$ by

$$F(f) = (\widehat{f}_m), \quad \widehat{f}_m(v) = \int_{\mathbb{T}^3} e^{-2\pi i m \cdot x} f(x, v) dx \quad \text{for } m \in \mathbb{Z}^3, \ v \in \mathbb{R}^3.$$

This mapping is bounded, invertible, and has a bounded inverse. We wish to show that $\widehat{G}_z = FG_zF^{-1}: W \to W$ is compact. Then also $G_z = F^{-1}\widehat{G}_zF$ is compact as a composition of a compact and two bounded operators. This idea is due to Ukai; see e.g. [135, Section 2.2.1].

Since K and Λ do not depend on x, it holds that $\hat{G}_z = K(\Lambda + \hat{T} - z)^{-1}$, where $\hat{T} = 2\pi i v \cdot m$. Let $(f^{(k)}) = (f_m^{(k)}) \subset W$ be a bounded sequence in W, i.e., there exists $c_0 > 0$ such that for all $k \in \mathbb{N}$,

$$||f^{(k)}||_W^2 = \sum_{m \in \mathbb{Z}^3} ||f_m^{(k)}||_{L_v^2}^2 \le c_0.$$
(4.3.7)

As $\Re(z) \in \mathbb{R} \setminus \overline{J}$, there is a constant $c_z > 0$ such that for all $i = 1, \ldots, n$ and $v \in \mathbb{R}^3$, $|\nu_i(v) + 2\pi i v \cdot m - z| \ge c_z$. Thus,

$$\|(\Lambda + \widehat{T} - z)^{-1} f_m^{(k)}\|_{L_v^2}^2 = \sum_{i=1}^n \int_{\mathbb{R}^3} \left| (\nu_i(v) + 2\pi i v \cdot m - z)^{-1} f_{m,i}^{(k)} \right|^2 dv$$

$$\leq c_z^{-2} \sum_{i=1}^n \int_{\mathbb{R}^3} |f_{m,i}^{(k)}|^2 dv = c_z^{-2} \|f_m^{(k)}\|_{L_v^2}^2.$$

Summing these inequalities over $m \in \mathbb{Z}^3$, we infer that

$$\|(\Lambda + \widehat{T} - z)^{-1} f^{(k)}\|_W^2 \le c_z^{-2} \|f^{(k)}\|_W^2 \le c_0 c_z^{-2}.$$

Consequently, the sequence $g^{(k)} := (\Lambda + \widehat{T} - z)^{-1} f^{(k)}$ is bounded in W. In particular, for any $s \in \mathbb{Z}^3$, $\|g_s^{(k)}\|_{L_v^2}^2 \leq \sum_{m \in \mathbb{Z}^3} \|g_m^{(k)}\|_{L_v^2}^2 = \|g^{(k)}\|_W^2 \leq c_0 c_z^{-2}$. Hence, for any $s \in \mathbb{Z}^3$, the sequence $(g_s^{(k)}) \subset L_v^2$ is bounded in L_v^2 . Since $K: L_v^2 \to L_v^2$ is compact and \mathbb{Z}^3 is countable, we may apply Cantor's diagonal argument to find a subsequence $(g^{(k_\ell)})$ of $(g^{(k)})$ such that $(K(g_m^{(k_\ell)}))$ is convergent in L_v^2 as $\ell \to \infty$, for all $m \in \mathbb{Z}^3$. We will show that $(\widehat{G}_z(f^{(k_\ell)}))$ is a Cauchy sequence in W. To this end, let ℓ , s, $N \in \mathbb{N}$.

We write

$$\|\widehat{G}_{z}(f^{(k_{\ell})}) - \widehat{G}_{z}(f^{(k_{s})})\|_{W}^{2} = \sum_{m \in \mathbb{Z}^{3}} \|K(g_{m}^{(k_{\ell})}) - K(g_{m}^{(k_{s})})\|_{L_{v}^{2}}^{2}$$

$$= \sum_{|m| \leq N} \|K(g_{m}^{(k_{\ell})}) - K(g_{m}^{(k_{s})})\|_{L_{v}^{2}}^{2} + \sum_{|m| > N} \|K(g_{m}^{(k_{\ell})}) - K(g_{m}^{(k_{s})})\|_{L_{v}^{2}}^{2},$$

$$(4.3.8)$$

where $|m| = \sum_{i=1}^{3} |m_i|$ for all $m \in \mathbb{Z}^3$. First, we consider the second sum on the right-hand side. Denote by $\|\cdot\|_{\mathscr{L}(L^2_v)}$ the norm in the space of linear bounded operators on L^2_v . By the definition of $g_m^{(k)}$, we obtain

$$\sum_{|m|>N} \|K(g_m^{(k_\ell)}) - K(g_m^{(k_s)})\|_{L_v^2}^2 = \sum_{|m|>N} \|K(\Lambda + 2\pi i v \cdot m - z)^{-1} (f_m^{(k_\ell)} - f_m^{(k_s)})\|_{L_v^2}^2 \quad (4.3.9)$$

$$\leq 2 \sum_{|m|>N} \|K(\Lambda + 2\pi i v \cdot m - z)^{-1}\|_{\mathcal{L}(L_v^2)}^2 (\|f_m^{(k_\ell)}\|_{L_v^2}^2 + \|f_m^{(k_s)}\|_{L_v^2}^2).$$

For the operator norm, we employ Prop. 2.2.6 in [135], which can be applied since $\Re(z) \in \mathbb{R} \setminus \overline{J}$:

$$||K(\Lambda + 2\pi i v \cdot m - z)^{-1}||_{\mathcal{L}(L^2)}^2 \le c_1(1 + |m|)^{-\alpha}$$
 for all $m \in \mathbb{Z}^3$

for some suitable constant $c_1 > 0$ (depending on z) and a suitable exponent $\alpha \in (0,1)$ (actually, $\alpha = 4/13$). Let $0 < \beta < 2\alpha/3$. By Hölder's inequality and (4.3.7), we estimate

$$\sum_{|m|>N} \|K(\Lambda + 2\pi i v \cdot m - z)^{-1}\|_{\mathcal{L}(L_v^2)}^2 \|f_m^{(k)}\|_{L_v^2}^2 \le c_0^{\beta/2} c_1 \sum_{|m|>N} (1 + |m|)^{-\alpha} \|f_m^{(k)}\|_{L_v^2}^{2-\beta}
\le c_0^{\beta/2} c_1 \left(\sum_{|m|>N} (1 + |m|)^{-2\alpha/\beta} \right)^{\beta/2} \left(\sum_{|m|>N} \|f_m^{(k)}\|_{L_v^2}^2 \right)^{1-\beta/2}
\le c_0 c_1 \left(\sum_{|m|>N} (1 + |m|)^{-2\alpha/\beta} \right)^{\beta/2}.$$

Using this estimate in (4.3.9), it follows that

$$\sup_{\ell,s\in\mathbb{N}} \sum_{|m|>N} \|K(g_m^{(k_\ell)}) - K(g_m^{(k_s)})\|_{L_v^2}^2 \le 2c_0c_1 \left(\sum_{|m|>N} (1+|m|)^{-2\alpha/\beta}\right)^{\beta/2}.$$

The choice of β implies that $2\alpha/\beta>3$ and hence, the sum over |m|>N is finite. In particular, $\sum_{|m|>N}(1+|m|)^{-2\alpha/\beta}\to 0$ as $N\to\infty$. As a consequence, for given $\varepsilon>0$, there exists $N_\varepsilon\in\mathbb{N}$ such that

$$\sup_{\ell,s\in\mathbb{N}} \sum_{|m|>N_{\varepsilon}} \|K(g_m^{(k_{\ell})}) - K(g_m^{(k_s)})\|_{L_v^2}^2 < \frac{\varepsilon}{2}.$$

Finally, since $(K(g_m^{(k_\ell)}))$ is convergent in L_v^2 for all $m \in \mathbb{Z}^3$, there is a number $\eta = \eta(\varepsilon) > 0$ such that for all $\ell, s > \eta$,

$$\sum_{|m| \le N_{\varepsilon}} \|K(g_m^{(k_{\ell})}) - K(g_m^{(k_s)})\|_{L_v^2}^2 < \frac{\varepsilon}{2}.$$

Thus, choosing $N = N_{\varepsilon}$ in (4.3.8), we deduce that $(\widehat{G}_z(f^{(k_s)}))$ is a Cauchy sequence in the Hilbert space W and consequently, it is convergent. This shows that $\widehat{G}_z : W \to W$ is a compact operator and (4.3.6) holds. This finishes the proof of Theorem 4.1.

Next, we show the spectral-gap estimate for the linearized collision operator $L = K - \Lambda$, i.e. the first statement of Theorem 4.2. Since K is compact on L_v^2 , it remains to prove that $\Lambda : \mathcal{D} \subset L_v^2 \to L_v^2$ is coercive.

Lemma 4.8. Let (A1)-(A4) hold. Then the embedding $\mathcal{H} \hookrightarrow L_v^2$ is continuous and $\Lambda : \mathcal{D} \to L_v^2$, defined in (4.3.1), is a linear unbounded operator with the property

$$(f, \Lambda(f))_{L_v^2} = ||f||_{\mathcal{H}}^2 \ge C||f||_{L_v^2}^2 \quad \text{for } f \in \mathcal{H}$$
 (4.3.10)

for some C>0. Moreover, Λ can be extended by density to a linear bounded operator $\Lambda: \mathcal{H} \to \mathcal{H}'$, where \mathcal{H}' is the dual of \mathcal{H} with respect to the L^2_v scalar product. In particular, the mapping $\mathcal{H} \to \mathbb{R}$, $f \mapsto \langle \Lambda(f), f \rangle$ is continuous, where $\langle \cdot, \cdot \rangle$ denotes the duality pairing between \mathcal{H}' and \mathcal{H} .

Proof. The strict positivity of ν_i in \mathbb{R}^3 (see Lemma 4.6) implies that the embedding $\mathcal{H} \hookrightarrow L_v^2$ is continuous. Then the definitions of Λ_i and \mathcal{H} show that for all $f \in \mathcal{H}$, (4.3.10) holds. For given $f \in \mathcal{H}$, the element $\Lambda(f) = (f_1\nu_1, \ldots, f_n\nu_n)$ can be identified with the linear bounded operator $\mathcal{H} \to \mathbb{R}$, $g \mapsto \sum_{i=1}^n \int_{\mathbb{R}^3} g_i f_i \nu_i dv$ and consequently, $\Lambda(f) \in \mathcal{H}'$. It is immediate to see that $\|\Lambda(f)\|_{\mathcal{H}'} = \|f\|_{\mathcal{H}}$, so that $\Lambda : \mathcal{H} \to \mathcal{H}'$ is isometric and thus bounded. Moreover, it follows that $\mathcal{H} \to \mathbb{R}$, $f \mapsto \langle \Lambda(f), f \rangle$, is continuous.

The following result provides a spectral gap for general operators which decompose into a compact and a coercive part.

Lemma 4.9. Let \mathcal{H}_0 and \mathcal{H} be Hilbert spaces such that $\mathcal{H} \hookrightarrow \mathcal{H}_0$ continuously and let $L: \mathcal{H} \to \mathcal{H}'$ be a linear bounded operator such that $L = K - \Lambda$ with linear bounded operators $\Lambda: \mathcal{H} \to \mathcal{H}'$ and $K: \mathcal{H}_0 \to \mathcal{H}_0$. Furthermore, assume that

- (i) for all $f \in \mathcal{H}$, $\langle L(f), f \rangle \leq 0$ with equality holding if and only if $f \in \text{Ker}(L)$;
- (ii) the operator $K: \mathcal{H}_0 \to \mathcal{H}_0$ is compact;
- (iii) there exists $C_0 > 0$ such that for all $f \in \mathcal{H}$, $\langle \Lambda(f), f \rangle \geq C_0 ||f||_{\mathcal{H}}^2$.

Then there exists a constant $C_1 > 0$ such that

$$-\langle L(f), f \rangle \ge C_1 ||f||_{\mathcal{H}}^2 \quad for \ all \ f \in \mathcal{H} \cap \operatorname{Ker}(L)^{\perp}.$$

Proof. We argue by contradiction. Let $(f_n) \subset \mathcal{H} \cap \operatorname{Ker}(L)^{\perp}$ be a sequence such that $||f_n||_{\mathcal{H}} = 1$ for $n \geq 1$ but $\langle L(f_n), f_n \rangle \to 0$ as $n \to \infty$. Since (f_n) is bounded in the Hilbert space \mathcal{H} , there exists a subsequence, which is not relabeled, such that $f_n \to f$ weakly in \mathcal{H} . Because of the continuous embedding $\mathcal{H} \hookrightarrow \mathcal{H}_0$, also $f_n \to f$ weakly in \mathcal{H}_0 . Since $f_n \in \operatorname{Ker}(L)^{\perp}$ and $\operatorname{Ker}(L)^{\perp}$ is weakly closed by Mazur's lemma, $f \in \operatorname{Ker}(L)^{\perp}$. As the operator $K: \mathcal{H}_0 \to \mathcal{H}_0$ is compact, by hypothesis (ii), the weak convergence of (f_n) in \mathcal{H}_0 implies that $K(f_n) \to K(f)$ strongly in \mathcal{H}_0 . Hence, $(f_n, K(f_n))_{\mathcal{H}_0} \to (f, K(f))_{\mathcal{H}_0}$. Since $\Lambda: \mathcal{H} \to \mathcal{H}'$ is bounded, the mapping $P: \mathcal{H} \to \mathbb{R}$, $f \mapsto \langle \Lambda(f), f \rangle$, is continuous. The linearity of Λ and property (iii) imply that P is also convex. Thus, P is weakly lower semicontinuous [24, Corollary 3.9]. Therefore,

$$-\langle L(f), f \rangle = \langle \Lambda(f), f \rangle - (K(f), f)_{\mathcal{H}_0} \le \liminf_{n \to \infty} \left(\langle \Lambda(f_n), f_n \rangle - (K(f_n), f_n)_{\mathcal{H}_0} \right) = 0,$$

because $\langle L(f_n), f_n \rangle \to 0$ as $n \to \infty$ by assumption. We infer from hypothesis (i) that $f \in \text{Ker}(L)$. But also $f \in \text{Ker}(L)^{\perp}$, so f = 0. Then, by hypothesis (iii),

$$0 < C_0 = C_0 ||f_n||_{\mathcal{H}}^2 \le \langle \Lambda(f_n), f_n \rangle = (K(f_n), f_n)_{\mathcal{H}_0} - \langle L(f_n), f_n \rangle \to 0,$$

which is a contradiction.

Let $\mathcal{H}_0 = L_v^2$. By [19, Prop. 2], assumption (ii) of Lemma 4.9 holds. Furthermore, Lemma 4.8 shows that (iii) holds true. Assumption (i) is a consequence of Lemma 4.4. Let $f \in \mathcal{D} \subset \mathcal{H}$ und set $\tilde{f} = f - \Pi^L(f) \in \text{Ker}(L)^{\perp}$. Then

$$-\langle L(f),f\rangle = \langle L(\widetilde{f}),\widetilde{f}\rangle \geq C\|\widetilde{f}\|_{\mathcal{H}}^2 = C\|f-\Pi^L(f)\|_{\mathcal{H}}^2,$$

since $L(f) \in L_v^2$ and $\langle L(f), f \rangle = (f, L(f))_{L_v^2}$ for $f \in \mathcal{D}$. This proves the first statement in Theorem 4.2.

4.4. Explicit spectral-gap estimate

We present a second proof of the spectral-gap estimate (4.1.7) with explicit constants. The idea is to decompose the collision operator L into a mono-species and a multi-species part and to exploit the fact that the conservation properties of L are different from those of the mono-species part L^m . Let assumptions (A1)-(A4) hold.

4.4.1. Decomposition

We decompose $L = L^m + L^b$, where $L^m = (L_1^m, \dots, L_n^m), L^b = (L_1^b, \dots, L_n^b)$, and

$$L_i^m(f_i) = L_{ii}(f_i, f_i), \quad L_i^b(f) = \sum_{j \neq i} L_{ij}(f_i, f_j).$$
 (4.4.1)

Denoting by Π^m the orthogonal projection onto $\operatorname{Ker}(L^m)$ (with respect to the scalar product in L_v^2), we can decompose f according to

$$f = f^{\parallel} + f^{\perp}$$
, where $f^{\parallel} := \Pi^m(f)$, $f^{\perp} := f - f^{\parallel}$. (4.4.2)

Lemma 4.4 shows that

$$f \in \text{Ker}(L)$$
 if and only if $f_i = M_i^{1/2}(\alpha_i + u \cdot v + e|v|^2)$ for $\alpha_i, e \in \mathbb{R}, u \in \mathbb{R}^3$, (4.4.3) $f \in \text{Ker}(L^m)$ if and only if $f_i = M_i^{1/2}(\alpha_i + u_i \cdot v + e_i|v|^2)$ for $\alpha_i, e_i \in \mathbb{R}, u_i \in \mathbb{R}^3$, (4.4.4)

and f^{\parallel} has clearly the form (4.4.4).

For later use, we define the following bilinear forms

$$-(f, L^{m}(f))_{L_{v}^{2}} = \frac{1}{4} \sum_{i=1}^{n} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ii} \Delta_{i} [h_{i}]^{2} M_{i} M_{i}^{*} dv dv^{*} d\sigma, \qquad (4.4.5)$$

$$-(f, L^b(f))_{L_v^2} = \frac{1}{4} \sum_{i=1}^n \sum_{j \neq i} \int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} A_{ij} [h_i, h_j]^2 M_i M_j^* dv dv^* d\sigma, \tag{4.4.6}$$

where $h_i = M_i^{-1/2} f_i$ and

$$\Delta_i[h_i] := h'_i + h'^*_i - h_i - h^*_i, \quad A_{ij}[h_i, h_j] := h'_i + h'^*_j - h_i - h^*_j.$$

4.4.2. Spectral-gap estimate for the mono-species part

Our starting point is the fact that the mono-species collision operator L^m has an explicitly computable spectral gap. A spectral-gap estimate for the linearized collision operator with n = 1 was proved in [110, Theorem 6.1, Remark 1]:

$$\frac{1}{4} \int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ii} \Delta_i[h_i]^2 M_i M_i^* dv dv^* d\sigma \ge \frac{\lambda_m}{\rho_{\infty,i}} \int_{\mathbb{R}^3} (f_i - \Pi^m(f_i))^2 \nu_{ii} dv, \tag{4.4.7}$$

where $\lambda_m = \lambda_m(\gamma, C_1, C^b) > 0$, only depending on γ , C_1 , and C^b (see (A3)-(A4)), can be computed explicitly,

$$\nu_{ii}(v) := \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ii}(|v - v^*|, \cos \vartheta) M_i^* dv^* d\sigma,$$

and $i \in \{1, ..., n\}$ is fixed. This yields the following estimate for L^m , where we recall that the space \mathcal{H} is defined in (4.1.6).

Lemma 4.10. With L^m defined in (4.4.1), we have

$$-(f, L^m(f))_{L^2_n} \ge C^m ||f - \Pi^m(f)||_{\mathcal{H}}^2 \quad \text{for all } f \in \text{Dom}(L^m),$$

where

$$C^{m} = \frac{\lambda^{m}(\gamma, C_{1}, C^{b})}{\beta \rho_{\infty}}, \tag{4.4.8}$$

and $\lambda^m = \lambda^m(\gamma, C_1, C^b)$ is given in (4.4.7).

Proof. We sum (4.4.7) over i = 1, ..., n and employ (4.4.5) to obtain

$$-(f, L^{m}(f))_{L_{v}^{2}} \ge \lambda^{m} \sum_{i=1}^{n} \int_{\mathbb{R}^{3}} (f_{i} - \Pi^{m}(f))^{2} \frac{\nu_{ii}}{\rho_{\infty, i}} dv.$$
 (4.4.9)

It remains to estimate ν_{ii} in terms of ν_i , defined in (4.1.4). The definition of M_i implies that $M_j = (\rho_{\infty,j}/\rho_{\infty,i})M_i$. This fact, as well as definition (4.1.4) of ν_i , the lower bound (4.2.3), and assumption (A6) give

$$\nu_i = \sum_{j=1}^n \frac{\rho_{\infty,j}}{\rho_{\infty,i}} \int_{\mathbb{R}^3} B_{ij} M_i^* dv^* d\sigma \le \beta \sum_{j=1}^n \frac{\rho_{\infty,j}}{\rho_{\infty,i}} \int_{\mathbb{R}^3} B_{ii} M_i^* dv^* d\sigma = \frac{\beta \rho_{\infty}}{\rho_{\infty,i}} \nu_{ii}.$$

We conclude that $\nu_{ii}/\rho_{\infty,i} \geq \nu_i/(\beta\rho_{\infty})$, and inserting this bound into (4.4.9) yields the result.

Lemma (4.10) and the inequality $-(f, L^b(f))_{L_2^v} \geq 0$ immediately show that

$$-(f, L(f))_{L^2_v} \ge C^m ||f - \Pi^m(f)||_{\mathcal{H}}^2$$
 for all $f \in \mathcal{D}$.

However, we need the projection onto $\operatorname{Ker}(L)^{\perp}$ instead of $\operatorname{Ker}(L^m)^{\perp}$, which is contained in $\operatorname{Ker}(L)^{\perp}$. Therefore, we will exploit the part $-(f, L^b(f))_{L^2}$ to derive a sharper estimate.

4.4.3. Absorption of the orthogonal parts

We prove that the contribution f^{\perp} (introduced in (4.4.2)) in the term $-(f, L^b(f))_{L_v^2} = -(f^{\parallel} + f^{\perp}, L^b(f^{\parallel} + f^{\perp}))_{L_v^2}$ can be absorbed by the \mathcal{H} norm of f^{\perp} .

Lemma 4.11. Let $\eta = \min\{1, C^m/8\}$, where $C^m > 0$ is given in Lemma 4.10. Then, for all $f \in \mathcal{D}$,

$$-(f, L(f))_{L_v^2} \ge (C^m - 4\eta) \|f - f^{\parallel}\|_{\mathcal{H}}^2 - \frac{\eta}{2} (f^{\parallel}, L^b(f^{\parallel}))_{L_v^2},$$

where $f^{\parallel} = \Pi^m(f)$ is the projection onto $\operatorname{Ker}(L^m)^{\perp}$.

Proof. By Lemma 4.10, we find that

$$-(f, L(f))_{L_n^2} \ge C^m \|f - f^{\parallel}\|_{\mathcal{H}}^2 - (f, L^b(f))_{L_n^2} \ge C^m \|f - f^{\parallel}\|_{\mathcal{H}}^2 - \eta(f, L^b(f))_{L_n^2}, \quad (4.4.10)$$

since $-(1-\eta)(f,L^b(f))_{L^2_v} \geq 0$ for $\eta \in (0,1]$. We estimate first the expression $A_{ij}[h_i,h_j]$ in definition (4.4.6), writing $h_i^{\parallel} = M_i^{-1/2} f_i^{\parallel}$ and $h_i^{\perp} = M_i^{-1/2} f_i^{\perp}$,

$$\begin{split} A_{ij}[h_i,h_j]^2 &= \left(A_{ij}[h_i^{\parallel},h_j^{\parallel}] + A_{ij}[h_i^{\perp},h_j^{\perp}]\right)^2 \\ &= A_{ij}[h_i^{\parallel},h_j^{\parallel}]^2 + A_{ij}[h_i^{\perp},h_j^{\perp}]^2 + 2A_{ij}[h_i^{\parallel},h_j^{\parallel}]A_{ij}[h_i^{\perp},h_j^{\perp}] \\ &\geq \frac{1}{2}A_{ij}[h_i^{\parallel},h_j^{\parallel}]^2 - A_{ij}[h_i^{\perp},h_j^{\perp}]^2. \end{split}$$

Inserting this estimate into (4.4.6) and (4.4.10) gives

$$-(f, L(f))_{L_{v}^{2}} \geq C^{m} \|f^{\perp}\|_{\mathcal{H}}^{2} + \frac{\eta}{8} \sum_{i=1}^{n} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} A_{ij} [h_{i}^{\parallel}, h_{j}^{\parallel}]^{2} M_{i} M_{j}^{*} dv dv^{*} d\sigma$$
$$- \frac{\eta}{4} \sum_{i=1}^{n} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} A_{ij} [h_{i}^{\perp}, h_{j}^{\perp}]^{2} M_{i} M_{j}^{*} dv dv^{*} d\sigma. \tag{4.4.11}$$

We claim that the last term on the right-hand side can be estimated from below by $||f^{\perp}||_{\mathcal{H}}^2$, up to a small factor. For this, we employ the invariance properties of B_{ij} and the identity $M_i M_i^* = M_i' M_i'^*$:

$$\int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij} A_{ij} [h_{i}^{\perp}, h_{j}^{\perp}]^{2} M_{i} M_{j}^{*} dv dv^{*} d\sigma$$

$$\leq 4 \int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij} \left(((h_{i}^{\perp})')^{2} + ((h_{j}^{\perp})'^{*})^{2} + (h_{i}^{\perp})^{2} + ((h_{j}^{\perp})^{*})^{2} \right) M_{i} M_{j}^{*} dv dv^{*} d\sigma$$

$$\leq 16 \int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij} (h_{i}^{\perp})^{2} M_{i} M_{j}^{*} dv dv^{*} d\sigma = 16 \int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij} (f_{i}^{\perp})^{2} M_{j}^{*} dv dv^{*} d\sigma.$$

Thus, the last term on the right-hand side of (4.4.11) can be estimated as

$$-\frac{\eta}{4} \sum_{i=1}^{n} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} A_{ij} [h_{i}^{\perp}, h_{j}^{\perp}]^{2} M_{i} M_{j}^{*} dv dv^{*} d\sigma$$

$$\geq -4\eta \sum_{i=1}^{n} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} (f_{i}^{\perp})^{2} M_{j}^{*} dv dv^{*} d\sigma \geq -4\eta \sum_{i=1}^{n} \int_{\mathbb{R}^{3}} (f_{i}^{\perp})^{2} \nu_{i} dv = -4\eta \|f^{\perp}\|_{\mathcal{H}}^{2},$$

taking into account definition (4.1.4) of ν_i . We infer from (4.4.11) that

$$-(f, L(f))_{L_v^2} \ge (C^m - 4\eta) \|f - f^{\parallel}\|_{\mathcal{H}}^2 + \frac{\eta}{8} \sum_{i=1}^n \sum_{j \ne i} \int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} A_{ij} [h_i^{\parallel}, h_j^{\parallel}]^2 M_i M_j^* dv dv^* d\sigma,$$

and definition (4.4.6) yields the conclusion.

4.4.4. Estimate for the remaining part

It remains to estimate the term $-(f^{\parallel}, L^b(f^{\parallel}))_{L^2_v}$.

Lemma 4.12. For $f^{\parallel} \in \text{Ker}(L^m)$, i.e. $f_i^{\parallel} = M_i^{1/2}(\alpha_i + u_i \cdot v + e_i|v|^2)$ for some α_i , $e_i \in \mathbb{R}$ and $u_i \in \mathbb{R}^3$, we have

$$-(f^{\parallel}, L^b(f^{\parallel}))_{L_v^2} \ge \frac{D^b}{4} \sum_{i,j=1}^n (|u_i - u_j|^2 + (e_i - e_j)^2),$$

where $D^b > 0$ is defined in (4.4.13).

Proof. Thanks to the momentum and energy conservation, we obtain differences of the momenta and energies, which will be crucial in the following:

$$u_i \cdot v' + u_j \cdot v'^* - u_i \cdot v - u_j \cdot v^* = (u_i - u_j) \cdot (v' - v),$$

$$e_i |v'|^2 + e_j |v'^*|^2 - e_i |v|^2 - e_j |v^*|^2 = (e_i - e_j)(|v'|^2 - |v|^2).$$

Using these identities in $A_{ij}[h_i^{\parallel}, h_j^{\parallel}]$, where $h_i^{\parallel} = \alpha_i + u_i \cdot v + e_i |v|^2$, we find that

$$-(f^{\parallel}, L^{b}(f^{\parallel}))_{L_{v}^{2}} = \frac{1}{4} \sum_{i=1}^{n} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} A_{ij} [h_{i}^{\parallel}, h_{j}^{\parallel}]^{2} M_{i} M_{j}^{*} dv dv^{*} d\sigma$$

$$= \frac{1}{4} \sum_{i=1}^{n} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} ((u_{i} - u_{j}) \cdot (v' - v) + (e_{i} - e_{j})(|v'|^{2} - |v|^{2}))^{2} M_{i} M_{j}^{*} dv dv^{*} d\sigma.$$

Using the symmetry of B_{ij} (thanks to assumption (A1)) and of $M_i M_j^*$ with respect to v, the function $G(v, v^*, \sigma) = B_{ij}(u_i - u_j) \cdot (v' - v)(|v'|^2 - |v|^2)$ is odd with respect to (v, v^*, σ) and thus, the mixed term of the square in the above integral vanishes. Therefore, we obtain

$$-(f^{\parallel}, L^{b}(f^{\parallel}))_{L_{v}^{2}} = \frac{1}{4} \sum_{i=1}^{n} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} (|(u_{i} - u_{j}) \cdot (v' - v)|^{2} + (e_{i} - e_{j})^{2} (|v'|^{2} - |v|^{2})^{2}) \times M_{i} M_{j}^{*} dv dv^{*} d\sigma.$$

$$(4.4.12)$$

Now, we claim that

$$\int_{\mathbb{R}^{6}\times\mathbb{S}^{2}}B_{ij}((u_{i}-u_{j})\cdot(v'-v))^{2}M_{i}M_{j}^{*}dvdv^{*}d\sigma = \frac{|u_{i}-u_{j}|^{2}}{3}\int_{\mathbb{R}^{6}\times\mathbb{S}^{2}}B_{ij}|v-v'|^{2}M_{i}M_{j}^{*}dvdv^{*}d\sigma.$$

To prove this identity, we write $u_{i,k}$ and v_k for the kth component of the vectors u_i and v, respectively. The transformation $(v_k, v_k^*, \sigma_k) \mapsto -(v_k, v_k^*, \sigma_k)$ for fixed k leaves B_{ij} , M_i , and M_i^* unchanged but $v_k' \mapsto -v_k'$ such that

$$\int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} v_k' v_\ell M_i M_j^* dv dv^* d\sigma = 0 \quad \text{for } \ell \neq k.$$

Furthermore,

$$\int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} v_k v_\ell M_i M_j^* dv dv^* d\sigma = 0 \quad \text{for } \ell \neq k,$$

since the integrand is odd. Therefore,

$$\int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij}((u_{i}-u_{j})\cdot(v'-v))^{2} M_{i} M_{j}^{*} dv dv^{*} d\sigma$$

$$= \sum_{k,\ell=1}^{3} (u_{i,k}-u_{j,k})(u_{i,\ell}-u_{j,\ell}) \int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij}(v'_{k}-v_{k})(v'_{\ell}-v_{\ell}) M_{i} M_{j}^{*} dv dv^{*} d\sigma$$

$$= \sum_{k=1}^{3} (u_{i,k}-u_{j,k})^{2} \int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij}(v_{k}-v'_{k})^{2} M_{i} M_{j}^{*} dv dv^{*} d\sigma.$$

In fact, we can see that the integral is independent of k, and we infer that

$$\int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} ((u_i - u_j) \cdot (v' - v))^2 M_i M_j^* dv dv^* d\sigma$$

$$= \frac{1}{3} \sum_{k=1}^3 (u_{i,k} - u_{j,k})^2 \int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} |v - v'|^2 M_i M_j^* dv dv^* d\sigma,$$

from which the claim follows.

Hence, (4.4.12) can be estimated as

$$-(f^{\parallel}, L^b(f^{\parallel}))_{L_v^2} \ge D^b \sum_{i,j=1}^3 (|u_i - u_j|^2 + (e_i - e_j)^2),$$

where

$$D^{b} = \min_{1 \le i, j \le n} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} \min \left\{ \frac{1}{3} |v - v'|^{2}, (|v'|^{2} - |v|^{2})^{2} \right\} M_{i} M_{j}^{*} dv dv^{*} d\sigma.$$
 (4.4.13)

It remains to show that $D^b > 0$. The integrand of (4.4.13) vanishes if and only if |v'| = |v|. However, the set

$$X = \{(v, v^*, \sigma) \in \mathbb{R}^3 \times \mathbb{R}^3 \times \mathbb{S}^2 : |v'| = |v|\}$$

is closed since it is the pre-image of $\{0\}$ of the continuous function $F(v, v^*, \sigma) = |v'|^2 - |v|^2$, i.e. $X = F^{-1}(\{0\})$, recalling that v' depends on (v, v^*, σ) through (2.1.2). Since $X \neq \mathbb{R}^3 \times \mathbb{R}^3 \times \mathbb{S}^2$, its complement X^c is open and nonempty and thus has positive Lebesgue measure. Since the integrand in (4.4.13) is positive on X^c , we infer that $D^b > 0$. This finishes the proof.

4.4.5. Estimate for the momentum and energy differences

The last step is to derive lower bounds for the differences $\sum_{i,j} (|u_i - u_j|^2 + (e_i - e_j)^2)$. First, we recall some moment identities:

$$\int_{\mathbb{R}^3} M_i dv = \rho_i, \quad \int_{\mathbb{R}^3} M_i v_j v_k dv = \rho_i \delta_{jk}, \quad \int_{\mathbb{R}^3} M_i |v|^4 dv = 15\rho_i$$
 (4.4.14)

for all $1 \le i \le n$ and $1 \le j, k \le 3$.

Lemma 4.13. Let $f \in L_v^2$ with $f_i^{\parallel} = M_i^{1/2}(\alpha_i + u_i \cdot v + e_i|v|^2)$ for $1 \le i \le n$. Then

$$\int_{\mathbb{R}^3} M_i^{1/2} f_i dv = \rho_i(\alpha_i + 3e_i), \ \int_{\mathbb{R}^3} M_i^{1/2} f_i v dv = \rho_i u_i, \ \int_{\mathbb{R}^3} M_i^{1/2} f_i |v|^2 dv = \rho_i (3\alpha_i + 15e_i).$$

Proof. Decomposing $f = f^{\parallel} + f^{\perp}$, where $f^{\parallel} = \Pi^m(f)$ and $f^{\perp} = f - \Pi^m(f)$, we infer from $M_i^{1/2} \in \text{Ker}(L^m)$ (see (4.4.4)) that $(M_i^{1/2}, f_i^{\perp})_{L_n^2} = 0$ and hence, by (4.4.4) again,

$$(M_i^{1/2}, f_i^{\parallel})_{L_v^2} = \int_{\mathbb{R}^3} M_i(\alpha_i + u_i \cdot v + e_i |v|^2) dv = \rho_i(\alpha_i + 3e_i).$$

The other identities can be shown in a similar way.

Lemma 4.14. For all $f \in \mathcal{D}$, we have

$$\sum_{i,j=1}^{n} (|u_i - u_j|^2 + (e_i - e_j)^2) \ge \frac{1}{C_k} (||f - \Pi^L(f)||_{\mathcal{H}}^2 - 2||f - \Pi^m(f)||_{\mathcal{H}}^2),$$

where u_i , e_i are the coefficients of the ith component of $\Pi^m(f)$ in (4.4.4), Π^L is the projection on Ker(L), $C_k > 0$ is given by

$$C_k = 60n\rho_{\infty} \max_{1 \le k, \ell \le 5n} \left| \sum_{i=1}^n \int_{\mathbb{R}^3} \psi_k \psi_\ell \nu_i dv \right|, \tag{4.4.15}$$

and (ψ_k) is an arbitrary orthonormal basis of $\operatorname{Ker}(L^m)$ in L_v^2 .

Proof. We again decompose $f = f^{\parallel} + f^{\perp}$ with $f^{\parallel} = \Pi^m(f)$ and $f^{\perp} = f - f^{\parallel}$. Then

$$||f - \Pi^{L}(f)||_{\mathcal{H}}^{2} \le 2(||f^{\perp}||_{\mathcal{H}}^{2} + ||f^{\parallel} - \Pi^{L}(f)||_{\mathcal{H}}^{2}). \tag{4.4.16}$$

We estimate first the difference $g:=f^{\parallel}-\Pi^L(f)=\Pi^m(f)-\Pi^L(f)\in \mathrm{Ker}(L^m)$ (note that $\mathrm{Ker}(L)\subset \mathrm{Ker}(L^m)$). Let (ψ_k) be an arbitrary orthonormal basis of $\mathrm{Ker}(L^m)$ in L^2_v . Because of (4.4.4) and $\nabla_v\nu_i\in L^\infty(\mathbb{R}^3)$, we have $\psi_k\in\mathcal{H}$. Then, by Young's inequality, we find that

$$\begin{split} \|g\|_{\mathcal{H}}^2 &= \sum_{i=1}^n \int_{\mathbb{R}^3} \left| \sum_{k=1}^{5n} (g, \psi_k)_{L_v^2} \psi_k \right|^2 \nu_i(v) dv = \sum_{k,\ell=1}^{5n} (g, \psi_k)_{L_v^2} (g, \psi_\ell)_{L_v^2} \sum_{i=1}^n \int_{\mathbb{R}^3} \psi_k \psi_\ell \nu_i(v) dv \\ &= \sum_{k,\ell=1}^{5n} (g, \psi_k)_{L_v^2} (g, \psi_\ell)_{L_v^2} (\psi_k, \psi_\ell)_{\mathcal{H}} \\ &\leq \frac{1}{2} \max_{1 \leq k,\ell \leq 5n} |(\psi_k, \psi_\ell)_{\mathcal{H}}| \sum_{k,\ell=1}^{5n} \left((g, \psi_k)_{L_v^2}^2 + (g, \psi_\ell)_{L_v^2}^2 \right) \\ &= 5n \max_{1 \leq k,\ell \leq 5n} |(\psi_k, \psi_\ell)_{\mathcal{H}}| \sum_{k=1}^{5n} (g, \psi_k)_{L_v^2}^2 = 5n \max_{1 \leq k,\ell \leq 5n} |(\psi_k, \psi_\ell)_{\mathcal{H}}| \|g\|_{L_v^2}^2. \end{split}$$

Thus, we infer from (4.4.16) that

$$||f - \Pi^{L}(f)||_{\mathcal{H}}^{2} \leq 2||f^{\perp}||_{\mathcal{H}}^{2} + 10n \max_{1 \leq k, \ell \leq 5n} |(\psi_{k}, \psi_{\ell})_{\mathcal{H}}| ||f^{\parallel} - \Pi^{L}(f)||_{L_{v}^{2}}^{2}.$$

Because of $\operatorname{Ker}(L) \subset \operatorname{Ker}(L^m)$, we have $\Pi^m \Pi^L = \Pi^L$ and

$$\begin{split} \|f^{\parallel} - \varPi^L(f)\|_{L^2_v}^2 &= \|f^{\parallel}\|_{L^2_v}^2 - 2(\varPi^m(f), \varPi^L(f))_{L^2_v} + \|\varPi^L(f)\|_{L^2_v}^2 \\ &= \|f^{\parallel}\|_{L^2_v}^2 - 2(f, \varPi^L(f))_{L^2_v} + \|\varPi^L(f)\|_{L^2_v}^2 = \|f^{\parallel}\|_{L^2_v}^2 - \|\varPi^L(f)\|_{L^2_v}^2. \end{split}$$

Consequently, setting $k_0 = 10n \max_{1 \le k, \ell \le n} |(\psi_k, \psi_\ell)_{\mathcal{H}}|$,

$$||f - \Pi^{L}(f)||_{\mathcal{H}}^{2} \leq 2||f^{\perp}||_{\mathcal{H}}^{2} + k_{0}(||f^{\parallel}||_{L_{x}^{2}}^{2} - ||\Pi^{L}(f)||_{L_{x}^{2}}^{2}). \tag{4.4.17}$$

Next, we compute the L_n^2 norms of f^{\parallel} and $\Pi^L(f)$. Moment identities (4.4.14) show that

$$||f^{\parallel}||_{L_{v}^{2}}^{2} = \sum_{i=1}^{n} \int_{\mathbb{R}^{3}} M_{i}(\alpha_{i} + u_{i} \cdot v + e_{i}|v|^{2})^{2} dv$$

$$= \sum_{i=1}^{n} \int_{\mathbb{R}^{3}} M_{i}(\alpha_{i}^{2} + (u_{i} \cdot v)^{2} + e_{i}^{2}|v|^{4} + 2\alpha_{i}e_{i}|v|^{2}) dv$$

$$= \sum_{i=1}^{n} \rho_{\infty,i}(\alpha_{i}^{2} + |u_{i}|^{2} + 15e_{i}^{2} + 6\alpha_{i}e_{i}).$$

For the computation of the L_v^2 norm of $\Pi^L(f)$, we choose the following orthonormal basis $(\phi_j) = (\phi_{j,i})_{i=1,\dots,n}$ of $\operatorname{Ker}(L)$ in L_v^2 :

$$\phi_{j,i} = \rho_{\infty}^{-1/2} M_j^{1/2} \delta_{ij}, \quad \phi_{n+k,i} = \rho_{\infty}^{-1/2} M_i^{1/2} v_k, \quad \phi_{n+4,i} = (6\rho_{\infty})^{-1/2} M_i^{1/2} (|v|^2 - 3),$$

where $1 \leq j \leq n$ and $1 \leq k \leq 3$. Then, using the moment identities of Lemma 4.13,

$$\|\Pi^{L}(f)\|_{L_{v}^{2}}^{2} = \sum_{i=1}^{n+4} (f, \phi_{j})_{L_{v}^{2}}^{2} = \sum_{i=1}^{n} \rho_{\infty, i} (\alpha_{i} + 3e_{i})^{2} + \rho_{\infty} \left| \sum_{i=1}^{n} \frac{\rho_{\infty, i}}{\rho_{\infty}} u_{i} \right|^{2} + 6\rho_{\infty} \left(\sum_{i=1}^{n} \frac{\rho_{\infty, i}}{\rho_{\infty}} e_{i} \right)^{2}.$$

Inserting the above identities for $||f^{\parallel}||_{L_v^2}^2$ and $||\Pi^L(f)||_{L_v^2}^2$ into (4.4.17), we conclude that

$$||f - \Pi^{L}(f)||_{\mathcal{H}}^{2} \leq 2||f - \Pi^{m}(f)||_{\mathcal{H}}^{2} + k_{0}\rho_{\infty} \left(\sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} |u_{i}|^{2} - \left| \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} u_{i} \right|^{2} \right) + 6k_{0}\rho_{\infty} \left(\sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} e_{i}^{2} - \left(\sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} e_{i} \right)^{2} \right).$$

Then, if the inequalities

$$\sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} |u_{i}|^{2} - \left| \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} u_{i} \right|^{2} \le \sum_{i,j=1}^{n} |u_{i} - u_{j}|^{2}, \tag{4.4.18}$$

$$\sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} e_i^2 - \left(\sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} e_i\right)^2 \le \sum_{i,j=1}^{n} (e_i - e_j)^2$$
(4.4.19)

hold, the lemma follows with $C_k = 6k_0\rho_{\infty}$.

It remains to prove (4.4.18) and (4.4.19). To this end, we define the following scalar product on \mathbb{R}^{3n} :

$$(u,v)_{\rho} = \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} u_i \cdot v_i, \quad u = (u_1, \dots, u_n), \ v = (v_1, \dots, v_n) \in \mathbb{R}^{3n},$$

where $u_i \cdot v_i$ denotes the usual scalar product in \mathbb{R}^3 . The corresponding norm is $||u||_{\rho} = (u, u)_{\rho}^{1/2}$. Then $\mathbf{1} = (1, \dots, 1) \in \mathbb{R}^{3n}$ satisfies $||\mathbf{1}||_{\rho} = 1$. The elementary identity

$$||u||_{\rho}^{2} - (u, \mathbf{1})_{\rho}^{2} = ||u - (u, \mathbf{1})_{\rho} \mathbf{1}||_{\rho}^{2}$$

can be equivalently written as

$$I := \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} |u_{i}|^{2} - \left| \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} u_{i} \right|^{2} = \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left| u_{i} - \sum_{j=1}^{n} \frac{\rho_{\infty,j}}{\rho_{\infty}} u_{j} \right|^{2}.$$

Then, using $\sum_{j=1}^{n} \rho_{\infty,j} = \rho_{\infty}$,

$$I = \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left| \left(1 - \frac{\rho_{\infty,i}}{\rho_{\infty}} \right) u_{i} - \sum_{j \neq i} \frac{\rho_{\infty,j}}{\rho_{\infty}} u_{j} \right|^{2} = \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left| \sum_{j \neq i} \frac{\rho_{\infty,j}}{\rho_{\infty}} (u_{i} - u_{j}) \right|^{2}$$

$$= \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left(\sum_{k \neq i} \frac{\rho_{\infty,k}}{\rho_{\infty}} \right)^{2} \left| \frac{\sum_{j \neq i} (\rho_{\infty,j}/\rho_{\infty}) (u_{i} - u_{j})}{\sum_{k \neq i} \rho_{\infty,k}/\rho_{\infty}} \right|^{2}$$

$$= \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left(\sum_{k \neq i} \frac{\rho_{\infty,k}}{\rho_{\infty}} \right)^{2} \left| \sum_{j \neq i} \lambda_{j} (u_{i} - u_{j}) \right|^{2},$$

where $\lambda_j = (\rho_{\infty,j}/\rho_\infty)(\sum_{k\neq i}(\rho_{\infty,k}/\rho_\infty))^{-1}$. Since $\sum_{j\neq i}\lambda_j = 1$, we may apply Jensen's inequality to this convex combination, leading to

$$I \leq \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left(\sum_{k \neq i} \frac{\rho_{\infty,k}}{\rho_{\infty}} \right)^{2} \sum_{j \neq i} \lambda_{j} |u_{i} - u_{j}|^{2}$$

$$= \sum_{i=1}^{n} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left(1 - \frac{\rho_{\infty,i}}{\rho_{\infty}} \right) \sum_{j \neq i} \frac{\rho_{\infty,j}}{\rho_{\infty}} |u_{i} - u_{j}|^{2} \leq \sum_{i,j=1}^{n} |u_{i} - u_{j}|^{2},$$

since $\rho_{\infty,j} \leq \rho_{\infty}$. This ends the proof.

Now, we are able to prove Theorem 4.2.

Proof of Theorem 4.2. By Lemmas 4.11, 4.12, and 4.14, we obtain

$$-(f, L(f))_{L_v^2} \ge (C^m - 4\eta) \|f - f^{\parallel}\|_{\mathcal{H}}^2 + \frac{\eta D^b}{8} \sum_{i,j=1}^n \left(|u_i - u_j|^2 + (e_i - e_j)^2 \right)$$
$$\ge \left(C^m - 4\eta - \frac{\eta D^b}{4C_k} \right) \|f - f^{\parallel}\|_{\mathcal{H}}^2 + \frac{\eta D^b}{8C_k} \|f - \Pi^L(f)\|_{\mathcal{H}}^2.$$

The first term on the right-hand side is nonnegative if we choose $\eta = \min\{1, 4C^mC_k/(16C_k + D^b)\}$, and estimate (4.1.7) follows with $\lambda = \eta D^b/(8C_k)$.

4.5. Convergence to equilibrium

In this section, we prove Theorem 4.3. The idea of the proof is to adapt the hypocoercivity method of [113] to the multi-species setting. To this end, we need to verify the structural assumptions (H1)-(H3) in the paper of C. Mouhot and L. Neumann [113, Theorem 1.1], which we call (M1)-(M3) here to avoid confusing with other assumptions in this thesis. The setting is as follows.

Let L be a closed, densely defined, and self-adjoint operator on $Dom(L) \subset L_v^2$ such that $L = K - \Lambda$ and the operators K and Λ satisfy the following assumptions:

(M1) The operator Λ is coercive in the following sense: There exist a norm $\|\cdot\|_{\mathcal{H}}$ on $\mathcal{H} \subset L^2_v$ and positive constants $\overline{\nu}_i$ $(0 \le i \le 4)$ such that for all $f \in \text{Dom}(L) \subset \mathcal{H}$,

$$\overline{\nu}_0 \|f\|_{L^2_v}^2 \le \overline{\nu}_1 \|f\|_{\mathcal{H}}^2 \le (f, \Lambda(f))_{L^2_v} \le \overline{\nu}_2 \|f\|_{\mathcal{H}}^2, \tag{4.5.1}$$

$$(\nabla_v f, \nabla_v \Lambda(f))_{L_v^2} \ge \overline{\nu}_3 \|\nabla_v f\|_{\mathcal{H}}^2 - \overline{\nu}_4 \|f\|_{L_v^2}^2. \tag{4.5.2}$$

Moreover, there exists a constant $C_L > 0$ such that for all $f, g \in Dom(L)$,

$$(L(f), g)_{L_n^2} \le C_L ||f||_{\mathcal{H}} ||g||_{\mathcal{H}}.$$
 (4.5.3)

(M2) The operator K has a regularizing effect in the following sense: For all $\varepsilon > 0$, there exists $C(\varepsilon) > 0$ such that for all $f \in H_v^1$,

$$(\nabla_v f, \nabla_v K(f))_{L_v^2} \le \varepsilon \|\nabla_v f\|_{L_v^2}^2 + C(\varepsilon) \|f\|_{L_v^2}^2.$$

(M3) The operator L has a finite-dimensional kernel and the following local spectral-gap assumption holds: There exists $\lambda > 0$ such that for all $f \in \text{Dom}(L)$,

$$-(f, L(f))_{L_v^2} \ge \lambda ||f - \Pi^L(f)||_{\mathcal{H}}^2,$$

where Π^L is the projection on Ker(L).

Assumption (M3) is a consequence of Theorem 4.2. Next, we verify assumption (M1). Using Lemma 4.8 and the continuous embedding $\mathcal{H} \hookrightarrow L_v^2$, we see that (4.5.1) holds. For the proof of (4.5.2), we employ Young's inequality:

$$(\nabla_v f, \nabla_v \Lambda(f))_{L_v^2} = \sum_{i=1}^n \int_{\mathbb{R}^3} \nabla_v f_i \cdot \nabla_v (\nu_i f_i) dv$$

$$\begin{split} &= \sum_{i=1}^{n} \left(\int_{\mathbb{R}^{3}} f_{i} \nabla_{v} f_{i} \cdot \nabla_{v} \nu_{i} dv + \int_{\mathbb{R}^{3}} |\nabla_{v} f_{i}|^{2} \nu_{i} dv \right) \\ &\geq \frac{1}{2} \sum_{i=1}^{n} \left(-\int_{\mathbb{R}^{3}} \frac{|\nabla_{v} \nu_{i}|^{2}}{\nu_{i}} f_{i}^{2} dv + \int_{\mathbb{R}^{3}} |\nabla_{v} f_{i}|^{2} \nu_{i} dv \right) \\ &\geq \overline{\nu}_{3} \|\nabla_{v} f\|_{\mathcal{H}}^{2} - \overline{\nu}_{4} \|f\|_{L_{x}^{2}}^{2}, \end{split}$$

where $\overline{\nu}_3 = 1/2$ and $\overline{\nu}_4 = \max_{1 \leq i \leq n} \sup_{v \in \mathbb{R}^3} |\nabla_v \nu_i|^2 / (2\nu_i)$. Note that $\overline{\nu}_4$ is finite since $\nabla_v \nu_i$ is bounded and ν_i is strictly positive (see Lemma 4.6). Finally, inequality (4.5.3) follows from the decomposition $L = K - \Lambda$, the compactness and hence continuity of K, the explicit expression for Λ , and the Cauchy-Schwarz inequality applied to $(L(f), g)_{L^2_v}$.

It remains to verify assumption (M2). Let $N := \rho_{\infty,i}^{-1/2} M_i^{1/2} = (2\pi)^{-3/4} \exp(-|v|^2/4)$. We decompose $K = K^{(1)} - K^{(2)}$, where $K^{(j)} = (K_1^{(j)}, \dots, K_n^{(j)})$ and

$$K_i^{(1)} = \sum_{j=1}^n \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} M_i^{1/2} M_j^* \left(\frac{f_i'}{(M_i')^{1/2}} + \frac{f_j'^*}{(M_j'^*)^{1/2}} \right) dv^* d\sigma,$$

$$K_i^{(2)} = \sum_{j=1}^n \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} (M_i M_j^*)^{1/2} f_j^* dv^* d\sigma$$

for $1 \leq i \leq n$. Because of $M'_k M'^*_k = M_k M^*_k$ for all k, we find that

$$K_{i}^{(1)}(f) = \sum_{j=1}^{n} \int_{\mathbb{R}^{3} \times \mathbb{S}^{2}} B_{ij} M_{i}^{1/2} M_{j}^{*} \left(\frac{(M_{i}^{\prime*})^{1/2} f_{i}^{\prime}}{(M_{i}^{\prime} M_{i}^{\prime*})^{1/2}} + \frac{(M_{j}^{\prime})^{1/2} f_{j}^{\prime*}}{(M_{j}^{\prime*} M_{j}^{\prime})^{1/2}} \right) dv^{*} d\sigma$$

$$= \sum_{j=1}^{n} \int_{\mathbb{R}^{3} \times \mathbb{S}^{2}} B_{ij} M_{i}^{1/2} M_{j}^{\prime} \left(\frac{(M_{i}^{\prime*})^{1/2} f_{i}^{\prime}}{(M_{i} M_{i}^{*})^{1/2}} + \frac{(M_{j}^{\prime})^{1/2} f_{j}^{\prime*}}{(M_{j}^{*} M_{j})^{1/2}} \right) dv^{*} d\sigma$$

$$= \sum_{j=1}^{n} \int_{\mathbb{R}^{3} \times \mathbb{S}^{2}} B_{ij} \left(\rho_{\infty,j}^{1/2} N^{\prime*} f_{i}^{\prime} + \rho_{\infty,i}^{1/2} N^{\prime} f_{j}^{\prime*} \right) \rho_{\infty,j}^{1/2} N^{*} dv^{*} d\sigma.$$

The transformation $\sigma \mapsto -\sigma$ leaves v and v^* unchanged and exchanges v' and v'^* . Assumption (A5) $(b_{ij}$ is an even function) ensures that B_{ij} is unchanged under this transformation. Therefore,

$$\int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} f_i' N'^* N^* dv^* d\sigma = \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} f_i'^* N' N^* dv^* d\sigma,$$

and we can write $K^{(1)}$ as

$$K_i^{(1)}(f) = \frac{1}{2} \sum_{j=1}^n \rho_{\infty,j}^{1/2} \left(\rho_{\infty,j}^{1/2} K_{ij}^{(1)}(f_i) + \rho_{\infty,i}^{1/2} K_{ij}^{(1)}(f_j) \right), \tag{4.5.4}$$

where

$$K_{ij}^{(1)}(f_k) = \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij}(N'^* f_k' + N' f_k'^*) N^* dv^* d\sigma, \quad 1 \le i, j, k \le n.$$

Note that $K_{ij}^{(1)} = K_{ji}^{(1)}$. In a similar way, we can decompose the operator $K^{(2)}$:

$$K_i^{(2)}(f) = \sum_{j=1}^n (\rho_{\infty,i}\rho_{\infty,j})^{1/2} N \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} f_j^* N^* dv^* d\sigma = \sum_{j=1}^n (\rho_{\infty,i}\rho_{\infty,j})^{1/2} K_{ij}^{(2)}(f_j),$$

where

$$K_{ij}^{(2)}(f_j) = N \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} N^* f_j^* dv^* d\sigma.$$

Next, we estimate the derivatives of $K_{ij}^{(\ell)}$. It is shown in [113, Eqs. (5.15)-(5.18)] that for all $\varepsilon > 0$, there exists $C(\varepsilon) > 0$ such that for any $f \in H_v^1$, $1 \le i, j, k \le n$, and $\ell = 1, 2$,

$$\|\nabla_v K_{ij}^{(\ell)}(f_k)\|_{L^2}^2 \le \varepsilon \|\nabla_v f_k\|_{L^2}^2 + C(\varepsilon) \|f_k\|_{L^2}^2. \tag{4.5.5}$$

Then we infer from (4.5.4) that

$$\begin{split} \|\nabla_{v}K^{(1)}(f)\|_{L_{v}^{2}}^{2} &= \sum_{i=1}^{n} \left\| \frac{1}{2} \sum_{j=1}^{n} \rho_{\infty,j}^{1/2} \left(\rho_{\infty,j}^{1/2} \nabla_{v} K_{ij}^{(1)}(f_{i}) + \rho_{\infty,i}^{1/2} \nabla_{v} K_{ij}^{(1)}(f_{j}) \right) \right\|_{L_{v}^{2}}^{2} \\ &\leq \frac{n}{4} \sum_{i,j=1}^{n} \left\| \rho_{\infty,j}^{1/2} \left(\rho_{\infty,j}^{1/2} \nabla_{v} K_{ij}^{(1)}(f_{i}) + \rho_{\infty,i}^{1/2} \nabla_{v} K_{ij}^{(1)}(f_{j}) \right) \right\|_{L_{v}^{2}}^{2} \\ &\leq n (\max_{1 \leq i \leq n} \rho_{\infty,i})^{2} \sum_{i,j=1}^{n} \|\nabla_{v} K_{ij}^{(1)}(f_{i})\|_{L_{v}^{2}}^{2}. \end{split}$$

Thus, by (4.5.5), it follows that for $\ell = 1$,

$$\|\nabla_v K^{(\ell)}(f)\|_{L_v^2}^2 \le n^2 (\max_{1 \le i \le n} \rho_{\infty,i})^2 \sum_{i=1}^n \left(\varepsilon \|\nabla_v f_i\|_{L_v^2}^2 + C(\varepsilon) \|f_i\|_{L_v^2}^2\right).$$

A similar computation shows that this estimate also holds for $\ell=2$. We infer that

$$\|\nabla K(f)\|_{L_v^2}^2 \le 4n^2 (\max_{1 \le i \le n} \rho_{\infty,i})^2 \sum_{i=1}^n \left(\varepsilon \|\nabla_v f_i\|_{L_v^2}^2 + C(\varepsilon) \|f_i\|_{L_v^2}^2 \right).$$

This proves assumption (M2) since $\varepsilon > 0$ is arbitrary.

Proof of Theorem 4.3. We have verified that (M1)-(M3) are satisfied. Then, using exactly the same arguments as in the proof of Theorem 1.1 in [113], but now for the multi-species case, we conclude the exponential decay (4.1.8) of the semigroup $S_G(t)$, which is the first property of the theorem.

It remains to show that the decay estimate (4.1.9) follows from (4.1.8). For this, we write the initial value f_I as $f_I = \Pi^G(f_I) + (I - \Pi^G)(f_I)$, where Π^G is the projection onto Ker(G) in $L^2_{x,v}$. Then the solution to (2.1.5) is given by

$$f(t) = S_G(t)f_I = S_G(t)\Pi^G(f_I) + S_G(t)(I - \Pi^G)(f_I), \quad t \ge 0.$$

We have already shown that

$$||S_G(t)(I - \Pi^G)g||_{H^1_{x,v}} \le Ce^{-\tau t}||g||_{H^1_{x,v}} \quad \text{for all } g \in H^1_{x,v}, \ t > 0.$$

In particular, the choice $g = (I - \Pi^G)(f_I)$ and the property $(I - \Pi^G)^2 = I - \Pi^G$ lead to

$$||S_G(t)(I - \Pi^G)(f_I)||_{H^1_{x,y}} \le Ce^{-\tau t}||(I - \Pi^G)(f_I)||_{H^1_{x,y}}.$$

It remains to prove that $f_{\infty} = \Pi^G(f_I) = S_G(t)\Pi^G(f_I)$ is the global equilibrium. Since $G\Pi^G(f_I) = 0$ and $\Pi^G(f_I)$ does not depend on time, the constant-in-time function $g = \Pi^G(f_I)$ is the unique solution to the Cauchy problem

$$\partial_t g = Gg, \quad t > 0, \quad g(0) = \Pi^G(f_I).$$

This shows that $\Pi^G(f_I) = S_G(t)g(0) = S_G(t)\Pi^G(f_I)$ and finishes the proof.

5. The nonlinear multi-species Boltzmann system close to global equilibrium

In this chapter we use a setting with different molar masses, and a different linearization compared to Chapter 4. Since we also slightly change the notation, let us briefly recap the model used in this chapter. A more detailed introduction for this model can be found in Section 2.2.1.

5.1. The model

5.1.1. Different molar masses

We want to study the evolution of a dilute gas on the torus \mathbb{T}^3 composed of N different species of chemically non-reacting mono-atomic particles with different molar masses, modeled by the system of Boltzmann equations on $\mathbb{R}^+ \times \mathbb{T}^3 \times \mathbb{R}^3$,

$$\forall 1 \le i \le N, \quad \partial_t F_i(t, x, v) + v \cdot \nabla_x F_i(t, x, v) = Q_i(\mathbf{F})(t, x, v) \tag{5.1.1}$$

with initial data

$$\forall 1 \le i \le N, \ \forall (x, v) \in \mathbb{T}^3 \times \mathbb{R}^3, \quad F_i(0, x, v) = F_{0,i}(x, v).$$

The distribution function of the system is given by the vector $\mathbf{F} = (F_1, \dots, F_N)$, with F_i describing the i^{th} species at time t, position x and velocity v.

The Boltzmann operator $\mathbf{Q}(\mathbf{F}) = (Q_1(\mathbf{F}), \dots, Q_N(\mathbf{F}))$ has the form

$$Q_i(\mathbf{F}) = \sum_{j=1}^{N} Q_{ij}(F_i, F_j),$$

where Q_{ij} models interactions between particles of either the same (i = j) or of different $(i \neq j)$ species and are local in time and space.

$$Q_{ij}(F_i, F_j)(v) = \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} \left(|v - v_*|, \cos \theta \right) \left[F_i' F_j'^* - F_i F_j^* \right] dv_* d\sigma,$$

with the shorthands $F'_i = F_i(v')$, $F_i = F_i(v)$, $F'_j = F_j(v'_*)$ and $F^*_j = F_j(v_*)$. Since now we allow also different molar masses for each species, the microscopic collision rules read

as follows:

$$\begin{cases} v' = \frac{1}{m_i + m_j} \left(m_i v + m_j v_* + m_j | v - v_* | \sigma \right) \\ v'_* = \frac{1}{m_i + m_j} \left(m_i v + m_j v_* - m_i | v - v_* | \sigma \right) \end{cases}, \text{ and } \cos \theta = \left\langle \frac{v - v_*}{|v - v_*|}, \sigma \right\rangle.$$

Note that these expressions imply that we deal with gases where only binary elastic collisions occur (the mass m_i of all molecules of species i remains the same, since there is no reaction). Indeed, v' and v'_* are the velocities of two molecules of species i and j before collision giving post-collisional velocities v and v_* respectively, with conservation of momentum and kinetic energy:

$$m_{i}v + m_{j}v_{*} = m_{i}v' + m_{j}v'_{*},$$

$$\frac{1}{2}m_{i}|v|^{2} + \frac{1}{2}m_{j}|v_{*}|^{2} = \frac{1}{2}m_{i}|v'|^{2} + \frac{1}{2}m_{j}|v'_{*}|^{2}.$$
(5.1.2)

5.1.2. The perturbative regime and its motivation

Using the standard changes of variables $(v, v_*) \mapsto (v', v'_*)$ and $(v, v_*) \mapsto (v_*, v)$ (note the lack of symmetry between v' and v'_* compared to v for the second transformation due to different masses) together with the symmetries of the collision operators (see [33][34][138] among others and [52][46] and in particular [21] for multi-species specifically), we recover the following weak forms:

$$\int_{\mathbb{R}^3} Q_{ij}(F_i, F_j)(v)\psi_i(v) dv = \int_{\mathbb{R}^6} \int_{\mathbb{S}^2} B_{ij}(|v - v_*|, \cos(\theta)) F_i F_j^* \left(\psi_i' - \psi_i\right) d\sigma dv dv_*$$

and

$$\int_{\mathbb{R}^{3}} Q_{ij}(F_{i}, F_{j})(v)\psi_{i}(v) dv + \int_{\mathbb{R}^{3}} Q_{ji}(F_{j}, F_{i})(v)\psi_{j}(v) dv =
- \frac{1}{2} \int_{\mathbb{R}^{6}} \int_{\mathbb{S}^{2}} B_{ij}(|v - v_{*}|, \cos(\theta)) \left(F'_{i}F^{*}_{j} - F_{i}F^{*}_{j}\right) \left(\psi'_{i} + \psi'^{*}_{j} - \psi_{i} - \psi^{*}_{j}\right) d\sigma dv dv_{*}.$$
(5.1.3)

Thus

$$\sum_{i,j=1}^{N} \int_{\mathbb{R}^3} Q_{ij}(F_i, F_j)(v)\psi_i(v) \, dv = 0$$
 (5.1.4)

if and only if $\psi(v)$ belongs to Span $\{\mathbf{e_1}, \dots, \mathbf{e_N}, v_1\mathbf{m}, v_2\mathbf{m}, v_3\mathbf{m}, |v|^2\mathbf{m}\}$, where $\mathbf{e_k}$ stands for the k^{th} unit vector in \mathbb{R}^N and $\mathbf{m} = (m_1, \dots, m_N)$. The fact that we need to sum over i has interesting consequences and implies a fundamental difference compared with the single-species Boltzmann equation. In particular it implies conservation of the total

number density $c_{\infty,i}$ of each species, of the total momentum of the gas $\rho_{\infty}u_{\infty}$ and its total energy $3\rho_{\infty}\theta_{\infty}/2$:

$$\forall t \geq 0, \quad c_{\infty,i} = \int_{\mathbb{T}^3 \times \mathbb{R}^3} F_i(t,x,v) \, dx dv \quad (1 \leq i \leq N)$$

$$u_{\infty} = \frac{1}{\rho_{\infty}} \sum_{i=1}^N \int_{\mathbb{T}^3 \times \mathbb{R}^3} m_i v F_i(t,x,v) \, dx dv$$

$$\theta_{\infty} = \frac{1}{3\rho_{\infty}} \sum_{i=1}^N \int_{\mathbb{T}^3 \times \mathbb{R}^3} m_i |v - u_{\infty}|^2 F_i(t,x,v) \, dx dv,$$

$$(5.1.5)$$

where $\rho_{\infty} = \sum_{i=1}^{N} m_i c_{\infty,i}$ is the global density of the gas. Note that this already shows intricate interactions between each species and the total mixture itself.

The operator $\mathbf{Q} = (Q_1, \dots, Q_N)$ also satisfies a multi-species version of the classical H-theorem [52] which implies that any local equilibrium, i.e. any function $\mathbf{F} = (F_1, \dots, F_N)$ being the maximum of the Boltzmann entropy, has the form of a local Maxwellian, that is

$$\forall \ 1 \le i \le N, \ F_i(t, x, v) = c_{\text{loc}, i}(t, x) \left(\frac{m_i}{2\pi k_B \theta_{\text{loc}}(t, x)}\right)^{3/2} \exp\left[-m_i \frac{|v - u_{\text{loc}}(t, x)|^2}{2k_B \theta_{\text{loc}}(t, x)}\right].$$

Here k_B is the Boltzmann constant and, denoting the total local mass density by $\rho_{\text{loc}} = \sum_{i=1}^{N} m_i c_{\text{loc},i}$, we used the following local definitions

$$\forall 1 \le i \le N, \quad c_{\text{loc},i}(t,x) = \int_{\mathbb{R}^3} F_i(t,x,v) \, dv,$$
$$u_{\text{loc}}(t,x) = \frac{1}{\rho_{\text{loc}}} \sum_{i=1}^N \int_{\mathbb{R}^3} m_i v F_i \, dv, \quad \theta_{\text{loc}}(t,x) = \frac{1}{3\rho_{\text{loc}}} \sum_{i=1}^N \int_{\mathbb{R}^3} m_i |v - u_{\text{loc}}|^2 F_i \, dv.$$

On the torus, this multi-species H-theorem also implies that the global equilibrium, i.e. a stationary solution \mathbf{F} to (5.1.1), associated to the initial data $\mathbf{F_0}(x, v) = (F_{0,1}, \dots, F_{0,N})$ is uniquely given by the global Maxwellian

$$\forall \ 1 \le i \le N, \quad F_i(t, x, v) = F_i(v) = c_{\infty, i} \left(\frac{m_i}{2\pi k_B \theta_\infty} \right)^{3/2} \exp \left[-m_i \frac{|v - u_\infty|^2}{2k_B \theta_\infty} \right].$$

By translating and rescaling the coordinate system we can always assume that $u_{\infty} = 0$ and $k_B \theta_{\infty} = 1$ so that the only global equilibrium is the normalized Maxwellian

$$\mu = (\mu_i)_{1 \le i \le N} \quad \text{with} \quad \mu_i(v) = c_{\infty,i} \left(\frac{m_i}{2\pi}\right)^{3/2} e^{-m_i \frac{|v|^2}{2}}.$$
(5.1.6)

The aim is to construct a Cauchy theory for the multi-species Boltzmann equation (5.1.1) around the global equilibrium μ . In other terms we study the existence, uniqueness and exponential decay of solutions of the form $F_i(t, x, v) = \mu_i(v) + f_i(t, x, v)$ for all i.

Under this perturbative regime, the Cauchy problem amounts to solving the perturbed multi-species Boltzmann system of equations

$$\partial_t \mathbf{f} + v \cdot \nabla_x \mathbf{f} = \mathbf{L}(\mathbf{f}) + \mathbf{Q}(\mathbf{f}), \tag{5.1.7}$$

or equivalently in the non-vectorial form

$$\forall 1 \leq i \leq N, \quad \partial_t f_i + v \cdot \nabla_x f_i = L_i(\mathbf{f}) + Q_i(\mathbf{f}),$$

where $\mathbf{f} = (f_1, \dots, f_N)$ and the operator $\mathbf{L} = (L_1, \dots, L_N)$ is the linear Boltzmann operator given for all $1 \le i \le N$ by

$$L_i(\mathbf{f}) = \sum_{j=1}^{N} L_{ij}(f_i, f_j),$$

with

$$L_{ij}(f_i, f_j) = Q_{ij}(\mu_i, f_j) + Q_{ij}(f_i, \mu_j).$$

Since we are looking for solutions \mathbf{F} preserving individual mass, total momentum and total energy (5.1.5) we have the equivalent perturbed conservation laws for $\mathbf{f} = \mathbf{F} - \boldsymbol{\mu}$ which are given by

$$\forall t \ge 0, \quad 0 = \int_{\mathbb{T}^3 \times \mathbb{R}^3} f_i(t, x, v) \, dx dv \quad (1 \le i \le N)$$

$$0 = \sum_{i=1}^N \int_{\mathbb{T}^3 \times \mathbb{R}^3} m_i v f_i(t, x, v) \, dx dv$$

$$0 = \sum_{i=1}^N \int_{\mathbb{T}^3 \times \mathbb{R}^3} m_i |v|^2 f_i(t, x, v) \, dx dv.$$
(5.1.8)

5.1.3. Notations and assumptions on the collision kernel

First, to avoid any confusion, vectors and vector-valued operators in \mathbb{R}^N will be denoted by a bold symbol, whereas their components by the same indexed symbol. For instance, **W** represents the vector or vector-valued operator (W_1, \ldots, W_N) .

We define the Euclidian scalar product in \mathbb{R}^N weighted by a vector **W** by

$$\langle \mathbf{f}, \mathbf{g} \rangle_{\mathbf{W}} = \sum_{i=1}^{N} f_i g_i W_i.$$

In the case $\mathbf{W} = \mathbf{1} = (1, \dots, 1)$ we may omit the index 1.

Function spaces. We define the following shorthand notation

$$\langle v \rangle = \sqrt{1 + |v|^2}.$$

The convention we choose is to index the space by the name of the concerned variable, so we have for p in $[1, +\infty]$

$$L_{\left[0,T\right]}^{p}=L^{p}\left(\left[0,T\right]\right),\quad L_{t}^{p}=L^{p}\left(\mathbb{R}^{+}\right),\quad L_{x}^{p}=L^{p}\left(\mathbb{T}^{3}\right),\quad L_{v}^{p}=L^{p}\left(\mathbb{R}^{3}\right).$$

For $\mathbf{W} = (W_1, \dots, W_N) : \mathbb{R}^3 \longrightarrow \mathbb{R}^+$ a strictly positive measurable function in v, we will use the following vector-valued weighted Lebesgue spaces defined by their norms

$$\begin{split} \|f\|_{L^2_v(\mathbf{W})} &= \left(\sum_{i=1}^N \|f_i\|_{L^2_v(W_i)}^2\right)^{1/2}, \qquad \|f_i\|_{L^2_v(W_i)} = \|f_iW_i(v)\|_{L^2_v}, \\ \|f\|_{L^2_{x,v}(\mathbf{W})} &= \left(\sum_{i=1}^N \|f_i\|_{L^2_{x,v}(W_i)}^2\right)^{1/2}, \quad \|f_i\|_{L^2_{x,v}(W_i)} = \left\|\|f_i\|_{L^2_x}W_i(v)\right\|_{L^2_v}, \\ \|f\|_{L^\infty_{x,v}(\mathbf{W})} &= \sum_{i=1}^N \|f_i\|_{L^\infty_{x,v}(W_i)}, \qquad \|f_i\|_{L^\infty_{x,v}(W_i)} = \sup_{(x,v)\in\mathbb{T}^3\times\mathbb{R}^3} \left(|f_i(x,v)|W_i(v)\right), \\ \|f\|_{L^1_vL^\infty_x(\mathbf{W})} &= \sum_{i=1}^N \|f_i\|_{L^1_vL^\infty_x(W_i)}, \qquad \|f_i\|_{L^1_vL^\infty_x(W_i)} = \left\|\sup_{x\in\mathbb{T}^3} |f_i(x,v)|W_i(v)\right\|_{L^1_v}. \end{split}$$

Note that $L_v^2(\mathbf{W})$ and $L_{x,v}^2(\mathbf{W})$ are Hilbert spaces with respect to the scalar products

$$\begin{split} \langle \mathbf{f}, \mathbf{g} \rangle_{L^2_v(\mathbf{W})} &= \sum_{i=1}^N \langle f_i, g_i \rangle_{L^2_v(W_i)} = \sum_{i=1}^N \int_{\mathbb{R}^3} f_i g_i W_i^2 dv, \\ \langle \mathbf{f}, \mathbf{g} \rangle_{L^2_{x,v}(\mathbf{W})} &= \sum_{i=1}^N \langle f_i, g_i \rangle_{L^2_{x,v}(W_i)} = \sum_{i=1}^N \int_{\mathbb{T}^3 \times \mathbb{R}^3} f_i g_i W_i^2 dx dv. \end{split}$$

Assumptions on the collision kernel.

We will use the following assumptions on the collision kernels B_{ij} .

(H1) The following symmetry holds

$$B_{ij}(|v-v_*|,\cos\theta) = B_{ji}(|v-v_*|,\cos\theta) \quad \text{for } 1 \le i,j \le N.$$

(H2) The collision kernels decompose into the product

$$B_{ii}(|v - v_*|, \cos \theta) = \Phi_{ij}(|v - v_*|)b_{ij}(\cos \theta), \quad 1 \le i, j \le N,$$

where the functions $\Phi_{ij} \geq 0$ are called kinetic part and $b_{ij} \geq 0$ angular part. This is a common assumption as it is technically more convenient and also covers a wide range of physical applications.

(H3) The kinetic part has the form of hard or Maxwellian ($\gamma = 0$) potentials, i.e.

$$\Phi_{ij}(|v-v_*|) = C_{ij}^{\Phi}|v-v_*|^{\gamma}, \quad C_{ij}^{\Phi} > 0, \quad \gamma \in [0,1], \quad \forall \ 1 \le i, j \le N.$$

(H4) For the angular part, we assume a strong form of Grad's angular cutoff (first introduced in [74]), that is: there exist constants C_{b1} , $C_{b2} > 0$ such that for all $1 \le i, j \le N$ and $\theta \in [0, \pi]$,

$$0 < b_{ij}(\cos \theta) \le C_{b1}|\sin \theta| |\cos \theta|, \quad b'_{ij}(\cos \theta) \le C_{b2}.$$

Furthermore,

$$C^b := \min_{1 \le i \le N} \inf_{\sigma_1, \sigma_2 \in \mathbb{S}^2} \int_{\mathbb{S}^2} \min \left\{ b_{ii}(\sigma_1 \cdot \sigma_3), b_{ii}(\sigma_2 \cdot \sigma_3) \right\} d\sigma_3 > 0.$$

We emphasize here that the important cases of Maxwellian molecules ($\gamma = 0$ and b = 1) and of hard spheres ($\gamma = b = 1$) are included in our study. We shall use the standard shorthand notations

$$b_{ij}^{\infty} = \|b_{ij}\|_{L_{[-1,1]}^{\infty}} \quad \text{and} \quad l_{b_{ij}} = \|b \circ \cos\|_{L_{\mathbb{S}^2}^1}.$$
 (5.1.9)

5.1.4. Main results

For the convenience of the reader, we summarize here the main results of this chapter. For a discussion of these results, we refer to Subsection 2.2.4.

Theorem 5.1. Let the collision kernels B_{ij} satisfy assumptions (H1) - (H4). Then the following holds.

(i) The operator **L** is a closed self-adjoint operator in $L_v^2(\mu^{-1/2})$ and there exists $\lambda_L > 0$ such that

$$\forall \mathbf{f} \in L_v^2\left(\boldsymbol{\mu}^{-1/2}\right), \quad \langle \mathbf{f}, \mathbf{L}\left(\mathbf{f}\right) \rangle_{L_v^2\left(\boldsymbol{\mu}^{-1/2}\right)} \leq -\lambda_L \left\| \mathbf{f} - \pi_{\mathbf{L}}\left(\mathbf{f}\right) \right\|_{L_v^2\left(\langle v \rangle^{\gamma/2}\boldsymbol{\mu}^{-1/2}\right)}^2;$$

(ii) Let $E = L_{x,v}^2(\boldsymbol{\mu}^{-1/2})$ or $E = L_{x,v}^{\infty}(\langle v \rangle^{\beta}\boldsymbol{\mu}^{-1/2})$ with $\beta > 3/2$. The linear perturbed operator $\mathbf{G} = \mathbf{L} - v \cdot \nabla_x$ generates a strongly continuous semigroup $S_{\mathbf{G}}(t)$ on E and there exist C_E , $\lambda_E > 0$ such that

$$\forall t \geq 0, \quad \|S_{\mathbf{G}}(t) \left(Id - \Pi_{\mathbf{G}}\right)\|_{E} \leq C_{E}e^{-\lambda_{E}t},$$

where $\pi_{\mathbf{L}}$ is the orthogonal projection onto $Ker(\mathbf{L})$ in $L_v^2(\boldsymbol{\mu}^{-1/2})$ and $\Pi_{\mathbf{G}}$ is the orthogonal projection onto $Ker(\mathbf{G})$ in $L_{x,v}^2(\boldsymbol{\mu}^{-1/2})$.

The constants λ_L , C_E and λ_E are explicit and depend on N, E, the different masses m_i and the collision kernels.

Remark 5.2. This Theorem is split into three parts, and handled separately in Theorem 5.6, Theorem 5.11 and Theorem 5.17.

The next result concerns the nonlinear equation close to global equilibrium.

Theorem 5.3. Let the collision kernels B_{ij} satisfy assumptions (H1) - (H4) and let $E = L_v^1 L_x^{\infty} (\langle v \rangle^k)$ with $k > k_0$, where k_0 is the minimal integer such that

$$C_{k} = \frac{2}{k+2} \frac{1 - \left[\max_{i,j} \frac{|m_{i} - m_{j}|}{m_{i} + m_{j}} \right]^{\frac{k+2}{2}} + \left[1 - \left(\max_{i,j} \frac{|m_{i} - m_{j}|}{m_{i} + m_{j}} \right) \right]^{\frac{k+2}{2}}}{1 - \max_{i,j} \frac{|m_{i} - m_{j}|}{m_{i} + m_{j}}} \max_{i,j} \frac{4\pi b_{ij}^{\infty}}{l_{b_{ij}}} < 1. \quad (5.1.10)$$

where $l_{b_{ij}}$ and b_{ij}^{∞} are angular kernel constants (5.1.9).

Then there exist η_E , C_E and $\lambda_E > 0$ such that for any $\mathbf{F_0} = \boldsymbol{\mu} + \mathbf{f_0} \ge 0$ satisfying the conservation of mass, momentum and energy (5.1.5) with $u_{\infty} = 0$ and $\theta_{\infty} = 1$, if

$$\|\mathbf{F_0} - \boldsymbol{\mu}\| \leq \eta_E$$

then there exists a unique solution $\mathbf{F} = \boldsymbol{\mu} + \mathbf{f}$ in E to the multi-species Boltzmann equation (5.1.1) with initial data $\mathbf{f_0}$. Moreover, \mathbf{F} is non-negative, satisfies the conservation laws and

$$\forall t \geq 0, \quad \|\mathbf{F} - \boldsymbol{\mu}\|_{E} \leq C_{E} e^{-\lambda_{E} t} \|\mathbf{F_{0}} - \boldsymbol{\mu}\|_{E}.$$

The constants are explicit and only depend on N, k, the different masses m_i and the collision kernels.

The proof of this theorem is pressented in Section 5.5.

5.1.5. Organisation of this chapter

Section 5.2 deals with the multi-species spectral gap of **L** in the case of different molar masses. The semigroup property in $L_{x,v}^2\left(\boldsymbol{\mu}^{-1/2}\right)$ is treated in Section 5.3. This property is then passed on to $L_{x,v}^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)$ in Section 5.4. At last, we work out the Cauchy problem for the full nonlinear equation in Section 5.5.

5.2. Spectral gap with different masses

5.2.1. First properties of the linear multi-species Boltzmann operator

We start by describing some properties of the linear multi-species Boltzmann operator $\mathbf{L} = (L_i)_{1 \leq i \leq N}$. First recall

$$L_i(\mathbf{f}) = \sum_{j=1}^{N} L_{ij}(f_i, f_j), \quad 1 \le i \le N,$$

with

$$L_{ij}(f_i, f_j) = Q_{ij}(\mu_i, f_j) + Q_{ij}(f_i, \mu_j)$$

$$= \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij}(|v - v_*|, \cos(\theta)) \left(\mu_j'^* f_i' + \mu_i' f_j'^* - \mu_j^* f_i - \mu_i f_j^* \right) dv_* d\sigma,$$

where we have used $\mu_i^{'*}\mu_j' = \mu_i^*\mu_j$ for any i, j, which follows from the laws of elastic collisions (5.1.2).

In Chapter 4 (see also [46]), we proved an explicite multi-species spectral gap in the case of species having same mass $(m_i = m_j)$. Most of the proofs are directly applicable in the case of different masses, and we therefore refer to this chapter for detailed proofs.

L is a self-adjoint operator in $L_v^2(\boldsymbol{\mu}^{-1/2})$ with $\langle \mathbf{f}, \mathbf{L}(\mathbf{f}) \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} = 0$ if and only if **f** belongs to $\text{Ker}(\mathbf{L})$.

$$\operatorname{Ker}(\mathbf{L}) = \operatorname{Span}\left\{\phi_1(v), \dots, \phi_{N+4}(v)\right\},\,$$

where $(\phi_i)_{1 \leq i \leq N+4}$ is an orthonormal basis of Ker (**L**) in $L_v^2(\boldsymbol{\mu}^{-1/2})$. More precisely, if we denote $\pi_{\mathbf{L}}$ the orthogonal projection onto Ker (**L**) in $L_v^2(\boldsymbol{\mu}^{-1/2})$:

$$\pi_{\mathbf{L}}(\mathbf{f}) = \sum_{k=1}^{N+4} \left(\int_{\mathbb{R}^3} \langle \mathbf{f}(v), \boldsymbol{\phi}_k(v) \rangle_{\boldsymbol{\mu}^{-1/2}} \, dv \right) \boldsymbol{\phi}_k(v),$$

and

$$\mathbf{e}_{\mathbf{k}} = (\delta_{ik})_{1 \le i \le N} \,,$$

we can write

$$\begin{cases}
\phi_{k}(v) = \frac{1}{\sqrt{c_{\infty,k}}} \mu_{k} \mathbf{e_{k}}, & 1 \leq k \leq N \\
\phi_{k}(v) = \frac{v_{k-N}}{\left(\sum_{i=1}^{N} m_{i} c_{\infty,i}\right)^{1/2}} (m_{i} \mu_{i})_{1 \leq i \leq N}, & N+1 \leq k \leq N+3. \\
\phi_{N+4}(v) = \frac{1}{\left(\sum_{i=1}^{N} c_{\infty,i}\right)^{1/2}} \left(\frac{|v|^{2} - 3m_{i}^{-1}}{\sqrt{6}} m_{i} \mu_{i}\right)_{1 \leq i \leq N}.
\end{cases} (5.2.1)$$

Finally, we denote $\pi_{\mathbf{L}}^{\perp} = \operatorname{Id} - \pi_{\mathbf{L}}$. The projection $\pi_{\mathbf{L}}(\mathbf{f}(t, x, \cdot))(v)$ of $\mathbf{f}(t, x, v)$ onto the kernel of \mathbf{L} is called its fluid part whereas $\pi_{\mathbf{L}}^{\perp}(\mathbf{f})$ is its microscopic part.

L can be written under the following form

$$\mathbf{L} = -\boldsymbol{\nu}(v) + \mathbf{K},\tag{5.2.2}$$

where $\boldsymbol{\nu} = (\nu_i)_{1 \leq i \leq N}$ is a multiplicative operator called the collision frequency

$$\nu_i(v) = \sum_{j=1}^{N} \nu_{ij}(v), \tag{5.2.3}$$

with

$$\nu_{ij}(v) = C_{ij}^{\Phi} \int_{\mathbb{R}^3 \times \mathbb{S}^2} b_{ij} (\cos \theta) |v - v_*|^{\gamma} \mu_j(v_*) d\sigma dv_*.$$

Each of the ν_{ij} could be seen as the collision frequency $\nu(v)$ of a single-species Boltzmann kernel with kernel B_{ij} . It is well-known (for instance [33][34][138][78]) that under our assumptions: $\nu(v) \sim (1+|v|^{\gamma}) \sim \langle v \rangle^{\gamma}$. This means that for all i, j there exist $\nu_{ij}^{(0)}, \nu_{ij}^{(1)} > 0$ (they are explicit, see the references above) such that

$$\forall v \in \mathbb{R}^3, \quad \nu_{ij}^{(0)} (1 + |v|^{\gamma}) \le \nu_{ij}(v) \le \nu_{ij}^{(1)} (1 + |v|^{\gamma}),$$

Every constant being strictly positive, the following lemma follows straightforwardly.

Lemma 5.4. There exists a constant $\beta > 0$, and for all i in $\{1, ..., N\}$ there exist $\nu_i^{(0)}, \nu_i^{(1)} > 0$ such that

$$\forall v \in \mathbb{R}^3, \quad \nu_i^{(0)} \left(1 + |v|^{\gamma} \right) \le \nu_i(v) \le \nu_i^{(1)} \left(1 + |v|^{\gamma} \right). \tag{5.2.4}$$

Thus, we get the following relation between the collision frequencies

$$\forall v \in \mathbb{R}^3, \quad \nu_i(v) \le \beta \nu_{ii}(v). \tag{5.2.5}$$

Remark 5.5. Estimate (5.2.5) is a crucial step in the proof of Lemma 5.7. In Chapter 4 (see also [46]), the additional assumption $B_{ij} \leq CB_{ii}$ for a constant C > 0 has been used in order to get (5.2.5). We want to point out that despite of even having different masses to handle, we manage to get rid of this assumption. The prize we have to pay is a slightly more restrictive assumption on the collision kernel B in assumption (H3).

Next we decompose the operator \mathbf{L} into its mono-species part $\mathbf{L}^{\mathbf{m}} = (L_i^m)_{1 \leq i \leq N}$ and its bi-species part $\mathbf{L}^{\mathbf{b}} = (L_i^b)_{1 \leq i \leq N}$ according to

$$\mathbf{L} = \mathbf{L}^{\mathbf{m}} + \mathbf{L}^{\mathbf{b}}, \quad L_i^m(f_i) = L_{ii}(f_i, f_i), \quad L_i^b(f) = \sum_{i \neq i} L_{ij}(f_i, f_j).$$
 (5.2.6)

Thus \mathbf{f} can be written as

$$\mathbf{f} = \pi_{\mathbf{L}^{\mathbf{m}}}(\mathbf{f}) + \pi_{\mathbf{L}^{\mathbf{m}}}^{\perp}(\mathbf{f}), \tag{5.2.7}$$

where $\pi_{\mathbf{L}^{\mathbf{m}}}$ is the orthogonal projection on $\operatorname{Ker}(\mathbf{L}^{\mathbf{m}})$ with respect to $L_v^2(\boldsymbol{\mu}^{-1/2})$, and

$$\pi_{\mathbf{L}^{\mathbf{m}}}^{\perp} := (1 - \pi_{\mathbf{L}^{\mathbf{m}}})$$
.

By employing the standard change of variables, the Dirichlet forms of $\mathbf{L^m}$ and $\mathbf{L^b}$ have the form

$$\langle \mathbf{f}, \mathbf{L}^{\mathbf{m}}(\mathbf{f}) \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} = -\frac{1}{4} \sum_{i=1}^{N} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ii} \mu_{i} \mu_{i}^{*} \left(A_{ii} \left[f_{i} \mu_{i}^{-1}, f_{i} \mu_{i}^{-1} \right] \right)^{2},$$
 (5.2.8)

$$\left\langle \mathbf{f}, \mathbf{L}^{\mathbf{b}}(\mathbf{f}) \right\rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} = -\frac{1}{4} \sum_{i=1}^N \sum_{j \neq i} \int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} \mu_i \mu_j^* \left(A_{ij} \left[f_i \mu_i^{-1}, f_j \mu_j^{-1} \right] \right)^2, \tag{5.2.9}$$

with the shorthands

$$A_{ij}\left[f_i\mu_i^{-1}, f_j\mu_j^{-1}\right] := \left(f_i\mu_i^{-1}\right)' + \left(f_j\mu_j^{-1}\right)'^* - \left(f_i\mu_i^{-1}\right) - \left(f_j\mu_j^{-1}\right)^*. \tag{5.2.10}$$

Since L^m describes a multi-species operator when all the cross-interactions are null,

$$\pi_{\mathbf{L}^{\mathbf{m}}}(\mathbf{f})_{i} = m_{i}\mu_{i}(v)(a_{i}(t, x) + u_{i}(t, x) \cdot v + e_{i}(t, x)|v|^{2}), \quad 1 \le i \le N,$$
(5.2.11)

where $a_i \in \mathbb{R}, u_i \in \mathbb{R}^3$ and $e_i \in \mathbb{R}$ are the coordinates of $\pi_{\mathbf{L}^{\mathbf{m}}}(\mathbf{f})$ with respect to a 5N-dimensional basis, while

$$\pi_{\mathbf{L}}(\mathbf{f})_i = m_i \mu_i(v) (a_i(t, x) + u(t, x) \cdot v + e(t, x)|v|^2) \quad 1 \le i \le N, \tag{5.2.12}$$

where $a_i \in \mathbb{R}$, $u \in \mathbb{R}^3$ and $e \in \mathbb{R}$ are the coordinates of $\pi_{\mathbf{L}}(\mathbf{f})$ with respect to an (N+4)-dimensional basis.

Finally, since

$$\int_{\mathbb{R}^3} \mu_i \, dv = c_i, \quad \int_{\mathbb{R}^3} \mu_i |v|^2 \, dv = 3c_i m_i^{-1}, \quad \int_{\mathbb{R}^3} \mu_i |v|^4 \, dv = 15c_i m_i^{-2}, \tag{5.2.13}$$

the following moment identities hold for a_i, u_i, e_i defined in (5.2.11)

$$\int_{\mathbb{R}^3} f_i \, dv = c_i (m_i a_i + 3e_i),$$

$$\int_{\mathbb{R}^3} f_i v \, dv = c_i u_i,$$

$$\int_{\mathbb{R}^3} f_i |v|^2 \, dv = c_i (3a_i + 15e_i m_i^{-1}).$$
(5.2.14)

5.2.2. Explicit spectral gap

This subsection is devoted to the proof of the following constructive spectral gap estimate for the multi-species linear operator \mathbf{L} with different masses.

Theorem 5.6. Let the collision kernels B_{ij} satisfy assumptions (H1)-(H4). Then there exists an explicit constant $\lambda_L > 0$ such that

$$\langle \mathbf{f}, \mathbf{L}(\mathbf{f}) \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} \le -\lambda_L \|\mathbf{f} - \pi_{\mathbf{L}}(\mathbf{f})\|_{L_v^2(\langle v \rangle^{\gamma/2} \boldsymbol{\mu}^{-1/2})}^2 \quad \forall \mathbf{f} \in Dom(\mathbf{L}),$$

where λ_L depends on the properties of the collision kernel, the number of species N and the different masses.

The next two lemmas are crucial for the proof of Theorem 5.6, generalizing the strategy of [46] presented in Chapter 4 to the case of different masses. The key idea is to decompose \mathbf{L} into $\mathbf{L} = \mathbf{L^m} + \mathbf{L^b}$ (see (5.2.6)), and to derive separately a spectral-gap estimate for the mono-species part $\mathbf{L^m}$ on its domain $\mathrm{Dom}(\mathbf{L^m})$ (see Lemma 5.7), and a spectral-gap type estimate for the bi-species part $\mathbf{L^b}$ on $\mathrm{Ker}(\mathbf{L^m})$ (see Lemma 5.8) measured in terms of the following functional

$$\mathcal{E}: \operatorname{Ker}(\mathbf{L^m}) \to \mathbb{R}^+, \quad \mathcal{E}(\mathbf{f}) := \sum_{i,j=1}^N \left(\left| u_i^{(f)} - u_j^{(f)} \right|^2 + \left(e_i^{(f)} - e_j^{(f)} \right)^2 \right),$$

where for a fixed $\mathbf{f} \in \mathrm{Ker}(\mathbf{L^m})$, $u_i^{(f)}$ and $e_i^{(f)}$ describe the coordinates of the i^{th} component of \mathbf{f} with respect to the basis defined in (5.2.11). To lighten computations, we introduce the following Hilbert space $\mathcal{H} := L_v^2 \left(\boldsymbol{\nu}^{1/2} \boldsymbol{\mu}^{-1/2} \right)$, which is equivalent to $L_v^2 \left(\langle v \rangle^{\gamma/2} \boldsymbol{\mu}^{-1/2} \right)$:

$$\mathcal{H} = \left\{ f \in L_v^2(\boldsymbol{\mu}^{-1/2}) : \|\mathbf{f}\|_{\mathcal{H}}^2 = \sum_{i=1}^N \int_{\mathbb{R}^3} f_i^2 \nu_i \mu_i^{-1} \, dv < \infty \right\}.$$
 (5.2.15)

Lemma 5.7. For all \mathbf{f} in $Dom(\mathbf{L^m})$ there exists an explicit constant $C_1 > 0$, such that

$$\langle \mathbf{f}, \mathbf{L}^{\mathbf{m}}(\mathbf{f}) \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} \leq -C_1 \|\mathbf{f} - \pi_{\mathbf{L}^{\mathbf{m}}}(\mathbf{f})\|_{L_v^2(\langle v \rangle^{\gamma/2}\boldsymbol{\mu}^{-1/2})}^2,$$

where C_1 depends on the properties of the collision kernel, the number of species N and the different masses.

Proof. By [111, Theorem 1.1, Remark 1] together with the shorthand introduced in (5.2.10),

$$\frac{1}{4} \int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ii} \left(A_{ii} \left[f_i \mu_i^{-1}, f_i \mu_i^{-1} \right] \right)^2 \mu_i \mu_i^* \, dv dv_* d\sigma \ge \lambda_m c_{\infty,i} \int_{\mathbb{R}^3} (f_i - \pi_{\mathbf{L}^{\mathbf{m}}}(\mathbf{f})_i)^2 \nu_{ii} \mu_i^{-1} \, dv,$$

where $\lambda_m > 0$ depends on the properties of the collision kernel, the number of species N and the different masses. Summing this estimate over i = 1, ..., N and employing (5.2.9) yields

$$-\left\langle \mathbf{f}, \mathbf{L}^{\mathbf{m}}(\mathbf{f}) \right\rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} \ge \lambda^m \sum_{i=1}^N c_{\infty,i} \int_{\mathbb{R}^3} (f_i - \pi_{\mathbf{L}^{\mathbf{m}}}(f_i))^2 \frac{\nu_{ii}}{\mu_i} \, dv. \tag{5.2.16}$$

Now we can estimate ν_{ii} in terms of ν_i by using (5.2.5), and plugging this bound into (5.2.16) together with the fact that \mathcal{H} is equivalent to $L_v^2(\langle v \rangle^{\gamma/2} \mu^{-1/2})$ finishes the proof.

Lemma 5.8. For all \mathbf{f} in $Ker(\mathbf{L^m}) \cap Dom(\mathbf{L^b})$ there exists an explicit $C_2 > 0$ such that

$$\left\langle \mathbf{f}, \mathbf{L}^{\mathbf{b}}(\mathbf{f}) \right\rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} \leq -C_{2} \, \mathcal{E}(\mathbf{f}),$$

with the functional \mathcal{E} defined by

$$\mathcal{E}: \operatorname{Ker}(\mathbf{L}^{\mathbf{m}}) \to \mathbb{R}^{+}, \quad \mathcal{E}(\mathbf{f}) := \sum_{i,j=1}^{N} \left(\left| u_{i}^{(f)} - u_{j}^{(f)} \right|^{2} + \left(e_{i}^{(f)} - e_{j}^{(f)} \right)^{2} \right), \tag{5.2.17}$$

where for fixed $\mathbf{f} \in Ker(\mathbf{L^m})$ it holds that $u_i^{(f)}$, $e_i^{(f)}$ describe the coordinates of the i^{th} component of \mathbf{f} with respect to the basis defined in (5.2.11), and $C_2 > 0$ is defined in (5.2.19).

Remark 5.9. Note that for f in Ker(L^m) it holds that

$$\mathcal{E}(\mathbf{f}) = 0 \quad \Leftrightarrow \quad \mathbf{f} \in Ker(\mathbf{L}^{\mathbf{b}}).$$

since $Ker(\mathbf{L}) = Ker(\mathbf{L^m}) \cap Ker(\mathbf{L^b})$. This fact together with a multi-species version of the H-theorem show that the left-hand side of the estimate in Lemma 5.8 is null if and only if the right-hand side is null.

Proof. Let $\mathbf{f} \in \text{Ker}(\mathbf{L^m}) \cap \text{Dom}(\mathbf{L^b})$. Writing \mathbf{f} in the form (5.2.11) and applying the microscopic conservation laws (5.1.2) yields

$$A_{ij}[f_i\mu_i^{-1}, f_j\mu_j^{-1}] = m_i(u_i - u_j) \cdot (v' - v) + m_i(e_i - e_j)(|v'|^2 - |v|^2),$$

and thus

$$-\langle \mathbf{f}, \mathbf{L}^{\mathbf{b}}(\mathbf{f}) \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})}$$

$$= \frac{1}{4} \sum_{\substack{i,j=1\\j\neq i}}^{N} m_{i}^{2} \int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij} \left[(u_{i} - u_{j}) \cdot (v' - v) + (e_{i} - e_{j})(|v'|^{2} - |v|^{2}) \right]^{2} \mu_{i} \mu_{j}^{*}.$$

Using the symmetry of B_{ij} and of $\mu_i \mu_j^*$ together with the oddity of the function $G(v, v_*, \sigma) = B_{ij}(u_i - u_j) \cdot (v' - v)(|v'|^2 - |v|^2)$ with respect to (v, v_*, σ) yields that the mixed term in the square of the integral above vanishes. Thus we obtain

$$-\langle \mathbf{f}, \mathbf{L}^{\mathbf{b}}(\mathbf{f}) \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} = \frac{1}{4} \sum_{\substack{i,j=1\\j\neq i}}^{N} m_{i}^{2} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij}$$

$$\times \left(|(u_{i} - u_{j}) \cdot (v' - v)|^{2} + (e_{i} - e_{j})^{2} (|v'|^{2} - |v|^{2})^{2} \right) \mu_{i} \mu_{i}^{*} \, dv dv_{*} d\sigma.$$
(5.2.18)

We claim that the following holds

$$\int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} ((u_i - u_j) \cdot (v' - v))^2 \mu_i \mu_j^* \, dv dv_* d\sigma = \frac{|u_i - u_j|^2}{3} \int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} |v - v'|^2 \mu_i \mu_j^* \, dv dv_* d\sigma.$$

To prove this identity, we write $u_{i,k}$ and v_k for the kth component of the vectors u_i and v, respectively. The change of variables $(v_k, v_k^*, \sigma_k) \mapsto -(v_k, v_k^*, \sigma_k)$ for fixed k leaves B_{ij} , μ_i , and μ_j^* unchanged but $v_k' \mapsto -v_k'$, such that

$$\int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} v_k' v_\ell \mu_i \mu_j^* \, dv dv_* d\sigma = 0 \quad \text{for } \ell \neq k.$$

Moreover,

$$\int_{\mathbb{R}^6 \times \mathbb{S}^2} B_{ij} v_k v_\ell \mu_i \mu_j^* \, dv dv_* d\sigma = 0 \quad \text{for } \ell \neq k,$$

since the integrand is odd. Thus,

$$\int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij}((u_{i}-u_{j})\cdot(v'-v))^{2}\mu_{i}\mu_{j}^{*} dv dv_{*} d\sigma$$

$$= \sum_{k,\ell=1}^{3} (u_{i,k}-u_{j,k})(u_{i,\ell}-u_{j,\ell}) \int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij}(v'_{k}-v_{k})(v'_{\ell}-v_{\ell})\mu_{i}\mu_{j}^{*} dv dv_{*} d\sigma$$

$$= \sum_{k=1}^{3} (u_{i,k}-u_{j,k})^{2} \int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij}(v_{k}-v'_{k})^{2}\mu_{i}\mu_{j}^{*} dv dv_{*} d\sigma.$$

Since the integral is independent of k, we get

$$\int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij}((u_{i}-u_{j})\cdot(v'-v))^{2}\mu_{i}\mu_{j}^{*} dv dv_{*} d\sigma$$

$$= \frac{1}{3} \sum_{k=1}^{3} (u_{i,k}-u_{j,k})^{2} \int_{\mathbb{R}^{6}\times\mathbb{S}^{2}} B_{ij}|v-v'|^{2}\mu_{i}\mu_{j}^{*} dv dv_{*} d\sigma,$$

which proves the claim.

This implies that for all \mathbf{f} in $\mathrm{Ker}(\mathbf{L^m}) \cap \mathrm{Dom}(\mathbf{L^b})$ it holds that

$$\langle \mathbf{f}, \mathbf{L}^{\mathbf{b}}(\mathbf{f}) \rangle_{L^{2}_{n}(\boldsymbol{\mu}^{-1/2})} \leq -C_{2} \, \mathcal{E}(\mathbf{f}),$$

where $\mathcal{E}(\cdot)$ is defined in (5.2.17) and

$$C_2 = \frac{1}{4} \min_{1 \le i, j \le n} \int_{\mathbb{R}^6 \times \mathbb{S}^2} m_i^2 B_{ij} \min \left\{ \frac{1}{3} |v - v'|^2, (|v'|^2 - |v|^2)^2 \right\} \mu_i \mu_j^* \, dv dv_* d\sigma. \tag{5.2.19}$$

The last part is to prove that $C_2 > 0$. For this we note that the integrand of (5.2.19) vanishes if and only if |v'| = |v|. However, the set

$$X = \{(v, v_*, \sigma) \in \mathbb{R}^3 \times \mathbb{R}^3 \times \mathbb{S}^2 : |v'| = |v|\}$$

is closed since it is the pre-image of $\{0\}$ of the function $H(v, v_*, \sigma) = |v'|^2 - |v|^2$ which is continuous. Now X^c is open and nonempty and thus has positive Lebesgue measure, and since the integrand in (5.2.19) is positive on X^c , we get that $C_2 > 0$, which finishes the proof.

Proof of Theorem 5.6. The proof will be performed in 4 steps, using the strategy worked out in Chapter 4 in order to get an explicit spectral-gap estimate, but now for different molar masses. To lighten notation, we will use the following shorthands for $\mathbf{f} \in \text{Dom}(\mathbf{L})$:

$$\mathbf{f}^{\parallel} = \pi_{\mathbf{L}^{\mathbf{m}}}(\mathbf{f}), \quad \mathbf{f}^{\perp} = \mathbf{f} - \mathbf{f}^{\parallel}, \quad h_i^{\parallel} = \mu_i^{-1} f_i^{\parallel}, \quad h_i^{\perp} = \mu_i^{-1} h_i^{\perp}.$$
 (5.2.20)

Step 1: Absorption of the orthogonal part.

The nonnegativity of $-\langle \mathbf{f}, \mathbf{L}^{\mathbf{b}}(\mathbf{f}) \rangle_{L_{n}^{2}(\boldsymbol{\mu}^{-1/2})} \geq 0$ and Lemma 5.7 imply that

$$-\langle (\mathbf{f}, \mathbf{L}(\mathbf{f})) \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} \ge C_1 \|\mathbf{f} - \mathbf{f}^{\parallel}\|_{\mathcal{H}}^2 - \eta \langle \mathbf{f}, \mathbf{L}^{\mathbf{b}}(\mathbf{f}) \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})}, \tag{5.2.21}$$

where $\eta \in (0,1]$ and $C_1 > 0$ was defined in Lemma 5.7. Now it holds that

$$A_{ij}[h_i, h_j]^2 \ge \frac{1}{2} A_{ij}[h_i^{\parallel}, h_j^{\parallel}]^2 - A_{ij}[h_i^{\perp}, h_j^{\perp}]^2,$$

and plugging this into (5.2.9) and (5.2.21) implies

$$-\langle \mathbf{f}, \mathbf{L}(\mathbf{f}) \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} \geq C_{1} \| f^{\perp} \|_{\mathcal{H}}^{2} + \frac{\eta}{8} \sum_{i=1}^{N} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} A_{ij} [h_{i}^{\parallel}, h_{j}^{\parallel}]^{2} \mu_{i} \mu_{j}^{*} \, dv dv_{*} d\sigma$$
$$- \frac{\eta}{4} \sum_{i=1}^{N} \sum_{j \neq i}^{N} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} A_{ij} [h_{i}^{\perp}, h_{j}^{\perp}]^{2} \mu_{i} \mu_{j}^{*} \, dv dv_{*} d\sigma. \tag{5.2.22}$$

Now we prove that (up to a small factor) the last term on the right-hand side can be estimated from below by $\|\mathbf{f}^{\perp}\|_{\mathcal{H}}^2$. For this we perform the standard change of variables $(v, v_*) \to (v_*, v)$ together with $i \leftrightarrow j$ and $(v, v_*) \to (v', v'_*)$, and by using the identity $\mu_i \mu_i^* = \mu'_i \mu'_i^*$ we obtain

$$\sum_{i=1}^{N} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} A_{ij} [h_{i}^{\perp}, h_{j}^{\perp}]^{2} \mu_{i} \mu_{j}^{*} dv dv_{*} d\sigma$$

$$\leq 4 \sum_{i=1}^{N} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} \left(((h_{i}^{\perp})')^{2} + ((h_{j}^{\perp})'^{*})^{2} + (h_{i}^{\perp})^{2} + ((h_{j}^{\perp})^{*})^{2} \right) \mu_{i} \mu_{j}^{*} dv dv_{*} d\sigma$$

$$\leq 16 \sum_{i=1}^{N} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} (h_{i}^{\perp})^{2} \mu_{i} \mu_{j}^{*} dv dv_{*} d\sigma.$$

Taking into account the definition (5.2.3) of ν_i , we get for the last term on the right-hand side of (5.2.22)

$$-\frac{\eta}{4} \sum_{i=1}^{N} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} A_{ij} [h_{i}^{\perp}, h_{j}^{\perp}]^{2} \mu_{i} \mu_{j}^{*} dv dv_{*} d\sigma$$

$$\geq -4\eta \sum_{i,j=1}^{N} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} (f_{i}^{\perp})^{2} \mu_{j}^{*} \mu_{i}^{-1} dv dv_{*} d\sigma$$

$$\geq -4\eta \sum_{i=1}^{N} \int_{\mathbb{R}^{3}} (f_{i}^{\perp})^{2} \nu_{i} \mu_{i}^{-1} dv = -4\eta \|f^{\perp}\|_{\mathcal{H}}^{2}.$$

Finally (5.2.22) yields

$$-\langle \mathbf{f}, \mathbf{L}(\mathbf{f}) \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} \geq (C_{1} - 4\eta) \left\| \mathbf{f} - \mathbf{f}^{\parallel} \right\|_{\mathcal{H}}^{2}$$
$$+ \frac{\eta}{8} \sum_{i=1}^{N} \sum_{j \neq i} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} B_{ij} A_{ij} \left[h_{i}^{\parallel}, h_{j}^{\parallel} \right]^{2} \mu_{i} \mu_{j}^{*} dv dv_{*} d\sigma.$$

Thus

$$\langle \mathbf{f}, \mathbf{L}(\mathbf{f}) \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} \le -(C_1 - 4\eta) \left\| \mathbf{f} - \mathbf{f}^{\parallel} \right\|_{\mathcal{H}}^2 + \frac{\eta}{2} \langle \mathbf{f}^{\parallel}, \mathbf{L}^{\mathbf{b}}(\mathbf{f}^{\parallel}) \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})}, \tag{5.2.23}$$

where $0 < \eta \le \min\{1, C_1/8\}$.

Step 2: Estimate for for the remaining part. Due to Lemma 5.8 there exists an explicit $C_2 > 0$ such that

$$\left\langle \mathbf{f}^{\parallel}, \mathbf{L}^{\mathbf{b}}(\mathbf{f}^{\parallel}) \right\rangle_{L^{2}_{v}\left(\boldsymbol{\mu}^{-1/2}\right)} \leq -C_{2}\,\mathcal{E}\left(\mathbf{f}^{\parallel}\right).$$

Step 3: Estimate for the momentum and energy differences.

We need to find a relation between $\mathcal{E}(\mathbf{f}^{\parallel})$, $\|\mathbf{f} - \mathbf{f}^{\parallel}\|$ and $\|\mathbf{f} - \pi_{\mathbf{L}}(\mathbf{f})\|$ respectively. To this end, we decompose $\mathbf{f} = \mathbf{f}^{\parallel} + \mathbf{f}^{\perp}$ recalling that $\mathbf{f}^{\parallel} = \pi_{\mathbf{L}^{\mathbf{m}}}(\mathbf{f})$ and $\mathbf{f}^{\perp} = \mathbf{f} - \mathbf{f}^{\parallel}$. Using an arbitrary orthonormal basis $(\psi_k)_{1 \leq k \leq 5N}$ of $\operatorname{Ker}(\mathbf{L}^m)$ in $L_v^2(\boldsymbol{\mu}^{-1/2})$, we first show that

$$\|\mathbf{f} - \pi_{\mathbf{L}}(\mathbf{f})\|_{\mathcal{H}}^{2} \leq 2\|\mathbf{f}^{\perp}\|_{\mathcal{H}}^{2} + k_{0} \left(\|\mathbf{f}^{\parallel}\|_{L_{n}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} - \|\pi_{\mathbf{L}}(\mathbf{f})\|_{L_{n}^{2}(\boldsymbol{\mu}^{-1/2})}^{2}\right), \tag{5.2.24}$$

where $k_0 = 10N \max_{1 \leq k, \ell \leq 5N} |\langle \boldsymbol{\psi_k}, \boldsymbol{\psi_\ell} \rangle_{\mathcal{H}}|$.

To this end, we start with

$$\|\mathbf{f} - \pi_{\mathbf{L}}(\mathbf{f})\|_{\mathcal{H}}^{2} \le 2\left(\|\mathbf{f}^{\perp}\|_{\mathcal{H}}^{2} + \|\mathbf{f}^{\parallel} - \pi_{\mathbf{L}}(\mathbf{f})\|_{\mathcal{H}}^{2}\right). \tag{5.2.25}$$

Denoting the last term by $\mathbf{g} := \mathbf{f}^{\parallel} - \pi_{\mathbf{L}}(\mathbf{f}) \in \text{Ker}(\mathbf{L}^m)$ (note that $\text{Ker}(\mathbf{L}) \subset \text{Ker}(\mathbf{L}^m)$) and using Young's inequality implies

$$\begin{aligned} \|\mathbf{g}\|_{\mathcal{H}}^{2} &= \sum_{i=1}^{N} \int_{\mathbb{R}^{3}} \left| \sum_{k=1}^{5N} \langle \mathbf{g}, \boldsymbol{\psi_{k}} \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} \psi_{k,i} \right|^{2} \nu_{i}(v) \, dv \\ &= \sum_{k,\ell=1}^{5N} \langle \mathbf{g}, \boldsymbol{\psi_{k}} \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} \langle \mathbf{g}, \boldsymbol{\psi_{\ell}} \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} \langle \boldsymbol{\psi_{k}}, \boldsymbol{\psi_{\ell}} \rangle_{\mathcal{H}} \\ &\leq \frac{1}{2} \max_{1 \leq k,\ell \leq 5N} \left| \langle \boldsymbol{\psi_{k}}, \boldsymbol{\psi_{\ell}} \rangle_{\mathcal{H}} \right| \sum_{k,\ell=1}^{5N} \left(\langle \mathbf{g}, \boldsymbol{\psi_{k}} \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} + \langle \mathbf{g}, \boldsymbol{\psi_{\ell}} \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} \right) \\ &= 5N \max_{1 \leq k,\ell \leq 5N} \left| \langle \boldsymbol{\psi_{k}}, \boldsymbol{\psi_{\ell}} \rangle_{\mathcal{H}} \right| \|\mathbf{g}\|_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2}. \end{aligned}$$

Thus, (5.2.25) implies

$$\|\mathbf{f} - \pi_{\mathbf{L}}(\mathbf{f})\|_{\mathcal{H}}^{2} \leq 2\|\mathbf{f}^{\perp}\|_{\mathcal{H}}^{2} + 10N \max_{1 \leq k, \ell \leq 5N} |\langle \boldsymbol{\psi}_{\boldsymbol{k}}, \boldsymbol{\psi}_{\boldsymbol{\ell}} \rangle_{\mathcal{H}}| \|\mathbf{f}^{\parallel} - \pi_{\mathbf{L}}(\mathbf{f})\|_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2}.$$

Now $Ker(\mathbf{L}) \subset Ker(\mathbf{L^m})$ implies $\pi_{\mathbf{L^m}}\pi_{\mathbf{L}} = \pi_{\mathbf{L}}$, thus

$$\|\mathbf{f}^{\parallel} - \pi_{\mathbf{L}}(\mathbf{f})\|_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} = \|\mathbf{f}^{\parallel}\|_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} - \|\pi_{\mathbf{L}}(\mathbf{f})\|_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2},$$

which indeed yields (5.2.24).

Now the moment identities (5.2.13) and (5.2.14) yield

$$\|\mathbf{f}^{\parallel}\|_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} = \sum_{i=1}^{N} c_{\infty,i}(m_{i}^{2}a_{i}^{2} + m_{i}|u_{i}|^{2} + 15e_{i}^{2} + 6m_{i}a_{i}e_{i}),$$

and

$$\|\pi_{\mathbf{L}}(\mathbf{f})\|_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} = \sum_{j=1}^{N+4} \langle \mathbf{f}, \phi_{\mathbf{j}} \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2}$$

$$= \sum_{i=1}^{N} c_{\infty,i} (m_{i}a_{i} + 3e_{i})^{2} + \rho_{\infty} \left| \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} u_{i} \right|^{2} + 6c_{\infty} \left(\sum_{i=1}^{N} \frac{c_{\infty,i}}{c_{\infty}} e_{i} \right)^{2},$$

where $(\phi_j)_{1 < j < N+4}$ is the orthonormal basis of $\operatorname{Ker}(\mathbf{L})$ in $L_v^2(\boldsymbol{\mu}^{-1/2})$ introduced in (5.2.1).

Inserting these expressions into (5.2.24), we conclude that

$$\|\mathbf{f} - \pi_{\mathbf{L}}(\mathbf{f})\|_{\mathcal{H}}^{2} \leq 2\|\mathbf{f} - \mathbf{f}^{\parallel}\|_{\mathcal{H}}^{2} + k_{0}\rho_{\infty} \left(\sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} |u_{i}|^{2} - \left| \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} u_{i} \right|^{2} \right) + 6k_{0}c_{\infty} \left(\sum_{i=1}^{N} \frac{c_{\infty,i}}{c_{\infty}} e_{i}^{2} - \left(\sum_{i=1}^{N} \frac{c_{\infty,i}}{c_{\infty}} e_{i} \right)^{2} \right).$$

The next step is to prove that the following estimates hold:

$$I_1 := \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} |u_i|^2 - \left| \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} u_i \right|^2 \le \sum_{i,j=1}^{N} |u_i - u_j|^2, \tag{5.2.26}$$

$$I_2 := \sum_{i=1}^{N} \frac{c_{\infty,i}}{c_{\infty}} e_i^2 - \left(\sum_{i=1}^{N} \frac{c_{\infty,i}}{c_{\infty}} e_i\right)^2 \le \sum_{i,j=1}^{N} (e_i - e_j)^2.$$
 (5.2.27)

Note that we only need to prove the estimate for I_1 , since the arguments for I_2 are exactly the same. In order to handle the expression I_1 , we define for $\mathbf{u} = (u_i)_{1 \leq i \leq N}$ and $\mathbf{v} = (v_i)_{1 \leq i \leq N} \in \mathbb{R}^{3N}$ the following scalar product on \mathbb{R}^{3N} with corresponding norm

$$\langle \mathbf{u}, \mathbf{v} \rangle_{\rho} = \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} u_i \cdot v_i, \quad \|\mathbf{u}\|_{\rho} = \langle \mathbf{u}, \mathbf{u} \rangle_{\rho}^{1/2},$$

where $u_i \cdot v_i$ denotes the standard Euclidean scalar product in \mathbb{R}^3 . Note that the vector $\mathbf{1} = (1, \dots, 1) \in \mathbb{R}^{3N}$ satisfies $\|\mathbf{1}\|_{\rho} = 1$. Now we use the following elementary identity

$$\|\mathbf{u}\|_{\rho}^2 - \langle \mathbf{u}, \mathbf{1} \rangle_{\rho}^2 = \|\mathbf{u} - \langle \mathbf{u}, \mathbf{1} \rangle_{\rho} \mathbf{1}\|_{\rho}^2$$

which can be written as

$$I_{1} = \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} |u_{i}|^{2} - \left| \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} u_{i} \right|^{2} = \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left| u_{i} - \sum_{j=1}^{N} \frac{\rho_{\infty,j}}{\rho_{\infty}} u_{j} \right|^{2}.$$

By using the fact that $\sum_{j=1}^{N} \rho_{\infty,j} = \rho_{\infty}$, we get

$$I_1 = \sum_{i=1}^N \frac{\rho_{\infty,i}}{\rho_{\infty}} \left| \left(1 - \frac{\rho_{\infty,i}}{\rho_{\infty}} \right) u_i - \sum_{j \neq i} \frac{\rho_{\infty,j}}{\rho_{\infty}} u_j \right|^2 = \sum_{i=1}^N \frac{\rho_{\infty,i}}{\rho_{\infty}} \left| \sum_{j \neq i} \frac{\rho_{\infty,j}}{\rho_{\infty}} (u_i - u_j) \right|^2.$$

Inserting the additional factor $(\sum_{j\neq i} \rho_{\infty,k}/\rho_{\infty})^2$ leads to a convex combination of λ_j such that $\sum_{j\neq i} \lambda_j = 1$:

$$I_{1} = \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left(\sum_{k \neq i} \frac{\rho_{\infty,k}}{\rho_{\infty}} \right)^{2} \left| \frac{\sum_{j \neq i} (\rho_{\infty,j}/\rho_{\infty})(u_{i} - u_{j})}{\sum_{k \neq i} \rho_{\infty,k}/\rho_{\infty}} \right|^{2}$$
$$= \sum_{i=1}^{N} \frac{\rho_{\infty,i}}{\rho_{\infty}} \left(\sum_{k \neq i} \frac{\rho_{\infty,k}}{\rho_{\infty}} \right)^{2} \left| \sum_{j \neq i} \lambda_{j} (u_{i} - u_{j}) \right|^{2},$$

where $\lambda_j = (\rho_{\infty,j}/\rho_\infty)(\sum_{k\neq i}(\rho_{\infty,k}/\rho_\infty))^{-1}$. Thus we can apply Jensen's inequality to this convex combination and obtain

$$I_1 = \sum_{i=1}^N \frac{\rho_{\infty,i}}{\rho_\infty} \left(\sum_{k \neq i} \frac{\rho_{\infty,k}}{\rho_\infty} \right)^2 \left| \sum_{j \neq i} \lambda_j (u_i - u_j) \right|^2 \le \sum_{i=1}^N \frac{\rho_{\infty,i}}{\rho_\infty} \left(\sum_{k \neq i} \frac{\rho_{\infty,k}}{\rho_\infty} \right)^2 \sum_{j \neq i} \lambda_j |u_i - u_j|^2.$$

Finally, we can estimate the right-hand side easily by using the definition of the λ_j and that $\rho_{\infty,j} \leq \rho_{\infty}$ to obtain

$$I_1 \le \sum_{i=1}^N \frac{\rho_{\infty,i}}{\rho_\infty} \left(1 - \frac{\rho_{\infty,i}}{\rho_\infty} \right) \sum_{j \ne i} \frac{\rho_{\infty,j}}{\rho_\infty} |u_i - u_j|^2 \le \sum_{i,j=1}^N |u_i - u_j|^2.$$

For I_2 in (5.2.27) exactly the same calculations hold. This implies that

$$-\mathcal{E}(\mathbf{f}^{\parallel}) \le -C_3(\|\mathbf{f} - \pi_{\mathbf{L}}(\mathbf{f})\|_{\mathcal{H}}^2 - 2\|\mathbf{f} - \mathbf{f}^{\parallel}\|_{\mathcal{H}}^2), \tag{5.2.28}$$

where $C_3 = 1/C_k > 0$, with

$$C_k = 10N \max_{1 \le k, \ell \le 5N} \left| \sum_{i=1}^N \int_{\mathbb{R}^3} \psi_{k,i} \psi_{\ell,i} \nu_i \ dv \right| \max \left\{ \rho_{\infty}, 6c_{\infty} \right\},$$

recalling that $(\psi_k)_{1 \le k \le 5N}$ is an arbitrary orthonormal basis of $\operatorname{Ker}(\mathbf{L}^{\mathbf{m}})$ in $L^2_v(\boldsymbol{\mu}^{-1/2})$.

Step 4: End of the proof.

Putting together (5.2.23), Lemma 5.8, and (5.2.28) yields

$$\begin{split} \langle \mathbf{f}, \mathbf{L}(\mathbf{f}) \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} &\leq -(C_1 - 4\eta) \|\mathbf{f} - \mathbf{f}^{\parallel}\|_{\mathcal{H}}^2 - C_2/2 \, \mathcal{E}(\mathbf{f}^{\parallel}) \\ &\leq -(C_1 - 4\eta - C_2 C_3 \eta) \, \|\mathbf{f} - \mathbf{f}^{\parallel}\|_{\mathcal{H}}^2 - (C_2 C_3 \eta)/2 \|\mathbf{f} - \pi_{\mathbf{L}}(\mathbf{f})\|_{\mathcal{H}}^2. \end{split}$$

The first term on the right-hand side is nonnegative if we choose

$$0 < \eta \le \min \left\{ 1, C_1/(4 + C_2 C_3) \right\},\,$$

and the desired spectral-gap estimate follows with $\lambda_L = (C_2 C_3 C_4 \eta)/2$, where the additional constant $C_4 > 0$ takes care of the fact that \mathcal{H} is equivalent to $L^2_v \left(\langle v \rangle^{\gamma/2} \boldsymbol{\mu}^{-1/2} \right)$.

Remark 5.10. (1) We obtain the following relation between the spectral-gap constant λ derived for same masses $m_i = m_j$ for $1 \le i, j \le N$ in Theorem 4.2 and our new constant λ_L for different masses in Theorem 5.6:

$$\lambda_L = \lambda \min_{1 \le i \le N} m_i^2 \frac{6\rho_\infty}{\max\{\rho_\infty, 6c_\infty\}},$$

where $\rho_{\infty} = \sum_{i=1}^{N} m_i c_{\infty,i}$ and $c_{\infty} = \sum_{i=1}^{N} c_{\infty,i}$. Thus, increasing the difference between the masses m_i makes the the spectral-gap constant λ_L smaller, while in the special case of identical masses the two spectral-gap constants λ and λ_L are equal.

(2) Furthermore, the spectral-gap result of Theorem 5.6 only holds for a finite number of species $1 \leq N < \infty$, since for $N \to \infty$ we get that $\lambda_L \to \infty$. It remains an open problem whether or not it is possible to extend the result of Theorem 5.6 to the limit $N \to \infty$.

5.3. L^2 -theory for the linear part with Maxwellian weight

This section is devoted to the study of the linear perturbed operator $\mathbf{G} = \mathbf{L} - v \cdot \nabla_x$ in $L_{x,v}^2(\boldsymbol{\mu}^{-1/2})$, which is the natural space for \mathbf{L} . We shall show that \mathbf{G} generates a strongly continuous semigroup on this space.

Theorem 5.11. We assume that assumptions (H1) - (H4) hold for the collision kernel. Then the linear perturbed operator $\mathbf{G} = \mathbf{L} - v \cdot \nabla_x$ generates a strongly continuous semigroup $S_{\mathbf{G}}(t)$ on $L^2_{x,v}\left(\boldsymbol{\mu}^{-1/2}\right)$ which satisfies

$$\forall t \ge 0, \quad \|S_{\mathbf{G}}(t) (Id - \Pi_{\mathbf{G}})\|_{L^{2}_{x,y}(\boldsymbol{\mu}^{-1/2})} \le C_{G}e^{-\lambda_{G}t},$$

where $\Pi_{\mathbf{G}}$ is the orthogonal projection onto $Ker(\mathbf{G})$ in $L_{x,v}^2(\boldsymbol{\mu}^{-1/2})$.

The constants C_G , $\lambda_G > 0$ are explicit and depend on N, the different masses m_i and the collision kernels.

Let us first make an important remark about $\Pi_{\mathbf{G}}$. Note that $\mathbf{G}(\mathbf{f}) = 0$ means

$$\forall i \in \{1, \dots, N\}, \ \forall (x, v) \in \mathbb{T}^3 \times \mathbb{R}^3, \quad v \cdot \nabla_x f_i(x, v) = L_i(\mathbf{f}(x, \cdot))(v)$$

Multiplying by $\mu_i^{-1}(v)f_i(x,v)$ and integrating over $\mathbb{T}^3 \times \mathbb{R}^3$ implies

$$0 = \int_{\mathbb{T}^3} \langle L_i(\mathbf{f}(x,\cdot)), f_i(x,\cdot) \rangle_{L_v^2(\mu_i^{-1/2})} dx$$

and therefore by summing over i in $\{1, \ldots, N\}$

$$0 = \int_{\mathbb{T}^3} \langle \mathbf{L}(\mathbf{f}(x,\cdot)), \mathbf{f}(x,\cdot) \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} dx.$$

The integrand is nonpositive thanks to the spectral gap of L and hence

$$\forall x \in \mathbb{T}^3, \ \forall v \in \mathbb{R}^3, \quad \mathbf{f}(x, v) = \pi_{\mathbf{L}}(\mathbf{f}(x, \cdot))(v)$$

and therefore $\mathbf{L}(\mathbf{f}(x,\cdot)) = 0$. The latter further implies that $v \cdot \nabla_x \mathbf{f}(x,v) = 0$ which in turn implies that \mathbf{f} does not depend on x [25, Lemma B.2].

We can thus define the projection in $L_{x,v}^2(\mu^{-1/2})$ onto the kernel of **G**

$$\Pi_{\mathbf{G}}(\mathbf{f}) = \sum_{k=1}^{N+4} \left(\int_{\mathbb{T}^3 \times \mathbb{R}^3} \langle \mathbf{f}(x, v), \boldsymbol{\phi}_k(v) \rangle_{\boldsymbol{\mu}^{-1/2}} \, dx dv \right) \boldsymbol{\phi}_k(v), \tag{5.3.1}$$

where the ϕ_k were defined in (5.2.1). Again we define $\Pi_{\mathbf{G}}^{\perp} = \mathrm{Id} - \Pi_{\mathbf{G}}$. Note that $\Pi_{\mathbf{G}}^{\perp}(\mathbf{f}) = 0$ amounts to saying that \mathbf{f} satisfies the multi-species perturbed conservation laws (5.1.8), *i.e.* null individual mass, sum of momentum and sum of energy.

In Subsection 5.3.1, we show the key lemma of the proof that is the *a priori* control of the fluid part of $S_{\mathbf{G}}(t)$ by its orthogonal part, thus recovering some coercivity for \mathbf{G} in the set of solutions to the linear perturbed equation. Subsection 5.3.2 is dedicated to the proof of Theorem 5.11.

5.3.1. A priori control of the fluid part by the microscopic part

As seen in the previous section, the operator \mathbf{L} is only coercive on the orthogonal part. The key argument is to show that we recover some coercivity for solutions to the differential equation. Namely, that for these specific functions, the microscopic part controls the fluid part. This is the purpose of the next lemma

Lemma 5.12. Let $\mathbf{f_0}(x, v)$ and $\mathbf{g}(t, x, v)$ be in $L_{x,v}^2\left(\boldsymbol{\mu}^{-1/2}\right)$ such that $\Pi_{\mathbf{G}}(\mathbf{f_0}) = \Pi_{\mathbf{G}}(\mathbf{g}) = 0$. Suppose that $\mathbf{f}(t, x, v)$ in $L_{x,v}^2\left(\boldsymbol{\mu}^{-1/2}\right)$ is solution to the equation

$$\partial_t \mathbf{f} = \mathbf{L}(\mathbf{f}) - v \cdot \nabla_x \mathbf{f} + \mathbf{g} \tag{5.3.2}$$

with initial value $\mathbf{f_0}$ and satisfying the multi-species conservation laws. Then there exist an explicit $C_{\perp} > 0$ and a function $N_{\mathbf{f}}(t)$ such that for all $t \geq 0$

(i)
$$|N_{\mathbf{f}}(t)| \le C_{\perp} \|\mathbf{f}(t)\|_{L^{2}_{x,v}(\boldsymbol{\mu}^{-1/2})}^{2};$$

(ii)

$$\begin{split} \int_{0}^{t} \|\pi_{\mathbf{L}}(\mathbf{f})\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2} \ ds \leq & N_{\mathbf{f}}(t) - N_{\mathbf{f}}(0) + C_{\perp} \int_{0}^{t} \left\|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\right\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2} \ ds \\ & + C_{\perp} \int_{0}^{t} \|\mathbf{g}\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)} \ ds. \end{split}$$

The constant C_{\perp} is independent of \mathbf{f} and \mathbf{g} .

The methods of the proof are a technical adaptation of the method proposed in [66] in the case of bounded domain with diffusive boundary conditions. The description of $Ker(\mathbf{L})$ associated with the global equilibrium $\boldsymbol{\mu}$ is given by orthogonal functions in L_v^2 but that are not of norm one. Unlike [66] where only mass conservation holds but boundary conditions overcome the lack of conservation laws, we strongly need the conservation of mass, momentum and energy.

Proof of Lemma 5.12. We recall (5.2.1) the definition of $\pi_{\mathbf{L}}(\mathbf{f}) = (\pi_i(\mathbf{f}))_{1 \leq i \leq N}$ and we define $(a_i(t,x))_{1 \leq i \leq N}$, b(t,x) and c(t,x) to be the coordinates of $\pi_{\mathbf{L}}(\mathbf{f})$:

$$\forall 1 \le i \le N, \quad \pi_i(\mathbf{f})(t, x, v) = \left[a_i(t, x) + b(t, x) \cdot v + c(t, x) \frac{|v^2| - 3m_i^{-1}}{2} \right] m_i \mu_i(v). \quad (5.3.3)$$

Note that we are working with an orthogonal but not orthonormal basis of $\operatorname{Ker}(\mathbf{L})$ in $L^2_{x,v}(\boldsymbol{\mu^{-1/2}})$ in order to lighten computations. We will denote by ρ_i the mass of $m_i\mu_i$.

The key idea of the proof is to choose suitable test functions $\psi = (\psi_i)_{1 \leq i \leq N}$ in $H^1_{x,v}$ that will catch the elliptic regularity of a_i , b and c and estimate them.

For a test function $\psi = \psi(t, x, v)$ integrated against the differential equation (5.3.2) we have by Green's formula on each coordinate

$$\int_{0}^{t} \frac{d}{dt} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \langle \boldsymbol{\psi}, \mathbf{f} \rangle_{\mathbf{1}} dx dv ds = \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \langle \boldsymbol{\psi}(t), \mathbf{f}(t) \rangle_{\mathbf{1}} dx dv - \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \langle \boldsymbol{\psi}_{0}, \mathbf{f}_{0} \rangle_{\mathbf{1}} dx dv ds
= \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \langle \mathbf{f}, \partial_{t} \boldsymbol{\psi} \rangle_{\mathbf{1}} dx dv ds + \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \langle \mathbf{L}(\mathbf{f}), \boldsymbol{\psi} \rangle_{\mathbf{1}} dx dv ds
+ \sum_{i=1}^{N} \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} f_{i} v \cdot \nabla_{x} \psi_{i} dx dv ds + \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \langle \boldsymbol{\psi}, \mathbf{g} \rangle_{\mathbf{1}} dx dv ds.$$

We decompose $\mathbf{f} = \pi_{\mathbf{L}}(\mathbf{f}) + \pi_{\mathbf{L}}^{\perp}(\mathbf{f})$ in the term involving $v \cdot \nabla_x$ and use the fact that $\mathbf{L}(\mathbf{f}) = \mathbf{L}[\pi_{\mathbf{L}}^{\perp}(\mathbf{f})]$ to obtain the weak formulation

$$-\sum_{i=1}^{N} \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \pi_{i}(\mathbf{f}) v \cdot \nabla_{x} \psi_{i} \, dx dv ds = \Psi_{1}(t) + \Psi_{2}(t) + \Psi_{3}(t) + \Psi_{4}(t) + \Psi_{5}(t) \quad (5.3.4)$$

with the following definitions

$$\Psi_1(t) = \int_{\mathbb{T}^3 \times \mathbb{R}^3} \langle \boldsymbol{\psi}_0, \mathbf{f_0} \rangle_1 \, dx dv - \int_{\mathbb{T}^3 \times \mathbb{R}^3} \langle \boldsymbol{\psi}(t), \mathbf{f}(t) \rangle_1 \, dx dv, \qquad (5.3.5)$$

$$\Psi_2(t) = \sum_{i=1}^N \int_0^t \int_{\mathbb{T}^3 \times \mathbb{R}^3} \pi_{\mathbf{L}}^{\perp}(\mathbf{f})_i v \cdot \nabla_x \psi_i \, dx dv ds, \qquad (5.3.6)$$

$$\Psi_3(t) = \sum_{i=1}^N \int_0^t \int_{\mathbb{T}^3 \times \mathbb{R}^3} \mathbf{L} \left(\pi_{\mathbf{L}}^{\perp}(\mathbf{f}) \right)_i \psi_i \, dx dv ds, \tag{5.3.7}$$

$$\Psi_4(t) = \sum_{i=1}^N \int_0^t \int_{\mathbb{T}^3 \times \mathbb{R}^3} f_i \partial_s \psi_i \, dx dv ds, \qquad (5.3.8)$$

$$\Psi_5(t) = \int_0^t \int_{\mathbb{T}^3 \times \mathbb{R}^3} \langle \boldsymbol{\psi}, \mathbf{g} \rangle_1 \, dx dv ds.$$
 (5.3.9)

For each of the functions $\mathbf{a} = (a_i)_{1 \leq i \leq N}$, b and c, we construct a ψ such that the left-hand side of (5.3.4) is exactly the L_x^2 -norm of the function and the rest of the proof is estimating the four different terms $\Psi_i(t)$. Note that $\Psi_1(t)$ is already under the desired form

$$\Psi_1(t) = N_{\mathbf{f}}(t) - N_{\mathbf{f}}(0) \tag{5.3.10}$$

with $|N_{\mathbf{f}}(s)| \leq C \|\mathbf{f}\|_{L^2_{x,v}(\boldsymbol{\mu}^{-1/2})}^2$ if $\psi_i(x,v)\mu_i^{1/2}(v)$ is in $L^2_{x,v}$ for all i and their norm is controlled by the one of \mathbf{f} (which will be the case in our next choices).

Remark 5.13. The linear perturbed equation (5.3.2) and the conservation laws are invariant under standard time mollification. We therefore consider for simplicity in the rest of the proof that all functions are smooth in the variable t. Exactly the same estimates can be derived for more general functions and the method would obviously be to study time mollified equation and then take the limit in the smoothing parameter.

For clarity, every positive constant will be denoted by C_k .

Estimate for a = $(a_i)_{1 \leq i \leq N}$. By assumption **f** preserves the mass which is equivalent to

$$0 = \int_{\mathbb{T}^3 \times \mathbb{R}^3} \mathbf{f}(t, x, v) \, dx dv = \int_{\mathbb{T}^3} \left(\int_{\mathbb{R}^3} \langle \mathbf{f}(t, x, v), \boldsymbol{\mu} \rangle_{\boldsymbol{\mu}^{-1/2}} \, dv \right) dx = \int_{\mathbb{T}^3} \mathbf{a}(t, x) \, dx,$$

where we used the fact that $\mu \in \text{Ker}(\mathbf{G})$, $\mathbf{f_0} \in \text{Ker}(\mathbf{G})^{\perp}$ and the orthogonality of the basis defined in (5.3.3). Define a test function $\psi_{\mathbf{a}} = (\psi_i)_{1 \le i \le N}$ by

$$\psi_i(t, x, v) = (|v|^2 - \alpha_i) v \cdot \nabla_x \phi_i(t, x)$$

where

$$-\Delta_x \phi_i(t, x) = a_i(t, x)$$

and $\alpha_i > 0$ is chosen such that for all $1 \le k \le 3$

$$\int_{\mathbb{R}^3} \left(|v|^2 - \alpha_i \right) \frac{|v|^2 - 3m_i^{-1}}{2} v_k^2 \mu_i(v) \, dv = 0.$$

The integral over \mathbb{T}^3 of $a_i(t,\cdot)$ is null and therefore standard elliptic estimate [67] yields:

$$\forall t \ge 0, \quad \|\phi_i(t)\|_{H_x^2} \le C_0 \|a_i(t)\|_{L_x^2}. \tag{5.3.11}$$

The latter estimate provides both the control of $\Psi_1 = N_{\mathbf{f}}^{(a)}(t) - N_{\mathbf{f}}^{(a)}(0)$, as discussed before, and the control of (5.3.9), using Cauchy-Schwarz and Young's inequality,

$$|\Psi_{5}(t)| \leq C \sum_{i=1}^{N} \int_{0}^{t} \|\sqrt{\rho_{i}}\phi_{i}\|_{L_{x}^{2}} \|g_{i}\|_{L_{x,v}^{2}(\mu_{i}^{-1/2})} ds$$

$$\leq \frac{C_{1}}{4} \int_{0}^{t} \|\mathbf{a}\|_{L_{x}^{2}(\rho^{1/2})}^{2} ds + C_{5} \int_{0}^{t} \|\mathbf{g}\|_{L_{x,v}^{2}(\mu^{-1/2})}^{2} ds, \qquad (5.3.12)$$

where $C_1 > 0$ is given in (5.3.13) below and where we defined $\boldsymbol{\rho} = (\rho_i)_{1 \leq i \leq N}$ the vector of the masses associated to $(m_i \mu_i)_{1 \leq i \leq N}$.

Firstly, we compute the term on the left-hand side of (5.3.4).

$$\begin{split} &-\sum_{i=1}^{N}\int_{0}^{t}\int_{\mathbb{T}^{3}\times\mathbb{R}^{3}}\pi_{i}(\mathbf{f})v\cdot\nabla_{x}\psi_{i}\,dxdvds\\ &=-\sum_{i=1}^{N}\sum_{1\leq j,k\leq 3}\int_{0}^{t}\int_{\mathbb{T}^{3}}a_{i}(s,x)\left(\int_{\mathbb{R}^{3}}\left(|v|^{2}-\alpha_{i}\right)v_{j}v_{k}m_{i}\mu_{i}(v)\,dv\right)\partial_{x_{j}}\partial_{x_{k}}\phi_{i}\,dxds\\ &-\sum_{i=1}^{N}\sum_{1\leq j,k\leq 3}\int_{0}^{t}\int_{\mathbb{T}^{3}}b(s,x)\cdot\left(\int_{\mathbb{R}^{3}}v\left(|v|^{2}-\alpha_{i}\right)v_{j}v_{k}m_{i}\mu_{i}(v)\,dv\right)\partial_{x_{j}}\partial_{x_{k}}\phi_{i}\,dxds\\ &-\sum_{i=1}^{N}\sum_{1\leq j,k\leq 3}\int_{0}^{t}\int_{\mathbb{T}^{3}}c(s,x)\left(\int_{\mathbb{R}^{3}}\left(|v|^{2}-\alpha_{i}\right)\frac{|v|^{2}-3m_{i}^{-1}}{2}v_{j}v_{k}m_{i}\mu_{i}dv\right)\partial_{x_{j}}\partial_{x_{k}}\phi_{i}.\end{split}$$

The second term is null as well as the first and last ones when $j \neq k$ thanks to the oddity in v. In the last term when j = k we recover our choice of α_i which makes the last term being null too. It remains the first term when k = j. In this case, the integral in v gives a constant C_1 independent of i times ρ_i . Direct computations give $\alpha_i = 10/m_i$ and $C_1 > 0$. It follows

$$-\sum_{i=1}^{N} \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \pi_{i}(\mathbf{f}) v \cdot \nabla_{x} \psi_{i} \, dx dv ds = -C_{1} \sum_{i=1}^{N} \int_{0}^{t} \int_{\mathbb{T}^{3}} a_{i}(s, x) \rho_{i} \Delta_{x} \phi_{i}(s, x) \, dx ds$$
$$= C_{1} \sum_{i=1}^{N} \int_{0}^{t} \int_{\mathbb{T}^{3}} a_{i}^{2} \rho_{i} \, ds$$

$$= C_1 \int_0^t \|\mathbf{a}(s)\|_{L_x^2(\boldsymbol{\rho}^{1/2})}^2 ds.$$
 (5.3.13)

We recall $\mathbf{L} = -\boldsymbol{\nu}(v) + \mathbf{K}$ where \mathbf{K} is a bounded operator in $L^2_v\left(\boldsymbol{\mu}^{-1/2}\right)$. Moreover, the H^2_x -norm of $\phi_i(t,x)$ is bounded by the L^2_x -norm of $a_i(t,x)$. Multiplying by $\mu_i^{1/2}(v)\mu_i(v)^{-1/2}$ inside the i^{th} integral of Ψ_2 (5.3.6) and of Ψ_3 (5.3.7) a mere Cauchy-Schwarz inequality yields

$$\forall k \in \{2,3\}, \quad |\Psi_{k}(t)| \leq C \sum_{i=1}^{N} \int_{0}^{t} \|\sqrt{\rho_{i}} a_{i}\|_{L_{x}^{2}} \|\pi_{i}^{\perp}(\mathbf{f})\|_{L_{x,v}^{2}(\mu_{i}^{-1/2})} ds \\
\leq \frac{C_{1}}{4} \int_{0}^{t} \|\mathbf{a}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds + C_{2} \int_{0}^{t} \|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} ds. \tag{5.3.14}$$

We used Young's inequality for the last inequality, with C_1 defined in (5.3.13).

It remains to estimate the term with time derivatives (5.3.8). It reads

$$\Psi_{4}(t) = \sum_{i=1}^{N} \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} f_{i} \left(|v|^{2} - \alpha_{i} \right) v \cdot \left[\partial_{t} \nabla_{x} \phi_{i} \right] dx dv ds$$

$$= \sum_{i=1}^{N} \sum_{k=1}^{3} \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \pi_{i}(\mathbf{f}) \left(|v|^{2} - \alpha_{i} \right) v_{k} \partial_{t} \partial_{x_{k}} \phi_{i} dx dv ds$$

$$+ \sum_{i=1}^{N} \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \pi_{i}^{\perp}(\mathbf{f}) \left(|v|^{2} - \alpha_{i} \right) v \cdot \left[\partial_{t} \nabla_{x} \phi_{i} \right] dx dv ds$$

Using oddity properties for the first integral on the right-hand side and then Cauchy-Schwarz with the following bound

$$\int_{\mathbb{D}^3} \left(|v|^2 - \alpha_i \right)^2 |v|^2 \,\mu_i(v) \, dv = C\rho_i < +\infty$$

we get

$$|\Psi_4(t)| \le C \sum_{i=1}^N \int_0^t \left[\sum_{k=1}^3 \|\rho_i b_k\|_{L_x^2} + \left\| \pi_i^{\perp}(\mathbf{f}) \right\|_{L_{x,v}^2 \left(\mu_i^{-1/2}\right)} \right] \|\partial_t \nabla_x \phi_i\|_{L_x^2} ds.$$
 (5.3.15)

Estimating $\|\partial_t \nabla_x \phi_a\|_{L^2_x}$ will come from elliptic estimates in negative Sobolev spaces. We use the decomposition of the weak formulation (5.3.4) between t and $t + \varepsilon$ (instead of between 0 and t) with $\psi(t, x, v) = \phi(x)\mathbf{e}_i \in H^1_x$, where $\mathbf{e}_i = (\delta_{ji})_{1 \leq j \leq N}$. We furthermore require that $\phi(x)$ has a null integral over \mathbb{T}^3 . ψ only depends on x and therefore $\Psi_4(t) = 0$. Moreover, multiplying by $\mu_i(v)\mu_i^{-1}(v)$ in the i^{th} integral of Ψ_3 yields

$$\Psi_3(t) = \int_t^{t+\varepsilon} \int_{\mathbb{T}^3} \langle \mathbf{L}(\mathbf{f}), \mu_i \mathbf{e}_i \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} \phi(x) \, dx dv ds = 0,$$

by definition of $Ker(\mathbf{L})$.

From the weak formulation (5.3.4) it therefore remains

$$\begin{split} &\int_{\mathbb{T}^{3}\times\mathbb{R}^{3}}\phi(x)\langle\mathbf{e}_{i},\mathbf{f}(t+\varepsilon)\rangle_{\mathbf{1}}\,dxdv - \int_{\mathbb{T}^{3}\times\mathbb{R}^{3}}\phi(x)\langle\mathbf{e}_{i},\mathbf{f}(t)\rangle_{\mathbf{1}}\,dxdv \\ &= \int_{t}^{t+\varepsilon}\int_{\mathbb{T}^{3}\times\mathbb{R}^{3}}\pi_{i}(\mathbf{f})v\cdot\nabla_{x}\phi(x)\,dxdvds + \int_{t}^{t+\varepsilon}\int_{\mathbb{T}^{3}\times\mathbb{R}^{3}}\pi_{i}^{\perp}(\mathbf{f})v\cdot\nabla_{x}\phi(x)\,dxdvds \\ &+ \int_{t}^{t+\varepsilon}\int_{\Omega\times\mathbb{R}^{3}}g_{i}(s,x,v)\phi(x)\,dxdvds \end{split}$$

which is equal to

$$\int_{\mathbb{T}^{3}} \rho_{i} \left[a_{i}(t+\varepsilon) - a_{i}(t) \right] \phi(x) dx = C \int_{t}^{t+\varepsilon} \int_{\mathbb{T}^{3}} \rho_{i} b(s,x) \cdot \nabla_{x} \phi(x) dx ds
+ \int_{t}^{t+\varepsilon} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \pi_{i}^{\perp}(\mathbf{f}) \mu_{i}(v)^{-1/2} \mu_{i}(v)^{1/2} v \cdot \nabla_{x} \phi(x)
+ \int_{t}^{t+\varepsilon} \int_{\Omega \times \mathbb{R}^{3}} g_{i}(s,x,v) \phi(x) dx dv ds,$$

where C does not depend on i.

Dividing by $\rho_i \varepsilon$ and taking the limit as ε goes to 0 yields, after a mere Cauchy-Schwarz inequality on the right-hand side

$$\left| \int_{\mathbb{T}^{3}} \partial_{t} a_{i}(s, x) \phi(x) \, dx \right| \leq C \left[\|b(t, x)\|_{L_{x}^{2}} + \left\| \pi_{i}^{\perp}(\mathbf{f}) \right\|_{L_{x, v}^{2} \left(\mu_{i}^{-1/2}\right)} \right] \|\nabla_{x} \phi(x)\|_{L_{x}^{2}}$$

$$+ C \|g_{i}\|_{L_{x, v}^{2} \left(\mu_{i}^{-1/2}\right)} \|\phi\|_{L_{x}^{2}}$$

$$\leq C \left[\|b(t, x)\|_{L_{x}^{2}} + \left\| \pi_{i}^{\perp}(\mathbf{f}) \right\|_{L_{x, v}^{2} \left(\mu_{i}^{-1/2}\right)} + \|g_{i}\|_{L_{x, v}^{2} \left(\mu_{i}^{-1/2}\right)} \right]$$

$$\times \|\nabla_{x} \phi(x)\|_{L_{x}^{2}}.$$

We used Poincaré inequality since $\phi(x)$ has a null integral over \mathbb{T}^d . The latter inequality is true for all ϕ in H_x^1 with a null integral and therefore implies for all $t \geq 0$

$$\|\partial_t a_i(t,x)\|_{(\mathcal{H}_x^1)^*} \le C \left[\|b(t,x)\|_{L_x^2} + \left\| \pi_i^{\perp}(\mathbf{f}) \right\|_{L_{x,v}^2(\mu_i^{-1/2})} + \|g_i\|_{L_{x,v}^2(\mu_i^{-1/2})} \right]$$
 (5.3.16)

where $(\mathcal{H}_x^1)^*$ is the dual of the set of functions in H_x^1 with null integral.

Thanks to the conservation of mass we have that $\partial_t a_i(t,x)$ have a zero integral on the torus and we can construct $\Phi_i(t,x)$ such that

$$-\Delta_x \Phi_i(t, x) = \partial_t a_i(t, x)$$

and by standard elliptic estimate [67]:

$$\|\Phi_i\|_{\mathcal{H}_x^1} \le \|\partial_t a_i\|_{(\mathcal{H}_x^1)^*} \le C \left[\|b(t,x)\|_{L_x^2} + \left\|\pi_i^{\perp}(\mathbf{f})\right\|_{L_{x,v}^2(\mu_i^{-1/2})} + \|g_i\|_{L_{x,v}^2(\mu_i^{-1/2})} \right],$$

where we used (5.3.16). Combining this estimate with

$$\|\partial_t \nabla_x \phi_i\|_{L_x^2} = \|\nabla_x \Delta^{-1} \partial_t a_i\|_{L_x^2} \le \|\Delta^{-1} \partial_t a_i\|_{H_x^1} = \|\Phi_i\|_{H_x^1}$$

we can further control Ψ_4 in (5.3.15) using $\rho_i = \sqrt{\rho_i}\sqrt{\rho_i}$

$$|\Psi_4(t)| \le C_5 \int_0^t \left(\sum_{i=1}^N \|\sqrt{\rho_i} b\|_{L_x^2}^2 + \left\| \pi_i^{\perp}(\mathbf{f}) \right\|_{L_{x,v}^2(\mu_i^{-1/2})}^2 + \|g_i\|_{L_{x,v}^2(\mu_i^{-1/2})}^2 \right) ds. \quad (5.3.17)$$

We now plug (5.3.13), (5.3.10), (5.3.14), (5.3.17) and (5.3.12) into (5.3.4)

$$\int_{0}^{t} \|\mathbf{a}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds \leq N_{\mathbf{f}}^{(a)}(t) - N_{\mathbf{f}}^{(a)}(0) + C_{a,b} \int_{0}^{t} \|b\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds
+ C_{a} \int_{0}^{t} \left[\left\| \pi_{\mathbf{L}}^{\perp}(\mathbf{f}) \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} + \|\mathbf{g}\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} \right] ds.$$
(5.3.18)

Estimate for b. The choice of function to integrate against to deal with the b term is more involved technically. We emphasize that b(t,x) is a vector $(b_1(t,x), b_2(t,x), b_3(t,x))$, thus we used the obvious short-hand notation

$$||b||_{L_x^2(\boldsymbol{\rho}^{1/2})}^2 = \sum_{i=1}^N \sum_{k=1}^3 ||\sqrt{\rho_i}b_k||_{L_x^2}^2.$$

Fix J in $\{1,2,3\}$ and the conservation of momentum implies that for all $t \geq 0$

$$\int_{\mathbb{T}^3} b_J(t, x) \, dx = 0.$$

Define $\psi_{b_J}(t, x, v) = (\psi_{iJ}(t, x, v))_{1 \le i \le N}$ with

$$\psi_{iJ}(t, x, v) = \sum_{j=1}^{3} \varphi_{ij}^{(J)}(t, x, v),$$

with

$$\varphi_{ij}^{(J)}(t,x,v) = \begin{cases} |v|^2 v_j v_J \partial_{x_j} \phi_J(t,x) - \frac{7}{2m_i} \left(v_j^2 - m_i^{-1} \right) \partial_{x_J} \phi_J(t,x), & \text{if } j \neq J \\ \frac{7}{2m_i} \left(v_J^2 - m_i^{-1} \right) \partial_{x_J} \phi_J(t,x), & \text{if } j = J. \end{cases}$$

where

$$-\Delta_x \phi_J(t, x) = b_J(t, x).$$

Since it will be important, we emphasize here that for all $j \neq k$

$$\int_{\mathbb{R}^3} \left(v_j^2 - m_i^{-1} \right) \mu_i(v) \, dv = 0 \quad \text{and} \quad \int_{\mathbb{R}^3} \left(v_j^2 - m_i^{-1} \right) v_k^2 \mu_i(v) \, dv = 0.$$
 (5.3.19)

The null integral of b_J implies by standard elliptic estimate [67]

$$\forall t \ge 0, \quad \|\phi_J(t)\|_{H^2_x} \le C_0 \|b_J(t)\|_{L^2_x}.$$
 (5.3.20)

Again, this estimate provides the control of $\Psi_1(t) = N_{\mathbf{f}}^{(J)}(t) - N_{\mathbf{f}}^{(J)}(0)$ and of $\Psi_5(t)$ as in (5.3.12):

$$|\Psi_5(t)| \le \frac{C_1}{4} \int_0^t \|b_J\|_{L_x^2(\rho^{1/2})}^2 ds + C_5 \int_0^t \|\mathbf{g}\|_{L_{x,v}^2(\boldsymbol{\mu}^{-1/2})}^2 ds, \tag{5.3.21}$$

where $C_1 > 0$ is given in (5.3.22) below.

We start by the left-hand side of (5.3.4). By oddity, there is neither contribution from any of the $a_i(s,x)$ nor from c(s,x). Hence, for all i in $\{1,\ldots,N\}$

$$-\int_{0}^{t} \int_{\Omega \times \mathbb{R}^{3}} \pi_{i}(\mathbf{f}) v \cdot \nabla_{x} \psi_{iJ} \, dx dv ds$$

$$= -\sum_{1 \leq k, l \leq 3} \sum_{\substack{j=1 \ j \neq J}}^{3} \int_{0}^{t} \int_{\Omega} b_{l}(s, x) \left(\int_{\mathbb{R}^{3}} \left| v^{2} \right| v_{l} v_{k} v_{j} v_{J} m_{i} \mu_{i}(v) \, dv \right) \partial_{x_{k}} \partial_{x_{j}} \phi_{J}(s, x) \, dx ds$$

$$+ \frac{7}{2m_{i}} \sum_{1 \leq k, l \leq 3} \sum_{\substack{j=1 \ j \neq J}}^{3} \int_{0}^{t} \int_{\Omega} b_{l}(s, x) \left(\int_{\mathbb{R}^{3}} \left(v_{j}^{2} - m_{i}^{-1} \right) v_{l} v_{k} m_{i} \mu_{i} dv \right) \partial_{x_{k}} \partial_{x_{J}} \phi_{J} \, dx ds$$

$$- \frac{7}{2m_{i}} \sum_{1 \leq k, l \leq 3} \int_{0}^{t} \int_{\Omega} b_{l}(s, x) \left(\int_{\mathbb{R}^{3}} \left(v_{J}^{2} - m_{i}^{-1} \right) v_{l} v_{k} m_{i} \mu_{i}(v) \, dv \right) \partial_{x_{k}} \partial_{x_{J}} \phi_{J} \, dx ds.$$

The last two integrals on \mathbb{R}^3 are zero if $l \neq k$. Moreover, when l = k and $l \neq J$ it is also zero by (5.3.19). We compute directly for l = J

$$\int_{\mathbb{R}^3} \left(v_J^2 - m_i^{-1} \right) v_J^2 m_i \mu_i(v) \, dv = \frac{2}{m_i^2} \rho_i.$$

The first term is composed by integrals in v of the form

$$\int_{\mathbb{R}^3} |v|^2 v_k v_j v_l v_J \mu_i(v) \, dv$$

which is always null unless two indices are equals to the other two. Therefore if j=l then k=J and if $j\neq l$ we only have two options: k=j and l=J or k=l and j=J. Hence, for all i in $\{1,\ldots,N\}$

$$-\int_{0}^{t} \int_{\Omega \times \mathbb{R}^{3}} \pi_{i}(\mathbf{f}) v \cdot \nabla_{x} \psi_{J} \, dx dv ds$$

$$= -\sum_{\substack{j=1\\j \neq J}}^{3} \int_{0}^{t} \int_{\Omega} b_{J}(s, x) \partial_{x_{j}x_{j}} \phi_{J} \left(\int_{\mathbb{R}^{3}} |v|^{2} v_{j}^{2} v_{J}^{2} m_{i} \mu_{i}(v) \, dv \right) dx ds$$

$$-\sum_{\substack{j=1\\j\neq J}}^{3} \int_{0}^{t} \int_{\Omega} b_{j}(s,x) \partial_{x_{j}x_{J}} \phi_{J} \left(\int_{\mathbb{R}^{3}} |v|^{2} v_{j}^{2} v_{J}^{2} m_{i} \mu_{i}(v) dv \right) dx ds$$

$$+ \frac{7}{m_{i}^{3}} \sum_{\substack{j=1\\j\neq J}}^{3} \int_{0}^{t} \int_{\Omega} \rho_{i} b_{j}(s,x) \partial_{x_{j}x_{J}} \phi_{J} dx ds - \frac{7}{m_{i}^{3}} \int_{0}^{t} \int_{\Omega} \rho_{i} b_{J}(s,x) \partial_{x_{J}} \phi_{J}(s,x) dx ds.$$

To conclude we compute for $j \neq J$

$$\int_{\mathbb{R}^3} |v^2| \, v_j^2 v_J^2 m_i \mu_i(v) \, dv = \frac{7}{m_i^3} \rho_i$$

and it thus only remains the following equality for all i in $\{1, \ldots, N\}$.

$$-\int_0^t \int_{\Omega \times \mathbb{R}^3} \pi_i(\mathbf{f}) v \cdot \nabla_x \psi_J \, dx dv ds = -\frac{7}{m_i^3} \int_0^t \int_{\Omega} \rho_i b_J(s, x) \Delta_x \phi_J(s, x) \, dx ds$$
$$= \frac{7}{m_i^3} \int_0^t \|\sqrt{\rho_i} b_J\|_{L_x^2}^2 \, ds.$$

Summing over i yields

$$-\sum_{i=1}^{N} \int_{0}^{t} \int_{\Omega \times \mathbb{R}^{3}} \pi_{j}(\mathbf{f}) v \cdot \nabla_{x} \psi_{J} = \frac{7}{m_{i}^{3}} \int_{0}^{t} \|b_{J}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})} dx dv ds.$$
 (5.3.22)

We recall $\boldsymbol{\rho} = (\rho_i)_{1 \leq i \leq N}$.

Then the terms Ψ_2 and Ψ_3 are dealt with as in (5.3.14)

$$\forall k \in \{2, 3\}, \quad |\Psi_k(t)| \le \frac{7}{4} \int_0^t \|b_J\|_{L_x^2(\boldsymbol{\rho}^{1/2})}^2 \ ds + C_2 \int_0^t \left\| \pi_{\mathbf{L}}^{\perp}(\mathbf{f}) \right\|_{L_{x,y}^2(\boldsymbol{\mu}^{-1/2})}^2 \ ds. \quad (5.3.23)$$

It remains to estimate Ψ_4 which involves time derivative (5.3.8):

$$\begin{split} \Psi_4(t) &= \sum_{i=1}^N \sum_{j=1}^3 \int_0^t \int_{\Omega \times \mathbb{R}^3} f_i \partial_t \varphi_{ij}^{(J)}(s,x,v) \, dx dv ds \\ &= \sum_{i=1}^N \sum_{j=1}^3 \int_0^t \int_{\Omega \times \mathbb{R}^3} \pi_i^{\perp}(\mathbf{f}) \partial_t \varphi_{ij}^{(J)}(s,x,v) \, dx dv ds \\ &+ \sum_{i=1}^N \sum_{j=1}^3 \int_0^t \int_{\Omega \times \mathbb{R}^3} \pi_i(\mathbf{f}) \, |v|^2 \, v_j v_J \partial_{x_j} \phi_J \, dx dv ds \\ &+ \sum_{i=1}^N \sum_{j=1}^3 \pm \frac{7}{2m_i} \int_0^t \int_{\Omega \times \mathbb{R}^3} \pi_i(\mathbf{f}) \left(v_j^2 - m_j^{-1} \right) \partial_{x_J} \phi_J \, dx dv ds. \end{split}$$

By oddity arguments, only terms in $a_i(s, x)$ and c(s, x) can contribute to the last two terms on the right-hand side. However, $j \neq J$ implies that the second term is zero as well as the contribution of $a_i(s, x)$ in the third term thanks to (5.3.19). Finally, a Cauchy-Schwarz inequality on both integrals yields as in (5.3.15)

$$|\Psi_4(t)| \le C \sum_{i=1}^N \int_0^t \left[\|\rho_i c\|_{L_x^2} + \left\| \pi_i^{\perp}(\mathbf{f}) \right\|_{L_{x,v}^2(\mu_i^{-1/2})} \right] \|\partial_t \nabla_x \phi_J\|_{L_x^2} ds.$$
 (5.3.24)

To estimate $\|\partial_t \nabla_x \phi_J\|_{L^2_x}$ we follow the idea developed for $\mathbf{a}(s,x)$ about negative Sobolev regularity. We apply the weak formulation (5.3.4) to a specific function between t and $t+\varepsilon$. The test function is $\psi(x,v) = \phi(x)v_J\mathbf{m}$ with ϕ in H^1_x with a zero integral over \mathbb{T}^3 . Note that ψ does not depend on t so $\Psi_4 = 0$ and multiplying by $\mu_i(v)\mu_i^{-1}(v)$ in the i^{th} integral of Ψ_3 yields

$$\Psi_3(t) = \int_0^t \int_{\mathbb{T}^3} \langle \mathbf{L}(\mathbf{f}), v_J(m_i \mu_i)_{1 \le i \le N} \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} \partial_{x_k} \phi(x) \, dx dv ds = 0,$$

by definition of $Ker(\mathbf{L})$.

It remains

$$C\sum_{i=1}^{N} \int_{\Omega} \rho_{i} \left[b_{J}(t+\varepsilon) - b_{J}(t) \right] \phi(x) dx$$

$$= \sum_{i=1}^{N} \int_{t}^{t+\varepsilon} \int_{\Omega \times \mathbb{R}^{3}} \pi_{i}(\mathbf{f}) v_{J} v \cdot \nabla_{x} \phi(x) dx dv ds$$

$$+ \sum_{i=1}^{N} \int_{t}^{t+\varepsilon} \int_{\Omega \times \mathbb{R}^{3}} \pi_{i}^{\perp}(\mathbf{f}) v_{J} v \cdot \nabla_{x} \phi(x) dx dv ds$$

$$+ \sum_{i=1}^{N} \int_{t}^{t+\varepsilon} \int_{\Omega \times \mathbb{R}^{3}} g_{i} v_{J} \phi(x) dx dv ds.$$

As for $a_i(t, x)$ we divide by ε and take the limit as ε goes to 0. By oddity, the first integral on the right-hand side only gives terms with $a_i(s, x)$ and c(s, x). The other two integrals are dealt with by a Cauchy-Schwarz inequality and Poincaré. This yields

$$\left| \int_{\Omega} \partial_{t} b_{J}(t, x) \phi(x) \, dx \right| \\
\leq C \left[\|\mathbf{a}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})} + \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})} + \|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})} + \|\mathbf{g}\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})} \right] \|\nabla_{x} \phi\|_{L_{x}^{2}}. \tag{5.3.25}$$

The latter is true for all $\phi(x)$ in H_x^1 with a null integral over \mathbb{T}^3 . We thus fix t and apply the inequality above to

$$-\Delta_x \phi(t,x) = \partial_t b_J(t,x)$$

which has a zero integral thanks to the conservation of momentum and obtain

$$\left\|\partial_t \nabla_x \phi_J\right\|_{L_x^2}^2 = \left\|\nabla_x \Delta^{-1} \partial_t b_J\right\|_{L_x^2}^2 = \int_{\Omega} \left(\nabla_x \Delta^{-1} \partial_t b_J\right) \nabla_x \phi(x) \, dx.$$

We integrate by parts

$$\|\partial_t \nabla_x \phi_J\|_{L_x^2}^2 = \int_{\Omega} \partial_t b_J(t, x) \phi(x) dx.$$

At last, we use (5.3.25)

$$\begin{split} &\|\partial_{t}\nabla_{x}\phi_{J}\|_{L_{x}^{2}}^{2} \\ &\leq C\left[\|\mathbf{a}\|_{L_{x}^{2}\left(\boldsymbol{\rho}^{1/2}\right)} + \|c\|_{L_{x}^{2}\left(\boldsymbol{\rho}^{1/2}\right)} + \left\|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\right\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)} + \|\mathbf{g}\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)}\right] \|\nabla_{x}\phi\|_{L_{x}^{2}} \\ &= C\left[\|\mathbf{a}\|_{L_{x}^{2}\left(\boldsymbol{\rho}^{1/2}\right)} + \|c\|_{L_{x}^{2}\left(\boldsymbol{\rho}^{1/2}\right)} + \left\|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\right\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)} + \|\mathbf{g}\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)}\right] \|\nabla_{x}\Delta_{x}^{-1}\partial_{t}b_{J}\|_{L_{x}^{2}} \\ &= C\left[\|\mathbf{a}\|_{L_{x}^{2}\left(\boldsymbol{\rho}^{1/2}\right)} + \|c\|_{L_{x}^{2}\left(\boldsymbol{\rho}^{1/2}\right)} + \left\|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\right\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)} + \|\mathbf{g}\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)}\right] \|\partial_{t}\nabla_{x}\phi_{J}\|_{L_{x}^{2}}. \end{split}$$

Combining this estimate with (5.3.24) and using Young's inequality with any $\varepsilon_b > 0$

$$|\Psi_{4}(t)| \leq \varepsilon_{b} \int_{0}^{t} \|\mathbf{a}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds + C_{5}(\varepsilon_{b}) \int_{0}^{t} \left[\|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + \left\|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} + \|\mathbf{g}\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} \right] ds.$$

$$(5.3.26)$$

We now gather (5.3.22), (5.3.10), (5.3.23), (5.3.26) and (5.3.21)

$$\int_{0}^{t} \|b_{J}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds \leq N_{\mathbf{f}}^{(J)}(t) - N_{\mathbf{f}}^{(J)}(0) + \varepsilon_{b} \int_{0}^{t} \|a\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds + C_{J,c}(\varepsilon_{b}) \int_{0}^{t} \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + C_{J}(\varepsilon_{b}) \int_{0}^{t} \left[\|\mathbf{g}\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} + \left\|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} \right] ds.$$

Finally, summing over all J in $\{1, 2, 3\}$

$$\int_{0}^{t} \|b\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds \leq N_{\mathbf{f}}^{(b)}(t) - N_{\mathbf{f}}^{(b)}(0) + \varepsilon_{b} \int_{0}^{t} \|\mathbf{a}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + C_{b,c} \int_{0}^{t} \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + C_{b,c} \int_{0}^{t} \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + C_{b,c} \int_{0}^{t} \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + C_{b,c} \int_{0}^{t} \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds, \qquad (5.3.27)$$

with $C_{b,c}$ and C_b depending on ε_b .

Estimate for c. The contribution of c(t, x) is really similar to the one of $\mathbf{a}(t, x)$. Since \mathbf{f} preserves mass and energy the following holds

$$\int_{\mathbb{T}^3} c(t, x) \ dx = 0.$$

Define the test function $\psi = (\psi_{ic}(t, x, v))_{1 \le i \le N}$ with

$$\psi_{ic}(t, x, v) = \left(|v|^2 - \alpha_{ic}\right) v \cdot \nabla_x \phi_c(t, x)$$

where

$$-\Delta_x \phi_c(t, x) = c(t, x)$$

and $\alpha_{ic} > 0$ is chosen such that for all $1 \le k \le 3$

$$\int_{\mathbb{R}^3} \left(|v|^2 - \alpha_{ic} \right) v_k^2 \,\mu_i(v) \, dv = 0.$$

Again, the null integral of c and standard elliptic estimate [67] show

$$\forall t \ge 0, \quad \|\phi_c(t)\|_{H_x^2} \le C_0 \|c(t)\|_{L_x^2}. \tag{5.3.28}$$

Again, this estimate provides the control of $\Psi_1 = N_{\bf f}^{(c)}(t) - N_{\bf f}^{(c)}(0)$ and of $\Psi_5(t)$ as in (5.3.12):

$$|\Psi_5(t)| \le \frac{C_1}{4} \int_0^t ||c||_{L_x^2(\boldsymbol{\rho}^{1/2})}^2 ds + C_5 \int_0^t ||\mathbf{g}||_{L_{x,v}^2(\boldsymbol{\mu}^{-1/2})}^2 ds, \tag{5.3.29}$$

where $C_1 > 0$ is given in (5.3.30) below.

We start by the left-hand side of (5.3.4).

$$\begin{split} &-\sum_{i=1}^{N}\int_{0}^{t}\int_{\mathbb{T}^{3}\times\mathbb{R}^{3}}\pi_{i}(\mathbf{f})v\cdot\nabla_{x}\psi_{c}\,dxdvds\\ &=-\sum_{i=1}^{N}\sum_{1\leq j,k\leq 3}\int_{0}^{t}\int_{\mathbb{T}^{3}}a_{i}(s,x)\left(\int_{\mathbb{R}^{3}}\left(|v|^{2}-\alpha_{ic}\right)v_{j}v_{k}m_{i}\mu_{i}(v)\,dv\right)\partial_{x_{j}}\partial_{x_{k}}\phi_{c}\,dxds\\ &-\sum_{i=1}^{N}\sum_{1\leq j,k\leq 3}\int_{0}^{t}\int_{\mathbb{T}^{3}}b(s,x)\cdot\left(\int_{\mathbb{R}^{3}}v\left(|v|^{2}-\alpha_{ic}\right)v_{j}v_{k}m_{i}\mu_{i}(v)\,dv\right)\partial_{x_{j}}\partial_{x_{k}}\phi_{c}\,dxds\\ &-\sum_{i=1}^{N}\sum_{1\leq j,k\leq 3}\int_{0}^{t}\int_{\mathbb{T}^{3}}c(s,x)\left(\int_{\mathbb{R}^{3}}\left(|v|^{2}-\alpha_{ic}\right)\frac{|v|^{2}-3m_{i}^{-1}}{2}v_{j}v_{k}m_{i}\mu_{i}dv\right)\partial_{x_{j}}\partial_{x_{k}}\phi_{c}.\end{split}$$

By oddity, the second integral vanishes, as well as all the others if $j \neq k$. Our choice of α_{ic} makes the first integral vanish even for j = k. It only remains the last integral with terms j = k and therefore the definition of $\Delta_x \phi_c(t, x)$ gives

$$-\sum_{i=1}^{N} \int_{0}^{t} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \pi_{i}(\mathbf{f}) v \cdot \nabla_{x} \psi_{c} \, dx dv ds = C_{1} \int_{0}^{t} \sum_{i=1}^{N} \int_{\mathbb{T}^{3}} \rho_{i} c(s, x)^{2} \, dx ds$$
$$= C_{1} \int_{0}^{t} \|c(s)\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})} \, ds. \qquad (5.3.30)$$

Again, direct computations give $\alpha_{ic} = 5/m_i$ and $C_1 > 0$.

Then the terms Ψ_2 and Ψ_3 are dealt with as in (5.3.14)

$$\forall k \in \{2,3\}, \quad |\Psi_k(t)| \le \frac{C_1}{4} \int_0^t \|c\|_{L_x^2(\boldsymbol{\rho}^{1/2})}^2 ds + C_2 \int_0^t \left\| \pi_{\mathbf{L}}^{\perp}(\mathbf{f}) \right\|_{L_{x,y}^2(\boldsymbol{\mu}^{-1/2})}^2 ds. \quad (5.3.31)$$

As for $\mathbf{a}(t,x)$ the estimate on Ψ_4 (5.3.8) will follow from elliptic regularity in negative Sobolev spaces. With exactly the same computations as for (5.3.15) we have

$$|\Psi_4(t)| \le C \int_0^t \|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\|_{L^2_{x,v}(\boldsymbol{\mu}^{-1/2})} \|\partial_t \nabla_x \phi_c\|_{L^2_x} ds.$$
 (5.3.32)

Note that the contribution of $\pi_{\mathbf{L}}$ was null by oddity for the $\mathbf{a}(t,x)$ and c(t,x) terms and also for the b(t,x) terms thanks to our choice of α_{ic} .

To estimate $\|\partial_t \nabla_x \phi_c\|_{L^2_x}$ we use the decomposition of the weak formulation (5.3.4) between t and $t + \varepsilon$ (instead of between 0 and t) with

$$\psi(t, x, v) = \left(m_i(|v|^2 - 3m_i^{-1})\phi(x)\right)_{1 \le i \le N}$$

where ϕ belongs to H_x^1 and has a zero integral on the torus. ψ does not depend on t and therefore $\Psi_4(t) = 0$. Moreover, multiplying by $\mu_i(v)\mu_i^{-1}(v)$ in the i^{th} integral of Ψ_3 yields

$$\Psi_3(t) = \int_0^t \int_{\mathbb{T}^3} \langle \mathbf{L}(\mathbf{f}), \left(\frac{|v|^2 - 3m_i^{-1}}{2} m_i \mu_i \right)_{1 \le i \le N} \rangle_{L_v^2(\boldsymbol{\mu}^{-1/2})} \partial_{x_k} \phi(x) \, dx dv ds = 0,$$

by definition of $Ker(\mathbf{L})$.

From the weak formulation (5.3.4) it therefore remains

$$C \int_{\mathbb{T}^{3}} \left[c(t+\varepsilon) - c(t) \right] \phi(x) dx = \sum_{i=1}^{N} \int_{t}^{t+\varepsilon} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \pi_{i}(\mathbf{f}) \frac{m_{i} |v|^{2} - 3}{2} v \cdot \nabla_{x} \phi(x)$$

$$+ \sum_{i=1}^{N} \int_{t}^{t+\varepsilon} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \pi_{i}^{\perp}(\mathbf{f}) \frac{m_{i} |v|^{2} - 3}{2} v \cdot \nabla_{x} \phi(x)$$

$$+ \sum_{i=1}^{N} \int_{t}^{t+\varepsilon} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} g_{i}(s, x, v) \frac{m_{i} |v|^{2} - 3}{2} \phi(x).$$

As for $\mathbf{a}(t,x)$ we divide by ε and take the limit as ε goes to 0. By oddity, the first integral on the right-hand side only gives terms with $\rho_i b(s,x)$. The last two terms are dealt with by multiplying by $\mu_i(v)^{-1/2}\mu_i(v)^{1/2}$ inside each integral and applying a Cauchy-Schwarz inequality. Note that again we also apply Poincaré inequality. This yields

$$\left| \int_{\mathbb{T}^3} \partial_t c(t, x) \phi(x) \, dx \right|$$

$$\leq C \left[\|b\|_{L_x^2(\boldsymbol{\rho}^{1/2})} + \left\| \pi_{\mathbf{L}}^{\perp}(\mathbf{f}) \right\|_{L_{x,v}^2(\boldsymbol{\mu}^{-1/2})} + \|\mathbf{g}\|_{L_{x,v}^2(\boldsymbol{\mu}^{-1/2})} \right] \|\nabla_x \phi\|_{L_x^2} \, .$$

That estimate holds for all $\phi(x)$ in H_x^1 with null integral over \mathbb{T}^3 . We copy the arguments made for $\mathbf{a}(t,x)$ or $b_J(t,x)$ and construct

$$-\Delta_x \Phi_c(t, x) = \partial_t c(t, x)$$

and obtain by elliptic estimates

$$\begin{split} \|\partial_t \nabla_x \phi_c\|_{L^2_x} &= \|\nabla_x \Delta^{-1} \partial_t c\|_{L^2_x} \le \|\Delta^{-1} \partial_t c\|_{H^1_x} = \|\Phi_c\|_{H^1_x} \\ &\le C \|\partial_t c(t, x)\|_{(H^1_x)^*} \\ &\le C \left[\|b\|_{L^2_x(\boldsymbol{\rho}^{1/2})} + \left\|\pi^{\perp}_{\mathbf{L}}(\mathbf{f})\right\|_{L^2_{x,v}(\boldsymbol{\mu}^{-1/2})} + \|\mathbf{g}\|_{L^2_{x,v}(\boldsymbol{\mu}^{-1/2})} \right]. \end{split}$$

Combining this estimate with (5.3.32) and using Young's inequality with any $\varepsilon_c > 0$

$$|\Psi_{4}(t)| \leq \varepsilon_{c} \int_{0}^{t} \|b\|_{L_{x}^{2}(\rho^{1/2})}^{2} ds + C_{5}(\varepsilon_{c}) \int_{0}^{t} \left[\left\| \pi_{\mathbf{L}}^{\perp}(\mathbf{f}) \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} + \left\| \mathbf{g} \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} \right] ds.$$

$$(5.3.33)$$

We now gather (5.3.30), (5.3.10), (5.3.31), (5.3.33) and (5.3.29) into (5.3.4):

$$\int_{0}^{t} \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds \leq N_{\mathbf{f}}^{(c)}(t) - N_{\mathbf{f}}^{(c)}(0) + \varepsilon_{c} \int_{0}^{t} \|b\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} ds
+ C_{c}(\varepsilon_{c}) \int_{0}^{t} \left[\left\| \pi_{\mathbf{L}}^{\perp}(\mathbf{f}) \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} + \|\mathbf{g}\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} \right] ds.$$
(5.3.34)

Conclusion of the proof. We gather together the estimates we derived for \mathbf{a} , b and c. We compute the linear combination $(5.3.18) + \alpha \times (5.3.27) + \beta \times (5.3.34)$. For all $\varepsilon_b > 0$ and $\varepsilon_c > 0$ this implies

$$\int_{0}^{t} \left[\|\mathbf{a}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + \alpha \|b\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + \beta \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} \right] ds$$

$$\leq N_{\mathbf{f}}(t) - N_{\mathbf{f}}(0) + C_{\perp} \int_{0}^{t} \left[\|\pi_{\mathbf{L}}^{\perp}(\mathbf{f})\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} + \|\mathbf{g}\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} \right] ds$$

$$+ \int_{0}^{t} \left[\alpha \varepsilon_{b} \|\mathbf{a}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + (C_{a,b} + \beta \varepsilon_{c}) \|b\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + \alpha C_{b,c}(\varepsilon_{b}) \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} \right] ds.$$

We first choose $\alpha > C_{a,b}$, then ε_b such that $\alpha \varepsilon_b < 1$ and then $\beta > \alpha C_{b,c}(\varepsilon_b)$. Finally, we fix ε_c small enough such that $C_{a,b} + \beta \varepsilon_c < \alpha$. With such choices we can absorb the last term on the right-hand side by the left-hand side. This concludes the proof of Lemma 5.12 since

$$\|\pi_{\mathbf{L}}(\mathbf{f})\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} = \|\mathbf{a}\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + \|b\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2} + \|c\|_{L_{x}^{2}(\boldsymbol{\rho}^{1/2})}^{2}.$$

5.3.2. Generation of a semigroup

We now have the tools to develop the hypocoercivity of G into a semigroup property.

Proof of Theorem 5.11. Let $\mathbf{f_0}$ be in $L_{x,v}^2(\boldsymbol{\mu}^{-1/2})$ and consider the following equation

$$\partial_t \mathbf{f} = \mathbf{L}(\mathbf{f}) - v \cdot \nabla_x \mathbf{f} \tag{5.3.35}$$

with initial data f_0 .

Since the transport part $-v \cdot \nabla_x$ is skew-symmetric in $L^2_{x,v}\left(\mu_i^{-1/2}\right)$ (mere integration by part) and **L** is self-adjoint, $\operatorname{Ker}(\mathbf{G})$ and $(\operatorname{Ker}(\mathbf{G}))^{\perp}$ are stable under (5.3.35). We therefore consider only the case $\mathbf{f_0}$ in $(\operatorname{Ker}(\mathbf{G}))^{\perp}$ and the associated solution stays in $(\operatorname{Ker}(\mathbf{G}))^{\perp}$ for all t.

Moreover, **L** has a spectral gap λ_L and so by Theorem 5.6, if $\mathbf{f} = (f_i)_{1 \leq i \leq N}$ is a solution to (5.3.35) we have the following

$$\frac{1}{2} \frac{d}{dt} \|\mathbf{f}\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} = \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \langle \partial_{t} \mathbf{f}, \mathbf{f} \rangle_{\boldsymbol{\mu}^{-1/2}} dx dv$$

$$= -\sum_{i=1}^{N} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} v \cdot \nabla_{x} \left(f_{i}(t, x, v)^{2} \right) \mu_{i}^{-1}(v) dx dv$$

$$+ \int_{\mathbb{T}^{3}} \langle \mathbf{L}(\mathbf{f})(t, x, \cdot), \mathbf{f}(t, x, \cdot) \rangle_{L_{v}^{2}(\boldsymbol{\mu}^{-1/2})} dx$$

$$\leq -\lambda_{L} \left\| \pi_{\mathbf{L}}^{\perp}(\mathbf{f}) \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2}. \tag{5.3.36}$$

We remind that $\pi_{\mathbf{L}}^{\perp} = \operatorname{Id} - \pi_{\mathbf{L}}$ where $\pi_{\mathbf{L}}$ is the orthogonal projection (5.2.1) onto $\operatorname{Ker}(L)$ in $L_v^2(\boldsymbol{\mu}^{-1/2})$. The norm is thus decreasing under the flow and it therefore follows that \mathbf{G} generates a strongly continuous semigroup on $L_v^2(\boldsymbol{\mu}^{-1/2})$, we refer the reader to [98] (general theory) or [134][135] (for the special case of single species Boltzmann equation).

Let $\mathbf{f} = S_{\mathbf{G}}(t)\mathbf{f_0}$ and define $\widetilde{\mathbf{f}}(t,x,v) = e^{\lambda t}\mathbf{f}(t,x,v)$ for $\lambda > 0$ to be defined later. $\widetilde{\mathbf{f}}$ satisfies the conservation laws and is solution in $L_{x,v}^2\left(\boldsymbol{\mu}^{-1/2}\right)$ to the following equation

$$\partial_t \widetilde{\mathbf{f}} = \mathbf{G}(\widetilde{\mathbf{f}}) + \lambda \widetilde{\mathbf{f}}.$$

As for (5.3.36) we obtain the following estimate

$$\left\|\widetilde{\mathbf{f}}\right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} \leq \left\|\mathbf{f_{0}}\right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} - 2\lambda_{L} \int_{0}^{t} \left\|\pi_{\mathbf{L}}^{\perp}(\widetilde{\mathbf{f}})\right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} + 2\lambda \int_{0}^{t} \left\|\widetilde{\mathbf{f}}\right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2}.$$
(5.3.37)

Along with the latter estimate, we have the following control given by Lemma 5.12 with $\mathbf{g}=\lambda\widetilde{\mathbf{f}}$

$$\int_{0}^{t} \left\| \pi_{\mathbf{L}}(\widetilde{\mathbf{f}}) \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} ds \leq N_{\widetilde{\mathbf{f}}}(t) - N_{\widetilde{\mathbf{f}}}(0) + C_{\perp} \int_{0}^{t} \left\| \pi_{\mathbf{L}}^{\perp}(\widetilde{\mathbf{f}}) \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} ds + C_{\perp} \lambda^{2} \int_{0}^{t} \left\| \widetilde{\mathbf{f}} \right\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})}^{2} ds$$
(5.3.38)

where $C_{\perp} > 0$ is independent of \mathbf{f} and $\left| N_{\widetilde{\mathbf{f}}}(s) \right| \leq C \left\| \widetilde{\mathbf{f}}(s) \right\|_{L_{x,v}^2(\boldsymbol{\mu}^{-1/2})}^2$, then $\varepsilon \times (5.3.38) + (5.3.37)$ yields

$$\begin{split} \left[\left\| \widetilde{\mathbf{f}} \right\|_{L^{2}_{x,v}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2} - \varepsilon N_{\widetilde{\mathbf{f}}}(t) \right] + C_{\varepsilon} \int_{0}^{t} \left(\left\| \pi_{\mathbf{L}}(\widetilde{\mathbf{f}}) \right\|_{L^{2}_{x,v}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2} + \left\| \pi_{\mathbf{L}}^{\perp}(\widetilde{\mathbf{f}}) \right\|_{L^{2}_{x,v}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2} \right) \\ & \leq \| \mathbf{f_{0}} \|_{L^{2}_{x,v}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2} - \varepsilon N_{\widetilde{\mathbf{f}}}(0) + \left(2\lambda + \varepsilon C_{\perp}\lambda^{2} \right) \int_{0}^{t} \left\| \widetilde{\mathbf{f}} \right\|_{L^{2}_{x,v}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2} ds. \end{split}$$

where $C_{\varepsilon} = \min \{2\lambda_L - \varepsilon C_{\perp}, \varepsilon\}$. By the control on $|N_{\tilde{\mathbf{f}}}(s)|$ and the fact that

$$\left\|\pi_{\mathbf{L}}(\widetilde{\mathbf{f}})\right\|_{L^{2}_{x,v}(\boldsymbol{\mu}^{-1/2})}^{2}+\left\|\pi_{\mathbf{L}}^{\perp}(\widetilde{\mathbf{f}})\right\|_{L^{2}_{x,v}(\boldsymbol{\mu}^{-1/2})}^{2}=\left\|\widetilde{\mathbf{f}}\right\|_{L^{2}_{x,v}(\boldsymbol{\mu}^{-1/2})}^{2}$$

we can choose ε small enough such that $C_{\varepsilon} > 0$ and then λ small enough such that $\left(2\lambda + \varepsilon C_{\perp}\lambda^{2}\right) < C_{\varepsilon}$. Such choices imply that $\left\|\tilde{\mathbf{f}}\right\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2}$ is uniformly bounded in time by $C\left\|\mathbf{f}_{\mathbf{0}}\right\|_{L_{x,v}^{2}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2}$.

By definition of $\tilde{\mathbf{f}}$, this shows an exponential decay for \mathbf{f} and concludes the proof of Theorem 5.11.

5.4. L^{∞} -theory for the linear part with Maxwellian weight

As explained in the introduction, the L^2 setting is not algebraic for the nonlinear operator \mathbf{Q} . We therefore need to work in an L^{∞} framework. We first give a pointwise control on the linear operator \mathbf{K} in Subsection 5.4.1 and then we prove that the linear part of the perturbed equation (5.1.7) generates a strongly continuous semigroup in $L_{x,v}^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)$ in Subsection 5.4.2.

5.4.1. Pointwise estimate on the compact part

We recall that L can be written under the following form

$$\mathbf{L} = -\boldsymbol{\nu}(v) + \mathbf{K},$$

where $\boldsymbol{\nu} = (\nu_i)_{1 \le i \le N}$ is a multiplicative operator satisfying (5.2.4):

$$\forall v \in \mathbb{R}^3, \quad \nu_i^{(0)}(1+|v|^\gamma) \le \nu_i(v) \le \nu_i^{(1)}(1+|v|^\gamma),$$

with
$$\nu_i^{(0)}$$
, $\nu_i^{(1)} > 0$.

In the case of single-species Boltzmann equation, the operator \mathbf{K} can be written as a kernel operator ([75] or [34] Section 7.2) and we give here a similar property where the different exponential decay rates, due to the different masses, are explicitly taken into account. These explicit bounds will be strongly needed for the L^{∞} theory.

Lemma 5.14. Let **f** be in $L_v^2(\boldsymbol{\mu}^{-1/2})$. Then for all i in $\{1,\ldots,N\}$ there exists $\mathbf{k}^{(i)}$ such that

$$K_i(\mathbf{f})(v) = \int_{\mathbb{R}^3} \langle \mathbf{k}^{(i)}(v, v_*), \mathbf{f}(v_*) \rangle dv_*.$$

Moreover there exist $m, C_K > 0$ such that for all i in $\{1, \ldots, N\}$ and for all $1 \leq j \leq N$

$$\left| k_j^{(i)}(v, v_*) \right| \le C_K \sqrt{\frac{\mu_i(v)}{\mu_j(v_*)}} \left[|v - v_*|^{\gamma} + |v - v_*|^{\gamma - 2} \right] e^{-m|v - v_*|^2 - m\frac{\left| |v|^2 - |v_*|^2 \right|^2}{|v - v_*|^2}}. \tag{5.4.1}$$

The constants m and C_K are explicit and depend only on $(m_i)_{1 \leq i \leq N}$ and the collision kernel B.

Proof of Lemma 5.14. By definition, $\mathbf{K} = (K_i)_{1 \le i \le N}$ with

$$K_{i}(\mathbf{f})(v) = \sum_{j=1}^{N} \int_{\mathbb{S}^{2} \times \mathbb{R}^{3}} B_{ij}(|v - v_{*}|, \cos \theta) \left[\mu_{j}^{'*} f_{i}^{'} + \mu_{i}^{'} f_{j}^{'*} - \mu_{i} f_{j}^{*} \right] d\sigma dv_{*}.$$
 (5.4.2)

We used the identity $\mu_i(v)\mu_j(v_*) = \mu_i(v')\mu_j(v'_*)$ that is a consequence of the conservation of energy during an elastic collision.

Step 1: A kernel form. The third term in the integral is already in the desired form. The first two terms require a new representation of the collision kernel where the integrand parameters will be v' and v'_* instead of v_* and σ . Such a representation has been obtained in the case of a single-species Boltzmann equation and is called the Carleman representation [30]. We derive below the Carleman representation associated with the multi-species Boltzmann operator. We follow the methods used in [30][34] Section 7.2 and [71]. However, the existence of different masses generates an asymmetry between v' and v'_* as we shall see.

The laws of elastic collisions gives

$$v' = V + \frac{m_j}{m_i + m_j} |v - v_*| \sigma$$
 and $v'_* = V - \frac{m_i}{m_i + m_j} |v - v_*| \sigma$

where V is the center of mass of the particles i and j:

$$V = \frac{m_i}{m_i + m_j} v + \frac{m_j}{m_i + m_j} v_*.$$

We can also express

$$v = V + \frac{m_j}{m_i + m_j} (v - v_*)$$
 and $v_* = V - \frac{m_i}{m_i + m_j} (v - v_*)$.

Note that

$$|v - v_*| = |v' - v_*'|,$$
 (5.4.3)

$$|v - v'| \le \frac{2m_j}{m_i + m_j} |v' - v_*'|,$$
 (5.4.4)

$$|v - v_*'| \le |v' - v_*'|.$$
 (5.4.5)

The points v, v_* , v' and v'_* therefore belong to the plane defined by V and $\mathrm{Span}(\sigma, v - v_*)$. We have the following geometric configuration, which gives a perfect circle in the case of equal masses.

Geometrically, $m_j^{-1}(v-V)$, $m_j^{-1}(v'-V)$, $m_i^{-1}(v_*-V)$ and $m_i^{-1}(v_*'-V)$ are on the same circle of diameter $\left|m_i^{-1}(v'^*-V) - m_j(v'-V)\right| = \frac{2}{m_i+m_j}|v-v_*|$. Therefore,

$$\left\langle \frac{1}{m_i}(v'_* - V) - \frac{1}{m_j}(v - V), \frac{1}{m_j}(v - V) - \frac{1}{m_j}(v' - V) \right\rangle = 0.$$

Using the laws of elasticity to see that

$$V = \frac{m_i}{m_i + m_j} v' + \frac{m_j}{m_i + m_j} v'_*$$

we end up with the following orthogonal property (that is also easily checked by direct computations)

$$\left\langle v'_* - \left(\frac{m_i + m_j}{2m_j} v - \frac{m_i - m_j}{2m_j} v' \right), v - v' \right\rangle = 0.$$
 (5.4.6)

We can now apply the change of variables $(v_*, \sigma) \mapsto (v', v'_*)$, where v' evolves in \mathbb{R}^3 and v'_* in $E^{ij}_{vv'}$. $E^{ij}_{vv'}$ is the hyperplane that passes through

$$V_E(v, v') = \frac{m_i + m_j}{2m_j} v - \frac{m_i - m_j}{2m_j} v'$$
(5.4.7)

and is orthogonal to v - v'; we denote $dE(v'_*)$ the Lebesgue measure on it. Note that $v_* = V(v', v'_*)$ is now a function of v' and v'_* :

$$V(v', v'_*) = v'_* + m_i m_j^{-1} v' - m_i m_j^{-1} v.$$

Up to the translation and dilatation (generating a constant $C_{ij} > 0$ only depending on m_i and m_j) from v to the origin of $E_{vv'}^{ij}$, this change of variables works as derived in [71]. Our operator thus reads

$$\int_{\mathbb{R}^{3}\times\mathbb{S}^{2}} B(v-v_{*},\sigma)f'g'^{*}dv_{*}d\sigma$$

$$= C_{ij} \int_{\mathbb{R}^{3}} \frac{1}{|v-v'|} \left(\int_{E_{vv'}^{ij}} \frac{B\left(v-V(v',v'_{*}),\frac{v'_{*}-v'}{|v'_{*}-v'|}\right)}{|v'_{*}-v'|} g'^{*}dE(v'_{*}) \right) f'dv'.$$
(5.4.8)

We can also give a Carleman representation where we first integrate against v'_* . In the case $m_i = m_j$ the orthogonal property (5.4.6) is entirely symmetric in v' and v'_* and we

reach the same representation (5.4.8) with the role of v' and v'_* swapped. This is the classical case of a single-species Boltzmann operator.

In the case $m_i \neq m_j$, (5.4.6) is equivalent to

$$|v'|^2 - 2\left\langle v', \frac{m_i}{m_i - m_j}v - \frac{m_j}{m_i - m_j}v'_* \right\rangle = \left\langle v, \frac{2m_j}{m_i - m_j}v'_* - \frac{m_i + m_j}{m_i - m_j}v \right\rangle$$

which is itself equivalent to

$$\left| v' + \left(\frac{m_j}{m_i - m_j} v'_* - \frac{m_i}{m_i - m_j} v \right) \right|^2 = \left| \frac{m_j}{m_i - m_j} v'_* - \frac{m_j}{m_i - m_j} v \right|^2.$$
 (5.4.9)

The same change of variables as before but $(v_*, \sigma) \mapsto (v'_*, v')$ instead of $(v_*, \sigma) \mapsto (v', v'_*)$ thus yields

$$\int_{\mathbb{R}^{3}\times\mathbb{S}^{2}} B(v-v_{*},\sigma)f'g'^{*} dv_{*}d\sigma$$

$$= C_{ij} \int_{\mathbb{R}^{3}} \frac{1}{|v-v'_{*}|} \left(\int_{\widetilde{E}_{vv'_{*}}^{ij}} \frac{B\left(v-V(v',v'_{*}), \frac{v'_{*}-v'}{|v'_{*}-v'|}\right)}{|v'_{*}-v'|} f' dE(v') \right) g'^{*} dv'_{*},$$
(5.4.10)

where $\widetilde{E}_{vv'_*}^{ij}$ stands for $E_{vv'_*}^{ij}$ if $m_i = m_j$ or for the sphere defined by (5.4.9); and dE is the Lebesgue measure on it.

We therefore conclude gathering (5.4.2), (5.4.8) and (5.4.10) with a relabelling of the integrated variables,

$$K_{i}(\mathbf{f})(v) = \sum_{j=1}^{N} C_{ij} \int_{\mathbb{R}^{3}} \left(\frac{1}{|v - v_{*}|} \int_{\widetilde{E}_{vv_{*}}^{ij}} \frac{B_{ij} \left(v - V(u, v_{*}), \frac{v_{*} - u}{|u - v_{*}|} \right)}{|u - v_{*}|} \mu_{i}(u) dE(u) \right) f_{j}^{*} dv_{*}$$

$$+ \sum_{j=1}^{N} C_{ji} \int_{\mathbb{R}^{3}} \left(\frac{1}{|v - v_{*}|} \int_{E_{vv_{*}}^{ij}} \frac{B_{ij} \left(v - V(v_{*}, u), \frac{u - v_{*}}{|u - v_{*}|} \right)}{|u - v_{*}|} \mu_{j}(u) dE(u) \right) f_{i}^{*} dv_{*}$$

$$- \sum_{j=1}^{N} \int_{\mathbb{R}^{3}} B_{ij} \left(|v - v_{*}|, \cos \theta \right) \mu_{i}(v) f_{j}^{*} dv_{*}.$$

$$(5.4.11)$$

This concludes the fact that K_i is a kernel operator.

Step 2: Pointwise estimate. It remains to show the pointwise estimate (5.4.1). The assumptions on B_{ij} imply that

$$\left| B_{ij} \left(v - V(v_*, u), \frac{u - v_*}{|u - v_*|} \right) \right| \le C |v - V(v_*, u)|^{\gamma},$$

where C denotes any positive constant independent of v and v_* . We shall bound each of the three terms in (5.4.11) separately.

From elastic collision laws (5.4.3), for u in $E_{vv_*}^{ij}$ one has $|v - V(v_*, u)| = |u - v_*|$, and hence

$$\left| \int_{E_{vv_*}^{ij}} \frac{B_{ij} \left(v - V(v_*, u), \frac{u - v_*}{|u - v_*|} \right)}{|u - v_*|} \mu_j(u) dE(u) \right| \le C \int_{E_{vv_*}^{ij}} \frac{1}{|u - v_*|^{1 - \gamma}} e^{-m_j \frac{|u|^2}{2}} dE(u).$$

We can further bound, since (5.4.4) is valid on $E_{vv_*}^{ij}$,

$$|u - v_*| \ge \frac{m_i + m_j}{2m_j} |v - v_*|,$$

and get

$$\left| \int_{E_{vv_*}^{ij}} \frac{B_{ij} \left(v - V(v_*, u), \frac{u - v_*}{|u - v_*|} \right)}{|u - v_*|} \mu_j(u) \ dE(u) \right| \le \frac{C}{|v - v_*|^{1 - \gamma}} \int_{E_{vv_*}} e^{-m_j \frac{|u|^2}{2}} \ dE(u).$$

To estimate the integral over $E^{ij}_{vv_*}$ we make the change of variables

$$u = V_E(v, v_*) + w$$

with $V_E(v, v_*)$ the origin (5.4.7) of $E_{vv_*}^{ij}$ and w in $(\operatorname{Span}(v - v_*))^{\perp}$. Using $\langle v, w \rangle = \langle v_*, w \rangle$ we compute

$$|u|^{2} = |V_{E}(v, v_{*}) + w|^{2} = \left| w + \frac{1}{2}(v + v_{*}) + \frac{m_{i}}{2m_{j}}(v - v_{*}) \right|^{2}$$

$$= \left| w + \frac{1}{2}(v + v_{*}) \right|^{2} + \frac{m_{i}^{2}}{4m_{j}^{2}}|v - v_{*}|^{2} + \frac{m_{i}}{2m_{j}}\left(|v|^{2} - |v_{*}|^{2}\right).$$

Now we decompose $v + v_* = V^{\perp} + V^{\parallel}$ where V^{\parallel} is the projection onto $\operatorname{Span}(v - v_*)$ and V^{\perp} is the orthogonal part. This implies

$$|u|^2 = \left| w + \frac{1}{2} V^{\perp} \right|^2 + \frac{1}{4} \left| V^{\parallel} \right|^2 + \frac{m_i^2}{4 m_i^2} |v - v_*|^2 + \frac{m_i}{2 m_i} \left(|v|^2 - |v_*|^2 \right).$$

By definition,

$$\left|V^{\parallel}\right|^{2} = \frac{\langle v + v_{*}, v - v_{*} \rangle^{2}}{\left|v - v_{*}\right|^{2}} = \frac{\left|\left|v\right|^{2} - \left|v_{*}\right|^{2}\right|^{2}}{\left|v - v_{*}\right|^{2}}$$

and therefore the following holds

$$\left| \frac{1}{|v - v_{*}|} \int_{E_{vv_{*}}^{ij}} \frac{B_{ij} \left(v - V(v_{*}, u), \frac{u - v_{*}}{|u - v_{*}|} \right)}{|u - v_{*}|} \mu_{j}(u) dE(u) \right| \\
\leq \frac{C}{|v - v_{*}|^{2 - \gamma}} e^{-\frac{m_{i}^{2}}{8m_{j}}|v - v_{*}|^{2} - \frac{m_{j}}{8}} \frac{||v|^{2} - |v_{*}|^{2}|^{2}}{|v - v_{*}|^{2}} \sqrt{\frac{\mu_{i}(v)}{\mu_{i}(v_{*})}} \left[\int_{(v - v_{*})^{\perp}} e^{-\frac{m_{j}}{2}} |w + \frac{1}{2}V^{\perp}|^{2} dE(w) \right].$$
(5.4.12)

The space $(v - v_*)^{\perp}$ is invariant by translation of vector $-2^{-1}V^{\perp}$ and the exponential term inside the integral only depends on the norm and therefore the integral term is a constant not depending on v or v_* .

We now turn to the term involving $\widetilde{E}_{vv'}^{ij}$ which is a bit more technical. In the case $m_i = m_j$ then $\widetilde{E}_{vv'}^{ij} = E_{vv'}^{ij}$. We therefore have the bound (5.4.12) to which we use $\mu_i(v)\mu_i^{-1}(v) = C_{ij}\mu_j(v)\mu_i^{-1}(v)$ since $m_i = m_j$.

Assume now that $m_i \neq m_j$. As for $E_{vv_*}^{ij}$, the elastic collision properties (5.4.3) and (5.4.5) give for all v_* in \mathbb{R}^3 and u in $\widetilde{E}_{vv_*}^{ij}$

$$\left| \int_{\widetilde{E}_{vv_*}^{ij}} \frac{B_{ij} \left(v - V(u, v_*), \frac{v_* - u}{|u - v_*|} \right)}{|u - v_*|} \mu_i(u) dE(u) \right| \leq \frac{C}{|v - v_*|^{1 - \gamma}} \int_{\widetilde{E}_{vv_*}^{ij}} e^{-m_i \frac{|u|^2}{2}} dE(u).$$

Since $\widetilde{E}_{vv_*}^{ij}$ is the sphere of radius

$$R_{vv_*} = \frac{m_j}{|m_i - m_j|} |v - v_*|$$

and centered at

$$O_{vv_*} = \frac{m_i}{m_i - m_j} v - \frac{m_j}{m_i - m_j} v_*.$$

We make a change of variables to end up on \mathbb{S}^2

$$\left| \frac{1}{|v - v_{*}|} \int_{\widetilde{E}_{vv_{*}}^{ij}} \frac{B_{ij} \left(v - V(u, v_{*}), \frac{v_{*} - u}{|u - v_{*}|} \right)}{|u - v_{*}|} \mu_{i}(u) dE(u) \right| \\
\leq C |v - v_{*}|^{\gamma} \int_{\mathbb{S}^{2}} e^{-\frac{m_{i}}{2} |R_{vv_{*}}u + O_{vv_{*}}|^{2}} d\sigma(u). \tag{5.4.13}$$

Decomposing the norm inside the integral and using Cauchy-Schwarz inequality yields

$$-\frac{m_{i}}{2} |R_{vv_{*}}u + O_{vv_{*}}|^{2} \le -\frac{m_{i}m_{j}^{2}}{2(m_{i} - m_{j})^{2}} |v - v_{*}|^{2} - \frac{m_{i}}{2(m_{i} - m_{j})^{2}} |m_{i}v - m_{j}v_{*}|^{2} + \frac{m_{i}m_{j}}{(m_{i} - m_{j})^{2}} |v - v_{*}| |m_{i}v - m_{j}v_{*}|$$

$$(5.4.14)$$

The idea is to express everything in terms of $|v-v_*|$ and $\frac{|v|^2-|v_*|^2}{|v-v_*|}$. We recall that we defined $v+v_*=V^\perp+V^\parallel$ with V^\perp orthogonal to $\mathrm{Span}(v-v_*)$ and $V^\parallel=\frac{\langle v+v_*,v-v_*\rangle}{|v-v_*|}(v-v_*)$. We first use the identity

$$|v - v_*| |m_i v - m_j v_*| = \frac{1}{4} \left[|(1 + m_i)v - (1 + m_j)v_*|^2 - |(1 - m_i)v - (1 - m_j)v_*|^2 \right]$$

and then the following equality that holds for all a and b,

$$|av - bv_*|^2 = \left| \frac{a-b}{2}(v+v_*) + \frac{a+b}{2}(v-v_*) \right|^2$$

$$= \frac{(a-b)^2}{4} \left| V^{\perp} \right|^2 + \frac{(a-b)^2}{4} \frac{\left| |v|^2 - |v_*|^2 \right|^2}{\left| v - v_* \right|^2} + \frac{(a-b)(a+b)}{2} \left(|v|^2 - |v_*|^2 \right) + \frac{(a+b)^2}{4} \left| v - v_* \right|^2.$$
 (5.4.15)

Direct computations from (5.4.14) then yield

$$-\frac{m_i}{2} |R_{vv_*} u + O_{vv_*}|^2 \le -\frac{m_i}{8} |V^{\perp}|^2 - \frac{m_i}{8} \frac{||v|^2 - |v_*|^2|^2}{|v - v_*|^2} - \frac{m_i}{4} (|v|^2 - |v_*|^2) - \frac{m_i}{8} |v - v_*|^2.$$

Taking (a, b) = (1, 0) and (a, b) = (0, 1) in (5.4.15) we have

$$\frac{m_i}{4} |v|^2 - \frac{m_j}{4} |v_*|^2 = \frac{m_i - m_j}{16} |V^{\perp}|^2 + \frac{m_i - m_j}{16} \frac{||v|^2 - |v_*|^2|^2}{|v - v_*|^2} + \frac{m_i + m_j}{8} (|v|^2 - |v_*|^2) + \frac{m_i - m_j}{16} |v - v_*|^2.$$

At last we obtain

$$-\frac{m_{i}}{2} |R_{vv_{*}}u + O_{vv_{*}}|^{2} \leq -\frac{m_{i}}{4} |v|^{2} + \frac{m_{j}}{4} |v_{*}|^{2} - \frac{m_{i} + m_{j}}{16} |V^{\perp}|^{2} + U\left(\frac{|v|^{2} - |v_{*}|}{|v - v_{*}|}, |v - v_{*}|\right)$$

$$(5.4.16)$$

where U(x,y) is a quadratic form defined by

$$U(x,v) = -\frac{m_i + m_j}{16}x^2 - \frac{m_i - m_j}{8}xy - \frac{m_i + m_j}{16}y^2.$$

The latter quadratic form is associated with the symmetric matrix

$$\left(\begin{array}{ccc}
-\frac{m_i + m_j}{16} & -\frac{m_i - m_j}{16} \\
-\frac{m_i - m_j}{16} & -\frac{m_i + m_j}{16}
\end{array}\right)$$

which has a negative trace and determinant $m_i m_j / 64 > 0$. It therefore is a negative definite symmetric matrix and thus, denoting by $-\lambda(m_i, m_j) < 0$ its largest eigenvalue we have

$$\forall (x,x) \in \mathbb{R}^2, \quad U(x,v) \le -\lambda(m_i,m_j) \left[x^2 + y^2\right].$$

Plugging the latter into (5.4.16) and going back to the integral of interest (5.4.13) we get

$$\left| \frac{1}{|v - v_{*}|} \int_{\widetilde{E}_{vv_{*}}^{ij}} \frac{B_{ij} \left(v - V(u, v_{*}), \frac{v_{*} - u}{|u - v_{*}|} \right)}{|u - v_{*}|} \mu_{i}(u) dE(u) \right| \\
\leq C |v - v_{*}|^{\gamma} e^{-\lambda (m_{i}, m_{j})|v - v_{*}|^{2} - \lambda (m_{i}, m_{j})} \frac{||v|^{2} - |v_{*}|^{2}|^{2}}{|v - v_{*}|^{2}} \sqrt{\frac{\mu_{i}(v)}{\mu_{j}(v_{*})}}.$$
(5.4.17)

To conclude we turn to the last integral term in (5.4.11) which is easily bounded by

$$|B_{ij}(|v - v_*|, \cos \theta) \mu_i(v)| \le C |v - v_*|^{\gamma} \mu_i(v)$$

$$\le C |v - v_*|^{\gamma} e^{-\frac{1}{4}(m_i|v|^2 + m_j|v_*|^2)} \sqrt{\frac{\mu_i(v)}{\mu_j(v_*)}}.$$

Using Cauchy-Schwartz

$$|v - v_*|^2 + \frac{\left||v|^2 - |v_*|^2\right|^2}{|v - v_*|^2} = |v - v_*|^2 + \frac{|\langle v - v_*, v + v_* \rangle|^2}{|v - v_*|^2}$$

$$\leq |v - v_*|^2 + |v + v_*|^2 = 2\left(|v|^2 + |v_*|^2\right),$$

this implies

$$|B_{ij}(|v-v_*|,\cos\theta)\mu_i(v)| \le C|v-v_*|^{\gamma} e^{-\frac{m_{ij}}{8}|v-v_*|^2 - \frac{m_{ij}}{8}\frac{||v|^2 - |v_*|^2}{|v-v_*|^2}} \sqrt{\frac{\mu_i(v)}{\mu_j(v_*)}}, \quad (5.4.18)$$

where $m_{ij} = \min\{m_i, m_j\}$.

Gathering (5.4.11)-(5.4.12)-(5.4.17)-(5.4.18) gives the desired estimate on
$$k_j^{(i)}$$
.

The pointwise estimate on $k_j^{(i)}$ can be transferred into a decay of the L_v^1 -norm with a relatively important weight. This has been proved in [83, Lemma 7] for the right-hand side of (5.4.1) with m = 1/8. The case of general m is identical and leads to

Lemma 5.15. Let $\beta > 0$ and θ in [0, 1/(32m)). There exists $C_{\theta,\beta} > 0$ and $\varepsilon_{\theta,\beta} > 0$ such that for all i, j in $\{1, \ldots, N\}$ and all ε in $[0, \varepsilon_{\theta,\beta})$,

$$\int_{\mathbb{R}^3} \left| k_j^{(i)}(v,v_*) \right| e^{\varepsilon m|v-v_*|^2 + \varepsilon m \frac{\left| |v|^2 - |v_*|^2 \right|^2}{|v-v_*|^2}} \frac{\langle v \rangle^\beta e^{\theta|v|^2} \mu_i(v)^{-1/2}}{\langle v_* \rangle^\beta e^{\theta|v_*|^2} \mu_j(v_*)^{-1/2}} \ dv_* \leq \frac{C_{\beta,\theta}}{1 + |v|}.$$

From Lemma 5.14 and 5.15 we conclude that **K** is a bounded operator on L_v^{∞} ($\langle v \rangle^{\beta} \mu^{-1/2}$).

5.4.2. Semigroup generated by the linear part

Following ideas developed in [83] in the case of bounded domains, the L^2 theory could be used to construct a L^{∞} one by using the flow of characteristics to transfer pointwise estimates at x-vt into integral in the space variable. Such a method is the core of the L^{∞} theory thanks to the following lemma.

Lemma 5.16. Let $\beta > 3/2$ and let (H1) - (H4) hold for the collision kernel. Assume that there exist $T_0 > 0$ and λ , $C_{T_0} > 0$ such that for all $\mathbf{f}(t, x, v)$ in $L_{x,v}^{\infty}(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2})$ solution to

$$\partial_t \mathbf{f} + v \cdot \nabla_x \mathbf{f} = \mathbf{L}(\mathbf{f}) \tag{5.4.19}$$

with initial data $\mathbf{f_0}$, the following holds for all t in $[0, T_0]$

$$\|\mathbf{f}(t)\|_{L^{\infty}_{x,v}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)} \leq e^{\lambda(T_{0}-2t)} \|\mathbf{f}_{\mathbf{0}}\|_{L^{\infty}_{x,v}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)} + C_{T_{0}} \int_{0}^{t} \|\mathbf{f}(s)\|_{L^{2}_{x,v}\left(\boldsymbol{\mu}^{-1/2}\right)} ds.$$

Then for all $0 < \widetilde{\lambda} < \min\{\lambda, \lambda_G\}$, defined in Theorem 5.11, there exists $C = C\left(\beta, \widetilde{\lambda}\right) > 0$ such that for all \mathbf{f} solution to (5.4.19) in $L_{x,v}^{\infty}\left(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2}\right)$ satisfying $\Pi_{\mathbf{G}}(\mathbf{f}) = 0$,

$$\forall t \geq 0, \quad \|\mathbf{f}(t)\|_{L^{\infty}_{x,v}\left(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2}\right)} \leq C e^{-\widetilde{\lambda} t} \|\mathbf{f}_{\mathbf{0}}\|_{L^{\infty}_{x,v}\left(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2}\right)}.$$

Proof of Lemma 5.16. To shorten the computations we use the following notation $\mathbf{w}_{\beta}(v) = \langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2}$.

Let **f** be a solution to (5.4.19) in $L_{x,v}^{\infty}(\mathbf{w}_{\beta})$ associated with the initial data **f**₀. Taking n in \mathbb{N} we can apply the assumption of the lemma to $\widetilde{\mathbf{f}}(t,x,v) = \mathbf{f}(t+nT_0,x,v)$. This yields, with a change of variables $t \mapsto t - nT_0$,

$$\|\mathbf{f}((n+1)T_0)\|_{L^{\infty}_{x,v}(\mathbf{w}_{\beta})} \leq e^{-\lambda T_0} \|\mathbf{f}(nT_0)\|_{L^{\infty}_{x,v}(\mathbf{w}_{\beta})} + C_{T_0} \int_{nT_0}^{(n+1)T_0} \|\mathbf{f}(s)\|_{L^{2}_{x,v}(\boldsymbol{\mu}^{-1/2})} ds.$$

We can iterate the process for $\mathbf{f}(nT_0)$ as long as $n \neq 0$. We thus obtain

$$\|\mathbf{f}((n+1)T_{0})\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} \leq e^{-(n+1)\lambda T_{0}} \|\mathbf{f}_{0}\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} + C_{T_{0}} \sum_{k=0}^{n} e^{-k\lambda T_{0}} \int_{(n-k)T_{0}}^{(n+1-k)T_{0}} \|\mathbf{f}(s)\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})} ds.$$

$$(5.4.20)$$

We see that multiplying and dividing by $\langle v \rangle^{\beta}$ gives

$$\|\mathbf{f}\|_{L^{2}_{x,v}\left(\boldsymbol{\mu}^{-1/2}\right)}^{2} = \sum_{i=1}^{N} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} f_{i}^{2} \mu_{i}^{-1} \, dx dv \leq \left| \mathbb{T}^{3} \right| \left(\int_{\mathbb{R}^{3}} \frac{dv}{\left(1 + \left| v \right|^{2}\right)^{\beta}} \right) \|\mathbf{f}\|_{L^{\infty}_{x,v}\left(\mathbf{w}_{\beta}\right)}^{2}.$$

Since $\beta > 3/2$, the integral is finite and \mathbf{f} also belongs to $L_{x,v}^2(\boldsymbol{\mu}^{-1/2})$. By Theorem 5.11 it follows that $\mathbf{f}(t) = S_{\mathbf{G}}(t)(\mathbf{f_0})$ and thus if $\Pi_{\mathbf{G}}(\mathbf{f}) = 0$ we have the following exponential decay

$$\forall t \geq 0, \quad \|\mathbf{f}(t)\|_{L^{2}_{x,v}(\boldsymbol{\mu}^{-1/2})} \leq C_{G}e^{-\lambda_{G}t} \|\mathbf{f_{0}}\|_{L^{2}_{x,v}(\boldsymbol{\mu}^{-1/2})} \leq C_{G,\beta}e^{-\lambda_{G}t} \|\mathbf{f_{0}}\|_{L^{\infty}_{x,v}(\mathbf{w}_{\beta})}.$$

Plugging the latter into (5.4.20) and taking $0 < \tilde{\lambda} < \lambda_1 \le \min \{\lambda, \lambda_G\}$

$$\begin{split} \|\mathbf{f}((n+1)T_{0})\|_{L^{\infty}_{x,v}\left(\mathbf{w}_{\beta}\right)} \\ &\leq \left[e^{-(n+1)\lambda T_{0}} + C_{\beta,G}\left(\sum_{k=0}^{n}e^{-k\lambda_{1}T_{0}}\int_{(n-k)T_{0}}^{(n+1-k)T_{0}}e^{-\lambda_{1}s}\,ds\right)\right] \|\mathbf{f_{0}}\|_{L^{\infty}_{x,v}\left(\mathbf{w}_{\beta}\right)} \\ &\leq \left[e^{-(n+1)\lambda T_{0}} + \frac{C_{\beta,G}e^{\lambda_{1}T_{0}}}{\lambda_{1}}(n+1)e^{-(n+1)\lambda_{1}T_{0}}\right] \|\mathbf{f_{0}}\|_{L^{\infty}_{x,v}\left(\mathbf{w}_{\beta}\right)} \\ &\leq C_{T_{0},\tilde{\lambda}}e^{-(n+1)\tilde{\lambda}T_{0}} \|\mathbf{f_{0}}\|_{L^{\infty}_{x,v}\left(\mathbf{w}_{\beta}\right)}, \end{split}$$

where we used $(n+1)e^{-(n+1)\lambda_1T_0} \leq Ce^{-(n+1)\widetilde{\lambda}T_0}$.

At last, for $t \geq 0$ there exists n in \mathbb{N} such that $nT_0 \leq t \leq (n+1)T_0$. Using the inequality satisfied by $\sup_{0 \leq t \leq T_0} \|\mathbf{f}(t - nT_0, x, v)\|_{L^{\infty}_{x,v}(\mathbf{w}_{\beta})}$, same computations as above gives

$$\|\mathbf{f}(t)\|_{L^{\infty}_{x,v}(\mathbf{w}_{\beta})} \leq C \|\mathbf{f}((n+1)T_{0})\|_{L^{\infty}_{x,v}(\mathbf{w}_{\beta})} \leq Ce^{-(n+1)\widetilde{\lambda}T_{0}} \|\mathbf{f}_{\mathbf{0}}\|_{L^{\infty}_{x,v}(\mathbf{w}_{\beta})}$$

$$\leq Ce^{-\widetilde{\lambda}t} \|\mathbf{f}_{\mathbf{0}}\|_{L^{\infty}_{x,v}(\mathbf{w}_{\beta})},$$

where C is any positive constants depending on T_0 . This concludes the proof.

We now state the theorem about the linear perturbed equation.

Theorem 5.17. Let $\beta > 3/2$ and let assumptions (H1) - (H4) hold for the collision kernel. The linear perturbed operator $\mathbf{G} = \mathbf{L} - v \cdot \nabla_x$ generates a semigroup $S_{\mathbf{G}}(t)$ on $L_{x,v}^{\infty}\left(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2}\right)$. Moreover, there exists λ_{∞} and $C_{\infty} > 0$ such that

$$\forall t \geq 0, \quad \|S_{\mathbf{G}}(t) \left(Id - \Pi_{\mathbf{G}}\right)\|_{L^{\infty}_{x,v}\left(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2}\right)} \leq C_{\infty} e^{-\lambda_{\infty} t},$$

where $\Pi_{\mathbf{G}}$ is the orthogonal projection onto $Ker(\mathbf{G})$ in $L^2_{x,v}(\boldsymbol{\mu}^{-1/2})$. The constants C_{∞} and λ_{∞} are explicit and depend on β , N and the collision kernel.

Proof of Theorem 5.17. As before, we use the shorthand notations $\mathbf{w}_{\beta} = \langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2}$ and $w_{\beta i} = \langle v \rangle^{\beta} \mu_i^{-1/2}$.

Let $\mathbf{f_0}$ be in $L_{x,v}^{\infty}(\mathbf{w}_{\beta})$ with $\beta > 3/2$. If \mathbf{f} is solution of (5.4.19):

$$\partial_t \mathbf{f} = \mathbf{G}(\mathbf{f})$$

in $L_{x,v}^{\infty}(\mathbf{w}_{\beta})$ with initial data $\mathbf{f_0}$ then because $\beta > 3/2$ we have that \mathbf{f} belongs to $L_{x,v}^2(\boldsymbol{\mu}^{-1/2})$ and $\mathbf{f}(t) = S_{\mathbf{G}}(t)\mathbf{f_0}$ in this space. This implies first that \mathbf{f} has to be unique and second that $\mathrm{Ker}(\mathbf{G})$ and $(\mathrm{Ker}(\mathbf{G}))^{\perp}$ are stable under the flow of the equation (5.4.19). It suffices to

consider $\mathbf{f_0}$ such that $\Pi_{\mathbf{G}}(\mathbf{f_0}) = 0$ and to prove existence and exponential decay of solutions to (5.4.19) in $L_{x,v}^{\infty}(\mathbf{w}_{\beta})$ with initial data $\mathbf{f_0}$.

We recall that $\boldsymbol{\nu}(v) = (\nu_i(v))_{1 \leq i \leq N}$ is a multiplicative operator and so the existence of solutions to equation (5.4.19) is equivalent to the existence of a fixed point to its Duhamel's form along the characteristics of the free transport equation. These characteristic trajectories are straight lines of constant speed. We thus need to have existence and exponential decay of a fixed point $\mathbf{f} = (f_i)_{1 \leq i \leq N}$ to the following problem for all i in $\{1, \ldots, N\}$:

$$f_i(t, x, v) = e^{-\nu_i(v)t} f_{0i}(x - vt, v) + \int_0^t e^{-\nu_i(v)(t-s)} K_i\left(\mathbf{f}(s, x - (t-s)v, \cdot)\right)(v) ds.$$

Thanks to Lemma 5.14, each operator K_i is a kernel operator and we thus have for all i in $\{1,\ldots,N\}$,

$$f_i(t, x, v) = e^{-\nu_i(v)t} f_{0i}(x - vt, v)$$

$$+ \sum_{i=1}^N \int_0^t \int_{\mathbb{R}^3} e^{-\nu_i(v)(t-s)} k_j^{(i)}(v, v_*) f_j(s, x - (t-s)v, v_*) dv_* ds.$$

Iterating this Duhamel's form we end up with the following formulation

$$f_i(t, x, v) = D_1^{(i)}(\mathbf{f_0})(t, x, v) + D_2^{(i)}(\mathbf{f_0})(t, x, v) + D_3^{(i)}(\mathbf{f})(t, x, v)$$
(5.4.21)

where we define

$$D_1^{(i)}(\mathbf{f_0}) = e^{-\nu_i(v)t} f_{0i}(x - vt, v), \tag{5.4.22}$$

$$D_2^{(i)}(\mathbf{f_0}) = \sum_{j=1}^N \int_0^t \int_{\mathbb{R}^3} e^{-\nu_i(v)(t-s)} e^{-\nu_j(v_*)s} k_j^{(i)}(v, v_*)$$
 (5.4.23)

$$\times f_{0j}(x - (t - s)v - sv_*, v_*) dv_* ds,$$

$$D_{3}^{(i)}(\mathbf{f}) = \sum_{j=1}^{N} \sum_{l=1}^{N} \int_{0}^{t} \int_{0}^{s} \int_{\mathbb{R}^{3}} \int_{\mathbb{R}^{3}} e^{-\nu_{i}(v)(t-s)} e^{-\nu_{j}(v_{*})(s-s_{1})} k_{j}^{(i)}(v, v_{*}) k_{l}^{(j)}(v_{*}, v_{**})$$

$$\times f_{l}(s_{1}, x - (t-s)v - (s-s_{1})v_{*}, v_{**}) dv_{**} dv_{*} ds_{1} ds.$$
(5.4.24)

Thanks to this Duhamel's formulation, the existence of a fixed point to (5.4.21) in $L_t^{\infty} L_{x,v}^{\infty}(\mathbf{w}_{\beta})$ follows from a contraction argument. The computations required to prove such a contraction property follow exactly the ones leading to the exponential decay of the latter fixed point. We therefore solely prove that if \mathbf{f} satisfies (5.4.21) then \mathbf{f} decreases exponentially in $L_{x,v}^{\infty}(\mathbf{w}_{\beta})$.

We shall bound each of the terms (5.4.22), (5.4.23) and (5.4.24) separately. From (5.2.4), for all i there exists $\nu_0^{(i)} = \min_{v \in \mathbb{R}^3} \{\nu_i(v)\} > 0$. We define by $\nu_0 > 0$ the minimum of the $\nu_0^{(i)}$ and every positive constant independent of i and \mathbf{f} will be denoted by C_k .

The first term (5.4.22) is straightforwardly bounded.

$$\left\| D_1^{(i)}(\mathbf{f_0})(t) \right\|_{L_{x,v}^{\infty}(w_{\beta i})} \le e^{-\nu_0 t} \left\| f_{0i} \right\|_{L_{x,v}^{\infty}(w_{\beta i})}. \tag{5.4.25}$$

In the second term (5.4.23) we multiply and divide inside the v_* integral by $w_{\beta j}(v_*)$ and take the supremum $\mathbb{T}^3 \times \mathbb{R}^3$.

$$\begin{split} \left| w_{\beta i}(v) D_2^{(i)}(\mathbf{f_0})(t) \right| \leq & Cte^{-\nu_0 t} \\ & \times \sum_{j=1}^N \left(\int_{\mathbb{R}^3} \left| k_j^{(i)}(v, v_*) \right| \frac{\langle v \rangle^\beta \mu_i^{-1/2}}{\langle v_* \rangle^\beta \mu_{j*}^{-1/2}} \, dv_* \right) \|f_{0j}\|_{L^{\infty}_{x,v}(w_{\beta j})} \, . \end{split}$$

Applying Lemma 5.15 with $\theta = \varepsilon = 0$, the integral term is bounded uniformly in i, j and v. Hence

$$\left\| D_2^{(i)}(\mathbf{f_0})(t) \right\|_{L_{x,v}^{\infty}(w_{\beta i})} \le C_2 t e^{-\nu_0 t} \left\| \mathbf{f_0} \right\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})}. \tag{5.4.26}$$

The third and last term (5.4.24) is more involved analytically and requires to consider the cases $|v| \ge R$ and $|v| \le R$, for R to be chosen later, separately.

Step 1: $|\mathbf{v}| \geq \mathbf{R}$. We multiply and divide by $w_{\beta l}(v_{**})$ inside the v_* integral of (5.4.24) and take the supremum in space and velocity for f_l . The exponential factor can be bounded by

$$e^{-\nu_i(v)(t-s)-\nu_j(v_*)(s-s_1)} < e^{-\frac{\nu_0}{2}t}e^{-\frac{\nu_0}{2}(t-s)}e^{-\frac{\nu_0}{2}(t-s_1)}e^{\frac{\nu_0}{2}s}.$$

Hence, for all t, x and v,

$$\left| w_{\beta i}(v) D_{3}^{(i)}\left(\mathbf{f}\right)(t, x, v) \right| \\
\leq e^{-\frac{\nu_{0}}{2}t} \sum_{1 \leq j, l \leq N} \int_{0}^{t} \int_{0}^{s} e^{-\frac{\nu_{0}}{2}(t-s_{1})} \left(e^{\frac{\nu_{0}}{2}s} \|f_{l}\|_{L_{x, v}^{\infty}\left(w_{\beta l}\right)} \right) \\
\times \left[\int_{\mathbb{R}^{3}} \left| k_{j}^{(i)}(v, v_{*}) \right| \frac{w_{\beta i}(v)}{w_{\beta i}(v_{*})} \left(\int_{\mathbb{R}^{3}} \left| k_{l}^{(j)}(v_{*}, v_{**}) \right| \frac{w_{\beta i}(v_{*})}{w_{\beta l}(v_{**})} dv_{**} \right) dv_{*} \right] ds_{1} ds. \tag{5.4.27}$$

We use Lemma 5.15 twice to bound the term inside bracket independently of j, l and v by

$$\frac{C_{\beta}^2}{1+|v|} \le \frac{C_{\beta}^2}{1+R}.$$

We conclude

$$\sup_{|v| \ge R} \left| w_{\beta i}(v) D_3^{(i)}(\mathbf{f})(t, x, v) \right| \le \frac{C_3}{1 + R} e^{-\frac{\nu_0}{2}t} \sup_{0 \le s \le t} \left[e^{\frac{\nu_0}{2}} \|\mathbf{f}\|_{L_{x, v}^{\infty}(\mathbf{w}_{\beta})} \right]. \tag{5.4.28}$$

Step 2: $|\mathbf{v}| \leq \mathbf{R}$. In order for the change of variables $y = x - (t - s)v - (s - s_1)v_*$ in the v_* integral to be well-defined we need $s - s_1$ bounded from below. Moreover, in order to make the L^2 -norm appearing we would need to have $k_j^{(i)}(v, v_*)$ uniformly bounded which is not the case. We therefore need to approximate it uniformly by compactly supported

functions, which is possible on compact domains. We take $\eta > 0$ and divide (5.4.24) into four parts

$$D_{3}^{(i)}(\mathbf{f}) = \int_{0}^{t} \int_{s-\eta}^{s} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3}} d_{3}^{(i)} + \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| \ge 2R} \int_{\mathbb{R}^{3}} d_{3}^{(i)} + \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| < 2R} \int_{|v_{**}| < 3R} d_{3}^{(i)} + \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| < 2R} \int_{|v_{**}| < 3R} d_{3}^{(i)},$$

$$(5.4.29)$$

where, using

$$e^{-\nu_i(v)(t-s)-\nu_j(v_*)(s-s_1)} < e^{-\nu_0(t-s_1)}$$
.

we have the following bound

$$d_3^{(i)} \le e^{-\nu_0(t-s_1)} \sum_{1 \le j,l \le N} k_j^{(i)}(v,v_*) k_l^{(j)}(v_*,v_{**})$$

$$\times |f_l(s_1,x-(t-s)v-(s-s_1)v_*,v_{**})| \ dv_{**} dv_* ds_1 ds.$$

The first integral in (5.4.29) is dealt with by using Lemma 5.15 twice, as for (5.4.27). We get

$$\left| w_{\beta i} \int_{0}^{t} \int_{s-\eta}^{s} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3}} d_{3}^{(i)} \right| \leq C e^{-\frac{\nu_{0}}{2}t} \left(\int_{0}^{t} \int_{s-\eta}^{s} e^{-\frac{\nu_{0}}{2}(t-s_{1})} ds_{1} ds \right) \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \|\mathbf{f}\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} \right]
\leq \eta C e^{-\frac{\nu_{0}}{2}t} \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \|\mathbf{f}\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} \right].$$
(5.4.30)

For the second and third terms in (5.4.29) we remark that for $|v| \leq R$ we always have either $|v - v_*| \geq R$ or $|v_* - v_{**}| \geq R$ in the domain of integration and therefore we have for any $\varepsilon > 0$ either one of the following inequalities

$$\begin{aligned} & \left| k_j^{(i)}(v, v_*) \right| & \leq & e^{-m\varepsilon R^2} \left| k_j^{(i)}(v, v_*) e^{m\varepsilon |v - v_*|^2} \right| \\ & \left| k_l^{(j)}(v_*, v_{**}) \right| & \leq & e^{-m\varepsilon R^2} \left| k_l^{(j)}(v_*, v_{**}) e^{m\varepsilon |v_* - v_{**}|^2} \right|. \end{aligned}$$

Now we take ε small enough to apply Lemma 5.15 as before but with the first inequality above for $|v_*| \ge 2R$ or the second inequality above for $|v_*| \le 2R$ and $|v_{**}| \ge 3R$. Exactly the same computations as (5.4.27) before yields

$$\left| w_{\beta i} \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| \geq 2R} \int_{\mathbb{R}^{3}} d_{3}^{(i)} \right| \leq C e^{-m\varepsilon R^{2}} e^{-\frac{\nu_{0}}{2}t} \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \|\mathbf{f}\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} \right] 5.4.31)$$

$$\left| w_{\beta i} \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| \geq 2R} \int_{|v_{**}| \geq 3} d_{3}^{(i)} \right| \leq C e^{-m\varepsilon R^{2}} e^{-\frac{\nu_{0}}{2}t} \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \|\mathbf{f}\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} \right] 5.4.32)$$

At last, the last term in (5.4.29) deals with a set included in the compact support

$$\Omega = \left\{ (v, v_*, v_{**}) \in \mathbb{R}^3, \quad |v| \le 3R, \ |v_*| \le 2R, \ |v_{**}| \le 3R \right\}.$$

As discussed earlier, Lemma 5.14 shows that $k_j^{(i)}(v, v_*)$ has a possible blow-up in $|v - v_*|^{\gamma}$. However, since Ω is compact we can approximate $k_j^{(i)}(v, v_*)$, for all i and j, by a smooth and compactly supported function $k_{R,j}^{(i)}(v, v_*)$ in the following uniform sense

$$\sup_{|v| \le 3R} \int_{|v_*| \le 3R} \left| k_j^{(i)}(v, v_*) - k_{R,j}^{(i)}(v, v_*) \right| \frac{w_{\beta i(v)}}{w_{\beta i}(v_*)} \, dv_* \le \frac{1}{R}. \tag{5.4.33}$$

Thanks to the following equality

$$\begin{split} k_{j}^{(i)}(v,v_{*})k_{l}^{(j)}(v_{*},v_{**}) &= \left(k_{j}^{(i)}(v,v_{*}) - k_{R,j}^{(i)}(v,v_{*})\right)k_{l}^{(j)}(v_{*},v_{**}) \\ &+ \left(k_{l}^{(j)}(v_{*},v_{**}) - k_{R,l}^{(j)}(v_{*},v_{**})\right)k_{R,j}^{(i)}(v,v_{*}) \\ &+ k_{R,j}^{(i)}(v,v_{*})k_{R,l}^{(j)}(v_{*},v_{**}) \end{split}$$

the last term in (5.4.29) is bounded by

$$\begin{aligned} & \left| w_{\beta i} \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| \leq 2R} \int_{|v_{**}| \leq 3R} d_{3}^{(i)} \right| \\ & \leq \frac{C}{R} e^{-\frac{\nu_{0}}{2}t} \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \left\| \mathbf{f} \right\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} \right] \sup_{1 \leq i,j,l \leq N} \left(\sup_{|v_{*}| \leq 2R} \int_{|v_{**}| \leq 3R} \left| k_{l}^{(j)}(v_{*}, v_{**}) \right| \frac{w_{\beta i}(v_{*})}{w_{\beta l}(v_{**})} dv_{**} \right) \\ & + \frac{C}{R} e^{-\frac{\nu_{0}}{2}t} \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \left\| \mathbf{f} \right\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} \right] \sup_{1 \leq i,j \leq N} \left(\sup_{|v| \leq 2R} \int_{|v_{*}| \leq 2R} \left| k_{R,j}^{(i)}(v, v_{*}) \right| \frac{w_{\beta i}(v)}{w_{\beta i}(v_{*})} dv_{*} \right) \\ & + \sum_{1 \leq i,l \leq N} \int_{0}^{t} \int_{0}^{s-\eta} e^{-\nu_{0}(t-s_{1})} \int_{|v_{*}| \leq 2R} \left| k_{R,j}^{(i)}(v, v_{*}) k_{R,l}^{(j)}(v_{*}, v_{**}) \right| \left| f_{l}(s_{1}, y(v_{*}), v_{**}) \right| \end{aligned}$$

where we made the usual controls (5.4.27) and used (5.4.33). We also defined $y(v_*) = x - (t - s)v - (s - s_1)v_*$. The first two terms are dealt with using Lemma 5.15 while we can bound $k_{R,j}^{(i)}k_{R,l}^{(j)}$ by a constant C_R depending only on R (note that all constants only depending on R will be denoted by C_R). This yields

$$\begin{aligned} & \left| w_{\beta i} \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| \leq 2R} \int_{|v_{**}| \leq 3R} d_{3}^{(i)} \right| \\ & \leq \frac{C}{R} e^{-\frac{\nu_{0}}{2}t} \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \left\| \mathbf{f} \right\|_{L_{x,v}^{\infty}\left(\mathbf{w}_{\beta}\right)} \right] + C_{R} \sum_{l=1}^{N} \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| \leq 2R} \left| f_{l}(s_{1}, y(v_{*}), v_{**}) \right|. \end{aligned}$$

We first integrate over v_* . We make the change of variables $y = y(v_*)$ which has a jacobian $|s - s_1|^{-3} \le \eta^{-3}$. Since we are on the periodic box, y has to be understood as the class of equivalence of $y(v_*)$ and is therefore not one-to-one. However, v_* being bounded by 2R we cover \mathbb{T}^3 only finitely many times (depending on R). Hence,

$$\begin{split} & \left| w_{\beta i} \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| \leq 2R} \int_{|v_{**}| \leq 3R} d_{3}^{(i)} \right| \\ & \leq \frac{C}{R} e^{-\frac{\nu_{0}}{2}t} \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \left\| \mathbf{f} \right\|_{L_{x,v}^{\infty}\left(\mathbf{w}_{\beta}\right)} \right] + \frac{C_{R}}{\eta^{3}} \sum_{l=1}^{N} \int_{0}^{t} \int_{0}^{s-\eta} \int_{\mathbb{T}^{3} \times \mathbb{R}^{3}} \left| f_{l}(s_{1}, y, v_{**}) \right|. \end{split}$$

Finally, a Cauchy-Schwarz inequality against $\mu_l^{-1/2}(v_{**})/\mu_l^{-1/2}(v_{**})$ yields the following estimate

$$\left| w_{\beta i} \int_{0}^{t} \int_{0}^{s-\eta} \int_{|v_{*}| \leq 2R} \int_{|v_{**}| \leq 3R} d_{3}^{(i)} \right| \\
\leq \frac{C}{R} e^{-\frac{\nu_{0}}{2}t} \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \|\mathbf{f}\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} \right] + \frac{C_{R}}{\eta^{3}} t \int_{0}^{t} \|\mathbf{f}(s_{1})\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})} ds_{1}. \tag{5.4.34}$$

Plugging (5.4.30), (5.4.31), (5.4.32) and (5.4.34) into (5.4.29) gives the final estimate

$$\sup_{|v| \le R} \left| w_{\beta i}(v) D_3^{(i)}(\mathbf{f}) \right| \le C_4 e^{-\frac{\nu_0}{2}t} \left(\eta + e^{-m\varepsilon R^2} + \frac{1}{R} \right) \sup_{0 \le s \le t} \left[e^{\frac{\nu_0}{2}} \|\mathbf{f}\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} \right] + C_{R,\eta} t \int_0^t \|\mathbf{f}(s)\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})} ds.$$
(5.4.35)

We can now conclude the proof by gathering (5.4.21), (5.4.25), (5.4.26), (5.4.28) and (5.4.35). We get that for all i in $\{1, \ldots, N\}$

$$e^{\frac{\nu_{0}}{2}t} \|f_{i}(t)\|_{L_{x,v}^{\infty}(w_{\beta i})} \leq (1 + C_{2}t) e^{-\frac{\nu_{0}}{2}t} \|\mathbf{f}_{0}\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})} + C_{R,\eta}t \int_{0}^{t} \|\mathbf{f}(s)\|_{L_{x,v}^{2}(\boldsymbol{\mu}^{-1/2})} ds + C_{5} \left(\eta + e^{-m\varepsilon R^{2}} + \frac{1}{R}\right) \sup_{0 \leq s \leq t} \left[e^{\frac{\nu_{0}}{2}} \|\mathbf{f}\|_{L_{x,v}^{\infty}(\mathbf{w}_{\beta})}\right].$$

$$(5.4.36)$$

We remind the reader that C_2 and C_5 are independent of η , R and t; moreover $\varepsilon > 0$ is fixed. We choose R large enough and η small enough such that

$$C_5 \left(\eta + e^{-m\varepsilon R^2} + \frac{1}{R} \right) \le \frac{1}{2}$$

and $T_0 > 0$ such that

$$2(1+C_2T_0)e^{-\nu_0T_0} = e^{-\frac{\nu_0}{2}T_0}.$$

Such choices with (5.4.36) yields that for all t in $[0, T_0]$,

$$\|\mathbf{f}(t)\|_{L^{\infty}_{x,v}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)} \leq e^{\frac{\nu_{0}}{2}(T_{0}-2t)} \|\mathbf{f}_{0}\|_{L^{\infty}_{x,v}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)} + CT_{0} \int_{0}^{t} \|\mathbf{f}(s)\|_{L^{2}_{x,v}\left(\boldsymbol{\mu}^{-1/2}\right)} ds.$$

Lemma 5.16 then concludes the proof of Theorem 5.17.

5.5. The full nonlinear equation in a perturbative regime

This section is devoted to the proof of Theorem 5.3. We divide our study in three steps. Subsection 5.5.1 deals with the existence of a solution with exponential decay to the perturbed multi-species Boltzmann equation that reads

$$\partial_t \mathbf{f} + v \cdot \nabla_x \mathbf{f} = \mathbf{L}(\mathbf{f}) + \mathbf{Q}(\mathbf{f}).$$
 (5.5.1)

Then Subsection 5.5.2 proves the uniqueness of such solutions and, at last, Subsection 5.5.3 shows the positivity of the latter.

5.5.1. Existence of a solution that decays exponentially

We refer to the definition of $\Pi_{\mathbf{G}}$ (5.3.1) and recall that $\Pi_{\mathbf{G}}(\mathbf{f}) = 0$ is a convenient way to say that \mathbf{f} satisfies the conservation laws (5.1.5) with $\theta_{\infty} = 1$ and $u_{\infty} = 0$.

This subsection is dedicated to the proof of the following proposition.

Proposition 5.18. Let assumptions (H1) - (H4) hold for the collision kernel, and let $k > k_0$, where k_0 is the smallest integer such that $C_{k_0} < 1$ where C_k was given by (5.1.10). There exists η_k , C_k and $\lambda_k > 0$ such that for any $\mathbf{f_0}$ in $L_v^1 L_x^{\infty} (\langle v \rangle^k)$ satisfying $\Pi_{\mathbf{G}}(\mathbf{f_0}) = 0$, if

$$\|\mathbf{f_0}\| \leq \eta_k$$

then there exists \mathbf{f} in $L_v^1 L_x^\infty \left(\langle v \rangle^k \right)$ with $\Pi_{\mathbf{G}}(\mathbf{f}) = 0$ solution to (5.5.1) with initial data $\mathbf{f_0}$ such that

$$\forall t \ge 0, \quad \|\mathbf{f}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \le C_k e^{-\lambda_k t} \|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}.$$

The constants are explicit and only depend on N, k and the collision kernels.

Decomposition of the perturbed equation and toolbox

As explained in the introduction, the main strategy is to find a decomposition of the perturbed Boltzmann equation (5.5.1) into a system of differential equations where we could make use of the L^{∞} semigroup theory developed in Section 5.4. More precisely, one would like to solve a somewhat simpler equation in $L_v^1 L_x^{\infty} (\langle v \rangle^k)$ and that the remainder part has regularising properties and could thus be handled in the more regular space $L_{x,v}^{\infty} (\langle v \rangle^{\beta} \mu^{-1/2})$. Then the exponential decay of $S_{\mathbf{G}}(t)$ in the more regular space could be carried up to the bigger space.

Remark that

$$L^\infty_{x,v}\left(\langle v\rangle^\beta \pmb{\mu}^{-1/2}\right)\subset L^1_vL^\infty_x\left(\langle v\rangle^k\right).$$

We propose here a decomposition of the mutli-species linear operator $\mathbf{G} = \mathbf{L} - v \cdot \nabla_x$ that follows the idea used in [78] for the single-species Boltzmann operator. We define for

 $\delta \in (0,1)$ to be chosen later the truncation function $\Theta(v,v^*,\sigma) \in C^{\infty}(\mathbb{R}^3 \times \mathbb{R}^3)$ bounded by one on the set

$$\{|v| \le \delta^{-1} \text{ and } 2\delta \le |v - v^*| \le \delta^{-1} \text{ and } |\cos \theta| \le 1 - 2\delta\},$$

and its support included in the set

$$\{|v| \le 2\delta^{-1} \text{ and } \delta \le |v - v^*| \le 2\delta^{-1} \text{ and } |\cos \theta| \le 1 - \delta\}.$$

Thus we can define the splitting

$$\mathbf{G} = \mathbf{L} - v \cdot \nabla_x = \mathbf{A}^{(\delta)} + \mathbf{B}^{(\delta)} - \boldsymbol{\nu} - v \cdot \nabla_x,$$

with the operators $\mathbf{A}^{(\delta)} = \left(A_i^{(\delta)}\right)_{1 \le i \le N}$ and $\mathbf{B}^{(\delta)} = \left(B_i^{(\delta)}\right)_{1 \le i \le N}$ defined as

$$A_i^{(\delta)}(\mathbf{f}(v)) = \sum_{j=1}^N C_{ij}^{\Phi} \int_{\mathbb{R}^3 \times \mathbb{S}^2} \Theta_{\delta} \left(\mu_j'^* f_i' + \mu_i' f_j'^* - \mu_i f_j^* \right) b_{ij}(\cos \theta) |v - v^*|^{\gamma} d\sigma dv^*,$$

$$B_{i}^{(\delta)}(\mathbf{f}(v)) = \sum_{i=1}^{N} C_{ij}^{\Phi} \int_{\mathbb{R}^{3} \times \mathbb{S}^{2}} (1 - \Theta_{\delta}) \left(\mu_{j}^{\prime *} f_{i}^{\prime} + \mu_{i}^{\prime} f_{j}^{\prime *} - \mu_{i} f_{j}^{*} \right) b_{ij}(\cos \theta) |v - v^{*}|^{\gamma} d\sigma dv^{*}.$$

Our goal is to show that $\mathbf{A}^{(\delta)}$ has some regularizing effects and that $\mathbf{G_1} := \mathbf{B}^{(\delta)} - \boldsymbol{\nu} - \boldsymbol{v} \cdot \nabla_x$ acts like a small perturbation of $\boldsymbol{G_{\nu}} := -\boldsymbol{\nu} - \boldsymbol{v} \cdot \nabla_x$ and is thus hypodissipative.

Lemma 5.19. For any k in \mathbb{N} , $\beta > 0$ and δ in (0,1), there exists $C_A > 0$ such that for all \mathbf{f} in $L_v^1 L_x^\infty \left(\langle v \rangle^k \right)$

$$\left\| \mathbf{A}^{(\delta)} \left(\mathbf{f} \right) \right\|_{L_{x,v}^{\infty} \left(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2} \right)} \leq C_A \left\| \mathbf{f} \right\|_{L_v^1 L_x^{\infty} \left(\langle v \rangle^k \right)}.$$

The constant C_A is constructive and only depends on k, β , δ , N and the collision kernels.

Proof of Lemma 5.19. As we proved it in Lemma 5.14, the operator $\mathbf{A}^{(\delta)}$ can be written as a kernel operator thanks to Carleman representation:

$$\forall i \in \{1, \dots, N\}, \quad A_i^{(\delta)}(\mathbf{f})(x, v) = \int_{\mathbb{R}^3} \langle \mathbf{k}_{\mathbf{A}}^{(\mathbf{i}), (\delta)}(v, v_*), \mathbf{f}(x, v_*) \rangle dv_*.$$

Moreover, by definition of $\mathbf{A}^{(\delta)}$, its kernels $\mathbf{k}_{\mathbf{A}}^{(\mathbf{i}),(\delta)}$ are of compact support which implies the desired estimate.

Thanks to the regularizing property above of the operator $\mathbf{A}^{(\delta)}$ we are looking for solutions to the perturbed Boltzmann equation

$$\partial_t \mathbf{f} = \mathbf{G}(\mathbf{f}) + \mathbf{Q}(\mathbf{f})$$

in the form of $\mathbf{f} = \mathbf{f_1} + \mathbf{f_2}$ with $\mathbf{f_1}$ in $L_v^1 L_x^\infty \left(\langle v \rangle^k \right)$ and $\mathbf{f_2}$ in $L_{x,v}^\infty \left(\langle v \rangle^\beta \boldsymbol{\mu}^{-1/2} \right)$ and $(\mathbf{f_1}, \mathbf{f_2})$ satisfying the following system of equation

$$\partial_t \mathbf{f_1} = \mathbf{G_1}^{(\delta)}(\mathbf{f_1}) + \mathbf{Q}(\mathbf{f_1} + \mathbf{f_2}) \quad \text{and} \quad \mathbf{f_1}(0, x, v) = \mathbf{f_0}(x, v),$$
 (5.5.2)

$$\partial_t \mathbf{f_2} = \mathbf{G}(\mathbf{f_2}) + \mathbf{A}^{(\delta)}(\mathbf{f_1}) \quad \text{and} \quad \mathbf{f_2}(0, x, v) = 0.$$
 (5.5.3)

The equation in the smaller space (5.5.3) will be treated thanks to the semigroup generated by \mathbf{G} in $L^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)$ whilst we expect an exponential decay for solutions in the larger space (5.5.2). Indeed, $\mathbf{B}^{(\delta)}$ can be controlled by the multiplicative operator $\boldsymbol{\nu}(v)$ thanks to the following lemma.

Lemma 5.20. Define

$$\overline{\mathbf{w_k}} = \left(1 + m_i^{k/2} |v|^k\right)_{1 \le i \le N} \quad and \quad \overline{\mathbf{w_k}} \boldsymbol{\nu} = \left((1 + m_i^{k/2} |v|^k) \nu_i(v)\right)_{1 \le i \le N}.$$

There exists k_0 in \mathbb{N} such that for any $k \geq k_0$ and δ in (0,1) there exists $C_B(k,\delta) > 0$ such that for all \mathbf{f} in $L_v^1 L_x^\infty(\overline{\mathbf{w_k}}\boldsymbol{\nu})$,

$$\left\| \mathbf{B}^{(\delta)}(\mathbf{f}) \right\|_{L_v^1 L_x^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}})} \leq C_B(k, \delta) \left\| \mathbf{f} \right\|_{L_v^1 L_x^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}} \boldsymbol{\nu})}.$$

Moreover we have the following formula

$$C_B(k,\delta) = C_k + \varepsilon_k(\delta)$$

where $\varepsilon_k(\delta)$ is an explicit function that tends to 0 as δ tends to 0 and C_k is defined by (5.1.10) and k_0 is the minimal integer such that $C_{k_0} < 1$.

We make an important remark.

Remark 5.21. We emphasize here that for $k > k_0$ we have that $\lim_{\delta \to 0} C_B(k, \delta) = C_k < 1$. Until the end we fix $\delta_k > 0$ such that $C_B(k, \delta_k) < 1$. For convenience we will drop the exponent and use the following notations: $\mathbf{B} = \mathbf{B}^{(\delta_k)}$, $\mathbf{A} = \mathbf{A}^{(\delta_k)}$, $\mathbf{G_1} = \mathbf{G_1}^{(\delta_k)}$ and finally $C_B = C_B(k, \delta_k)$. The equivalent of this result for the mono-species Boltzmann equation can be found in [78, Lemma 4.4] for k > 2 which is recovered here when $m_i = m_j$ (note that our Lemma deals with more general collision kernels).

We also notice here that the weighted norm $\overline{\mathbf{w_k}}$ required for this sharp lemma is equivalent to $\langle v \rangle^k$.

The proof of Lemma 5.20 relies on a Povzner-type inequality. Such inequalities are now common in the mono-species Boltzmann literature (for both elastic and inelastic collisions) [124][109][8][9][78] and state that the integral on \mathbb{S}^2 of $\left[|v'|^k + |v_*'|^k\right]$ can be controlled strictly by the integral on \mathbb{S}^2 of $C_k\left[|v|^k + |v_*|^k\right]$ with $C_k = 4/(k+2) < 1$ (for hard sphere

collision kernels) and a remainder term of lower order when k > 2. As we shall see, the asymmetry brought by the difference of masses generates a larger constant C_k that can still be less than 1 if k is large enough.

The method proposed here to prove such a Povzner inequality is inspired by [9, Lemma 1 and Corollary 3]. The main idea is to consider kinetic energies $m_i |v'|^2$ and $m_j |v'_*|^2$ to exhibit the problematic term arising from $m_i - m_j$ which can be non-zero. We state our result, which covers the mono-species case when $m_i = m_j$.

Proposition 5.22 (Povzner-type inequality). Let i and j in $\{1, ..., N\}$. Then for all k > 2,

$$\int_{\mathbb{S}^{2}} \left[m_{i}^{k/2} \left| v' \right|^{k} + m_{j}^{k/2} \left| v_{*}' \right|^{k} \right] d\sigma \leq \frac{l_{b_{ij}}}{b_{ij}^{\infty}} C_{k} \left[m_{i} \left| v \right|^{2} + m_{j} \left| v_{*} \right|^{2} \right]^{k/2}$$

where C_k was defined by (5.1.10) and $l_{b_{ij}}$, b_{ij}^{∞} by (5.1.9).

Proof of Proposition 5.22. By definition of v' and v'_* we can expand $|v'|^2$ and $|v'_*|^2$ as follows

$$m_i |v'|^2 = E \frac{1 + a_{ij} + b_{ij} \langle e, \sigma \rangle}{2}$$

 $m_i |v'_*|^2 = E \frac{1 - a_{ij} - b_{ij} \langle e, \sigma \rangle}{2}$

where we denoted by e the direction of the vector $m_i v + m_j v_*$ and we defined

$$E = m_{i} |v|^{2} + m_{j} |v_{*}|^{2},$$

$$a_{ij} = \frac{1}{E} \frac{m_{i} - m_{j}}{m_{i} + m_{j}} \left[m_{i} \frac{m_{i} - m_{j}}{m_{i} + m_{j}} |v|^{2} + m_{j} \frac{m_{j} - m_{i}}{m_{i} + m_{j}} |v_{*}|^{2} + 4 \frac{m_{i} m_{j}}{m_{i} + m_{j}} \langle v, v_{*} \rangle \right],$$

$$b_{ij} = \frac{1}{E} \frac{4 m_{i} m_{j}}{(m_{i} + m_{j})^{2}} |v - v_{*}| |m_{i}v + m_{j}v_{*}|.$$
(5.5.4)

We will drop the dependencies on v and v_* The first important property to notice is that for all σ on \mathbb{S}^2 , $m_i |v'|^2$ and $m_j |v'_*|^2$ are positive and this implies

$$|a_{ij}| \le 1$$
, $|a_{ij} + b_{ij}| \le 1$ and $|a_{ij} - b_{ij}| \le 1$. (5.5.5)

Plugging these equalities inside the integral yields

$$\int_{\mathbb{S}^{2}} \left[m_{i}^{k/2} \left| v' \right|^{k} + m_{j}^{k/2} \left| v_{*}' \right|^{k} \right] d\sigma$$

$$= E^{k/2} \int_{\mathbb{S}^{2}} \left[\left(\frac{1 + a_{ij} + b_{ij} \langle e, \sigma \rangle}{2} \right)^{k/2} + \left(\frac{1 - a_{ij} - b_{ij} \langle e, \sigma \rangle}{2} \right)^{k/2} \right] d\sigma$$

$$= 2\pi E^{k/2} \int_{-1}^{1} \left[\left(\frac{1 + a_{ij} + b_{ij} z}{2} \right)^{k/2} + \left(\frac{1 - a_{ij} - b_{ij} z}{2} \right)^{k/2} \right] dz$$

$$= \frac{8\pi}{k+2} E^{k/2} \left[F_{k/2}(|a_{ij}|, b_{ij}) + F_{k/2}(-|a_{ij}|, b_{ij}) \right]$$
 (5.5.6)

where

$$F_p(a,b) = \frac{\left(\frac{1+a+b}{2}\right)^{p+1} - \left(\frac{1+a-b}{2}\right)^{p+1}}{b}.$$

When $|a| \leq 1$, a mere study of the function $F_p(a,\cdot)$ shows that the latter function is increasing on [0,1+a] if $p \geq 0$. Therefore, since $|a_{ij}| \leq 1$, for $k \geq 2$ we can bound $F_{k/2}(|a_{ij}|,b_{ij})$ and $F_{k/2}(-|a_{ij}|,b_{ij})$ with their value at an upper bound on b_{ij} . Using (5.5.5) we see that $0 \leq b_{ij} \leq 1 - |a_{ij}|$. Bounding into (5.5.6), this gives

$$\int_{\mathbb{S}^2} \left[m_i^{k/2} \left| v' \right|^k + m_j^{k/2} \left| v_*' \right|^k \right] d\sigma \le \frac{8\pi}{k+2} E^{k/2} \frac{1 - |a_{ij}|^{k/2+1} + (1 - |a_{ij}|)^{k/2+1}}{1 - |a_{ij}|}.$$
 (5.5.7)

To conclude the proof it suffices to see that the function

$$a \mapsto \frac{1 - |a|^{k/2+1} + (1 - |a|)^{k/2+1}}{1 - |a|}$$

is increasing on [0, 1]. Proposition 5.22 will follow if $|a_{ij}| \leq |m_i - m_j|/(m_i + m_j)$.

Going back to the definition of a_{ij} and decomposing v_* as $v_* = \lambda v + v^{\perp}$ with v^{\perp} orthogonal to v we see that

$$|a_{ij}| = \frac{1}{E} \frac{|m_i - m_j|}{m_i + m_j} \left| \left(\frac{m_i^2 + m_i m_j (4\lambda - \lambda^2 - 1) + \lambda^2 m_j^2}{m_i + m_j} \right) |v|^2 + m_j \frac{m_i - m_j}{m_i + m_j} \left| v^{\perp} \right|^2 \right|.$$

But then, direct computations show first

$$\left| m_j \frac{m_i - m_j}{m_i + m_j} \right| \le m_j$$

and second

$$\left| m_i^2 + m_i m_j (4\lambda - \lambda^2 - 1) + \lambda^2 m_j^2 \right|^2 - (m_i + m_j)^2 (m_i + \lambda^2 m_j)^2$$

= $-4m_i m_j (1 - \lambda)^2 (m_i + \lambda m_j)^2 \le 0.$

Hence

$$|a_{ij}| \le \frac{|m_i - m_j|}{m_i + m_j} \frac{(m_i + \lambda^2 m_j) |v|^2 + m_j |v^{\perp}|^2}{E}$$

which terminates the proof of the proposition.

Now we can prove the estimate on $\mathbf{B}^{(\delta)}$.

Proof of Lemma 5.20. We use the definition $\overline{\mathbf{w_k}} = (1 + m_i^{k/2} |v|^k)_{1 \le i \le N}$. Moreover, as we will drastically bound $\mathbf{B}^{(\delta)}(\mathbf{f})$ by the absolute value inside the integral in v, it is enough to show Lemma 5.20 only for f = f(v).

With the multi-species Povzner inequality (Proposition 5.22), the proof follows closely the proof of [78, Lemma 4.4] with appropriate characteristic functions that fit the invariance of the elastic collisions (5.1.2).

First we bound the truncation function from above by cutting the integral in the following way

$$\begin{split} & \left\| \mathbf{B}^{(\delta)}(\mathbf{f}) \right\|_{L_{v}^{1}(\overline{\mathbf{w_{k}}})} \\ & \leq \sum_{i,j=1}^{N} C_{ij}^{\Phi} \int_{\mathbb{R}^{6} \times \mathbb{S}^{2}} \left(1 - \Theta_{\delta} \right) \left[\mu_{j}^{'*} | f_{i}^{'} | + \mu_{i}^{'} | f_{j}^{'*} | + \mu_{i} | f_{j}^{*} | \right] b_{ij}(\cos \theta) | v - v_{*}|^{\gamma} \overline{w_{k}}_{i} dv_{*} d\sigma \\ & \leq \sum_{i,j=1}^{N} C_{ij}^{\Phi} \int_{\{|\cos \theta| \in [1-\delta,1]\}} b_{ij}(\cos \theta) | v - v_{*}|^{\gamma} \mu_{j}^{*} | f_{i} | (\overline{w_{k}^{'}} + \overline{w_{k}^{'}}_{j}^{*} + \overline{w_{k}^{*}}_{j}^{*}) dv dv_{*} d\sigma \\ & + \sum_{i,j=1}^{N} C_{ij}^{\Phi} \int_{|v - v_{*}| \leq \delta} b_{ij}(\cos \theta) | v - v_{*}|^{\gamma} \mu_{j}^{*} | f_{i} | (\overline{w_{k}^{'}} + \overline{w_{k}^{'}}_{j}^{*} + \overline{w_{k}^{*}}_{j}^{*}) dv dv_{*} d\sigma \\ & + \sum_{i,j=1}^{N} C_{ij}^{\Phi} \int_{\{|v| \geq \delta^{-1} \text{ or } |v - v_{*}| \geq \delta^{-1}\}} \left[\mu_{j}^{'*} | f_{i}^{'} | + \mu_{i}^{'} | f_{j}^{'*} | + \mu_{i} | f_{j}^{*} | \right] b_{ij}(\cos \theta) | v - v_{*}|^{\gamma} \overline{w_{k}}_{i}. \end{split}$$

Note that we used the change of variables $(v, v_*, \sigma) \to (v', v'^*, v - v_* / |v - v_*|)$ for $\mu_j'^* f_i'$. Then for $\mu_i' f_j'^*$ we used first $(v, v_*, \sigma) \to (v_*, v, -\sigma)$ which sends (v'_{ij}, v'_{ij}) to (v'_{ji}, v'_{ji}) and then relabelling i and j we come back to the first term $\mu_i'^* f_i'$.

Defining the characteristic function χ_A on the set

$$A = \left\{ \sqrt{m_i |v|^2 + m_j |v_*|^2} \ge \min\left\{ \sqrt{m_i}, \sqrt{m_j} \right\} \delta^{-1} \text{ or } |v - v_*| \ge \delta^{-1} \right\}$$

we can bound $b(\cos \theta)$ by its supremum b_{∞} and use the equivalence between ν_i and $1 + |v|^{\gamma}$ to get

$$\begin{aligned} & \left\| \mathbf{B}^{(\delta)}(\mathbf{f}) \right\|_{L_{v}^{1}(\overline{\mathbf{w}_{k}})} \\ & \leq \delta C(k) \left\| \mathbf{f} \right\|_{L_{v}^{1}(\overline{\mathbf{w}_{k}}\nu)} \\ & + \sum_{i,j=1}^{N} C_{ij}^{\Phi} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3} \times \mathbb{S}^{2}} \chi_{A} \left[\mu_{j}^{\prime*} |f_{i}^{\prime}| + \mu_{i}^{\prime} |f_{j}^{\prime*}| + \mu_{i} |f_{j}^{*}| \right] b_{ij}(\cos \theta) |v - v_{*}|^{\gamma} \overline{w_{k_{i}}} \, dv dv_{*} d\sigma \end{aligned}$$

$$(5.5.8)$$

where C(k) will denote any positive constant independent on δ and \mathbf{f} .

We shall deal with the second term on the right-hand side of (5.5.8) thanks to the Povzner inequality. Indeed, the set A is invariant by the changes of variables already mentioned (remember that when changing v to v_* we also change i and j) and therefore

$$\sum_{i,j=1}^{N} C_{ij}^{\Phi} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3} \times \mathbb{S}^{2}} \chi_{A} \left[\mu_{j}^{\prime *} | f_{i}^{\prime} | + \mu_{i}^{\prime} | f_{j}^{\prime *} | + \mu_{i} | f_{j}^{*} | \right] b_{ij}(\cos \theta) | v - v_{*}|^{\gamma} \overline{w_{ki}} \, dv dv_{*} d\sigma$$

$$= \sum_{i,j=1}^{N} C_{ij}^{\Phi} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3} \times \mathbb{S}^{2}} \chi_{A} b_{ij}(\cos \theta) | v - v_{*}|^{\gamma} \mu_{j}^{*} | f_{i} | \left(\overline{w_{ki}^{*}}' + \overline{w_{ki}^{\prime}} + \overline{w_{ki}^{*}} \right) dv dv_{*} d\sigma$$

$$\leq \sum_{i,j=1}^{N} C_{ij}^{\Phi} b_{ij}^{\infty} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3}} \chi_{A} | v - v_{*}|^{\gamma} \mu_{j}^{*} | f_{i} | \left(\int_{\mathbb{S}^{2}} \left[\overline{w_{ki}^{\prime}} + \overline{w_{kj}^{\prime}} - \overline{w_{kj}^{*}} - \overline{w_{ki}} \right] d\sigma \right) dv dv_{*}$$

$$+ 8\pi \sum_{i,j=1}^{N} C_{ij}^{\Phi} b_{ij}^{\infty} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3}} \chi_{A} | v - v_{*}|^{\gamma} \mu_{j}^{*} | f_{i} | \overline{w_{ki}^{*}} \, dv dv_{*}$$

$$+ 4\pi \sum_{i,j=1}^{N} C_{ij}^{\Phi} b_{ij}^{\infty} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3}} \chi_{A} | v - v_{*}|^{\gamma} \mu_{j}^{*} | f_{i} | \overline{w_{ki}^{*}} \, dv dv_{*}$$

$$(5.5.9)$$

We can use Proposition 5.22 for the first term on the right-hand side of the inequality. Indeed,

$$\int_{\mathbb{S}^{2}} \left[\overline{w_{k'j}} + \overline{w_{k'j}}^{*} - \overline{w_{k}}_{j}^{*} - \overline{w_{k}}_{i} \right] d\sigma$$

$$\leq \frac{l_{b_{ij}}}{b_{ij}^{\infty}} C_{k} \left(m_{i} |v|^{2} + m_{j} |v_{*}|^{2} \right)^{k/2} - 4\pi m_{i}^{k/2} |v|^{k} - 4\pi m_{j}^{k/2} |v_{*}|^{k}$$

$$\leq 2^{k/2} \frac{l_{b_{ij}}}{b_{ij}^{\infty}} C_{k} \left[(m_{i} |v|^{2})^{k/2 - 1/2} (m_{j} |v_{*}|^{2})^{1/2} + (m_{i} |v|^{2})^{1/2} (m_{j} |v_{*}|^{2})^{k/2 - 1/2} \right]$$

$$- 4\pi \left(1 - \frac{l_{b_{ij}}}{4\pi b_{ij}^{\infty}} C_{k} \right) \left[m_{i}^{k/2} |v|^{k} + m_{j}^{k/2} |v_{*}|^{k} \right]$$

For $k \geq k_0$ we have that $C_k < 1$, hence $\frac{l_{b_{ij}}}{4\pi b_{ij}^{\infty}}C_k < 1$. We can thus plug this back into

(5.5.9) we find, recalling that $\overline{w_{k}}_{i} = 1 + m_{i}^{k/2} |v|^{k}$

$$\begin{split} \sum_{i,j=1}^{N} C_{ij}^{\Phi} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3} \times \mathbb{S}^{2}} \chi_{A} \left[\mu_{j}^{\prime *} |f_{i}^{\prime}| + \mu_{i}^{\prime} |f_{j}^{\prime *}| + \mu_{i} |f_{j}^{*}| \right] b_{ij}(\cos \theta) |v - v_{*}|^{\gamma} \overline{w_{k_{i}}} \, dv dv_{*} d\sigma \\ & \leq C(k) \sum_{i,j=1}^{N} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3}} \chi_{A} |v - v_{*}|^{\gamma} \mu_{j}^{*} |f_{i}| \left[|v|^{k-1} |v_{*}| + |v| |v_{*}|^{k-1} \right] \, dv dv_{*} \\ & + 12\pi \sum_{i,j=1}^{N} C_{ij}^{\Phi} b_{ij}^{\infty} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3}} \chi_{A} |v - v_{*}|^{\gamma} \mu_{j}^{*} |f_{i}| \, dv dv_{*} + \\ & + 8\pi \sum_{i,j=1}^{N} C_{ij}^{\Phi} b_{ij}^{\infty} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3}} \chi_{A} |v - v_{*}|^{\gamma} \mu_{j}^{*} |f_{i}| m_{j}^{k/2} |v_{*}|^{k} \, dv dv_{*} \\ & + C_{k} \sum_{i,j=1}^{N} C_{ij}^{\Phi} l_{b_{ij}} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3}} \chi_{A} |v - v_{*}|^{\gamma} \mu_{j}^{*} |f_{i}| m_{i}^{k/2} |v|^{k} \, dv dv_{*} \end{split}$$

From here we can use that

$$\chi_A(v, v_*) \le 2 \max_{i,j} \{m_i, m_j\} \delta(m_i |v|^2 + m_j |v^*|^2)$$

and the fact that $\gamma+1 < k_0 \le k$ to bound the first, second and third term on the right-hand side by $\delta C(k) \|\mathbf{f}\|_{L^1_v(\overline{\mathbf{w}_k})}$. And finally, we exactly have the definition of $\nu_i(v)$ in the last term on the right-hand side. This gives

$$\sum_{i,j=1}^{N} C_{ij}^{\Phi} \int_{\mathbb{R}^{3} \times \mathbb{R}^{3} \times \mathbb{S}^{2}} \chi_{A} \left[\mu_{j}^{\prime *} |f_{i}^{\prime}| + \mu_{i}^{\prime} |f_{j}^{\prime *}| + \mu_{i} |f_{j}^{*}| \right] b_{ij}(\cos \theta) |v - v_{*}|^{\gamma} \overline{w_{k}}_{i} \, dv dv_{*} d\sigma$$

$$\leq C_{k} \sum_{i=1}^{N} \|f_{i}\|_{L_{v}^{1}(\overline{w_{k}}_{i}\nu_{i})} + \delta C(k) \|\mathbf{f}\|_{L_{v}^{1}(\overline{\mathbf{w_{k}}})}. \tag{5.5.10}$$

Combining (5.5.8) and (5.5.10) yields the desired estimate.

We conclude this subsection with a control on the nonlinear term.

Lemma 5.23. Define $\widetilde{\mathbf{Q}}(\mathbf{f}, \mathbf{g})$ by

$$\forall 1 \le i \le N, \quad \widetilde{Q}_i(\mathbf{f}, \mathbf{g}) = \frac{1}{2} \sum_{j=1}^{N} (Q_{ij}(f_i, g_j) + Q_{ij}(g_i, f_j)). \tag{5.5.11}$$

Then for all \mathbf{f} , \mathbf{g} such that $\widetilde{\mathbf{Q}}(\mathbf{f},\mathbf{g})$ is well-defined, the latter belongs to $[Ker(L)]^{\perp}$:

$$\pi_{\mathbf{L}}\left(\widetilde{\mathbf{Q}}(\mathbf{f},\mathbf{g})\right) = 0.$$

Moreover, there exists $C_Q > 0$ such that for all i in $\{1, \ldots, N\}$ and every \mathbf{f} and \mathbf{g} ,

$$\begin{aligned} \left\| \widetilde{Q}_{i}(\mathbf{f}, \mathbf{g}) \right\|_{L_{v}^{1} L_{x}^{\infty} \left(\langle v \rangle^{k} \right)} &\leq C_{Q} \left[\left\| f_{i} \right\|_{L_{v}^{1} L_{x}^{\infty} \left(\langle v \rangle^{k} \right)} \left\| \mathbf{g} \right\|_{L_{v}^{1} L_{x}^{\infty} \left(\langle v \rangle^{k} \boldsymbol{\nu} \right)} \\ &+ \left\| f_{i} \right\|_{L_{v}^{1} L_{x}^{\infty} \left(\nu_{i} \langle v \rangle^{k} \right)} \left\| \mathbf{g} \right\|_{L_{v}^{1} L_{x}^{\infty} \left(\langle v \rangle^{k} \right)} \right], \end{aligned}$$

The constant C_Q is explicit and depends only on k, N and the kernel of the collision operator.

Proof of Lemma 5.23. The orthogonality property is well-known for the single-species case [25, Appendix A.2],[21] and follows in the same way as (5.1.3), (5.1.4). The estimate also follows standard computations from the mono-species case, we adapt them to the case of multi-species for the sake of completeness. Since we are dealing with hard potential kernels, we can decompose the bilinear operator $Q_{ij}(f_i, g_j)$, for any i, j in $\{1, \ldots, N\}$, as

$$Q_{ij}(f_i, g_j) = \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} (|v - v_*|, \cos \theta) f_i' g_j'^* dv_* d\sigma$$
$$- \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} (|v - v_*|, \cos \theta) f_i g_j^* dv_* d\sigma.$$

By Minkowski integral inequality we have for all q in $[1, \infty)$,

$$\int_{\mathbb{R}^{3}} \langle v \rangle^{k} \left[\int_{\mathbb{T}^{3}} |Q_{ij}(f_{i}, g_{j})|^{q} dx \right]^{1/q} dv \leq \int_{\mathbb{S}^{2} \times \mathbb{R}^{3} \times \mathbb{R}^{3}} \langle v \rangle^{k} \left[\int_{\mathbb{T}^{3}} \left| B_{ij} f'_{i} g'_{j}^{*} \right|^{q} dx \right]^{1/q} d\sigma dv_{*} dv + \int_{\mathbb{S}^{2} \times \mathbb{R}^{3} \times \mathbb{R}^{3}} \langle v \rangle^{k} \left[\int_{\mathbb{T}^{3}} \left| B_{ij} f_{i} g_{j}^{*} \right|^{q} dx \right]^{1/q} d\sigma dv_{*} dv.$$

Since the function $(v, v_*) \mapsto (v', v'_*)$ is its own inverse and does not change the value of $B_{ij}(|v-v_*|, \cos\theta)$, we make the latter change of variables in the first integral and we obtain

$$\int_{\mathbb{R}^{3}} \langle v \rangle^{k} \left[\int_{\mathbb{T}^{3}} |Q_{ij}(f_{i}, g_{j})|^{q} dx \right]^{1/q} dv$$

$$\leq \int_{\mathbb{S}^{2} \times \mathbb{R}^{3} \times \mathbb{R}^{3}} \left(\langle v \rangle^{k} + \langle v' \rangle^{k} \right) \left[\int_{\mathbb{T}^{3}} |B_{ij} f_{i} g_{j}^{*}|^{q} dx \right]^{1/q} d\sigma dv_{*} dv$$

$$\leq C_{ij} \int_{\mathbb{S}^{2} \times \mathbb{R}^{3} \times \mathbb{R}^{3}} \langle v \rangle^{k} \langle v_{*} \rangle^{k} |v - v_{*}|^{\gamma} \left[\int_{\mathbb{T}^{3}} |f_{i} g_{j}^{*}|^{q} dx \right]^{1/q} d\sigma dv_{*} dv.$$

The constant $C_{ij} > 0$ will stand for any constant depending only on m_i , m_j , the integral over the sphere of b_{ij} and C_{ij}^{Φ} (see assumptions on the kernel B_{ij}). Finally we use the fact that $|v - v_*|^{\gamma} \leq \langle v \rangle^{\gamma} + \langle v^* \rangle^{\gamma}$.

$$\int_{\mathbb{R}^{3}} \langle v \rangle^{k} \left[\int_{\mathbb{T}^{3}} |Q_{ij}(f_{i}, g_{j})|^{q} dx \right]^{1/q} dv$$

$$\leq C_{ij} \int_{\mathbb{S}^{2} \times \mathbb{R}^{3} \times \mathbb{R}^{3}} \left(\langle v \rangle^{k+\gamma} \langle v_{*} \rangle^{k} + \langle v \rangle^{k} \langle v_{*} \rangle^{k+\gamma} \right) \left[\int_{\mathbb{T}^{3}} |f_{i} g_{j}^{*}|^{q} dx \right]^{1/q} d\sigma dv_{*} dv.$$

We take the limit as q tends to infinity and conclude

$$||Q_{ij}(f_{i},g_{j})||_{L_{v}^{1}L_{x}^{\infty}(\langle v\rangle^{k})} \leq C_{ij} \Big[||f_{i}||_{L_{v}^{1}L_{x}^{\infty}(\langle v\rangle^{k})} ||g_{j}||_{L_{v}^{1}L_{x}^{\infty}(\langle v\rangle^{k+\gamma})} + ||f_{i}||_{L_{v}^{1}L_{x}^{\infty}(\langle v\rangle^{k+\gamma})} ||g_{j}||_{L_{v}^{1}L_{x}^{\infty}(\langle v\rangle^{k})} \Big].$$

We remind (5.2.4) which states that $\nu_i(v) \sim \langle v \rangle^{\gamma}$ and the lemma follows after summing over j, C_Q being the maximum of all the C_{ij} .

Study of the equations in $L^1_v L^\infty_x\left(\langle v \rangle^k\right)$

We start with the well-posedness of the system (5.5.2) in $L_v^1 L_x^{\infty} (\langle v \rangle^k)$.

Proposition 5.24. Let $k > k_0$. Let $\mathbf{f_0}$ be in $L_v^1 L_x^{\infty} \left(\langle v \rangle^k \right)$ and \mathbf{g} in $L_v^{\infty} L_v^1 L_x^{\infty} \left(\boldsymbol{\nu} \langle v \rangle^k \right)$. There exist $\eta_1, \lambda_1 > 0$ such that if

$$\|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}\left(\langle v \rangle^k\right)} \leq \eta_1 \quad and \quad \exists C, \ \lambda > 0 \quad \|\mathbf{g}(t)\|_{L_v^1 L_x^{\infty}\left(\boldsymbol{\nu}\langle v \rangle^k\right)} \leq C \|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}\left(\langle v \rangle^k\right)} e^{-\lambda t}$$

then there exists a function $\mathbf{f_1}$ in $L_t^{\infty} L_v^1 L_x^{\infty} \left(\langle v \rangle^k \right)$ such that

$$\partial_t \mathbf{f_1} = \mathbf{G_1}(\mathbf{f_1}) + \mathbf{Q}(\mathbf{f_1} + \mathbf{g})$$
 and $\mathbf{f_1}(0, x, v) = \mathbf{f_0}(x, v)$.

Moreover, any solution f_1 satisfies

$$\forall t \ge 0, \quad \|\mathbf{f_1}(t)\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \le C_1 e^{-\lambda_1 t} \|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}.$$

The constants C_1 , δ_1 , η_1 and λ_1 are independent of $\mathbf{f_0}$ and \mathbf{g} and depends on N, k and the collision kernel.

Proof of Proposition 5.24. We start by showing the exponential decay and then prove existence. As a matter of fact, we saw in Lemma 5.20 that the natural weight to estimate **B** is $\overline{\mathbf{w}_{\mathbf{k}}} = 1 + \mathbf{m}^{k/2} |v|^k$ which is equivalent to $\langle v \rangle^k$. We will therefore rather work in $L_v^1 L_x^{\infty} (\overline{\mathbf{w}_{\mathbf{k}}})$ which just modifies the definition for C_1 , δ_1 and η_1 .

Step 1: a priori exponential decay. Suppose that $\mathbf{f_1}$ is a solution to the differential equation in $L_v^1 L_x^{\infty}(\overline{\mathbf{w_k}})$ with initial data $\mathbf{f_0}$.

We recall that for q in $[1, \infty)$,

$$\|\mathbf{f_1}\|_{L_v^1 L_x^q(\overline{\mathbf{w_k}})} = \sum_{i=1}^N \int_{\mathbb{R}^3} \left(1 + m_i^{k/2} |v|^k \right) \left(\int_{\mathbb{T}^3} |f_{1i}|^q dx \right)^{1/q} dv.$$

Therefore we can compute for all i in $\{1, \ldots, N\}$

$$\frac{d}{dt} \|f_{1i}\|_{L_v^1 L_x^q \left(1 + m_i^{k/2} |v|^k\right)}
= \int_{\mathbb{R}^3} \left(1 + m_i^{k/2} |v|^k\right) \|f_{1i}\|_{L_x^q}^{1-q} \left(\int_{\mathbb{T}^3} \operatorname{sgn}(f_{i1}) |f_{1i}|^{q-1} \partial_t f_{1i} dx\right) dv.$$

Observing that

$$\partial_t f_{1i} = -v \cdot \nabla_x f_{1i} - \nu_i(v) f_{1i} + B_i(\mathbf{f_1}) + Q_i(\mathbf{f_1} + \mathbf{g}),$$

that the transport gives null contribution

$$\int_{\mathbb{T}^3} \operatorname{sgn}(f_{1i}) |f_{1i}|^{q-1} v \cdot \nabla_x f_{1i} \, dx = \frac{1}{q} v \cdot \int_{\mathbb{T}^3} \nabla_x (|f_{1i}|^q) \, dx = 0,$$

that the multiplicative part gives a negative contribution,

$$-\int_{\mathbb{T}^3} \nu_i(v) f_{1i} \operatorname{sgn}(f_{1i}) |f_{1i}|^{q-1} dx \le -\nu_i(v) \|f_{1i}\|_{L_x^q}^q$$

and that by Hölder inequality with q and q/(q-1),

$$\left| \int_{\mathbb{T}^3} \operatorname{sgn}(f_{1i}) |f_{1i}|^{q-1} g_i dx \right| \le \|f_{1i}\|_{L_x^q}^{q-1} \|g_i\|_{L_x^q}, \tag{5.5.12}$$

we deduce

$$\frac{d}{dt} \|f_{i1}\|_{L_{v}^{1}L_{x}^{q}\left(1+m_{i}^{k/2}|v|^{k}\right)} \leq -\|f_{1i}\|_{L_{v}^{1}L_{x}^{q}\left(\nu_{i}\left(1+m_{i}^{k/2}|v|^{k}\right)\right)} + \|B_{i}\left(\mathbf{f_{1}}\right)\|_{L_{v}^{1}L_{x}^{q}\left(1+m_{i}^{k/2}|v|^{k}\right)} + \|Q_{i}\left(\mathbf{f_{1}}+\mathbf{g}\right)\|_{L_{v}^{1}L_{x}^{q}\left(1+m_{i}^{k/2}|v|^{k}\right)}.$$

First sum over i in $\{1, ..., N\}$ and then let q tend to infinity (on the torus the limit is thus the L^{∞} -norm). This yields for all $t \geq 0$.

$$\frac{d}{dt} \|\mathbf{f_1}\|_{L_v^1 L_x^{\infty}(\overline{\mathbf{w_k}})} \le -\|\mathbf{f_1}\|_{L_v^1 L_x^{\infty}(\nu \overline{\mathbf{w_k}})} + \|\mathbf{B}(\mathbf{f_1})\|_{L_v^1 L_x^{\infty}(\overline{\mathbf{w_k}})}
+ \|\mathbf{Q}(\mathbf{f_1} + \mathbf{g})\|_{L_v^1 L_x^{\infty}(\overline{\mathbf{w_k}})}.$$
(5.5.13)

We use Lemma 5.20 to control **B**, recalling that $0 < C_B < 1$, and the control of **Q** given in Lemma 5.23 for **Q** (of course, since $\overline{\mathbf{w_k}} \sim \langle v \rangle^k$ the Lemma still holds with a different C_Q). We get that for all $t \geq 0$,

$$\frac{d}{dt} \|\mathbf{f_1}\|_{L_v^1 L_x^{\infty}(\overline{\mathbf{w_k}})} \le -\left[1 - C_B - 2C_Q\left(\|\mathbf{f_1}\|_{L_v^1 L_x^{\infty}(\overline{\mathbf{w_k}})} + 2\|\mathbf{g}\|_{L_v^1 L_x^{\infty}(\overline{\mathbf{w_k}})}\right)\right] \|\mathbf{f_1}\|_{L_v^1 L_x^{\infty}(\boldsymbol{\nu}\overline{\mathbf{w_k}})} + C_Q \|\mathbf{g}(t)\|_{L_v^1 L_x^{\infty}(\boldsymbol{\nu}\overline{\mathbf{w_k}})}^2.$$

Since $C_B < 1$, if $\|\mathbf{f_1}(0)\|_{L_v^1 L_x^{\infty}(\overline{\mathbf{w_k}})}$ is sufficiently small and thanks to the exponential decay of $\|\mathbf{g}(t)\|_{L_v^{\infty} L_v^1 L_x^{\infty}(\boldsymbol{\nu}\overline{\mathbf{w_k}})}$, a direct application of Grönwall lemma yields the desired exponential decay.

Step 2: existence. Let $f^{(0)} = 0$ and consider the following iterative scheme

$$\partial_t \mathbf{f}^{(\mathbf{n}+\mathbf{1})} + v \cdot \nabla_x \mathbf{f}^{(\mathbf{n}+\mathbf{1})} = -\boldsymbol{\nu}(v) \left(\mathbf{f}^{(\mathbf{n}+\mathbf{1})} \right) + \mathbf{B} \left(\mathbf{f}^{(\mathbf{n})} \right) + \widetilde{\mathbf{Q}} \left(\mathbf{f}^{(\mathbf{n})} + \mathbf{g} \right)$$

with the initial data $\mathbf{f}^{(\mathbf{n+1})}(0, x, v) = \mathbf{f_0}$.

For each n in \mathbb{N} , $\mathbf{f^{(n+1)}}$ is well-defined by induction since we have the explicit Duhamel formula along the characteristics for all i in $\{1, \ldots, N\}$

$$f_i^{(n+1)}(t, x, v) = e^{-\nu_i(v)t} f_{0i} + \int_0^t e^{-\nu_i(v)(t-s)} \left[B_i \left(\mathbf{f}^{(\mathbf{n})} \right) + Q_i \left(\mathbf{f}^{(\mathbf{n})} + \mathbf{g} \right) \right] (x - sv, v) \, ds.$$

We are about to show that $(\mathbf{f^{(n)}})_{n\in\mathbb{N}}$ is a Cauchy sequence in $L_t^{\infty}L_x^1L_x^{\infty}(\overline{\mathbf{w_k}})$.

Direct computations on the nonlinear operator gives

$$\begin{split} \partial_t \left(\mathbf{f^{(n+1)}} - \mathbf{f^{(n)}} \right) &= -\boldsymbol{\nu}(v) \left(\mathbf{f^{(n+1)}} - \mathbf{f^{(n)}} \right) + \mathbf{B} \left(\mathbf{f^{(n)}} - \mathbf{f^{(n-1)}} \right) \\ &+ \widetilde{\mathbf{Q}} \left(\mathbf{f^{(n)}} - \mathbf{f^{(n-1)}}, \mathbf{f^{(n-1)}} + \mathbf{g} \right) + \widetilde{\mathbf{Q}} \left(\mathbf{f^{(n)}} + \mathbf{g}, \mathbf{f^{(n)}} - \mathbf{f^{(n-1)}} \right), \end{split}$$

where we remind that $\widetilde{\mathbf{Q}}$ was defined by (5.5.11) and that $\widetilde{\mathbf{Q}}(\mathbf{a}, \mathbf{a}) - \widetilde{\mathbf{Q}}(\mathbf{b}, \mathbf{b}) = \widetilde{\mathbf{Q}}(\mathbf{a} - \mathbf{b}, \mathbf{b}) + \widetilde{\mathbf{Q}}(\mathbf{a}, \mathbf{a} - \mathbf{b})$.

Taking the $L_v^1 L_x^{\infty}(\overline{\mathbf{w_k}})$ -norm of $(\mathbf{f^{(n+1)}} - \mathbf{f^{(n)}})$ and summing over i from 1 to N gives for all $t \ge 0$

$$\begin{split} \left\| \mathbf{f}^{(\mathbf{n}+\mathbf{1})}(t) - \mathbf{f}^{(\mathbf{n})}(t) \right\|_{L_{v}^{1} L_{x}^{\infty}(\overline{\mathbf{w}_{k}})} \\ &\leq \sum_{i=1}^{N} \int_{0}^{t} ds \int_{\mathbb{R}^{3}} dv \, e^{-\nu_{i}(v)(t-s)} \left(1 + m_{i}^{k/2} |v|^{k} \right) \left\| \Delta_{ni} \left(\mathbf{f}^{(\mathbf{n})} - \mathbf{f}^{(\mathbf{n}-\mathbf{1})} \right) \right\|_{L_{x}^{\infty}}. \end{split}$$

where we defined

$$\begin{split} & \boldsymbol{\Delta}_n \left(\mathbf{f^{(n)}} - \mathbf{f^{(n-1)}} \right) \\ & = \mathbf{B} \left(\mathbf{f^{(n)}} - \mathbf{f^{(n-1)}} \right) + \widetilde{\mathbf{Q}} \left(\mathbf{f^{(n)}} - \mathbf{f^{(n-1)}}, \mathbf{f^{(n-1)}} + \mathbf{g} \right) + \widetilde{\mathbf{Q}} \left(\mathbf{f^{(n)}} + \mathbf{g}, \mathbf{f^{(n)}} - \mathbf{f^{(n-1)}} \right). \end{split}$$

As $\nu_i(v) \geq \nu_0$ for all i and v we further get

$$\begin{split} \left\| \mathbf{f}^{(\mathbf{n}+\mathbf{1})}(t) - \mathbf{f}^{(\mathbf{n})}(t) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{k}})} &\leq \int_{0}^{t} e^{-\nu_{0}(t-s)} \left\| \boldsymbol{\Delta}_{n} \left(\mathbf{f}^{(\mathbf{n})} - \mathbf{f}^{(\mathbf{n}-1)} \right) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{k}})} ds \\ &\leq \left[C_{B} + C_{Q} \left(\left\| \mathbf{f}^{(\mathbf{n})} \right\|_{L_{v}^{\infty}L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{k}})} + \left\| \mathbf{f}^{(\mathbf{n}-1)} \right\|_{L_{v}^{\infty}L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{k}})} + 2 \left\| \mathbf{g} \right\|_{L_{v}^{\infty}L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{k}})} \right) \right] \\ &\times \int_{0}^{t} e^{-\nu_{0}(t-s)} \left\| \mathbf{f}^{(\mathbf{n})}(s) - \mathbf{f}^{(\mathbf{n}-1)}(s) \right\|_{L_{v}^{1}L_{x}^{\infty}(\nu\overline{\mathbf{w}_{k}})} ds \\ &+ C_{Q} \left[\int_{0}^{t} e^{-\nu_{0}(t-s)} \left(\left\| \mathbf{f}^{(\mathbf{n})} \right\|_{L_{v}^{1}L_{x}^{\infty}(\nu\overline{\mathbf{w}_{k}})} + \left\| \mathbf{f}^{(\mathbf{n}-1)} \right\|_{L_{v}^{1}L_{x}^{\infty}(\nu\overline{\mathbf{w}_{k}})} \right) ds \right] \\ &\times \sup_{s \in [0,t]} \left\| \mathbf{f}^{(\mathbf{n})}(s) - \mathbf{f}^{(\mathbf{n}-1)}(s) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{k}})} . \end{split}$$

$$(5.5.14)$$

where, as above, we used Lemma 5.20 and the estimate of Lemma 5.23.

Let us look at the terms inside the time integrals. Thus, we take the $L_t^1 L_v^1 L_x^{\infty} (\nu \overline{\mathbf{w_k}})$ norm of $(\mathbf{f^{(n+1)}} - \mathbf{f^{(n)}})$ and sum over i.

$$\int_{0}^{t} \left\| \mathbf{f}^{(\mathbf{n}+\mathbf{1})}(s) - \mathbf{f}^{(\mathbf{n})}(s) \right\|_{L_{v}^{1} L_{x}^{\infty}(\langle v \rangle^{k} \boldsymbol{\nu})} ds$$

$$\leq \sum_{i=1}^{N} \int_{0}^{t} \int_{0}^{s} \int_{\mathbb{R}^{3}} e^{-\nu_{i}(v)(s-s_{1})} \nu_{i}(v) \overline{w}_{ki}(v) \left\| \boldsymbol{\Delta}_{n} \left(\mathbf{f}^{(\mathbf{n})} - \mathbf{f}^{(\mathbf{n}-\mathbf{1})} \right) \right\|_{L_{x}^{\infty}} (s_{1}) ds_{1} ds.$$

We exchange the integration domains in s and s_1 , which implies

$$\int_{0}^{t} \left\| \mathbf{f}^{(\mathbf{n}+\mathbf{1})}(s) - \mathbf{f}^{(\mathbf{n})}(s) \right\|_{L_{v}^{1} L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}} \boldsymbol{\nu})} ds$$

$$\leq \sum_{i=1}^{N} \int_{0}^{t} \int_{\mathbb{R}^{3}} \left(\int_{s_{1}}^{t} e^{-\nu_{i}(v)(s-s_{1})} \nu_{i}(v) ds \right) \overline{w}_{ki}(v) \left\| \boldsymbol{\Delta}_{n} \left(\mathbf{f}^{(\mathbf{n})} - \mathbf{f}^{(\mathbf{n}-\mathbf{1})} \right) \right\|_{L_{x}^{\infty}} (s_{1}) ds_{1}.$$

Since the integral in s is bounded by 1, we use Lemma 5.20 and Lemma 5.23 again and obtain

$$\int_{0}^{t} \left\| \mathbf{f}^{(\mathbf{n}+\mathbf{1})}(s) - \mathbf{f}^{(\mathbf{n})}(s) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}}\nu)} ds$$

$$\leq \left[C_{B} + C_{Q} \left(\left\| \mathbf{f}^{(\mathbf{n})} \right\|_{L_{v}^{\infty}L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}})} + \left\| \mathbf{f}^{(\mathbf{n}-\mathbf{1})} \right\|_{L_{v}^{\infty}L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}})} + 2 \left\| \mathbf{g} \right\|_{L_{v}^{\infty}L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}})} \right) \right]$$

$$\times \int_{0}^{t} \left\| \mathbf{f}^{(\mathbf{n})}(s_{1}) - \mathbf{f}^{(\mathbf{n}-\mathbf{1})}(s_{1}) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}}\nu)} ds_{1}$$

$$+ C_{Q} \left[\int_{0}^{t} \left(\left\| \mathbf{f}^{(\mathbf{n})} \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}}\nu)} + \left\| \mathbf{f}^{(\mathbf{n}-\mathbf{1})} \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}}\nu)} \right) ds_{1} \right]$$

$$\times \sup_{s \in [0,t]} \left\| \mathbf{f}^{(\mathbf{n})}(s) - \mathbf{f}^{(\mathbf{n}-\mathbf{1})}(s) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}})}$$

$$(5.5.15)$$

We now conclude the proof of existence. Indeed, exact same computations but subtracting $e^{-\nu(v)t}\mathbf{f_0}$ instead of $\mathbf{f^{(n)}}$ lead to (5.5.14) and (5.5.15) with $\mathbf{f^{(n-1)}}$ replaced by 0. Therefore, since $C_B < 1$ it follows that for $\|\mathbf{f_0}\|_{L_v^1 L_x^\infty(\overline{\mathbf{w_k}})}$ and $\|\mathbf{g}\|_{L_v^\infty L_v^1 L_x^\infty(\overline{\mathbf{w_k}}\nu)}$ sufficiently small we have that there exists C > 0 such that for all n in \mathbb{N} and all $t \ge 0$,

$$\left\|\mathbf{f^{(n)}}(t)\right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w_{k}}})} \leq C \left\|\mathbf{f_{0}}\right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w_{k}}})}$$

and

$$\int_0^t \left\| \mathbf{f^{(n)}}(s) \right\|_{L^1_v L^\infty_x(\overline{\mathbf{w_k}}\boldsymbol{\nu})} \ ds \leq C \int_0^t \left\| \mathbf{f^{(1)}} \right\|_{L^1_v L^\infty_x(\overline{\mathbf{w_k}}\boldsymbol{\nu})} \ ds \leq C \left\| \mathbf{f_0} \right\|_{L^1_v L^\infty_x(\overline{\mathbf{w_k}})}.$$

Therefore, denoting by C any positive constant independent of $\mathbf{f^{(n)}}$ and \mathbf{g} , adding (5.5.14) and (5.5.15) yields

$$\begin{aligned} & \left\| \mathbf{f}^{(\mathbf{n}+\mathbf{1})}(t) - \mathbf{f}^{(\mathbf{n})}(t) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}})} + \int_{0}^{t} \left\| \mathbf{f}^{(\mathbf{n}+\mathbf{1})}(s) - \mathbf{f}^{(\mathbf{n})}(s) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}}\boldsymbol{\nu})} ds \\ & \leq C\eta_{1} \sup_{s \in [0,t]} \left\| \mathbf{f}^{(\mathbf{n})}(s) - \mathbf{f}^{(\mathbf{n}-\mathbf{1})}(s) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}})} \\ & + \left[C_{B} + C\eta_{1} \right] \int_{0}^{t} \left\| \mathbf{f}^{(\mathbf{n})}(s) - \mathbf{f}^{(\mathbf{n}-\mathbf{1})}(s) \right\|_{L_{v}^{1}L_{x}^{\infty}(\overline{\mathbf{w}_{\mathbf{k}}}\boldsymbol{\nu})} ds. \end{aligned}$$

Since $C_B < 1$, choosing η_1 such that $C_B + C\eta_1 < 1$ implies that $(\mathbf{f}^{(n)})_{n \in \mathbb{N}}$ is a Cauchy sequence in $L^{\infty}_t L^1_v L^{\infty}_x(\overline{\mathbf{w_k}})$. Hence, $(\mathbf{f}^{(n)})_{n \in \mathbb{N}}$ converges to a function $\mathbf{f_1}$ in $L^{\infty}_t L^1_v L^{\infty}_x(\overline{\mathbf{w_k}})$ and since $k > k_0 > \gamma$ we can take the limit inside the iterative scheme and $\mathbf{f_1}$ is thus a solution of our differential equation.

Study of the equations in $L^{\infty}_{x,v}\left(\langle v\rangle^{\beta}\mu^{-1/2}\right)$

We turn to the system (5.5.3) in $L_{x,v}^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)$ with $\beta>3/2$ so that Theorem 5.17 holds.

Proposition 5.25. Let $k > k_0$, $\beta > 3/2$ and let assumptions (H1) - (H4) hold for the collision kernel. Let $\mathbf{g} = \mathbf{g}(t, x, v)$ be in $L_t^{\infty} L_v^1 L_x^{\infty} (\langle v \rangle^k)$. Then there exists a unique function $\mathbf{f_2}$ in $L_t^{\infty} L_{x,v}^{\infty} (\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2})$ such that

$$\partial_t \mathbf{f_2} = \mathbf{G}(\mathbf{f_2}) + \mathbf{A}(\mathbf{g})$$
 and $\mathbf{f_2}(0, x, v) = 0$.

Moreover, if $\Pi_{\mathbf{G}}(\mathbf{f_2} + \mathbf{g}) = 0$ and if

$$\exists \lambda_g, \, \eta_g > 0, \, \forall t \ge 0, \, \|\mathbf{g}(t)\|_{L^1_v L^\infty_\omega(\langle v \rangle^k)} \le \eta_g e^{-\lambda_g t},$$

then for any $0 < \lambda_2 < \min \{\lambda_g, \lambda_\infty\}$, with λ_∞ defined in Theorem 5.17, there exist $C_2 > 0$ such that

$$\forall t \geq 0, \quad \|\mathbf{f_2}(t)\|_{L^{\infty}_{x,v}(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2})} \leq C_2 \eta_g e^{-\lambda_2 t}.$$

The constant C_2 only depends on λ_2 .

Proof of Proposition 5.25. Thanks to the regularising property of **A**, Lemma 5.19, **A**(g) belongs to $L_t^{\infty} L_{x,v}^{\infty} \left(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2} \right)$. Theorem 5.17 implies that there is indeed a unique $\mathbf{f_2}$ solution to the differential system, given by

$$\mathbf{f_2} = \int_0^t S_{\mathbf{G}}(t-s) \left[\mathbf{A} \left(\mathbf{g} \right) \left(s \right) \right] ds,$$

where $S_{\mathbf{G}}(t)$ is the semigroup generated by \mathbf{G} in $L_{x,v}^{\infty}(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2})$.

Suppose that $\Pi_{\mathbf{G}}(\mathbf{f_2} + \mathbf{g}) = 0$ and that $\exists \eta_2 > 0$ such that $\|\mathbf{g}(t)\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \leq \eta_2 e^{-\lambda t}$.

Using the definition of $\Pi_{\mathbf{G}}$ (5.3.1), the projection part of $\mathbf{f_2}$ is straightforwardly bounded for all $t \geq 0$:

$$\|\Pi_{\mathbf{G}}\left(\mathbf{f_{2}}\right)\left(t\right)\|_{L_{x,v}^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)} = \|\Pi_{\mathbf{G}}\left(\mathbf{g}\right)\left(t\right)\|_{L_{x,v}^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)} \leq C_{\Pi_{G}} \|\mathbf{g}\|_{L_{v}^{1}L_{x}^{\infty}\left(\langle v\rangle^{k}\right)}$$

$$\leq C_{\Pi_{G}}\eta_{g} e^{-\lambda_{g}t}.$$

$$(5.5.16)$$

Applying $\Pi_{\mathbf{G}}^{\perp} = \mathrm{Id} - \Pi_{\mathbf{G}}$ to the equation satisfied by $\mathbf{f_2}$ we get, thanks to the fact the definition of $\Pi_{\mathbf{G}}$ (5.3.1) which is independent of t,

$$\partial_t \left[\Pi_{\mathbf{G}}^{\perp} \left(\mathbf{f_2} \right) \right] = \mathbf{G} \left[\Pi_{\mathbf{G}}^{\perp} \left(\mathbf{f_2} \right) \right] + \Pi_{\mathbf{G}}^{\perp} \left(\mathbf{A} \left(\mathbf{g} \right) \right).$$

This yields

$$\Pi_{\mathbf{G}}^{\perp}\left(\mathbf{f_{2}}\right) = \int_{0}^{t} S_{\mathbf{G}}(t-s) \left[\Pi_{\mathbf{G}}^{\perp}\left(\mathbf{A}\left(\mathbf{g}\right)\right)(s) \right] ds.$$

We now use the exponential decay of $S_{\mathbf{G}}(t)$ on $(\mathrm{Ker}(\mathbf{G}))^{\perp}$, see Theorem 5.17.

$$\left\|\Pi_{\mathbf{G}}^{\perp}\left(\mathbf{f_{2}}\right)\right\|_{L_{x,v}^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)} \leq C_{\infty} \int_{0}^{t} e^{-\lambda_{\infty}(t-s)} \left\|\Pi_{\mathbf{G}}^{\perp}\left(\mathbf{A}\left(\mathbf{g}\right)\right)\left(s\right)\right\|_{L_{x,v}^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)} \ ds.$$

Using the definition of $\Pi_{\mathbf{G}}$ (5.3.1) and then the regularising property of **A** Lemma 5.19 we further bound, for a fixed $\lambda_2 < \min \{\lambda_{\infty}, \lambda_g\}$,

$$\|\Pi_{\mathbf{G}}^{\perp}(\mathbf{f_{2}})\|_{L_{x,v}^{\infty}(\langle v \rangle^{\beta}\mu^{-1/2})} \leq C_{\infty}C_{\Pi_{G}}C_{A}C_{g}\eta_{g} \int_{0}^{t} e^{-\lambda_{\infty}(t-s)}e^{-\lambda_{g}s} ds$$

$$\leq C_{G}C_{\infty}C_{\Pi_{G}}C_{A}C_{g}\eta_{g} te^{-\min\{\lambda_{g},\lambda_{\infty}\}t}$$

$$\leq C_{2}(\lambda_{2})\eta_{g}e^{-\lambda_{2}t}. \tag{5.5.17}$$

Gathering (5.5.16) and (5.5.17) yields the desired exponential decay.

Proof of Proposition 5.18

Take $\mathbf{f_0}$ in $L_v^1 L_x^{\infty} (\langle v \rangle^k)$ such that $\Pi_{\mathbf{G}}(\mathbf{f_0}) = 0$.

The existence will be proved by an iterative scheme. We start with $\mathbf{f_1^{(0)}} = \mathbf{f_2^{(0)}} = 0$ and we approximate the system of equation (5.5.2) - (5.5.3) as follows.

$$\begin{array}{lcl} \partial_t \mathbf{f_1^{(n+1)}} & = & \mathbf{G_1} \left(\mathbf{f_1^{(n+1)}} \right) + \mathbf{Q} \left(\mathbf{f_1^{(n+1)}} + \mathbf{f_2^{(n)}} \right) \\ \partial_t \mathbf{f_2^{(n+1)}} & = & \mathbf{G} \left(\mathbf{f_2^{(n+1)}} \right) + \mathbf{A}^{(\delta)} \left(\mathbf{f_1^{(n+1)}} \right), \end{array}$$

with the following initial data

$$\mathbf{f_1^{(n+1)}}(0, x, v) = \mathbf{f_0}(x, v)$$
 and $\mathbf{f_2^{(n+1)}}(0, x, v) = 0$.

Assume that $(1 + C_1C_2) \|\mathbf{f_0}\| \leq \eta_1$, where C_1 , η_1 were defined in Proposition 5.24 and C_2 was defined in Proposition 5.25. Thanks to Proposition 5.24 and Proposition 5.25, an induction proves first that $\left(\mathbf{f_1^{(n)}}\right)_{n\in\mathbb{N}}$ and $\left(\mathbf{f_2^{(n)}}\right)_{n\in\mathbb{N}}$ are well-defined sequences and second that for all n in \mathbb{N} and all $t\geq 0$

$$\left\|\mathbf{f}_{1}^{(\mathbf{n})}(t)\right\|_{L_{v}^{1}L_{\infty}^{\infty}(\langle v\rangle^{k})} \leq e^{-\lambda_{1}t} \left\|\mathbf{f}_{0}\right\|_{L_{v}^{1}L_{x}^{\infty}(\langle v\rangle^{k})}$$

$$(5.5.18)$$

$$\left\| \mathbf{f_{1}^{(n)}}(t) \right\|_{L_{v}^{1} L_{x}^{\infty}(\langle v \rangle^{k})} \leq e^{-\lambda_{1} t} \left\| \mathbf{f_{0}} \right\|_{L_{v}^{1} L_{x}^{\infty}(\langle v \rangle^{k})}$$

$$\left\| \mathbf{f_{2}^{(n)}}(t) \right\|_{L_{x,v}^{\infty}(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2})} \leq C_{1} C_{2} e^{-\lambda_{2} t} \left\| \mathbf{f_{0}} \right\|_{L_{v}^{1} L_{x}^{\infty}(\langle v \rangle^{k})},$$

$$(5.5.18)$$

with $\lambda_2 < \min \{\lambda_1, \lambda_\infty\}$. Indeed, if we constructed $\mathbf{f_1^{(n)}}$ and $\mathbf{f_2^{(n)}}$ satisfying the exponential decay above then we can construct $\mathbf{f_1^{(n+1)}}$ with Proposition 5.24 and $\mathbf{g} = \mathbf{f_2^{(n)}}$, which has the required exponential decay (5.5.18), and then construct $\mathbf{f_2^{(n+1)}}$ with Proposition 5.25 and $\mathbf{g} = \mathbf{f_1^{(n+1)}}$. Finally, we have the following equality

$$\partial_t \left(f_1^{(n+1)} + f_2^{(n+1)} \right) = G \left(f_1^{(n+1)} + f_2^{(n+1)} \right) + Q \left(f_1^{(n+1)} + f_2^{(n)} \right).$$

Thanks to orthogonality property of \mathbf{Q} in Lemma 5.23 and the definition of $\Pi_{\mathbf{G}}$ (5.3.1) we obtain that the projection is constant with time and thus

$$\Pi_{\mathbf{G}}\left(\mathbf{f}_{1}^{(n+1)} + \mathbf{f}_{2}^{(n+1)}\right) = \Pi_{\mathbf{G}}(\mathbf{f}_{0}) = 0.$$

Applying Proposition 5.25 we obtain the exponential decay (5.5.19) for $\mathbf{f_2^{(n+1)}}$.

We recognize exactly the same iterative scheme for $\mathbf{f_1^{n+1}}$ as in the proof of Proposition 5.24 with \mathbf{g} replaced by $\mathbf{f_2^{(n)}}$. Moreover, the uniform bound (5.5.19) allows us to derive the same estimates as in the latter proof independently of $\mathbf{f_2^{(n)}}$. As a conclusion, $\left(\mathbf{f_1^{(n)}}\right)_{n\in\mathbb{N}}$ is a Cauchy sequence in $L_t^{\infty} L_v^1 L_x^{\infty} (\langle v \rangle^k)$ and therefore converges strongly towards a function

By (5.5.19), the sequence $\left(\mathbf{f_2^{(n)}}\right)_{n\in\mathbb{N}}$ is bounded in $L_t^{\infty}L_{x,v}^{\infty}\left(\langle v\rangle^{\beta}\boldsymbol{\mu}^{-1/2}\right)$ and is therefore weakly-* compact and therefore converges, up to a subsequence, weakly-* towards $\mathbf{f_2}$ in $L_t^{\infty} L_{x,v}^{\infty} \left(\langle v \rangle^{\beta} \boldsymbol{\mu}^{-1/2} \right).$

Since the function inside the collision operator behaves like $|v-v_*|^{\gamma}$ and that in our weighted spaces $k > k_0 > \gamma$, we can take the weak limit inside the iterative scheme. This implies that $(\mathbf{f_1}, \mathbf{f_2})$ is solution to the system (5.5.2) - (5.5.3) and thus $\mathbf{f} = \mathbf{f_1} + \mathbf{f_2}$ is solution to the perturbed multi-species equation (5.5.1). Moreover, taking the limit inside the exponential decays (5.5.18) and (5.5.19) yields the expected exponential decay for f.

5.5.2. Uniqueness of solutions in the perturbative regime

As said in Remark 2.2.5, we are solely interested in the uniqueness of solutions to the multispecies Boltzmann equation (5.1.1) in the perturbative setting. In other terms, uniqueness of solutions of the form $\mathbf{F} = \mu + \mathbf{f}$ as long as $\mathbf{F_0}$ is close enough to the global equilibrium μ . This is equivalent to proving the uniqueness of solutions to the perturbed multi-species equation

$$\partial_t \mathbf{f} = \mathbf{G}(\mathbf{f}) + \mathbf{Q}(\mathbf{f}) \tag{5.5.20}$$

for f_0 small.

Proposition 5.26. Let $k > k_0$ and let assumptions (H1) - (H4) hold for the collision kernel. There exists $\eta_k > 0$ such that for any $\mathbf{f_0}$ in $L_v^1 L_x^\infty \left(\langle v \rangle^k \right)$; if $\|\mathbf{f_0}\|_{L_v^1 L_x^\infty \left(\langle v \rangle^k \right)} \leq \eta_k$ then there exists at most one solution to the perturbed multi-species equation (5.5.20). The constant η_k only depends on k, N and the collision kernels.

The uniqueness will follow from the study of the semigroup generated by G in a dissipative norm as well as a new *a priori* stability estimate for solutions to (5.5.20) in the latter norm. They are the purpose of the next two lemmas.

Lemma 5.27. Let $k > k_0$ and let assumptions (H1) - (H4) hold for the collision kernel. The operator \mathbf{G} generates a semigroup in $L_v^1 L_x^{\infty} (\langle v \rangle^k)$. Moreover, there exist C_k , $\lambda_k > 0$ such that for all $\mathbf{f_0}$ in $L_v^1 L_x^{\infty} (\langle v \rangle^k)$ with $\Pi_{\mathbf{G}}(\mathbf{f_0}) = 0$

$$\forall t \ge 0, \quad \|S_{\mathbf{G}}(\mathbf{f})\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \le C_k e^{-\lambda_k t} \|\mathbf{f}_{\mathbf{0}}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}.$$

Proof of Lemma 5.27. From Proposition 5.18 with a collision operator $\mathbf{Q} = 0$ we have the existence of a solution to the equation

$$\partial_t \mathbf{f} = \mathbf{G}(\mathbf{f})$$

with initial data $\mathbf{f_0}$ in $L_v^1 L_x^{\infty} (\langle v \rangle^k)$. Moreover, that solution satisfies $\Pi_{\mathbf{G}}(\mathbf{f}) = 0$ and it decays exponentially fast with rate λ_k .

Let g be another solution to the linear equation then

$$\partial_t (\mathbf{f} - \mathbf{g}) = [-v \cdot \nabla_x - \boldsymbol{\nu} + \mathbf{B} + \mathbf{A}] (\mathbf{f} - \mathbf{g}).$$

Similar computations as to obtain (5.5.13) yield

$$\frac{d}{dt} \|\mathbf{f} - \mathbf{g}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \le -\|\mathbf{f} - \mathbf{g}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k \boldsymbol{\nu})} + \|\mathbf{A}(\mathbf{f} - \mathbf{g})\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} + \|\mathbf{B}(\mathbf{f} - \mathbf{g})\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}.$$

Using Lemma 5.19 and Lemma 5.20, there exists $0 < C_B < 1$ such that

$$\frac{d}{dt} \|\mathbf{f} - \mathbf{g}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \le -(1 - C_B) \|\mathbf{f} - \mathbf{g}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k \boldsymbol{\nu})} + C_A \|\mathbf{f} - \mathbf{g}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}. \quad (5.5.21)$$

Since $(1 - C_B) > 0$ we can further bound

$$\frac{d}{dt} \|\mathbf{f} - \mathbf{g}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \le [C_A - (1 - C_B)] \|\mathbf{f} - \mathbf{g}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}$$

and a Grönwall lemma therefore yields $\mathbf{f} = \mathbf{g}$ if $\mathbf{g_0} = \mathbf{f_0}$.

We thus obtain existence and uniqueness of solution to the linear equation which means that \mathbf{G} generates a semigroup in $L_v^1 L_x^\infty \left(\langle v \rangle^k \right)$. Moreover it has an exponential decay of rate $\lambda_k > 0$ for functions in $(\operatorname{Ker}(\mathbf{G}))^{\perp}$.

We now derive a stability estimate in an equivalent norm that catches the dissipativity of the linear operator.

Lemma 5.28. Let $k > k_0$ and let assumptions (H1) - (H4) hold for the collision kernel. For $\alpha > 0$, we define

$$\|\mathbf{f}\|_{\alpha,k} = \alpha \|\mathbf{f}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} + \int_0^{+\infty} \|S_{\mathbf{G}}(s)(\mathbf{f})\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} ds.$$

There exist η , α , C_1 , C_2 and $\lambda > 0$ such that $\|\cdot\|_{\alpha,k} \sim \|\cdot\|_{L^1_v L^\infty_x(\langle v \rangle^k)}$ and for all $\mathbf{f_0}$ in $L^1_v L^\infty_x(\langle v \rangle^k)$ with $\Pi_{\mathbf{G}}(\mathbf{f_0}) = 0$ and such that

$$\|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \le \eta;$$

if \mathbf{f} in $L_v^1 L_x^\infty \left(\langle v \rangle^k \right)$ with $\Pi_{\mathbf{G}}(\mathbf{f}) = 0$ is solution to the perturbed equation (5.5.20) with initial data $\mathbf{f_0}$ then

$$\frac{d}{dt} \|\mathbf{f}\|_{\alpha,k} \le -\left(C_1 - C_2 \|\mathbf{f}\|_{\alpha,k}\right) \|\mathbf{f}\|_{\alpha,k,\nu},$$

where the subscript ν refers to the fact that the weight is multiplied by $\nu_i(v)$ on each coordinate.

Proof of Lemma 5.28. Start with the new norm. Lemma 5.27 proved that for all $\mathbf{f_0}$ such that $\Pi_{\mathbf{G}}(\mathbf{f_0}) = 0$ and all $s \geq 0$,

$$\|S_{\mathbf{G}}(s)\left(\mathbf{f_0}\right)\|_{L_v^1 L_x^{\infty}\left(\langle v \rangle^k\right)} \le C_k e^{-\lambda_k s} \|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}\left(\langle v \rangle^k\right)}$$

and hence

$$\alpha \|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \leq \|\mathbf{f_0}\|_{\alpha,k} \leq \left(\alpha + \frac{C_k}{\lambda_k}\right) \|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}.$$

Suppose that \mathbf{f} is the solution described in Lemma 5.28. Same computations as to obtain (5.5.13) and (5.5.21) yields

$$\frac{d}{dt} \|\mathbf{f}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \le -(1 - C_B) \|\mathbf{f}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k \boldsymbol{\nu})} + C_A \|\mathbf{f}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} + \|\mathbf{Q}(\mathbf{f})\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}.$$

To which we can apply Lemma 5.23:

$$\frac{d}{dt} \|\mathbf{f}\|_{L_{v}^{1} L_{x}^{\infty}(\langle v \rangle^{k})} \leq -\left(1 - C_{B} - C_{Q} \|\mathbf{f}\|_{L_{v}^{1} L_{x}^{\infty}(\langle v \rangle^{k})}\right) \|\mathbf{f}\|_{L_{v}^{1} L_{x}^{\infty}(\langle v \rangle^{k} \boldsymbol{\nu})} + C_{A} \|\mathbf{f}\|_{L_{v}^{1} L_{x}^{\infty}(\langle v \rangle^{k})}.$$
(5.5.22)

We now turn to the second term in the $\|\cdot\|_{\alpha,k}$ norm. For q in $[1,\infty)$ we denote $\Phi_q(\mathbf{F}) = \operatorname{sgn}(\mathbf{F}) |\mathbf{F}|^{q-1}$, where it has to be understood component by component. We thus have

$$\frac{d}{dt} \int_{0}^{+\infty} \|S_{\mathbf{G}}(s) \left(\mathbf{f}(t)\right)\|_{L_{v}^{1} L_{x}^{q}\left(\langle v \rangle^{k}\right)} ds$$

$$= \int_{0}^{+\infty} \int_{\mathbb{R}^{3}} \langle v \rangle^{k} \|S_{\mathbf{G}}(s) \left(\mathbf{f}\right)\|_{L_{x}^{q}}^{1-q} \left(\int_{\mathbb{T}^{3}} \Phi_{q} \left(S_{\mathbf{G}}(s) \left(\mathbf{f}\right)\right) S_{\mathbf{G}}(s) \left[\mathbf{G}(\mathbf{f})\right] dx\right) dv ds$$

$$+ \int_{0}^{+\infty} \int_{\mathbb{R}^{3}} \langle v \rangle^{k} \|S_{\mathbf{G}}(s) \left(\mathbf{f}\right)\|_{L_{x}^{q}}^{1-q} \left(\int_{\mathbb{T}^{3}} \Phi_{q} \left(S_{\mathbf{G}}(s) \left(\mathbf{f}\right)\right) S_{\mathbf{G}}(s) \left[\mathbf{Q}(\mathbf{f})\right] dx\right) dv ds$$

First, by definition of $S_{\mathbf{G}}(s)$ we have that

$$\Phi_q\left(S_{\mathbf{G}}(s)(\mathbf{f}(t))\right)S_{\mathbf{G}}(s)\left[\mathbf{G}(\mathbf{f}(t))\right] = \frac{d}{ds}\left|S_{\mathbf{G}}(s)(\mathbf{f}(t))\right|^q.$$

Second, by Hölder inequality with q and q/(q-1) (see (5.5.12)):

$$\int_{\mathbb{T}^3} \Phi_q \left(S_{\mathbf{G}}(s)(\mathbf{f}(t)) \right) S_{\mathbf{G}}(s) \left[\mathbf{Q}(\mathbf{f}(t)) \right] dx \le \| S_{\mathbf{G}}(s)(\mathbf{f}(t)) \|_{L_x^q}^{q-1} \| S_{\mathbf{G}}(s)(\mathbf{Q}(\mathbf{f}(t))) \|_{L_x^q}.$$

We therefore get

$$\frac{d}{dt} \int_{0}^{+\infty} \|S_{\mathbf{G}}(s) \left(\mathbf{f}(t)\right)\|_{L_{v}^{1} L_{x}^{q}\left(\langle v \rangle^{k}\right)} ds \leq \int_{0}^{+\infty} \frac{d}{ds} \|S_{\mathbf{G}}(\mathbf{f}(t))\|_{L_{v}^{1} L_{x}^{q}\left(\langle v \rangle^{k}\right)} ds
+ \int_{0}^{+\infty} \|S_{\mathbf{G}}(s) \left(\mathbf{Q}(\mathbf{f}(t))\right)\|_{L_{v}^{1} L_{x}^{q}\left(\langle v \rangle^{k}\right)} ds.$$

We make q tend to infinity. Then we have $\Pi_{\mathbf{G}}(\mathbf{Q}(\mathbf{f}(t))) = 0$ by Lemma 5.23 so we are able to use the exponential decay of $S_{\mathbf{G}}(s)$ Lemma 5.27. This yields

$$\begin{split} \frac{d}{dt} \int_{0}^{+\infty} \|S_{\mathbf{G}}(s)\left(\mathbf{f}(t)\right)\|_{L_{v}^{1} L_{x}^{\infty}\left(\langle v \rangle^{k}\right)} \ ds &\leq -\|\mathbf{f}(t)\|_{L_{v}^{1} L_{x}^{\infty}\left(\langle v \rangle^{k}\right)} \\ &+ C_{k} \left(\int_{0}^{+\infty} e^{-\lambda_{k} s} \ ds\right) \|\mathbf{Q}(\mathbf{f}(t))\|_{L_{v}^{1} L_{x}^{\infty}\left(\langle v \rangle^{k}\right)}. \end{split}$$

With Lemma 5.23 we control $\mathbf{Q}(\mathbf{f})$:

$$\frac{d}{dt} \int_{0}^{+\infty} \|S_{\mathbf{G}}(s)\left(\mathbf{f}(t)\right)\|_{L_{v}^{1}L_{x}^{\infty}\left(\langle v\rangle^{k}\right)} ds \leq -\|\mathbf{f}(t)\|_{L_{v}^{1}L_{x}^{\infty}\left(\langle v\rangle^{k}\right)} + \frac{C_{k}C_{Q}}{\lambda_{k}} \|\mathbf{f}\|_{L_{v}^{1}L_{x}^{\infty}\left(\langle v\rangle^{k}\right)} \|\mathbf{f}\|_{L_{v}^{1}L_{x}^{\infty}\left(\langle v\rangle^{k}\boldsymbol{\nu}\right)}.$$
(5.5.23)

To conclude we add $\alpha \times (5.5.22) + (5.5.23)$,

$$\frac{d}{dt} \|\mathbf{f}\|_{k,\alpha} \le -\left[\alpha(1 - C_B) - \left(\alpha C_Q + \frac{C_k C_Q}{\lambda_k}\right) \|\mathbf{f}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}\right] \|\mathbf{f}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k \boldsymbol{\nu})} + \left[\alpha C_A - 1\right] \|\mathbf{f}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}.$$
(5.5.24)

Choosing α such that $(\alpha C_A - 1) < 0$ yields the desired estimate.

We now prove the uniqueness proposition.

Proof of Proposition 5.26. Let **f** and **g** in $L_v^1 L_x^\infty (\langle v \rangle^k)$, $k > k_0$, be two solutions of the perturbed equation with initial datum $\mathbf{f_0}$.

Thanks to Lemma 5.28, if $\|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)}$ is small enough we can deduce from the differential inequality that for all $t \geq 0$,

$$\frac{d}{dt} \|\mathbf{f}\|_{\alpha,k} \le -C_k \|\mathbf{f}\|_{\alpha,k,\boldsymbol{\nu}}$$

and the same holds for \mathbf{g} with the same constant $C_k > 0$. We therefore have two estimates on \mathbf{f} and \mathbf{g} . Either by integrating from 0 to t:

$$\forall t \ge 0, \quad \int_0^t \|\mathbf{f}(s)\|_{\alpha,k,\nu} \ ds \le C_k^{-1} \|\mathbf{f_0}\|_{\alpha,k};$$
 (5.5.25)

or by Grönwall lemma:

$$\forall t \ge 0, \quad \|\mathbf{f}(t)\|_{\alpha,k} \le e^{-C_k t} \|\mathbf{f_0}\|_{\alpha,k}.$$
 (5.5.26)

The same estimate holds for \mathbf{g} .

Recalling the definition (5.5.11) of the operator $\widetilde{\mathbf{Q}}$, we find the differential equation satisfied by $\mathbf{f} - \mathbf{g}$:

$$\partial_t \left(\mathbf{f} - \mathbf{g} \right) = \mathbf{G} \left(\mathbf{f} - \mathbf{g} \right) + \widetilde{\mathbf{Q}} \left(\mathbf{f} - \mathbf{g}, \mathbf{f} \right) + \widetilde{\mathbf{Q}} \left(\mathbf{g}, \mathbf{f} - \mathbf{g} \right).$$

Using controls on **B** (Lemma 5.20), **A** (Lemma 5.19), $\widetilde{\mathbf{Q}}$ (Lemma 5.23) and the semigoup property (Lemma 5.27), exact same computations as for (5.5.24), gives

$$\frac{d}{dt} \|\mathbf{f} - \mathbf{g}\|_{k,\alpha} \le -\left[\alpha(1 - C_B) - \left(\alpha C_Q + \frac{C_k C_Q}{\lambda_k}\right) \left(\|\mathbf{f}\|_{\alpha,k} + \|\mathbf{g}\|_{\alpha,k}\right)\right] \|\mathbf{f} - \mathbf{g}\|_{\alpha,k,\nu} + \left[\alpha C_A - 1 + C_Q \left(\|\mathbf{f}\|_{\alpha,k,\nu} + \|\mathbf{g}\|_{\alpha,k,\nu}\right)\right] \|\mathbf{f} - \mathbf{g}\|_{\alpha,k}.$$

Note that we used the equivalence of the $\|\cdot\|_{\alpha,k}$ norm and our usual norm (see Lemma 5.28).

First, by (5.5.26) and $C_B < 1$, if $\mathbf{f_0}$ is small enough then for all $t \ge 0$,

$$\alpha(1 - C_B) - \left(\alpha C_Q + \frac{C_k C_Q}{\lambda_k}\right) \left(\|\mathbf{f}\|_{\alpha,k} + \|\mathbf{g}\|_{\alpha,k}\right) \le 0.$$

Second we take α small enough so that $(\alpha C_A - 1) < 0$. Hence, integrating the differential inequality from 0 to t:

$$\|\mathbf{f}(t) - \mathbf{g}(t)\|_{k,\alpha} \le C_Q \left[\int_0^t \left(\|\mathbf{f}(s)\|_{\alpha,k,\nu} + \|\mathbf{g}(\mathbf{s})\|_{\alpha,k,\nu} \right) ds \right] \sup_{s \in [0,t]} \|\mathbf{f}(s) - \mathbf{g}(s)\|_{k,\alpha}.$$

To conclude we use (5.5.25) to obtain

$$\forall t \geq 0, \quad \|\mathbf{f}(t) - \mathbf{g}(t)\|_{k,\alpha} \leq \frac{2C_Q}{C_k} \|\mathbf{f_0}\|_{\alpha,k} \left(\sup_{s \in [0,t]} \|\mathbf{f}(s) - \mathbf{g}(s)\|_{k,\alpha} \right),$$

which implies $\mathbf{f} = \mathbf{g}$ if $\|\mathbf{f_0}\|_{\alpha,k}$ is small enough.

5.5.3. Positivity of solutions

This last subsection is dedicated to the positivity of the solution to the multi-species Boltzmann equation

$$\partial_t \mathbf{F} + v \cdot \nabla_x \mathbf{F} = \mathbf{Q} \left(\mathbf{F} \right) \tag{5.5.27}$$

in the perturbative setting studied above.

Proposition 5.29. Let $k > k_0$, let assumptions (H1) - (H4) hold for the collision kernel, and let $\mathbf{f_0}$ be in $L_v^1 L_x^{\infty} (\langle v \rangle^k)$ with $\Pi_{\mathbf{G}}(\mathbf{f_0}) = 0$ and

$$\|\mathbf{f_0}\|_{L_v^1 L_x^{\infty}(\langle v \rangle^k)} \le \eta_k,$$

where $\eta_k > 0$ is chosen such that Proposition 5.18 and Proposition 5.26 hold and denote \mathbf{f} the unique solution of the perturbed multi-species equation associated to $\mathbf{f_0}$. Suppose that $\mathbf{F_0} = \boldsymbol{\mu} + \mathbf{f_0} \geq 0$ then $\mathbf{F} = \boldsymbol{\mu} + \mathbf{f} \geq 0$.

Proof of Proposition 5.29. Since we are working with the Grad's cutoff assumption we can decompose the nonlinear operator into

$$\mathbf{Q}(\mathbf{F}) = -\mathbf{Q_1}(\mathbf{F}) + \mathbf{Q_2}(\mathbf{F})$$

where

$$Q_{1i}(\mathbf{F}) = \sum_{j=1}^{N} \int_{\mathbb{R}^{3} \times \mathbb{S}^{2}} B_{ij} \left(|v - v_{*}|, \cos \theta \right) F_{i} F_{j}^{*} dv_{*} d\sigma$$

$$Q_{2i}(\mathbf{F}) = \sum_{j=1}^{N} \int_{\mathbb{R}^3 \times \mathbb{S}^2} B_{ij} \left(|v - v_*|, \cos \theta \right) F_i' F_j'^* dv_* d\sigma.$$

Following the idea of [83], we construct an interative scheme for the multi-species Boltzmann equation

$$\partial_t \mathbf{F^{(n+1)}} + v \cdot \nabla_x \mathbf{F^{(n+1)}} + \overline{\mathbf{Q}}_1(\mathbf{F^{(n+1)}}, \mathbf{F^{(n)}}) = \mathbf{Q}_2(\mathbf{F^{(n)}}),$$

with the non-symmetriized bilinear form $\overline{\mathbf{Q}}_1$ defined as

$$\overline{Q}_{1i}(\mathbf{F}, \mathbf{G}) = \sum_{j=1}^{N} \int_{\mathbb{R}^{3} \times \mathbb{S}^{2}} B_{ij} (|v - v_{*}|, \cos \theta) F_{i} G_{j}^{*} dv_{*} d\sigma$$

$$\overline{Q}_{2i}(\mathbf{F}, \mathbf{G}) = \sum_{j=1}^{N} \int_{\mathbb{R}^{3} \times \mathbb{S}^{2}} B_{ij} \left(|v - v_{*}|, \cos \theta \right) F_{i}' G_{j}'^{*} dv_{*} d\sigma.$$

Defining $\mathbf{f^{(n)}} = \mathbf{F^n} - \boldsymbol{\mu}$ we have the following differential iterative scheme

$$\partial_t \mathbf{f^{(n+1)}} + v \cdot \nabla_x \mathbf{f^{(n+1)}} = -\boldsymbol{\nu}(v) \left(\mathbf{f^{(n+1)}} \right) + \mathbf{K} \left(\mathbf{f^{(n)}} \right) + \mathbf{Q_2} \left(\mathbf{f^{(n)}} \right) - \widetilde{\mathbf{Q}_1} \left(\mathbf{f^{(n+1)}}, \mathbf{f^{(n)}} \right).$$

As before, we can prove that $(\mathbf{f^{(n)}})_{n\in\mathbb{N}}$ is well-defined and converges in $L^1_vL^\infty_x\left(\langle v\rangle^k\right)$ towards \mathbf{f} , the unique solution of the perturbed multi-species equation and thus the same holds for $\mathbf{F^n}$ converging towards \mathbf{F} the unique perturbed solution of the original multi-species Boltzmann equation.

We prove that $\mathbf{F}^{\mathbf{n}} \geq 0$ by an induction on N. By definition we see that

$$\widetilde{\mathbf{Q}}_{\mathbf{1}}(\mathbf{F}^{(\mathbf{n}+\mathbf{1})}, \mathbf{F}^{(\mathbf{n})}) = q_1(\mathbf{F}^{(\mathbf{n})})\mathbf{F}^{(\mathbf{n}+\mathbf{1})}$$

and thus applying the Duhamel formula along the characteristics gives

$$\begin{aligned} \mathbf{F}^{(\mathbf{n}+\mathbf{1})}(t,x,v) \\ &= \exp\left[-\int_0^t q_1(\mathbf{F}^{(\mathbf{n})})(s,x-(t-s)v,v)\,ds\right]\mathbf{F_0}(x-tv,v) \\ &+ \int_0^t \exp\left[-\int_s^t q_1(\mathbf{F}^{(\mathbf{n})})(s_1,x-(t-s_1)v,v)\,ds_1\right]\mathbf{Q_2}(\mathbf{F}^{(\mathbf{n})})(s,x-(t-s)v,v)\,ds. \end{aligned}$$

By positivity of $\mathbf{F^{(n)}}$, all the terms on the right-hand side are positive and therefore $\mathbf{F^{n+1}} \geq 0$.

Part III. The diffusive models

6. Cross-diffusion population systems for multiple species

Our goal in this chapter is the proof of global existence of weak solutions to reaction-cross-diffusion systems for an arbitrary number of competing population species in a bounded domain with homogeneous Neumann boundary conditions. In the case of linear transition rates, the model considered extends the two-species population model of Shigesada, Kawasaki, and Teramoto. The existence proof uses a refined entropy method with the help of a new approximation scheme. Global existence can be proved under a detailed balance or weak cross-diffusion condition. The detailed balance condition is related to the symmetry of the mobility matrix, which mirrors Onsager's principle in thermodynamics. Under detailed balance (and without reaction), we can show that the entropy is nonincreasing in time, but counter-examples illustrate that the entropy can increase initially if detailed balance is not satisfied.

6.1. The model

We briefly recall the model. As already introduced in Section 3.1, we consider the reactioncross-diffusion equations

$$\partial_t u_i - \operatorname{div}\left(\sum_{j=1}^n A_{ij}(u)\nabla u_j\right) = f_i(u) \quad \text{in } \Omega, \ t > 0, \quad i = 1, \dots, n,$$
(6.1.1)

with homogeneous Neumann boundary conditions and initial conditions

$$\sum_{i=1}^{n} A_{ij}(u) \nabla u_j \cdot \nu = 0 \quad \text{on } \partial \Omega, \ t > 0, \quad u_i(\cdot, 0) = u_i^0 \quad \text{in } \Omega,$$
 (6.1.2)

where the variable u_i describes the density of the *i*th species of $u = (u_1, \ldots, u_n)$, $\Omega \subset \mathbb{R}^d$ $(d \geq 1)$ is a bounded domain with Lipschitz boundary, and ν is the exterior unit normal vector to $\partial\Omega$. The diffusion coefficients have the form

$$A_{ij}(u) = \delta_{ij}p_i(u) + u_i \frac{\partial p_i}{\partial u_j}(u), \quad p_i(u) = a_{i0} + \sum_{k=1}^n a_{ik}u_k^s, \quad i, j = 1, \dots, n,$$
 (6.1.3)

with a_{i0} , $a_{ij} \ge 0$ and s > 0. The reaction terms f_i are of Lotka-Volterra type,

$$f_i(u) = u_i \left(b_{i0} - \sum_{j=1}^n b_{ij} u_j \right), \quad i = 1, \dots, n,$$
 (6.1.4)

and we assume that we are in the competition case, thus b_{i0} , $b_{ij} \geq 0$. We notice that equations (6.1.1) can be also written in the following form

$$\partial_t u_i - \Delta(u_i p_i(u)) = f_i(u), \quad 1 < i < n. \tag{6.1.5}$$

The entropy reads

$$\mathcal{H}[u] = \int_{\Omega} h(u)dx = \int_{\Omega} \sum_{i=1}^{n} \pi_i h_s(u_i) dx, \tag{6.1.6}$$

where $\pi_i > 0$ are some positive numbers and

$$h_s(z) = \begin{cases} z(\log z - 1) + 1 & \text{for } s = 1, \\ \frac{z^s - sz}{s - 1} + 1 & \text{for } s \neq 1. \end{cases}$$
 (6.1.7)

If we use the so-called entropy variable $w = (w_1, \ldots, w_n)$ by

$$w_i = \frac{\partial h}{\partial u_i}(u) = \begin{cases} \pi_i \log u_i & \text{for } s = 1, \\ \frac{s\pi_i}{s-1}(u_i^{s-1} - 1) & \text{for } s \neq 1, \end{cases}$$

we can rewrite equations (6.1.1) as

$$\partial_t u(w) - \text{div}(B(w)\nabla w) = f(u(w)), \quad B(w) = A(u)H(u)^{-1},$$
 (6.1.8)

with $u(w) := (h')^{-1}(w)$ describing the inverse transformation and H(u) = h''(u) the Hessian of the entropy density.

Now we introduce two important conditions, a detailed-balance condition and a weak cross-diffusion condition, which are important in the context of deriving the crucial gradient estimate in Section 6.2.1, which has the form

$$\frac{d}{dt}\mathcal{H}[u] + 4\int_{\Omega} \sum_{i=1}^{n} \pi_i a_{i0} |\nabla \sqrt{u_i}|^2 dx + 2\int_{\Omega} \sum_{i=1}^{n} \pi_i a_{ii} |\nabla u_i|^2 dx \le 0.$$
 (6.1.9)

This inequality implies an H^1 estimate for $\sqrt{u_i}$ if $a_{i0} > 0$, and an H^1 estimate for u_i if $a_{ii} > 0$. First, it turns out that (6.1.6) is a Lyapunov functional and the condition H(u)A(u) is symmetric and positive definite if

$$\pi_i a_{ij} = \pi_j a_{ji}$$
 for all $i, j = 1, \dots, n$, (6.1.10)

which we will call detailed balance condition.

Moreover, we show that if self-diffusion dominates cross-diffusion in the sense

$$\eta_0 := \min_{i=1,\dots,n} \left(a_{ii} - \frac{s}{2(s+1)} \sum_{j=1}^n \left(\sqrt{a_{ij}} - \sqrt{a_{ji}} \right)^2 \right) > 0, \tag{6.1.11}$$

and detailed balance may be not satisfied, then the estimate leading to (6.1.9) still holds true, but with different constants, leading to global existence. We remark that we set $\pi_i = 1$ whenever detailed balance does not hold.

6.1.1. Main results

Theorem 6.1 (Global existence for linear transition rates). Let T > 0, s = 1 and $u^0 = (u_1^0, \ldots, u_n^0)$ be such that $u_i^0 \ge 0$ for $i = 1, \ldots, n$ and $\int_{\Omega} h(u^0) dx < \infty$. Let either detailed balance and $a_{ii} > 0$ for $i = 1, \ldots, n$; or (6.1.11) hold. Then there exists a weak solution $u = (u_1, \ldots, u_n)$ to (6.1.1)-(6.1.2) satisfying $u_i \ge 0$ in Ω , t > 0, and

$$u_i \in L^2(0, T; H^1(\Omega)), \quad u_i \in L^{\infty}(0, T; L^1(\Omega)),$$

 $u_i \in L^{2+2/d}(Q_T), \quad \partial_t u_i \in L^{q'}(0, T; W^{1,q}(\Omega)'), \quad i = 1, \dots, n,$

where q = 2(d+1) and q' = (2d+2)/(2d+1). The solution u solves (6.1.1) in the weak sense

$$\int_{0}^{T} \langle \partial_{t} u, \phi \rangle dt + \int_{0}^{T} \int_{\Omega} \nabla \phi : A(u) \nabla u dx dt = \int_{0}^{T} \int_{\Omega} f(u) \cdot \phi dx dt$$
 (6.1.12)

for all test functions $\phi \in L^q(0,T;W^{1,q}(\Omega))$, and the initial condition in (6.1.2) is satisfied in the sense of $W^{1,q}(\Omega)'$.

For a generalization of this theorem to the case of vanishing self-diffusion, see Remark 6.12

The second result works for nonlinear transition rates $(s \neq 1)$. In the sublinear case (i.e. s < 1), we face the probem that the entropy inequality only yields the regularity $u_i \in L^{2s+2/d}(Q_T)$. This doe not yield an L^2 estimate for "small" exponents s < 1 and large dimensions d. Thus, we need to assume the lower bound s > 1 - 2/d and a weaker growth of the Lotka-Volterra terms:

$$f_i(u) = u_i \left(b_{i0} - \sum_{j=1}^n b_{ij} u_j^{\sigma} \right), \quad i = 1, \dots, n, \quad 0 \le \sigma < 2s - 1 + 2/d.$$
 (6.1.13)

The superlinear case (i.e. s > 1) is in some sense easier than the sublinear one, because the entropy inequality gives the higher regularity $u_i \in L^p(Q_T)$ with p > 2. On the other hand, we need a weak cross-diffusion constraint. Thus, under detailed balance, we require that

$$\eta_1 := \min_{i=1,\dots,n} \left(a_{ii} - \frac{s-1}{s+1} \sum_{j=1, j \neq i}^n a_{ij} \right) > 0, \tag{6.1.14}$$

and if detailed balance is not satisfied, we assume that

$$\eta_2 := \min_{i=1,\dots,n} \left(a_{ii} - \frac{1}{2(s+1)} \sum_{j=1,j \neq i} \left(s(a_{ij} + a_{ji}) - 2\sqrt{a_{ij}a_{ji}} \right) \right) > 0.$$
 (6.1.15)

For $m \geq 2$ and $1 \leq q \leq \infty$ we introduce the space

$$W_{\nu}^{m,q}(\Omega) = \{ \phi \in W^{m,q}(\Omega) : \nabla \phi \cdot \nu = 0 \text{ on } \partial \Omega \}.$$
 (6.1.16)

Theorem 6.2 (Global existence for nonlinear transition rates). Assume that T > 0, $s > \max\{0, 1 - 2/d\}$, and let the initial data u^0 be such that $u_i^0 \ge 0$ for i = 1, ..., n and $\int_{\Omega} h(u^0) dx < \infty$. If s < 1, we suppose that (6.1.13) and either detailed balance and $a_{ii} > 0$ for i = 1, ..., n; or (6.1.11) hold. If s > 1, we suppose that (6.1.4) and either detailed balance and (6.1.14) or (6.1.15) hold. Then there exist a number $2 \le q < \infty$ and a weak solution $u = (u_1, ..., u_n)$ to (6.1.1)-(6.1.2) satisfying $u_i \ge 0$ in Ω , t > 0, and

$$u_i^s \in L^2(0,T; H^1(\Omega)), \quad u_i \in L^{\infty}(0,T; L^{\max\{1,s\}}(\Omega)),$$

 $u_i \in L^{p(s)}(Q_T), \quad \partial_t u_i \in L^{q'}(0,T; W_{\nu}^{m,q}(\Omega)'), \quad i = 1, \dots, n,$

where $p(s) = 2s + (2/d) \max\{1, s\}$, 1/q + 1/q' = 1, and $m > \max\{1, d/2\}$. The solution u solves (6.1.1) in the "very weak" sense

$$\int_0^T \langle \partial_t u, \phi \rangle dt - \int_0^T \int_{\Omega} \sum_{i=1}^n u_i p_i(u) \Delta \phi_i dx dt = \int_0^T \int_{\Omega} f(u) \cdot \phi dx dt$$
 (6.1.17)

for all $\phi = (\phi_1, \dots, \phi_n) \in L^q(0, T; W_{\nu}^{m,q}(\Omega))$, and the initial condition holds in the sense of $W_{\nu}^{m,q}(\Omega)'$.

Again, we can generalize the theorem to the case of vanishing self-diffusion if either $s > \max\{1, d/2\}$; or 0 < s < 1, d = 1, and $\sigma < s + 1$ hold; which is sketched in Remark 6.17.

The chapter is organized as follows. Section 6.2 is concerned with the positive definiteness of the matrices H(u)A(u) and $H_{\varepsilon}(u)A_{\varepsilon}(u)$. The existence theorems are proved in Sections 6.3 and 6.4, respectively. In the final Section 6.5, we detail the connection between the detailed balance condition and the symmetry of H(u)A(u), prove a nonlinear Aubin-Lions compactness lemma needed in the proof of Theorem 6.2, and show that the entropy may be increasing initially for special initial data.

6.2. Positive definiteness of the mobility matrix

We derive sufficient conditions for the positive definiteness of the matrix H(u)A(u). Let $\mathbb{R}_+ = (0, \infty)$. Recall that

$$A_{ij}(u) = \delta_{ij} \left(a_{i0} + \sum_{k=1}^{n} a_{ik} u_k^s \right) + s a_{ij} u_i u_j^{s-1}, \quad H_{ij}(u) = \delta_{ij} s \pi_i u_i^{s-2}.$$

The following result is valid for any s > 0.

Lemma 6.3. Let s > 0. Then, for any $z \in \mathbb{R}^n$ and $u \in \mathbb{R}^n_+$,

$$z^{\top}H(u)A(u)z \ge s \sum_{i=1}^{n} \pi_{i}a_{i0}u_{i}^{s-2}z_{i}^{2} + s(1-s) \sum_{i,j=1, i\neq j}^{n} \pi_{i}a_{ij}u_{j}^{s}u_{i}^{s-2}z_{i}^{2}$$

$$+ s \sum_{i=1}^{n} \left((s+1)\pi_{i}a_{ii} - \frac{s}{2} \sum_{j=1}^{n} \left(\sqrt{\pi_{i}a_{ij}} - \sqrt{\pi_{j}a_{ji}} \right)^{2} \right) u_{i}^{2(s-1)}z_{i}^{2}.$$
 (6.2.1)

Proof. The elements of the matrix H(u)A(u) equal

$$(H(u)A(u))_{ij} = \delta_{ij}s\pi_i \left(a_{i0}u_i^{s-2} + \sum_{k=1}^n a_{ik}u_k^s u_i^{s-2} \right) + s^2 a_{ij}(u_i u_j)^{s-1}$$

$$= \delta_{ij} \left(s\pi_i a_{i0}u_i^{s-2} + s(s+1)\pi_i a_{ii}u_i^{2(s-1)} \right)$$

$$+ \delta_{ij}s\pi_i \sum_{k=1, k\neq i}^n a_{ik}u_k^s u_i^{s-2} + (1-\delta_{ij})s^2\pi_i a_{ij}(u_i u_j)^{s-1}.$$

Therefore, for $z \in \mathbb{R}^n$,

$$z^{\top}H(u)A(u)z = s \sum_{i=1}^{n} \pi_{i}a_{i0}u_{i}^{s-2}z_{i}^{2} + s(s+1) \sum_{i=1}^{n} \pi_{i}a_{ii}u_{i}^{2(s-1)}z_{i}^{2}$$

$$+ s \sum_{i,j=1, i\neq j}^{n} \pi_{i}a_{ij}u_{j}^{s}u_{i}^{s-2}z_{i}^{2} + s^{2} \sum_{i,j=1, i\neq j}^{n} \pi_{i}a_{ij}(u_{i}u_{j})^{s-1}z_{i}z_{j}$$

$$=: I_{1} + \dots + I_{4}.$$

$$(6.2.2)$$

The sum I_1 is the same as the first term on the right-hand side of (6.2.1), and I_2 equals the first part of the last term on this right-hand side. The remaining terms are written as

$$I_{3} + I_{4} = s^{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} u_{j}^{s} u_{i}^{s-2} z_{i}^{2} + s(1-s) \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} u_{j}^{s} u_{i}^{s-2} z_{i}^{2}$$
$$+ s^{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{i} u_{j})^{s-1} z_{i} z_{j}.$$

The second term corresponds to the second term on the right-hand side of (6.2.1). Thus, it remains to prove that

$$J := s^{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} u_{j}^{s} u_{i}^{s-2} z_{i}^{2} + s^{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{i} u_{j})^{s-1} z_{i} z_{j}$$

$$\geq -\frac{s^{2}}{2} \sum_{i=1}^{n} \left(\sqrt{\pi_{i} a_{ij}} - \sqrt{\pi_{j} a_{ji}} \right)^{2} u_{i}^{2(s-1)} z_{i}^{2}.$$

For this, we employ twice the inequality $b^2 + c^2 \ge 2bc$:

$$J = s^{2} \sum_{i,j=1, i < j}^{n} \pi_{i} a_{ij} u_{j}^{s} u_{i}^{s-2} z_{i}^{2} + s^{2} \sum_{i,j=1, i > j}^{n} \pi_{i} a_{ij} u_{j}^{s} u_{i}^{s-2} z_{i}^{2}$$

$$+ s^{2} \sum_{i,j=1, i < j}^{n} \pi_{i} a_{ij} (u_{i} u_{j})^{s-1} z_{i} z_{j} + s^{2} \sum_{i,j=1, i > j}^{n} \pi_{i} a_{ij} (u_{i} u_{j})^{s-1} z_{i} z_{j}$$

$$= s^{2} \sum_{i,j=1, i < j}^{n} \left(\pi_{i} a_{ij} u_{j}^{s} u_{i}^{s-2} z_{i}^{2} + \pi_{j} a_{ji} u_{i}^{s} u_{j}^{s-2} z_{j}^{2} + (\pi_{i} a_{ij} + \pi_{j} a_{ji}) (u_{i} u_{j})^{s-1} z_{i} z_{j} \right)$$

$$\geq s^{2} \sum_{i,j=1, i < j}^{n} \left(2\sqrt{\pi_{i}a_{ij}\pi_{j}a_{ji}}(u_{i}u_{j})^{s-1}|z_{i}z_{j}| - (\pi_{i}a_{ij} + \pi_{j}a_{ji})(u_{i}u_{j})^{s-1}|z_{i}z_{j}| \right)$$

$$= -s^{2} \sum_{i,j=1, i < j}^{n} \left(\sqrt{\pi_{i}a_{ij}} - \sqrt{\pi_{j}a_{ji}} \right)^{2} \left| (u_{i}^{s-1}z_{i})(u_{j}^{s-1}z_{j}) \right|$$

$$\geq -\frac{s^{2}}{2} \sum_{i,j=1, i < j}^{n} \left(\sqrt{\pi_{i}a_{ij}} - \sqrt{\pi_{j}a_{ji}} \right)^{2} \left((u_{i}^{s-1}z_{i})^{2} + (u_{j}^{s-1}z_{j})^{2} \right)$$

$$= -\frac{s^{2}}{2} \sum_{i,j=1, i \neq j}^{n} \left(\sqrt{\pi_{i}a_{ij}} - \sqrt{\pi_{j}a_{ji}} \right)^{2} (u_{i}^{s-1}z_{i})^{2}.$$

This finishes the proof.

6.2.1. Sublinear and linear transition rates

For $s \leq 1$, Lemma 6.3 provides immediately the positive definiteness of H(u)A(u) if detailed balance (6.1.10) holds. However, we can derive a sharper result.

Lemma 6.4 (Detailed balance). Let $0 < s \le 1$ and $\pi_i a_{ij} = \pi_j a_{ji}$ for all $i \ne j$. Then, for all $z \in \mathbb{R}^n$ and $u \in \mathbb{R}^n_+$,

$$z^{\top}H(u)A(u)z \ge s \sum_{i=1}^{n} \pi_{i} u_{i}^{s-2} \left(a_{i0} + (s+1)a_{ii}u_{i}^{s}\right) z_{i}^{2}$$

$$+ \frac{s^{2}}{2} \sum_{i,j=1, i \ne j}^{n} \pi_{i} a_{ij} (u_{i}u_{j})^{s-1} \left(\sqrt{\frac{u_{j}}{u_{i}}} z_{i} + \sqrt{\frac{u_{i}}{u_{j}}} z_{j}\right)^{2}.$$

$$(6.2.3)$$

Proof. The sum of the terms I_1 and I_2 in (6.2.2) is exactly the first term on the right-hand side of (6.2.3). Using detailed balance, we find that

$$I_{3} + I_{4} = \frac{s}{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{i}u_{j})^{s-1} \frac{u_{j}}{u_{i}} z_{i}^{2} + \frac{s}{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{j}u_{i})^{s-1} \frac{u_{i}}{u_{j}} z_{j}^{2}$$

$$+ s^{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{i}u_{j})^{s-1} z_{i} z_{j}$$

$$= \frac{s^{2}}{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{i}u_{j})^{s-1} \frac{u_{j}}{u_{i}} z_{i}^{2} + \frac{s^{2}}{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{j}u_{i})^{s-1} \frac{u_{i}}{u_{j}} z_{j}^{2}$$

$$+ s^{2} \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{i}u_{j})^{s-1} z_{i} z_{j} + \frac{s}{2} (1-s) \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{i}u_{j})^{s-1} \frac{u_{j}}{u_{i}} z_{i}^{2}$$

$$+ \frac{s}{2} (1-s) \sum_{i,j=1, i \neq j}^{n} \pi_{i} a_{ij} (u_{j}u_{i})^{s-1} \frac{u_{i}}{u_{j}} z_{j}^{2}.$$

The sum of the first three terms equal the second term on the right-hand side of (6.2.3), and the remaining two terms are nonnegative since $s \le 1$.

Remark 6.5. In the existence proof, we will choose $z_i = \nabla u_i$ (with a slight abuse of notation). Then the first term in (6.2.3) gives an estimate for $\nabla u_i^{s/2}$ in L^2 (if $a_{i0} > 0$) and the better bound $\nabla u_i^s \in L^2$ (if $a_{ii} > 0$). If $a_{ii} = 0$, we lose the latter regularity. This loss can be compensated by the last term in (6.2.3) giving

$$(u_i u_j)^{s-1} \left| \sqrt{\frac{u_j}{u_i}} \nabla u_i + \sqrt{\frac{u_i}{u_j}} \nabla u_j \right|^2 = \frac{4}{s^2} |\nabla (u_i u_j)^{s/2}|^2, \quad i \neq j,$$

and consequently a bound for $\nabla (u_i u_j)^{s/2}$ in L^2 . This observation is used in Remark 6.12.

Lemma 6.6 (Non detailed balance). Let $0 < s \le 1$. If

$$\eta_0 := \min_{i=1,\dots,n} \left(a_{ii} - \frac{s}{2(s+1)} \sum_{j=1}^n \left(\sqrt{a_{ij}} - \sqrt{a_{ji}} \right)^2 \right) \ge 0,$$

then H(u)A(u) is positive definite. Under the slightly stronger condition $\eta_0 > 0$, it holds for all $z \in \mathbb{R}^n$ and $u \in \mathbb{R}^n_+$ that

$$z^{\top}H(u)A(u)z \ge s\sum_{i=1}^{n} a_{i0}u_i^{s-2}z_i^2 + \eta_0 s(s+1)\sum_{i=1}^{n} u_i^{2(s-1)}z_i^2.$$

The lemma follows from Lemma 6.3 after choosing $\pi_i = 1$ for i = 1, ..., n. Observe that $\eta_0 > 0$ holds if $a_{ii} > 0$ for all i and (a_{ij}) is symmetric.

It is possible to show the positive definiteness of H(u)A(u) without any restriction on (a_{ij}) (except positivity) if we restrict the choice of the parameter s; see the following lemma.

Lemma 6.7. Let $a_{ij} + a_{ji} > 0$ for i, j = 1, ..., n and $0 < s \le s_0$, where

$$s_0 := \min_{i,j=1,\dots,n} \frac{2\sqrt{a_{ij}a_{ji}}}{a_{ij} + a_{ji}} \le 1.$$

Then, for all $z \in \mathbb{R}^n$ and $u \in \mathbb{R}^n_+$,

$$z^{\top}H(u)A(u)z \ge s\sum_{i=1}^{n} a_{i0}u_i^{s-2}z_i^2 + s(s+1)\sum_{i=1}^{n} a_{ii}u_i^{2(s-1)}z_i^2.$$

Proof. We choose $\pi_i = 1$ for i = 1, ..., n. With the notation of the proof of Lemma 6.3, we only need to show that $I_3 + I_4 \ge 0$. Employing the inequality $b^2 + c^2 \ge 2bc$, we find that

$$I_{3} + I_{4} = s \sum_{i,j=1, i < j}^{n} \left(a_{ij} u_{j}^{s} u_{i}^{s-2} z_{i}^{2} + a_{ji} u_{i}^{s} u_{j}^{s-2} z_{j}^{2} + s(a_{ij} + a_{ji})(u_{i}u_{j})^{s-1} z_{i} z_{j} \right)$$

$$\geq s \sum_{i,j=1, i < j}^{n} \left(2\sqrt{a_{ij}a_{ji}}(u_{i}u_{j})^{s-1} |z_{i}z_{j}| - s(a_{ij} + a_{ji})(u_{i}u_{j})^{s-1} |z_{i}z_{j}| \right)$$

$$= s \sum_{i,j=1, i < j}^{n} (a_{ij} + a_{ji}) \left(\frac{2\sqrt{a_{ij}a_{ji}}}{a_{ij} + a_{ji}} - s \right) (u_{i}u_{j})^{s-1} |z_{i}z_{j}|,$$

and this expression is nonnegative if $s \leq s_0$.

6.2.2. Superlinear transition rates

Again, we assume first that detailed balance holds.

Lemma 6.8 (Detailed balance). Let s > 1 and $\pi_i a_{ij} = \pi_j a_{ji}$ for all $i \neq j$. If

$$\eta_1 := \min_{i=1,\dots,n} \left(a_{ii} - \frac{s-1}{s+1} \sum_{j=1, j \neq i}^n a_{ij} \right) \ge 0,$$

then H(u)A(u) is positive definite. Furthermore, if $\eta_1 > 0$, then, for all $z \in \mathbb{R}^n$ and $u \in \mathbb{R}^n_+$

$$z^{\top}H(u)A(u)z \ge s\sum_{i=1}^{n} \pi_i a_{i0} u_i^{s-2} z_i^2 + \eta_1 s(s+1) \sum_{i=1}^{n} \pi_i u_i^{2(s-1)} z_i^2.$$

Proof. It is sufficient to estimate the sum $I_3 + I_4$, defined in the proof of Lemma 6.3:

$$I_{3} + I_{4} = s \sum_{i,j=1,i < j}^{n} \left(\pi_{i} a_{ij} u_{j}^{s} u_{i}^{s-2} z_{i}^{2} + \pi_{j} a_{ji} u_{i}^{s} u_{j}^{s-2} z_{j}^{2} + s (\pi_{i} a_{ij} + \pi_{j} a_{ji}) (u_{i} u_{j})^{s-1} z_{i} z_{j} \right)$$

$$\geq s \sum_{i,j=1,i < j}^{n} \left(2 \sqrt{\pi_{i} a_{ij} \pi_{j} a_{ji}} (u_{i} u_{j})^{s-1} |z_{i} z_{j}| - s (\pi_{i} a_{ij} + \pi_{j} a_{ji}) (u_{i} u_{j})^{s-1} |z_{i} z_{j}| \right)$$

$$= -s \sum_{i,j=1,i < j}^{n} \left(s (\pi_{i} a_{ij} + \pi_{j} a_{ji}) - 2 \sqrt{\pi_{i} a_{ij} \pi_{j} a_{ji}} \right) (u_{i} u_{j})^{s-1} |z_{i} z_{j}|$$

$$\geq -\frac{s}{2} \sum_{i,j=1,i < j}^{n} \left(s (\pi_{i} a_{ij} + \pi_{j} a_{ji}) - 2 \sqrt{\pi_{i} a_{ij} \pi_{j} a_{ji}} \right) \left((u_{i}^{s-1} z_{i})^{2} + (u_{j}^{s-1} z_{j})^{2} \right)$$

$$= -\frac{s}{2} \sum_{i,j=1,i \neq j}^{n} \left(s (\pi_{i} a_{ij} + \pi_{j} a_{ji}) - 2 \sqrt{\pi_{i} a_{ij} \pi_{j} a_{ji}} \right) (u_{i}^{s-1} z_{i})^{2}.$$

This expression simplifies because of the detailed balance condition:

$$I_3 + I_4 \ge -s(s-1) \sum_{i,j=1, i \ne j}^n \pi_i a_{ij} (u_i^{s-1} z_i)^2,$$

and we end up with

$$z^{\top}H(u)A(u)z \ge s\sum_{i=1}^{n} \pi_{i}a_{i0}u_{i}^{s-2}z_{i}^{2} + s(s+1)\sum_{i=1}^{n} \pi_{i}\left(a_{ii} - \frac{s-1}{s+1}\sum_{j=1, j\neq i}^{n} a_{ij}\right)u_{i}^{2(s-1)}z_{i}^{2},$$

from which we conclude the result.

Remark 6.9. Let n=2. Then the condition $\eta_1 \geq 0$ on the coefficients (a_{ij}) becomes $a_{11} \geq a_{12}(s-1)/(s+1)$ and $a_{22} \geq a_{21}(s-1)/(s+1)$. The product

$$a_{11}a_{22} \ge \left(\frac{s-1}{s+1}\right)^2 a_{12}a_{21}$$

is the same as the condition imposed in [51, Section 5.1] but weaker than

$$a_{11}a_{22} \ge \left(\frac{s-1}{s}\right)^2 a_{12}a_{21},$$

which was needed in [94, Lemma 11]. Furthermore, under the slightly stronger condition $\eta_1 > 0$, that is

$$a_{11}a_{22} > \left(\frac{s-1}{s+1}\right)^2 a_{12}a_{21},$$

our weak solution satisfies the stronger estimate $u_i^s \in L^2(0,T;H^1(\Omega))$ than that in [51, Section 5.1].

Lemma 6.10 (Non detailed balance). Let s > 1 and let

$$\eta_2 := \min_{i=1,\dots,n} \left(a_{ii} - \frac{1}{2(s+1)} \sum_{j=1, j \neq i} \left(s(a_{ij} + a_{ji}) - 2\sqrt{a_{ij}a_{ji}} \right) \right) \ge 0.$$

Then H(u)A(u) is positive definite. Moreover, if $\eta_2 > 0$, then, for all $z \in \mathbb{R}^n$ and $u \in \mathbb{R}^n_+$,

$$z^{\top} H(u) A(u) z \ge s \sum_{i=1}^{n} a_{i0} u_i^{s-2} z_i^2 + \eta_2 s(s+1) \sum_{i=1}^{n} u_i^{2(s-1)} z_i^2.$$

Proof. We choose $\pi_i = 1$ for $i = 1, \ldots, n$. Then, as in the previous proof,

$$I_3 + I_4 \ge -\frac{s}{2} \sum_{i,j=1, i \ne j}^{n} \left(s(a_{ij} + a_{ji}) - 2\sqrt{a_{ij}a_{ji}} \right) u_i^{2(s-1)} z_i^2$$

and

$$z^{\top}H(u)A(u)z \ge s \sum_{i=1}^{n} a_{i0}u_{i}^{s-2}z_{i}^{2} + s(s+1) \sum_{i=1}^{n} a_{ii}u_{i}^{2(s-1)}z_{i}^{2}$$

$$-\frac{s}{2} \sum_{i,j=1, i \ne j}^{n} \left(s(a_{ij} + a_{ji}) - 2\sqrt{a_{ij}a_{ji}}\right)u_{i}^{2(s-1)}z_{i}^{2}$$

$$= s \sum_{i=1}^{n} a_{i0}u_{i}^{s-2}z_{i}^{2}$$

$$+ s(s+1) \sum_{i=1}^{n} \left(a_{ii} - \frac{1}{2(s+1)} \sum_{i,j=1, i \ne j}^{n} \left(s(a_{ij} + a_{ji}) - 2\sqrt{a_{ij}a_{ji}}\right)\right)u_{i}^{2(s-1)}z_{i}^{2}.$$

By definition of η_2 , the result follows.

6.2.3. Approximate matrices

Our theory requires that the range of the derivative h' equals \mathbb{R}^n . Since this is not the case if $s \neq 1$, we need to approximate the entropy density and consequently also the diffusion matrix. The approximate entropy density

$$h_{\varepsilon}(u) = h(u) + \varepsilon \sum_{i=1}^{n} \left(u_i (\log u_i - 1) + 1 \right)$$
(6.2.4)

possesses the property that the range of its derivative is \mathbb{R}^n . We set $H(u) = h''(u) = (\delta_{ij} s \pi_i u_i^{s-2})_{i,j=1,\dots,n}$ for its Hessian and

$$H_{\varepsilon}(u) = H(u) + \varepsilon H^{0}(u), \quad H_{ij}^{0}(u) = \delta_{ij} u_{i}^{-1},$$

$$A_{\varepsilon}(u) = A(u) + \varepsilon A^{0}(u) + \varepsilon^{\eta} A^{1}(u), \tag{6.2.5}$$

where $\eta < 1/2$ and

$$A_{ij}^{0}(u) = \delta_{ij} \frac{u_{i}}{\pi_{i}} \mu_{i} - (1 - \delta_{ij}) \frac{u_{i}}{\pi_{i}} a_{ji}, \quad A_{ij}^{1}(u) = \delta_{ij} u_{i},$$
$$\mu_{i} := \frac{\pi_{i}}{2} \sum_{j=1, j \neq i}^{n} \left(\frac{a_{ji}}{\pi_{i}} + \frac{a_{ij}}{\pi_{j}} \right), \quad i = 1, \dots, n.$$

The approximation $\varepsilon^{\eta}A^{1}(u)$ is needed to achieve bounds for $\varepsilon^{(\eta+1)/2}\nabla u_{i}$ in L^{2} , which are necessary for the limit $\varepsilon \to 0$. The off-diagonal terms in $A^{0}(u)$ are needed to preserve the entropy structure in the sense that $H_{\varepsilon}(u)A_{\varepsilon}(u)$ is still positive definite. This is shown in the following lemma.

Lemma 6.11. Let s > 0. Then, for all $z \in \mathbb{R}^n$ and $u \in \mathbb{R}^n_+$,

$$z^{\top} H_{\varepsilon}(u) A_{\varepsilon}(u) z \ge z^{\top} H(u) A(u) z + \varepsilon^{\eta} s \sum_{i=1}^{n} \pi_{i} u_{i}^{s-1} z_{i}^{2} + \varepsilon^{\eta+1} \sum_{i=1}^{n} z_{i}^{2}.$$

Proof. We decompose the product $H_{\varepsilon}(u)A_{\varepsilon}(u)$ as

$$H_{\varepsilon}(u)A_{\varepsilon}(u) = H(u)A(u) + \varepsilon^{\eta}H_{\varepsilon}(u)A^{1}(u) + \varepsilon(H^{0}(u)A(u) + H(u)A^{0}(u))$$
$$+ \varepsilon^{2}H^{0}(u)A^{0}(u).$$

The ε^2 -term becomes

$$(H^{0}(u)A^{0}(u))_{ij} = \sum_{k=1}^{n} \delta_{ik} u_{k}^{-1} \left(\delta_{kj} \frac{u_{k}}{\pi_{k}} \mu_{k} - (1 - \delta_{kj}) \frac{u_{k}}{\pi_{k}} a_{jk} \right)$$
$$= \delta_{ij} \frac{\mu_{i}}{\pi_{i}} - (1 - \delta_{ij}) \frac{a_{ji}}{\pi_{i}}.$$

We obtain for $z \in \mathbb{R}^n$:

$$z^{\top} H^{0}(u) A^{0}(u) z = \sum_{i=1}^{n} \frac{\mu_{i}}{\pi_{i}} z_{i}^{2} - \sum_{i,j=1, i \neq j}^{n} \frac{a_{ji}}{\pi_{i}} z_{i} z_{j}$$

$$\geq \sum_{i=1}^{n} \frac{\mu_{i}}{\pi_{i}} z_{i}^{2} - \frac{1}{2} \sum_{i,j=1, i \neq j}^{n} \frac{a_{ji}}{\pi_{i}} (z_{i}^{2} + z_{j}^{2})$$

$$= \sum_{i=1}^{n} \frac{\mu_{i}}{\pi_{i}} z_{i}^{2} - \frac{1}{2} \sum_{i=1}^{n} \left(\sum_{j=1, j \neq i}^{n} \frac{a_{ji}}{\pi_{i}} \right) z_{i}^{2} - \frac{1}{2} \sum_{i=1}^{n} \left(\sum_{j=1, j \neq i}^{n} \frac{a_{ij}}{\pi_{j}} \right) z_{i}^{2}$$

$$= 0.$$

Next, we consider the ε -terms:

$$(H^{0}(u)A(u))_{ij} = \sum_{k=1}^{n} \delta_{ik} u_{i}^{-1} \left(\delta_{kj} \left(a_{k0} + \sum_{\ell=1}^{n} a_{k\ell} u_{\ell}^{s} + s a_{kk} u_{k}^{s} \right) + (1 - \delta_{kj}) s a_{kj} u_{j}^{s-1} u_{k} \right)$$

$$= \delta_{ij} \left(a_{i0} u_{i}^{-1} + \sum_{\ell=1}^{n} a_{i\ell} u_{\ell}^{s} u_{i}^{-1} + s a_{ii} u_{i}^{s-1} \right) + (1 - \delta_{ij}) s a_{ij} u_{j}^{s-1},$$

$$(H(u)A^{0}(u))_{ij} = \sum_{k=1}^{n} \delta_{ik} s \pi_{i} u_{i}^{s-2} \left(\delta_{kj} \frac{u_{k}}{\pi_{k}} \mu_{k} - (1 - \delta_{kj}) \frac{u_{k}}{\pi_{k}} a_{jk} \right)$$

$$= \delta_{ij} s u_{i}^{s-1} \mu_{i} - (1 - \delta_{ij}) s a_{ji} u_{i}^{s-1}.$$

Summing these expressions and neglecting some positive contributions, we find that

$$z^{\top} (H^{0}(u)A(u) + H(u)A^{0}(u))z \geq \sum_{i=1}^{n} (a_{i0}u_{i}^{-1} + sa_{ii}u_{i}^{s-1})z_{i}^{2}$$

$$+ s \sum_{i,j=1}^{n} (1 - \delta_{ij})a_{ij}u_{j}^{s-1}z_{i}z_{j} - s \sum_{i,j=1}^{n} (1 - \delta_{ij})a_{ji}u_{i}^{s-1}z_{i}z_{j}$$

$$= \sum_{i=1}^{n} (a_{i0}u_{i}^{-1} + sa_{ii}u_{i}^{s-1})z_{i}^{2} \geq s \sum_{i=1}^{n} a_{ii}u_{i}^{s-1}z_{i}^{2}.$$

Here we see how we constructed $A_{ij}^0(u)$: The off-diagonal coefficients are chosen in such a way that the mixed terms in $z_i z_j$ cancel, and the diagonal elements (namely μ_i) are sufficiently large to obtain positive definiteness of $H^0(u)A^0(u)$. Finally, we have $(H_{\varepsilon}(u)A^1(u))_{ij} = \delta_{ij}(s\pi_i u_i^{s-1} + \varepsilon)$ and

$$z^{\top} H_{\varepsilon}(u) A^{1}(u) z = \sum_{i=1}^{n} (s \pi_{i} u_{i}^{s-1} + \varepsilon) z_{i}^{2},$$

which proves the lemma.

6.3. Linear transition rates: proof of Theorem 6.1

In this section, we prove Theorem 6.1. Let T > 0, $N \in \mathbb{N}$, $\tau = T/N$, $\varepsilon > 0$, and $m \in \mathbb{N}$ with m > d/2. This ensures that the embedding $H^m(\Omega) \hookrightarrow L^{\infty}(\Omega)$ is compact. We assume that $u_i^0(x) \in [a,b]$ for $x \in \Omega$, $i = 1, \ldots, n$, where $0 < a < b < \infty$. Then, clearly, $w^0 = 1$

 $h'(u^0) \in L^{\infty}(\Omega; \mathbb{R}^n)$. For general $u_i^0 \geq 0$, we may first consider $u_{\varepsilon}^0 = (Q_{\varepsilon}(u_1^0), \dots, Q_{\varepsilon}(u_n^0))$, where $0 < \varepsilon < 1$ and Q_{ε} is the cut-off function

$$Q_{\varepsilon}(z) = \begin{cases} \varepsilon & \text{for } 0 \le z < \varepsilon, \\ z & \text{for } \varepsilon \le z < \varepsilon^{-1/2}, \\ \varepsilon^{-1/2} & \text{for } z \ge \varepsilon^{-1/2}, \end{cases}$$

and then pass to the limit $\varepsilon \to 0$. We leave the details to the reader.

Step 1: solution of an approximated problem. Given $w^{k-1} \in L^{\infty}(\Omega; \mathbb{R}^n)$ for $k \in \mathbb{N}$, we wish to find $w^k \in H^m(\Omega; \mathbb{R}^n)$ such that

$$\frac{1}{\tau} \int_{\Omega} (u(w^k) - u(w^{k-1})) \cdot \phi dx + \int_{\Omega} \nabla \phi : B(w^k) \nabla w^k dx
+ \varepsilon \int_{\Omega} \left(\sum_{|\alpha| = m} D^{\alpha} w^k \cdot D^{\alpha} \phi + w^k \cdot \phi \right) dx = \int_{\Omega} f(u(w^k)) \cdot \phi dx$$
(6.3.1)

for all $\phi \in H^m(\Omega; \mathbb{R}^n)$. Here, $u(w^k) = (h')^{-1}(w^k)$, $B(w^k) = A(u(w^k))H(u(w^k))^{-1}$, $\alpha = (\alpha_1, \dots, \alpha_d) \in \mathbb{N}_0^n$ with $|\alpha| = \alpha_1 + \dots + \alpha_d = m$ is a multiindex, and $D^{\alpha} = \partial^{|\alpha|}/(\partial x_1^{\alpha_1} \cdots \partial x_d^{\alpha_d})$ is a partial derivative of order m. If k = 1, we define $w^0 = h'(u^0)$. Equation (6.3.1) is an implicit Euler discretization of (6.1.1) including an H^m regularization term.

We recall that the entropy is given by

$$\mathcal{H}[u] = \int_{\Omega} h(u)dx = \int_{\Omega} \sum_{i=1}^{n} \pi_i h_1(u_i) dx, \quad h_1(u_i) = u_i(\log u_i - 1) + 1.$$

Then the entropy variables equal $w_i = \partial h/\partial u_i = \pi_i \log u_i$. In particular, $h': \mathbb{R}^n_+ \to \mathbb{R}^n$ is invertible on \mathbb{R}^n , i.e., Hypothesis (H1) in [94] is satisfied. By Lemmas 6.4 and 6.6, H(u)A(u) is positive definite, i.e., Hypothesis (H2) in [94] holds as well. (At this step, we only need that H(u)A(u) is positive semi-definite.) Furthermore, f_i grows at most linearly which implies that

$$\sum_{i=1}^{n} f_i(u)\pi_i \log u_i \le C_f(1 + h(u)),$$

where $C_f > 0$ depends only on (b_{ij}) and π . This means that Hypothesis (H3) in [94] is also satisfied. Thus, we can apply Lemma 5 in [94] giving a weak solution $w^k \in H^m(\Omega; \mathbb{R}^n)$ to (6.3.1) satisfying the discrete entropy inequality

$$(1 - C_f \tau) \int_{\Omega} h(u(w^k)) dx + \tau \int_{\Omega} \nabla w^k : B(w^k) \nabla w^k dx$$

$$+ \varepsilon \tau \int_{\Omega} \left(\sum_{|\alpha| = m} |D^{\alpha} w^k|^2 + |w^k|^2 \right) dx \le \int_{\Omega} h(u(w^{k-1})) dx + C_f \tau \operatorname{meas}(\Omega).$$

$$(6.3.2)$$

Step 2: uniform estimates. We set $u^k = u(w^k)$ and introduce the piecewise in time constant functions $w^{(\tau)}(x,t) = w^k(x)$ and $u^{(\tau)}(x,t) = u^k(x)$ for $x \in \Omega$, $t \in ((k-1)\tau, k\tau]$.

At time t = 0, we set $w^{(\tau)}(\cdot, 0) = h'(u^0) = w^0$ and $u^{(\tau)}(\cdot, 0) = u^0$. Let $u^{(\tau)} = (u_1^{(\tau)}, \dots, u_n^{(\tau)})$. We define the backward shift operator $(\sigma_\tau u^{(\tau)})(x, t) = u(w^{k-1}(x))$ for $x \in \Omega$, $t \in ((k-1)\tau, k\tau]$. Then $u^{(\tau)}$ solves

$$\frac{1}{\tau} \int_{0}^{T} \int_{\Omega} (u^{(\tau)} - \sigma_{\tau} u^{(\tau)}) \cdot \phi dx dt + \int_{0}^{T} \int_{\Omega} \nabla \phi : B(w^{(\tau)}) \nabla w^{(\tau)} dx dt
+ \varepsilon \int_{0}^{T} \int_{\Omega} \left(\sum_{|\alpha| = m} D^{\alpha} w^{(\tau)} \cdot D^{\alpha} \phi + w^{(\tau)} \cdot \phi \right) dx dt = \int_{0}^{T} \int_{\Omega} f(u^{(\tau)}) \cdot \phi dx dt \quad (6.3.3)$$

for piecewise constant functions $\phi:(0,T)\to H^m(\Omega;\mathbb{R}^n)$. By a density argument, this equation also holds for all $\phi\in L^2(0,T;H^m(\Omega;\mathbb{R}^n))$ [126, Prop. 1.36].

By Lemmas 6.4 and 6.6, we have

$$\nabla w^k : B(w^k) \nabla w^k = \nabla u^k : H(u^k) A(u^k) \nabla u^k \ge 2\eta_0 \sum_{i=1}^n |\nabla u_i^k|^2,$$

where $\eta_0 = \min_{i=1,...,n} \pi_i a_{ii} > 0$ if detailed balance holds, and $\eta_0 > 0$ is given by (6.1.11) otherwise. By the generalized Poincaré inequality [132, Chapter 2, Section 1.4], it holds that

$$\int_{\Omega} \left(\sum_{|\alpha|=m} |D^{\alpha} w^{k}|^{2} + |w^{k}|^{2} \right) dx \ge C_{P} ||w^{k}||_{H^{m}(\Omega)}^{2},$$

where $C_P > 0$ is the Poincaré constant. Then the discrete entropy inequality (6.3.2) gives

$$(1 - C_f \tau) \int_{\Omega} h(u^k) dx + 2\eta_0 \tau \int_{\Omega} |\nabla u^k|^2 dx + \varepsilon C_P \tau ||w^k||^2_{H^m(\Omega)}$$

$$\leq \int_{\Omega} h(u^{k-1}) dx + C_f \tau \operatorname{meas}(\Omega).$$

Summing these inequalities over k = 1, ..., j, it follows that

$$(1 - C_f \tau) \int_{\Omega} h(u^j) dx + 2\eta_0 \tau \sum_{j=1}^k \int_{\Omega} |\nabla u^k|^2 dx + \varepsilon C_P \tau \sum_{j=1}^k ||w^k||^2_{H^m(\Omega)}$$

$$\leq \int_{\Omega} h(u^0) dx + C_f \tau \sum_{k=1}^{j-1} \int_{\Omega} h(u^k) dx + C_f T \text{meas}(\Omega).$$

By the discrete Gronwall inequality [42], if $\tau < 1/C_f$,

$$\int_{\Omega} h(u^j) dx + \tau \sum_{j=1}^k \int_{\Omega} |\nabla u^k|^2 dx + \varepsilon \tau \sum_{j=1}^k ||w^k||_{H^m(\Omega)}^2 \le C,$$

where here and in the following, C > 0 denotes a generic constant independent of τ and ε . Then, observing that the entropy density dominates the L^1 norm and consequently, $u^{(\tau)}$ is uniformly bounded in $L^{\infty}(0,T;L^1(\Omega;\mathbb{R}^n))$, we obtain

$$||u^{(\tau)}||_{L^{\infty}(0,T;L^{1}(\Omega))} + ||u^{(\tau)}||_{L^{2}(0,T;H^{1}(\Omega))} + \varepsilon^{1/2}||w^{(\tau)}||_{L^{2}(0,T;H^{m}(\Omega))} \le C.$$
 (6.3.4)

We wish to derive more a priori estimates. Set $Q_T = \Omega \times (0, T)$. The Gagliardo-Nirenberg inequality with p = 2 + 2/d and $\theta = 2d(p-1)/(dp+2p) \in [0,1]$ (such that $\theta p = 2$) yields for $i = 1, \ldots, n$,

$$||u_{i}^{(\tau)}||_{L^{p}(Q_{T})}^{p} = \int_{0}^{T} ||u_{i}^{(\tau)}||_{L^{p}(\Omega)}^{p} dt \leq C \int_{0}^{T} ||u_{i}^{(\tau)}||_{H^{1}(\Omega)}^{\theta p} ||u_{i}^{(\tau)}||_{L^{1}(\Omega)}^{(1-\theta)p} dt$$

$$\leq C ||u_{i}^{(\tau)}||_{L^{\infty}(0,T;L^{1}(\Omega))}^{(1-\theta)p} ||u_{i}^{(\tau)}||_{L^{2}(0,T;H^{1}(\Omega))}^{\theta p} \leq C.$$
(6.3.5)

In order to apply a compactness result, we need a uniform estimate for the discrete time derivative of $u^{(\tau)}$. Let q=2(d+1) and $\phi\in L^q(0,T;W^{m,q}(\Omega;\mathbb{R}^n))$. Then 1/p+1/q+1/2=1 and, by Hölder's inequality,

$$\begin{split} \frac{1}{\tau} \bigg| \int_{0}^{T} \int_{\Omega} (u^{(\tau)} - \sigma_{\tau} u^{(\tau)}) \cdot \phi dx dt \bigg| &\leq \sum_{i,j=1}^{n} \|A_{ij}(u^{(\tau)})\|_{L^{p}(Q_{T})} \|\nabla u_{j}^{(\tau)}\|_{L^{2}(Q_{T})} \|\nabla \phi_{i}\|_{L^{q}(Q_{T})} \\ &+ \varepsilon \|w^{(\tau)}\|_{L^{2}(0,T;H^{m}(\Omega))} \|\phi\|_{L^{2}(0,T;H^{m}(\Omega))} \\ &+ \|f(u^{(\tau)})\|_{L^{q'}(Q_{T})} \|\phi\|_{L^{q}(Q_{T})}, \end{split}$$

where q' = (2d+2)/(2d+1). Estimate (6.3.5) and the linear growth of $A_{ij}(u^{(\tau)})$ with respect to $u^{(\tau)}$ show that the first term on the right-hand side is bounded. The second term is bounded because of (6.3.4). Finally, $|f_i(u^{(\tau)})|$ is growing at most like $(u_i^{(\tau)})^2$ such that

$$||f(u^{(\tau)})||_{L^{q'}(Q_T)} \le C(1 + ||u^{(\tau)}||_{L^{2q'}(Q_T)}^2) \le C,$$

since $2q' \leq p$. We conclude that

$$\tau^{-1} \| u^{(\tau)} - \sigma_{\tau} u^{(\tau)} \|_{L^{q'}(0,T;W^{m,q}(\Omega)')} \le C.$$
(6.3.6)

Step 3: the limit $(\varepsilon, \tau) \to 0$. In view of (6.3.4) and (6.3.6), we can apply the Aubin-Lions lemma in the version of [61], which yields the existence of a subsequence, which is not relabeled, such that, as $(\tau, \varepsilon) \to 0$,

$$u^{(\tau)} \to u$$
 strongly in $L^2(Q_T)$ and a.e., (6.3.7)

$$u^{(\tau)} \rightharpoonup u \quad \text{weakly in } L^2(0, T; H^1(\Omega)),$$
 (6.3.8)

$$\varepsilon w^{(\tau)} \to 0$$
 strongly in $L^2(0,T;H^m(\Omega)),$ (6.3.9)

$$\tau^{-1}(u^{(\tau)} - \sigma_{\tau}u^{(\tau)}) \rightharpoonup \partial_t u \quad \text{weakly in } L^{q'}(0, T; W^{m, q}(\Omega)'), \tag{6.3.10}$$

where $u = (u_1, \ldots, u_n)$. In view of the a.e. convergence (6.3.7) and the uniform bound (6.3.5), we have

$$u^{(\tau)} \to u$$
 strongly in $L^{\gamma}(Q_T)$ for all $\gamma < 2 + 2/d$. (6.3.11)

Then, together with (6.3.8),

$$u_i^{(\tau)} \nabla u_j^{(\tau)} \rightharpoonup u_i \nabla u_j$$
 weakly in $L^1(Q_T)$.

We deduce from the $L^{q'}(Q_T)$ bound for $A(u^{(\tau)})\nabla u^{(\tau)}$ that

$$B(w^{(\tau)})\nabla w^{(\tau)} = A(u^{(\tau)})\nabla u^{(\tau)} \rightharpoonup A(u)\nabla u$$
 weakly in $L^{q'}(Q_T)$.

Furthermore, taking into account (6.3.11) and the uniform bound for $f_i(u^{(\tau)})$ in $L^{q'}(Q_T)$,

$$f_i(u^{(\tau)}) \rightharpoonup f_i(u)$$
 weakly in $L^{q'}(Q_T)$.

Then (6.3.9) and (6.3.10) allow us to perform the limit $(\varepsilon,\tau) \to 0$ in (6.3.3) with $\phi \in L^q(0,T;W^{m,q}(\Omega))$, which directly yields (6.1.12). Since $\partial_t u = \operatorname{div}(A(u)\nabla u) + f(u) \in L^{q'}(0,T;W^{1,q}(\Omega)')$, a density argument shows that the weak formulation holds for all $\phi \in L^q(0,T;W^{1,q}(\Omega))$. Moreover, $u_i \in W^{1,q'}(0,T;W^{1,q}(\Omega)') \hookrightarrow C^0([0,T];W^{1,q}(\Omega)')$, which shows that the initial condition is satisfied in $W^{1,q}(\Omega)'$. This ends the proof.

Remark 6.12 (Detailed balance and vanishing self-diffusion). In the case of detailed balance, we may allow for vanishing self-diffusion. If $a_{ii} = 0$ but $a_{i0} > 0$, Lemma 6.4 implies that only $\nabla (u_i^{(\tau)})^{1/2}$ is bounded in $L^2(Q_T)$. This situation was considered in [38] for the two-species case, and we sketch the generalization to the n-species case.

Applying the Gagliardo-Nirenberg inequality similarly as in Step 2 of the previous proof, we conclude that $(u_i^{(\tau)})^{1/2} \in L^{\widetilde{p}}(Q_T)$ with $\widetilde{p} = 2 + 4/d$. Then

$$\|\nabla u_i^{(\tau)}\|_{L^{\widetilde{q}}(Q_T)} = 2\|(u_i^{(\tau)})^{1/2}\|_{L^{\widetilde{p}}(Q_T)} \|\nabla (u_i^{(\tau)})^{1/2}\|_{L^2(Q_T)} \le C, \quad \widetilde{q} = \frac{d+2}{d+1},$$

and thus, $(u_i^{(\tau)})$ is bounded in $L^{\widetilde{q}}(0,T;W^{1,\widetilde{q}}(\Omega))$ instead of $L^2(0,T;H^1(\Omega))$. This loss of regularity is problematic for the estimate of the discrete time derivative of $u_i^{(\tau)}$. In order to compensate this, we need the last sum in (6.2.3). Indeed, Remark 6.5 shows that for any $i \neq j$, $(u_i^{(\tau)}u_j^{(\tau)})^{1/2}$ is bounded in $L^2(0,T;H^1(\Omega))$. Moreover, $(u_i^{(\tau)}u_j^{(\tau)})^{1/2}$ is bounded in $L^\infty(0,T;L^1(\Omega))$. We infer from the Gagliardo-Nirenberg inequality that $(u_i^{(\tau)}u_j^{(\tau)})^{1/2}$ is bounded in $L^p(Q_T)$ with p=2+2/d.

Next we exploit the structure of the equations,

$$\sum_{j=1}^{n} A_{ij}(u^{(\tau)}) \nabla u_j^{(\tau)} = \nabla (u_i^{(\tau)} p_i(u^{(\tau)})), \quad p_i(u^{(\tau)}) = a_{i0} + \sum_{j=1}^{n} a_{ij} u_j^{(\tau)}.$$

Thus, to show that $A_{ij}(u^{(\tau)})\nabla u_j^{(\tau)}$ is bounded, we only need to verify that $\nabla(u_i^{(\tau)}u_j^{(\tau)})$ is bounded:

$$\|\nabla (u_i^{(\tau)}u_j^{(\tau)})\|_{L^{q'}(Q_T)} \leq 2\|(u_i^{(\tau)}u_j^{(\tau)})^{1/2}\|_{L^p(Q_T)}\|\nabla (u_i^{(\tau)}u_j^{(\tau)})^{1/2}\|_{L^2(Q_T)} \leq C,$$

where q'=(2d+2)/(2d+1). The estimate for the Lotka-Volterra term is more delicate since we have only the regularity $u_i^{(\tau)} \in L^{1+1/d}(Q_T)$. Here, we need to suppose that $b_{ii}>0$, since this assumption provides an estimate for $(u_i^{(\tau)})^2 \log u_i^{(\tau)}$ in $L^1(Q_T)$. Then the discrete time derivative of $u_i^{(\tau)}$ is bounded in $L^1(0,T;W^{m,q}(\Omega)')$ – but not in $L^{q'}(0,T;W^{m,q}(\Omega)')$. By the Aubin-Lions lemma, there exists a subsequence (not relabeled) such that, as $(\varepsilon,\tau)\to 0$,

$$u_i^{(\tau)} \to u_i$$
 strongly in $L^{q'}(Q_T)$.

The problem now is to show that (a subsequence of) the discrete time derivative of $u_i^{(\tau)}$ converges to $\partial_t u_i$ since $L^1(0,T;W^{m,q}(\Omega)')$ is not reflexive. The idea is to apply a result from [142] which provides a criterium for weak compactness in $L^1(0,T;X)$, where X is a reflexive Banach space. For details, we refer to [38].

6.4. Nonlinear transition rates: proof of Theorem 6.2

The strategy of the proof is similar to the proof of Theorem 6.1 but the nonlinear transition rates complicate the proof significantly. As outlined in Section 6.2.3, we approximate the entropy density by (6.2.4) and the diffusion matrix by (6.2.5). Again, we assume without loss of generality that $u_i^0(x) \in [a, b]$ for $x \in \Omega$, $i = 1, \ldots, n$, where $0 < a < b < \infty$.

Step 1: solution of an approximated problem. We employ the transformation $w_i = \partial h_{\varepsilon}/\partial u_i$ and define $B_{\varepsilon}(w) = A_{\varepsilon}(u(w))H_{\varepsilon}(u(w))^{-1}$. Given $w^{k-1} \in L^{\infty}(\Omega; \mathbb{R}^n)$, we wish to find $w^k \in H^m(\Omega; \mathbb{R}^n)$ solving

$$\frac{1}{\tau} \int_{\Omega} \left(u(w^k) - u(w^{k-1}) \right) \cdot \phi dx + \int_{\Omega} \nabla \phi : B_{\varepsilon}(w^k) \nabla w^k dx
+ \varepsilon \int_{\Omega} \left(\sum_{|\alpha| = m} D^{\alpha} w^k \cdot D^{\alpha} \phi + w^k \cdot \phi \right) dx = \int_{\Omega} f(u(w^k)) \cdot \phi dx$$
(6.4.1)

for all $\phi \in H^m(\Omega; \mathbb{R}^n)$. If k = 1, we define $w^0 = h_{\varepsilon}'(u^0)$ such that $u(w^0) = u^0$.

The construction of h_{ε} ensures that Hypothesis (H1) of [94] is satisfied. By Lemma 6.11, Hypothesis (H2) holds as well. Also Hypothesis (H3) holds true since, for some $C_f > 0$,

$$f(u) \cdot w = \sum_{i=1}^{n} \left(b_{i0} - \sum_{j=1}^{n} b_{ij} u_j^{\sigma} \right) (s u_i^s + \varepsilon u_i \log u_i) \le C_f (1 + h_{\varepsilon}(u)),$$

where $\sigma = 1$ if s > 1 and $0 \le \sigma \le \max\{0, 2s - 1 + 2/d\}$ if s < 1. We apply Lemma 5 in [94] to deduce the existence of a weak solution $w^k \in H^m(\Omega; \mathbb{R}^n)$ to the above problem, which satisfies the discrete entropy inequality

$$(1 - C_f \tau) \int_{\Omega} h_{\varepsilon}(u(w^k)) dx + \tau \int_{\Omega} \nabla w^k : B_{\varepsilon}(w^k) \nabla w^k dx$$

$$+ \varepsilon \tau \int_{\Omega} \left(\sum_{|\alpha| = m} |D^{\alpha} w^k|^2 + |w^k|^2 \right) dx \le \int_{\Omega} h_{\varepsilon}(u(w^{k-1})) dx + C_f \tau \operatorname{meas}(\Omega).$$

$$(6.4.2)$$

Setting $u^k := u(w^k)$ and employing Lemma 6.11, the second integral can be estimated as follows:

$$\begin{split} \int_{\Omega} \nabla w^k : B_{\varepsilon}(w^k) \nabla w^k dx &= \int_{\Omega} u^k : H_{\varepsilon}(u^k) A_{\varepsilon}(u^k) \nabla u^k dx \\ &\geq s(s+1) \int_{\Omega} \sum_{i=1}^n \min\{a_{ii} \pi_i, \eta_0, \eta_1 \pi_i, \eta_2\} (u_i^k)^{2(s-1)} |\nabla u_i^k|^2 dx \end{split}$$

$$\begin{split} & + \varepsilon^{\eta} s \int_{\Omega} \sum_{i=1}^{n} \pi_{i}(u_{i}^{k})^{s-1} |\nabla u_{i}^{k}|^{2} dx + \varepsilon^{\eta+1} \int_{\Omega} \sum_{i=1}^{n} |\nabla u_{i}^{k}|^{2} dx \\ & \geq C_{s} \int_{\Omega} \sum_{i=1}^{n} |\nabla (u_{i}^{k})^{s}|^{2} dx \\ & + \frac{4\varepsilon^{\eta} s}{(s+1)^{2}} \int_{\Omega} \sum_{i=1}^{n} \pi_{i} |\nabla (u_{i}^{k})^{(s+1)/2}|^{2} dx + \varepsilon^{\eta+1} \int_{\Omega} \sum_{i=1}^{n} |\nabla u_{i}^{k}|^{2} dx, \end{split}$$

where $C_s = s^{-1}(s+1) \min\{a_{11}\pi_1, \dots, a_{nn}\pi_n, \eta_0, \eta_1\pi_1, \dots, \eta_1\pi_n, \eta_2\}.$ To finish this step, we wish to write the "very weak" formulation for the solution $u^{(\tau)}$, which is defined from u^k as in the previous section. First, we observe that

$$(B_{\varepsilon}(w^k)\nabla w^k)_i = (A_{\varepsilon}(u^k)\nabla u^k)_i = \varepsilon(A^0(u^k)\nabla u^k)_i + \varepsilon^{\eta}(A^1(u^k)\nabla u^k)_i + \nabla(u_i^k p_i(u^k))$$
$$= \varepsilon(A^0(u^k)\nabla u^k)_i + \frac{\varepsilon^{\eta}}{2}\nabla(u_i^k)^2 + \nabla(u_i^k p_i(u^k)).$$

Next, we choose a test function $\phi = (\phi_1, \dots, \phi_n) \in L^q(0,T; W_{\nu}^{m,q}(\Omega))$, where m > 1 $\max\{1,d/2\}$ and $q\geq 2$ will be determined below. Recall that $W^{m,q}_{\nu}(\Omega)$ is defined in (6.1.16). Integrating by parts in (6.4.1), $u^{(\tau)}$ solves

$$\frac{1}{\tau} \int_{0}^{T} \int_{\Omega} \left(u^{(\tau)} - \sigma_{\tau} u^{(\tau)} \right) \cdot \phi dx dt - \int_{0}^{T} \int_{\Omega} \sum_{i=1}^{n} u_{i}^{(\tau)} p_{i}(u^{(\tau)}) \Delta \phi_{i} dx dt
+ \varepsilon \int_{0}^{T} \int_{\Omega} \nabla \phi : A^{0}(u^{(\tau)}) \nabla u^{(\tau)} dx dt - \frac{\varepsilon^{\eta}}{2} \int_{0}^{T} \int_{\Omega} \sum_{i=1}^{n} (u_{i}^{(\tau)})^{2} \Delta \phi_{i} dx dt
+ \varepsilon \int_{0}^{T} \int_{\Omega} \left(\sum_{|\alpha|=m} D^{\alpha} w^{(\tau)} \cdot D^{\alpha} \phi + w^{(\tau)} \cdot \phi \right) dx dt = \int_{0}^{T} \int_{\Omega} f(u^{(\tau)}) \cdot \phi dx dt.$$
(6.4.4)

Step 2: uniform estimates. Arguing as in Step 2 of the proof of Theorem 6.1, we obtain from (6.4.2) and (6.4.3) for sufficiently small $\tau > 0$ the following uniform estimates.

Lemma 6.13. It holds for i = 1, ..., n that

$$||u_i^{(\tau)}||_{L^{\infty}(0,T;L^{\max\{1,s\}}(\Omega))} + ||(u_i^{(\tau)})^s||_{L^2(0,T;H^1(\Omega))} \le C, \tag{6.4.5}$$

$$\varepsilon^{\eta/2} \| (u_i^{(\tau)})^{(s+1)/2} \|_{L^2(0,T;H^1(\Omega))} + \varepsilon^{(\eta+1)/2} \| u_i^{(\tau)} \|_{L^2(0,T;H^1(\Omega))} \le C, \tag{6.4.6}$$

$$\varepsilon^{1/2} \| w_i^{(\tau)} \|_{L^2(0,T;H^m(\Omega))} \le C.$$
 (6.4.7)

Here, we used the fact that $\int_{\Omega} h_{\varepsilon}(u^0) dx$ is uniformly bounded and that s < 1 implies that $u^{(\tau)} \leq C(1+h(u^{(\tau)}))$ for some C>0, from which we deduce that $(u_i^{(\tau)})$ is bounded in $L^{\infty}(0,T;L^{1}(\Omega))$. We need more a priori estimates.

Lemma 6.14. Let $s > \max\{0, 1 - 2/d\}$. It holds that

$$||u^{(\tau)}||_{L^{p(s)}(Q_T)} + \varepsilon^{\eta/r(s)}||u^{(\tau)}||_{L^{r(s)}(Q_T)} \le C, \tag{6.4.8}$$

where $p(s) = 2s + (2/d) \max\{1, s\}$ and $r(s) = s + 1 + (2/d) \max\{1, s\} > 2$.

Proof. The estimates are consequences of Lemma 6.13 and the Gagliardo-Nirenberg inequality. First, let s < 1. We employ the Gagliardo-Nirenberg inequality, with $\theta = ds/(ds+1) \in (0,1)$:

$$\begin{aligned} \|u_i^{(\tau)}\|_{L^{p(s)}(Q_T)}^{p(s)} &= \int_0^T \|(u_i^{(\tau)})^s\|_{L^{p(s)/s}(\Omega)}^{p(s)/s} dt \le C \int_0^T \|(u_i^{(\tau)})^s\|_{H^1(\Omega)}^{\theta p(s)/s} \|(u_i^{(\tau)})^s\|_{L^{1/s}(\Omega)}^{(1-\theta)p(s)/s} dt \\ &\le C \|u_i^{(\tau)}\|_{L^{\infty}(0,T;L^1(\Omega))}^{(1-\theta)p(s)} \int_0^T \|(u_i^{(\tau)})^s\|_{H^1(\Omega)}^{\theta p(s)/s} dt, \quad i = 1,\dots, n. \end{aligned}$$

It holds that $\theta p(s)/s = 2$. By (6.4.5), $\|u^{(\tau)}\|_{L^{p(s)}(Q_T)} \leq C$. Next, let s > 1. Then, with $\theta = d/(d+1) \in (0,1)$,

$$\|(u_i^{(\tau)})^s\|_{L^{2+2/d}(Q_T)}^{2+2/d} \le C \int_0^T \|(u_i^{(\tau)})^s\|_{H^1(\Omega)}^{\theta(2+2/d)} \|(u_i^{(\tau)})^s\|_{L^1(\Omega)}^{(1-\theta)(2+2/d)} dt$$

$$\le C \|(u_i^{(\tau)})^s\|_{L^2(0,T;H^1(\Omega))}^2 \|u_i^{(\tau)}\|_{L^\infty(0,T;L^s(\Omega))}^{s(1-\theta)(2+2/d)} \le C,$$

again taking into account estimate (6.4.5). This shows that $(u^{(\tau)})$ is bounded in $L^{p(s)}(Q_T)$. Finally, let $\max\{0, 1-2/d\} < s < 1$. Then r(s) = s+1+2/d. We apply the Gagliardo-Nirenberg inequality with $\theta = d(s+1)/(2+d(s+1)) \in (0,1)$ such that $\theta \cdot 2r(s)/(s+1) = 2$,

$$\begin{split} \varepsilon^{\eta} \|u_i^{(\tau)}\|_{L^{r(s)}(Q_T)}^{r(s)} &= \varepsilon^{\eta} \|(u_i^{(\tau)})^{(s+1)/2}\|_{L^{2r(s)/(s+1)}(Q_T)}^{2r(s)/(s+1)} \\ &\leq \varepsilon^{\eta} C \int_0^T \|(u_i^{(\tau)})^{(s+1)/2}\|_{H^1(\Omega)}^{2r(s)\theta/(s+1)} \|(u_i^{(\tau)})^{(s+1)/2}\|_{L^{2/(s+1)}(\Omega)}^{2r(s)(1-\theta)/(s+1)} dt \\ &\leq C \varepsilon^{\eta} \|(u_i^{(\tau)})^{(s+1)/2}\|_{L^2(0,T;H^1(\Omega))}^2 \|u_i^{(\tau)}\|_{L^{\infty}(0,T;L^1(\Omega))}^{(1-\theta)r(s)} \leq C, \end{split}$$

using (6.4.5) and (6.4.6). If s > 1, we have r(s) = s + 1 + 2s/d, and applying the Gagliardo-Nirenberg inequality with $\theta = d(s+1)/(2s+d(s+1)) \in (0,1)$, we obtain in a similar way as above

$$\varepsilon^{\eta} \| u_i^{(\tau)} \|_{L^{r(s)}(Q_T)}^{r(s)} = \varepsilon^{\eta} \| (u_i^{(\tau)})^{(s+1)/2} \|_{L^{2r(s)/(s+1)}(Q_T)}^{2r(s)/(s+1)} \\
\leq \varepsilon^{\eta} C \int_0^T \| (u_i^{(\tau)})^{(s+1)/2} \|_{H^1(\Omega)}^{2r(s)\theta/(s+1)} \| (u_i^{(\tau)})^{(s+1)/2} \|_{L^{2s/(s+1)}(\Omega)}^{2r(s)(1-\theta)/(s+1)} dt \\
\leq C \varepsilon^{\eta} \| (u_i^{(\tau)})^{(s+1)/2} \|_{L^2(0,T;H^1(\Omega))}^2 \| u_i^{(\tau)} \|_{L^{\infty}(0,T;L^s(\Omega))}^{(1-\theta)r(s)} \leq C.$$

This shows the lemma.

Lemma 6.15. Let $s > \max\{0, 1-2/d\}$ and $m > \max\{1, d/2\}$. Then there exist $2 \le q < \infty$ and C > 0 such that

$$\tau^{-1} \| u^{(\tau)} - \sigma_{\tau} u^{(\tau)} \|_{L^{q'}(0,T;W^{m,q}(\Omega)')} \le C, \tag{6.4.9}$$

 \Box

and 1/q + 1/q' = 1.

Proof. Let $\phi \in L^q(0,T;W^{m,q}_{\nu}(\Omega))$, where $q \geq 2$ has to be determined. Recall that $W^{m,q}_{\nu}(\Omega)$ is defined in (6.1.16) and that $m > \max\{1, d/2\}$. Then, by (6.4.4),

$$\tau^{-1} \left| \int_{\Omega} (u^{(\tau)} - \sigma_{\tau} u^{(\tau)}) \cdot \phi dx \right| \le \varepsilon \|w^{(\tau)}\|_{L^{2}(0,T;H^{m}(\Omega))} \|\phi\|_{L^{2}(0,T;H^{m}(\Omega))}$$

$$+ \sum_{i=1}^{n} \|u_{i}^{(\tau)} p_{i}(u^{(\tau)})\|_{L^{q'}(Q_{T})} \|\Delta \phi_{i}\|_{L^{q}(Q_{T})} + \varepsilon \sum_{i,j=1}^{n} \|A_{ij}^{0}(u^{(\tau)}) \nabla u_{j}^{(\tau)}\|_{L^{q'}(Q_{T})} \|\nabla \phi_{j}\|_{L^{q}(Q_{T})}$$

$$+ \frac{\varepsilon^{\eta}}{2} \sum_{i=1}^{n} \|(u_{i}^{(\tau)})^{2}\|_{L^{q'}(Q_{T})} \|\Delta \phi_{i}\|_{L^{q}(Q_{T})} + \|f(u^{(\tau)})\|_{L^{q'}(Q_{T})} \|\phi\|_{L^{q}(Q_{T})}$$

$$=: I_{1} + \dots + I_{5},$$

$$(6.4.10)$$

where 1/q + 1/q' = 1.

By (6.4.7), I_1 is bounded. We deduce from (6.4.8) that $u_i^{(\tau)}(u_j^{(\tau)})^s$ is uniformly bounded in $L^{p(s)/(s+1)}(Q_T)$, and so does $u_i^{(\tau)}p_i(u^{(\tau)})$. As s>1-2/d, we have $q_1:=p(s)/(s+1)>1$. We conclude that I_2 is bounded with $q'\leq \min\{2,q_1\}$.

Since $A_{ij}^0(u^{(\tau)})$ depends linearly on $u^{(\tau)}$, it is sufficient to prove that $\varepsilon u_i^{(\tau)} \nabla u_j^{(\tau)}$ is uniformly bounded in some $L^{q_2}(Q_T)$ for all i,j. Let $q_2 = 2r(s)/(r(s)+2)$, where r(s) = s+1+2/d is defined in Lemma 6.14. As r(s) > 2, it holds that $q_2 > 1$. Then, by Hölder's inequality, (6.4.6), and (6.4.8),

$$\varepsilon^{\eta/r(s)+(\eta+1)/2}\|u_i^{(\tau)}\nabla u_i^{(\tau)}\|_{L^{q_2}(Q_T)}\leq \varepsilon^{\eta/r(s)}\|u_i^{(\tau)}\|_{L^{r(s)}(Q_T)}\cdot \varepsilon^{(\eta+1)/2}\|\nabla u_i^{(\tau)}\|_{L^2(Q_T)}\leq C.$$

The property r(s) > 2 also implies that $\eta/r(s) + (\eta + 1)/2 < 1$. This shows the bound on I_3 with $q' \le \min\{2, q_2\}$.

Set $q_3 = r(s)/2 > 1$. Using the second estimate in (6.4.8) and 1 - 2/r(s) > 0, we find that

$$\varepsilon^{\eta} \| (u_i^{(\tau)})^2 \|_{L^{q_3}(Q_T)} = \varepsilon^{(1 - 2/r(s))\eta} \big(\varepsilon^{\eta/r(s)} \| u_i^{(\tau)} \|_{L^{r(s)}(Q_T)} \big)^2 \leq C,$$

proving that I_4 is bounded with $q' \leq \min\{2, q_3\}$.

Finally, in view of (6.1.13), $|f_i(u^{(\tau)})|$ grows at most like $(u_i^{(\tau)})^{1+\sigma}$, where $\sigma=1$ if s>1 and $\sigma<2s-1+2/d$ if s<1. Therefore, we have $q_4:=p(s)/(1+\sigma)>1$ and

$$||f(u^{(\tau)})||_{L^{q_4}(Q_T)} \le C(1 + ||u^{(\tau)}||_{L^{(1+\sigma)q_4}(Q_T)}^{1+\sigma}) = C(1 + ||u^{(\tau)}||_{L^{p(s)}(Q_T)}^{1+\sigma}) \le C.$$

Hence, I_5 is bounded with $q' \leq \min\{2, q_4\}$. We conclude that the lemma follows with $q' := \min\{2, q_1, q_2, q_3, q_4\} > 1$ and q = q'/(q'-1).

Step 3: the limit $(\varepsilon, \tau) \to 0$. Estimates (6.4.5) and (6.4.9) allow us to apply the nonlinear Aubin-Lions lemma (Theorem 6.21 if $s \ge 1/2$ or Theorem 6.22 if s < 1/2) to obtain the existence of a subsequence which is not relabeled such that, as $(\varepsilon, \tau) \to 0$,

$$u^{(\tau)} \to u$$
 strongly in $L^{\gamma}(Q_T)$ for all $1 \le \gamma < p(s)$.

In particular, $u^{(\tau)} \to u$ a.e. in Q_T . By estimates (6.4.5), (6.4.7), and (6.4.9), we have, up to subsequences,

$$\begin{split} (u_i^{(\tau)})^s &\rightharpoonup u_i^s \quad \text{weakly in } L^2(0,T;H^1(\Omega)), \\ \varepsilon w^{(\tau)} &\to 0 \quad \text{strongly in } L^2(0,T;H^m(\Omega)), \\ \tau^{-1}(u^{(\tau)} - \sigma_\tau u^{(\tau)}) &\rightharpoonup \partial_t u \quad \text{weakly in } L^{q'}(0,T;W^{m,q}(\Omega)'). \end{split}$$

We have shown in the proof of Lemma 6.15 that $(u_i^{(\tau)}p_i(u^{(\tau)}))$ is bounded uniformly in $L^{p(s)/(s+1)}(Q_T)$. Taking into account the a.e. convergence $u_i^{(\tau)}p_i(u^{(\tau)}) \to u_ip_i(u)$ in Q_T , we infer that

$$u_i^{(\tau)} p_i(u^{(\tau)}) \to u_i p_i(u)$$
 strongly in $L^1(Q_T)$.

Furthermore, we proved that $(\varepsilon^{\eta/r(s)+(\eta+1)/2}A^0_{ij}(u^{(\tau)})\nabla u^{(\tau)}_j)$ is bounded in $L^{q_2}(Q_T)$ with $q_2=2r(s)/(r(s)+2)$ such that

$$\begin{split} \varepsilon A^0_{ij}(u^{(\tau)}) \nabla u^{(\tau)}_j &= \varepsilon^{1-\eta/r(s)-(\eta+1)/2} \cdot \varepsilon^{\eta/r(s)+(\eta+1)/2} A^0_{ij}(u^{(\tau)}) \nabla u^{(\tau)}_j \\ &\to 0 \quad \text{strongly in } L^1(Q_T). \end{split}$$

Here, we used the fact that $\eta/r(s) + (\eta+1)/2 < 1$ such that $\varepsilon^{1-\eta/r(s)-(\eta+1)/2} \to 0$ as $\varepsilon \to 0$. We know from (6.4.8) that $(\varepsilon^{\eta/r(s)}u_i^{(\tau)})$ is bounded in $L^2(Q_T)$. Consequently,

$$\varepsilon^{\eta}(u_i^{(\tau)})^2 = \varepsilon^{\eta(1-2/r(s))} (\varepsilon^{\eta/r(s)} u_i^{(\tau)})^2 \to 0$$
 strongly in $L^1(Q_T)$,

since $\varepsilon^{\eta(1-2/r(s))} \to 0$ as $\varepsilon \to 0$ because of r(s) > 2. Finally, $f_i(u^{(\tau)}) \to f_i(u)$ a.e. and the uniform bound $||f_i(u^{(\tau)})||_{L^{q_4}(Q_T)} \le C$ with $q_4 = p(s)/(1+\sigma) > 1$ imply that

$$f_i(u^{(\tau)}) \to f_i(u)$$
 strongly in $L^1(Q_T)$.

Then, performing the limit $(\varepsilon, \tau) \to 0$ in (6.4.4) with $\phi \in L^{\infty}(0, T; W_{\nu}^{m,\infty}(\Omega))$, it follows that u solves (6.1.17) for such test functions. A density argument shows that, in fact, u solves (6.1.17) for $\phi \in L^{q}(0, T; W_{\nu}^{m,q}(\Omega))$, finishing the proof.

Remark 6.16 (Weak formulation). In the superlinear case s > 1, the solution constructed in the previous proof satisfies (6.1.1) even in the weak sense (6.1.12) with test functions $\phi \in L^q(0,T;W^{1,q}(\Omega))$. In order to see this, it is sufficient to show that

$$A_{ij}(u^{(\tau)})\nabla u_j^{(\tau)} \rightharpoonup A_{ij}(u)\nabla u_j$$
 weakly in $L^{q'}(Q_T)$

for some $1 < q' \le 2$. Because of the structure of A_{ij} , we only need to verify that

$$u_i^{(\tau)}(u_j^{(\tau)})^{s-1} \nabla u_j^{(\tau)} \rightharpoonup u_i u_j^{s-1} \nabla u_j \quad \text{weakly in } L^{q'}(Q_T),$$
$$(u_i^{(\tau)})^s \nabla u_j^{(\tau)} \rightharpoonup u_i^s \nabla u_j \quad \text{weakly in } L^{q'}(Q_T).$$

Indeed, we have the convergences $u_i^{(\tau)} \to u_i$ strongly in $L^{\gamma}(Q_T)$ for any $2 < \gamma < p(s)$ and $(u_i^{(\tau)})^s \rightharpoonup u_i^s$ weakly in $L^2(0,T;H^1(\Omega))$ and hence,

$$u_i^{(\tau)}(u_j^{(\tau)})^{s-1} \nabla u_j^{(\tau)} = s^{-1} u_i^{(\tau)} \nabla (u_j^{(\tau)})^s \rightharpoonup s^{-1} u_i \nabla u_j^s = u_i u_j^{s-1} \nabla u_j \quad \text{weakly in } L^{q'}(Q_T),$$

choosing $q' = 2\gamma/(\gamma+2) > 1$. For the remaining convergence, we need to integrate by parts. It holds for $\phi_i \in L^q(0,T;W^{2,q}_{\nu}(\Omega))$ that

$$\int_{0}^{T} \int_{\Omega} (u_{i}^{(\tau)})^{s} \nabla u_{j}^{(\tau)} \cdot \nabla \phi_{i} dx dt$$

$$= -\int_0^T \int_{\Omega} u_j^{(\tau)} \nabla (u_i^{(\tau)})^s \cdot \nabla \phi_i dx dt - \int_0^T \int_{\Omega} (u_i^{(\tau)})^s u_j^{(\tau)} \Delta \phi_i dx dt$$

$$\to -\int_0^T u_j \nabla u_i^s \cdot \nabla \phi_i dx dt - \int_0^T \int_{\Omega} u_i^s u_j \Delta \phi_i dx dt = \int_0^t \int_{\Omega} u_i^s \nabla u_j \cdot \nabla \phi_i dx dt.$$

A density argument shows that the weak formulation also holds for $\phi_i \in L^q(0,T;W^{1,q}(\Omega))$.

Remark 6.17 (Vanishing self-diffusion). Assume that $a_{i0} > 0$ and $a_{ii} = 0$. The difficulty is that we obtain a uniform bound only for $\nabla (u_i^{(\tau)})^{s/2}$ instead for $\nabla (u_i^{(\tau)})^s$ in $L^2(Q_T)$. In order to compensate this loss of regularity, we need additional assumptions, namely either $s > \max\{1, d/2\}$ (superlinear rates); or 0 < s < 1, d = 1, and $\sigma < s + 1$ (sublinear rates). Under these conditions, the statement of Theorem 6.2 holds true.

For the proof, we remark that the regularity for $(u_i^{(\tau)})^s$ in $L^2(0,T;H^1(\Omega))$ is employed in the estimate of $u_i^{(\tau)}$ in $L^{p(s)}(Q_T)$. If only $(u_i^{(\tau)})^{s/2}$ is bounded in $L^2(0,T;H^1(\Omega))$, the Gagliardo-Nirenberg inequality gives a weaker result: for 0 < s < 1 with $\theta = ds/(ds+2)$ and $\rho = s + 2/d$,

$$\begin{aligned} \|u_i^{(\tau)}\|_{L^{\rho}(Q_T)}^{\rho} &= \|(u_i^{(\tau)})^{s/2}\|_{L^{2\rho/s}(Q_T)}^{2\rho/s} \leq C \int_0^T \|(u_i^{(\tau)})^{s/2}\|_{H^1(\Omega)}^{2\theta\rho/s} \|(u_i^{(\tau)})^{s/2}\|_{L^{2/s}(\Omega)}^{2(1-\theta)\rho/s} dt \\ &\leq C \|(u_i^{(\tau)})^{s/2}\|_{L^2(0,T;H^1(\Omega))}^2 \|u_i^{(\tau)}\|_{L^{\infty}(0,T;L^1(\Omega))}^{(1-\theta)\rho} \leq C, \end{aligned}$$

since $2\theta \rho/s = 2$; and for s > 1 with $\theta = d/(d+2)$ and $\rho = s + 2s/d$,

$$\|u_i^{(\tau)}\|_{L^\rho(Q_T)}^\rho \le C \|(u_i^{(\tau)})^{s/2}\|_{L^2(0,T;H^1(\Omega))}^2 \|u_i^{(\tau)}\|_{L^\infty(0,T;L^s(\Omega))}^{(1-\theta)\rho} \le C,$$

since $2\theta \rho/s = 2$. Consequently, $(u_i^{(\tau)})$ is bounded in $L^{\rho}(Q_T)$ with $\rho = s + (2/d) \max\{1, s\}$. We claim that this estimate is sufficient to derive a bound for the discrete time derivative. Since the ε -terms in (6.4.10) do not need the estimate for $u_i^{(\tau)}$ in $L^{\rho}(Q_T)$, it is sufficient to bound $u_i^{(\tau)}p_i(u^{(\tau)})$ and $(u_i^{(\tau)})^{\sigma+1}$ in some $L^{q'}(Q_T)$ with q' > 1. This is possible as long as $\rho > s + 1$ and $\rho > \sigma + 1$, respectively. If 0 < s < 1, these two inequalities are equivalent to d = 1 and $\sigma < s - 1 + d/2 = s + 1$. If s > 1 (in this case $\sigma = 1$), they give the restriction s > d/2, thus $s > \max\{1, d/2\}$. This shows the claim.

6.5. Additional and auxiliary results

6.5.1. Detailed balance condition

We wish to interpret the detailed balance condition (6.1.10) and to explain how the numbers π_i can be computed from the coefficients (a_{ij}) . We assume that the coefficients are normalized in the sense that $a_{ij} \geq 0$ and $\sum_{k=1, k\neq j} a_{kj} \leq 1$ for all i, j. The idea is to use a probabilistic approach, interpreting the coefficients a_{ij} as the transition rates between two discrete states i and j of the state space $S := \{1, \ldots, n\}$. Then

$$a_{ij} = P(X_k = j | X_{k-1} = i)$$

is the conditional probability for a random variable $X: \mathbb{N} \to S$. This variable represents the Markov chain associated to the stochastic matrix $Q = (Q_{ij})_{i,j} \in \mathbb{R}^{n \times n}$, defined by $Q_{ij} = a_{ij}$ for $i \neq j$ and $Q_{ii} = 1 - \sum_{i=1, i \neq j} a_{ij}$ for $i = 1, \ldots, n$. A Markov chain is called reversible if there exists a probability distribution $\pi = (\pi_1, \ldots, \pi_n)$ on S (called an invariant measure) such that

$$\pi_i a_{ij} = \pi_j a_{ji}, \quad i, j = 1, \dots, n.$$
 (6.5.1)

The Markov chain can be interpreted as a directed graph, where the states $i \in S$ are the nodes and the edges are labeled by the probabilities a_{ij} going from state i to state j.

The state space S can be partitioned into so-called communicating classes. We write $i \to j$ if there exist $i_0, i_1, \ldots, i_{n+1} \in S$ such that $a_{i_0,i_1}a_{i_1,i_2}\cdots a_{i_n,i_{n+1}} > 0$ for $i_0 = i$ and $i_{n+1} = j$. We say that i communicates with j if both $i \to j$ and $j \to i$. A set of states $\sigma \subset S$ is a communicating class if every pair in σ communicates with each other. This defines an equivalence relation, and communicating classes are the equivalence classes.

Consider the following properties:

- (A1) For all $i, j \in S$, it holds that either $a_{ij} = a_{ji} = 0$ or $a_{ij}a_{ji} > 0$.
- (A2) For any periodic cycle $i_0, i_1, \ldots, i_{m+1} = i_0$,

$$\prod_{k=0}^{m} a_{i_k, i_{k+1}} = \prod_{k=0}^{m} a_{i_{k+1}, i_k}.$$

The detailed balance condition (6.5.1) implies (A1) and (A2). It is shown in [131] that the converse is true and that the invariant measure π can be constructed explicitly.

Proposition 6.18. Let (A1)-(A2) hold. Then there exists an invariant measure $\pi = (\pi_1, \ldots, \pi_n)$ such that the detailed balance condition (6.5.1) is satisfied. Moreover, π can be computed explicitly by choosing an i_0 in each communicating class and defining π_j for i_0 and j belonging in the same class by

$$\pi_j := \prod_{k=1}^{n-1} \frac{a_{i_k, i_{k+1}}}{a_{i_{k+1}, i_k}}$$

depending only on i_0 and j, where $i_1, i_2, \ldots, i_n = j$ are such that $a_{i_k, i_{k+1}} > 0$ for $k = 0, \ldots, n-1$.

For instance, if n = 3, we need to suppose (according to (A2)) that

$$a_{12}a_{23}a_{31} = a_{13}a_{32}a_{21}, (6.5.2)$$

and the invariant measure is given by $\pi = c(1, a_{12}/a_{21}, a_{13}/a_{31})$, where $c = (1 + a_{12}/a_{21} + a_{13}/a_{31})^{-1}$.

The following result relates the detailed balance condition and the symmetry of the matrix H(u)A(u).

Proposition 6.19. The following three properties are equivalent:

- (i) Graph-theoretical condition: (A1) and (A2) hold.
- (ii) Detailed balance condition: $\pi_i a_{ij} = \pi_j a_{ji}$ for $i \neq j$.
- (iii) Symmetry: The matrix H(u)A(u) is symmetric.

Proof. The implication (i) \Leftarrow (ii) is shown in Proposition 6.18. The converse can be proved directly using the detailed balance condition. Finally, the equivalence (ii) \Leftrightarrow (iii) follows from an explicit calculation of H(u)A(u).

Remark 6.20. The equivalence of the symmetry of H(u)A(u) and the detailed balance condition is related to the Onsager principle of thermodynamics. Indeed, the diffusion matrix $B = A(u)H(u)^{-1}$ in

$$\partial_t u - \operatorname{div}(B\nabla w) = f(u),$$

where w = h'(u) is the vector of entropy variables, is the Onsager matrix which is symmetric, according to Onsager, if and only if the thermodynamic system is time-reversible. Time-reversibility means that the Markov chain associated to the matrix (a_{ij}) is reversible, and the symmetry of B is equivalent to the symmetry of H(u)A(u). Thus, the equivalence $(ii) \Leftrightarrow (iii)$ corresponds to the equivalence of the symmetry of B and the time-reversibility. For details on the detailed balance principle in thermodynamics, we refer to [47].

6.5.2. Nonlinear Aubin-Lions lemmas

Let $\Omega \subset \mathbb{R}^d$ $(d \ge 1)$ be a bounded domain with Lipschitz boundary. Let $(u^{(\tau)})$ be a family of nonnegative functions which are piecewise constant in time with uniform time step size $\tau > 0$. We introduce the time shift operator $(\sigma_{\tau}u^{(\tau)})(t) = u^{(\tau)}(t-\tau)$ for $t \ge \tau$.

If there exist uniform estimates for the gradient $(\nabla u^{(\tau)})$ and the discrete time derivative $\tau^{-1}(u^{(\tau)} - \sigma_{\tau}u^{(\tau)})$, then, by the Aubin-Lions theorem and under suitable conditions on the spaces, $(u^{(\tau)})$ is relatively compact in some L^q space. In the case of nonlinear transition rates, we obtain uniform estimates only for $(\nabla(u^{(\tau)})^s)$, where s > 0. Then relative compactness follows from a nonlinear version of the Aubin-Lions theorem [40]. We recall a special case of this result.

Theorem 6.21 (Nonlinear Aubin-Lions lemma for $s \ge 1/2$). Let $s \ge 1/2$, $m \ge 0$, $1 \le q < \infty$, and there exists C > 0 such that for all $\tau > 0$,

$$\|(u^{(\tau)})^s\|_{L^2(0,T;W^{1,q}(\Omega))} + \tau^{-1}\|u^{(\tau)} - \sigma_\tau u^{(\tau)}\|_{L^1(\tau,T;H^m(\Omega)')} \le C.$$

Then there exists a subsequence of $(u^{(\tau)})$, which is not relabeled, such that, as $\tau \to 0$,

$$u^{(\tau)} \to u \quad \text{strongly in } L^{2s}(0,T;L^{ps}(\Omega)),$$

where $p \ge \max\{1, 1/s\}$ is such that the embedding $W^{1,q}(\Omega) \hookrightarrow L^p(\Omega)$ is compact.

Theorem 6.21 can be extended to the case s < 1/2 if $(u^{(\tau)})$ is additionally bounded in $L^{\infty}(0,T;L^{1}(\Omega))$ which generally follows from the entropy inequality. This result is new.

Theorem 6.22 (Nonlinear Aubin-Lions lemma for s < 1/2). Let $\max\{0, 1/2 - 1/d\} < s < 1/2$, $m \ge 0$, and there exists C > 0 such that for all $\tau > 0$,

$$||u^{(\tau)}||_{L^{\infty}(0,T;L^{1}(\Omega))} + ||(u^{(\tau)})^{s}||_{L^{2}(0,T;H^{1}(\Omega))} + \tau^{-1}||u^{(\tau)} - \sigma_{\tau}u^{(\tau)}||_{L^{1}(\tau,T;H^{m}(\Omega)')} \le C.$$

Then there exists a subsequence of $(u^{(\tau)})$, which is not relabeled, such that, as $\tau \to 0$,

$$u^{(\tau)} \to u$$
 strongly in $L^1(0,T;L^1(\Omega))$.

Proof. The result follows from Theorem 6.21 and the Hölder inequality. Indeed, we have

$$\begin{split} \|\nabla(u^{(\tau)})^{1/2}\|_{L^{2}(0,T;L^{1/(1-s)}(\Omega))} &= (2s)^{-1}\|(u^{(\tau)})^{1/2-s}\nabla(u^{(\tau)})^{s}\|_{L^{2}(0,T;L^{1/(1-s)}(\Omega))} \\ &\leq (2s)^{-1}\|(u^{(\tau)})^{(1-2s)/2}\|_{L^{\infty}(0,T;L^{2/(1-2s)}(\Omega))}\|\nabla(u^{(\tau)})^{s}\|_{L^{2}(0,T;L^{2}(\Omega))} \\ &= \|u^{(\tau)}\|_{L^{\infty}(0,T;L^{1}(\Omega))}^{(1-2s)/2}\|\nabla(u^{(\tau)})^{s}\|_{L^{2}(0,T;L^{2}(\Omega))} \leq C. \end{split}$$

Therefore, $(u^{(\tau)})^{1/2}$ is uniformly bounded in $L^2(0,T;W^{1,1/(1-s)}(\Omega))$. By Rellich-Kondrachov's theorem, the embedding $W^{1,1/(1-s)}(\Omega) \hookrightarrow L^2(\Omega)$ is compact for s>0 if $d\leq 2$ and s>1/2-1/d if $d\geq 3$. Applying Theorem 6.21 with s=1/2, q=1/(1-s), and p=2, we infer that $(u^{(\tau)})$ is relatively compact in $L^1(0,T;L^1(\Omega))$.

6.5.3. Increasing entropies

If detailed balance or a weak cross-diffusion condition hold, we have shown that the entropy is nonincreasing in time along solutions to (6.1.1)-(6.1.2). In this section, we show that the entropy may be increasing for small times if these conditions do not hold. To simplify the presentation, we restrict ourselves to the case n=3 (three species), s=1 (linear transition rates), and $\Omega=(0,1)$.

Lemma 6.23 (Vanishing diffusion coefficients a_{i0}). Let $a_{13} = a_{32} = a_{21} = 1$ and $a_{ij} = 0$ else. For any $\varepsilon > 0$, there exist initial data u^0 such that

$$\frac{d\mathcal{H}}{dt}[u^0] \ge \frac{1}{\varepsilon}.$$

In particular, if $t \mapsto \mathcal{H}[u(t)]$ is continuous, there exists $t_0 > 0$ such that $t \mapsto \mathcal{H}[u(t)]$ is increasing on $[0, t_0]$.

Proof. Observe that (6.5.2) is not satisfied, and hence detailed balance does not hold. Furthermore, we have

$$H(u)A(u) = \begin{pmatrix} 1/u_1 & 0 & 0 \\ 0 & 1/u_2 & 0 \\ 0 & 0 & 1/u_3 \end{pmatrix} \begin{pmatrix} u_3 & 0 & u_1 \\ u_2 & u_1 & 0 \\ 0 & u_3 & u_2 \end{pmatrix} = \begin{pmatrix} u_3/u_1 & 0 & 1 \\ 1 & u_1/u_2 & 0 \\ 0 & 1 & u_2/u_3 \end{pmatrix}.$$

Let $0 < \varepsilon < 0.5$ and define $u^0 = (u_1^0, u_2^0, u_3^0)$ by $u_1^0(x) = 1$ for $x \in (0, 1)$ and

$$u_2^0(x) = \begin{cases} 3 & \text{for } 0 < x < 0.5, \\ 3 - \varepsilon^{-1}(x - 0.5) & \text{for } 0.5 < x < 0.5 + \varepsilon, \\ 2 & \text{for } 0.5 + \varepsilon < x < 1, \end{cases}$$

$$u_3^0(x) = \begin{cases} 9 & \text{for } 0 < x < 0.5, \\ 9 + \varepsilon^{-1}(x - 0.5) & \text{for } 0.5 < x < 0.5 + \varepsilon, \\ 10 & \text{for } 0.5 + \varepsilon < x < 1, \end{cases}$$

Then

$$\int_{0}^{1} (\partial_{x} u^{0})^{\top} H(u^{0}) A(u^{0}) \partial_{x} u^{0} dx = \frac{1}{\varepsilon^{2}} \int_{0.5}^{0.5+\varepsilon} \left(\frac{1}{u_{2}^{0}(x)} - 1 + \frac{u_{2}^{0}(x)}{u_{3}^{0}(x)} \right) dx$$
$$\leq \frac{1}{\varepsilon} \left(\frac{1}{2} - 1 + \frac{3}{9} \right) = -\frac{1}{6\varepsilon},$$

which implies that $(d\mathcal{H}/dt)[u^0] \geq 1/(6\varepsilon)$.

One may ask if a similar result as above holds if the diffusion coefficients a_{i0} do not vanish, since they give positive contributions to the entropy production. The next lemma shows that the entropy may be increasing even if $a_{i0} > 0$ is chosen arbitrarily.

Lemma 6.24 (Positive diffusion coefficients a_{i0}). Let $a_{13} = a_{32} = a_{21} = 1$, $a_{i0} > 0$ for i = 1, 2, 3, and $a_{ij} = 0$ else. For any $\varepsilon > 0$, there exist initial data u^0 such that

$$\frac{d\mathcal{H}}{dt}[u^0] \ge \frac{1}{\varepsilon}.$$

In particular, if $t \mapsto \mathcal{H}[u(t)]$ is continuous, there exists $t_0 > 0$ such that $t \mapsto \mathcal{H}[u(t)]$ is increasing on $[0, t_0]$.

Proof. We choose the initial datum

$$u_1^0(x) = \frac{a_{20}(2a_{20} + a_{30})}{8a_{20} + 4a_{30}},$$

$$u_2^0(x) = \begin{cases} 4a_{20} & \text{for } 0 < x < 0.5, \\ a_{20}(4 - \varepsilon^{-1}(x - 0.5)) & \text{for } 0.5 < x < 0.5 + \varepsilon, \\ 3a_{20} & \text{for } 0.5 + \varepsilon < x < 1, \end{cases}$$

$$u_3^0(x) = \begin{cases} 8a_{20} + 4a_{30} & \text{for } 0 < x < 0.5, \\ a_{20}(8 - \varepsilon^{-1}(x - 0.5)) + 4a_{30} & \text{for } 0.5 < x < 0.5 + \varepsilon, \\ 9a_{20} + 4a_{30} & \text{for } 0.5 < x < 0.5 + \varepsilon, \end{cases}$$

Then

$$\begin{split} & \int_0^1 (\partial_x u^0)^\top H(u^0) A(u^0) \partial_x u^0 dx \\ & = \int_{0.5}^{0.5+\varepsilon} \left(\frac{u_1^0}{u_2^0} (\partial_x u_2^0)^2 + \frac{a_{20}}{u_2^0} (\partial_x u_2^0)^2 + \partial_x u_2^0 \partial_x u_3^0 + \frac{u_2^0 + a_{30}}{u_3^0} (\partial_x u_3^0)^2 \right) dx \\ & \leq \frac{a_{20}^2}{\varepsilon^2} \int_{0.5}^{0.5+\varepsilon} \left(\frac{2a_{20} + a_{30}}{3(8a_{20} + 4a_{30})} \frac{a_{20}}{3a_{20}} - 1 + \frac{4a_{20} + a_{30}}{8a_{20} + 4a_{30}} \right) dx \\ & \leq \frac{a_{20}^2}{\varepsilon^2} \left(\frac{1}{12} - \frac{1}{3} - 1 + \frac{1}{2} \right) = -\frac{a_{20}^2}{12\varepsilon}, \end{split}$$

which proves the result.

 $6. \ \ Cross-diffusion \ population \ systems \ for \ multiple \ species$

7. From reaction diffusion to cross diffusion in the fast-reaction limit

7.1. Motivation

Our goal is to derive new entropies to new classes of cross-diffusion systems by using a fast-reaction limit. For a detailed presentation of our strategy, see Section 3.2.

7.1.1. Outline

The first model consists of three species with one fast reaction and general reaction and diffusion functions q_i and f_i , which we will study in Section 7.2. The second model is a "hybrid" reaction-cross diffusion system in four species with linear diffusivities and quadratic reaction terms, studied in Section 7.3.

7.1.2. Notation

We will use the following notation in this chapter:

$$\mathbb{R}_{+} := [0, \infty), \quad \mathbb{R}_{+}^{*} := (0, \infty), \quad \Omega_{T} := \Omega \times (0, T),$$

$$||u_i||_{(L\log L)(\Omega_T)} := \int_0^T \int_{\Omega} u_i \log(u_i) \, dx dt,$$

$$||u_i||_{(L^2(\log L)^2)(\Omega_T)} := \int_0^T \int_{\Omega} (u_i)^2 (\log(u_i))^2 \, dx dt.$$

7.2. Rigorous limit from three species with one fast reaction

In this section we prove the rigorous limit of a quite general quasilinear reaction-diffusion system consisting of three species with one fast reaction to a limiting cross-diffusion system when the fast-reaction rate $1/\varepsilon$ tends to infinity.

7.2.1. Model for three species with one fast reaction

The model

For the concentrations $u_i^{\varepsilon} := u_i^{\varepsilon}(t, x) \ge 0$ of species A_i , i = 1, 2, 3, we consider the following system of reaction-diffusion equations with fast-reaction term

$$\begin{cases}
\partial_t u_1^{\varepsilon} - \Delta_x f_1(u_1^{\varepsilon}) = -\frac{1}{\varepsilon} \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right), \\
\partial_t u_2^{\varepsilon} - \Delta_x f_2(u_2^{\varepsilon}) = +\frac{1}{\varepsilon} \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right), \\
\partial_t u_3^{\varepsilon} - \Delta_x f_3(u_3^{\varepsilon}) = +\frac{1}{\varepsilon} \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right),
\end{cases} (7.2.1)$$

together with initial data

$$u_i^{\varepsilon}(0,x) = u_i^{in}(x) \ge 0, \quad x \in \Omega, \quad i = 1,\ldots,3,$$

and homogeneous Neumann boundary conditions on the boundary $\partial\Omega\in C^2$ of a bounded smooth open subset Ω of \mathbb{R}^N

$$n(x) \cdot \nabla_x u_i^{\varepsilon}(t, x) = 0, \quad x \in \partial\Omega, \quad i = 1, \dots, 3.$$

Limiting cross-diffusion system

Our goal is to study the fast-reaction limit $\varepsilon \to 0$ such that $u_i^{\varepsilon} \to u_i$, with u_i satisfying the limiting cross-diffusion system:

$$\begin{cases} q_1(u_1) - q_2(u_2)q_3(u_3) = 0, \\ \partial_t (u_1 + u_2) - \Delta_x (f_1(u_1) + f_2(u_2)) = 0, \\ \partial_t (u_1 + u_3) - \Delta_x (f_1(u_1) + f_3(u_3)) = 0. \end{cases}$$
(7.2.2)

7.2.2. Existence of solutions

Assumptions for subsection 7.2.2

Our first goal is to prove global existence of weak solutions to system (7.2.1) for fixed $\varepsilon > 0$. To this end, we impose the following set of assumptions for this subsection.

(C1) The functions $f_i : \mathbb{R}_+ \to \mathbb{R}_+$ are in $C^2(\mathbb{R}_+)$ with $f_i(0) = 0$, and there exists a constant $C_1 > 0$ such that

$$f_i'(u_i) \ge C_1 > 0$$
 for all $u_i \ge 0$, $i = 1, 2, 3$.

Consequently,

$$f_i(u_i) \ge \widetilde{C}_1 u_i > 0$$
 for all $u_i \ge 0$, $i = 1, 2, 3$.

(C2) The functions $q_i : \mathbb{R}_+ \to \mathbb{R}_+$ are continuous. Moreover, there exists a constant $C_2 > 0$, such that $q_i(u_i) \leq C_2 \frac{u_i}{1+u_i}$ for all $u_i \geq 0$, i=1,2,3.

(C3) There exists a constant $\mu > 0$, such that $u_i^{in}(x) \ge \mu > 0$, for all $x \in \Omega$, i = 1, 2, 3.

Existence theorem

Theorem 7.1. Assume that (C1)-(C3) are satisfied with $u_i^{in}(x) \in L^{\infty}(\Omega)$, i = 1, 2, 3.

Then for any fixed $\varepsilon > 0$, there exists a global weak solution $u^{\varepsilon} = (u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon})$ to system (7.2.1), satisfying

$$u_i^{\varepsilon} \ge 0$$
, $u_i^{\varepsilon} \in L^2(0,T;H^1(\Omega))$, $u_i^{\varepsilon} \in L^{\infty}(\Omega_T)$, $\partial_t u_i^{\varepsilon} \in L^2(0,T;H^{-1}(\Omega))$,

and u^{ε} satisfies for all $\phi \in C_0^{\infty}([0,T) \times \overline{\Omega}; \mathbb{R}^3)$ and for $\beta = (1,-1,-1)$ that

$$\int_{0}^{T} \int_{\Omega} u^{\varepsilon} \partial_{t} \phi \, dx dt + \int_{0}^{T} \int_{\Omega} \nabla_{x} \phi : f'(u^{\varepsilon}) \nabla_{x} u^{\varepsilon} \, dx dt =$$

$$\frac{(-1)^{\beta}}{\varepsilon} \int_{0}^{T} \int_{\Omega} \left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon}) q_{3}(u_{3}^{\varepsilon}) \right) \phi \, dx dt + \int_{\Omega} u^{in}(x) \phi(0, x) \, dx.$$
(7.2.4)

Proof. Since $\varepsilon > 0$ is fixed, we will omit for simplicity the ε -dependency of u for this proof, and assume without loss of generality that $\varepsilon = 1$ for the right-hand side.

Step 1: solution for an approximated problem. Let $1 \le k \le N$ and T > 0. We consider the time step $\tau = T/N$. Then, for $u_i^{k-1} \in L^{\infty}(\Omega)$ we can introduce the following time-discretized scheme in $\tau > 0$:

Given $u^{k-1}=(u_1^{k-1},u_2^{k-1},u_3^{k-1})\in L^\infty(\Omega;\mathbb{R}^3)$ nonnegative and bounded, find the k-th approximated solution $u^k=(u_1^k,u_2^k,u_3^k)\in L^\infty(\Omega;\mathbb{R}^3)$, such that (in the strong sense) it holds

$$\begin{cases}
\tau^{-1}(u_i^k - u_i^{k-1}) - \Delta_x f_i(u_i^k) = Q_i(u^k) & \text{in } \Omega, \quad 1 \le i \le 3, \\
\nabla_x u_i^k(x) \cdot n(x) = 0 & \text{on } \partial\Omega,
\end{cases} (7.2.5)$$

where $Q_i(u^k) = (-1)^{\beta} (q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon}))$ with $\beta = (1, -1, -1)$ is the *i*-th component of the reaction term on the right-hand side. If k = 1, we set $u_i^{k-1}(x) := u_i^{in}(x)$ for the initial condition.

In order to prove existence of smooth solutions for the scheme, we first show that due to assumptions (C1) - (C3) there exists a constant C > 0 uniform in time step $\tau > 0$, such that

$$u_i^k(x) \ge C > 0$$
, for all $x \in \Omega$, $i = 1, 2, 3$.

To this end, we assume without loss of generality that i=2, and define $\widetilde{x}\in\overline{\Omega}$ such that $u_2^k(\widetilde{x})=\inf_{x\in\overline{\Omega}}u_2^k(x)$. We need to distinguish the cases $\widetilde{x}\in\Omega$ or $\widetilde{x}\in\partial\Omega$. If $\widetilde{x}\in\partial\Omega$, we can use a strong Hopf lemma to obtain that $\nabla_x u_2^k \cdot n(x)>0$, in contradiction to our assumption of homogeneous Neumann boundary conditions. If $\widetilde{x}\in\Omega$, then $\nabla_x u_2^k(\widetilde{x})=0$ and $\Delta_x u_2^k(\widetilde{x})\geq 0$. Thus, by (C1),

$$\Delta_{x}(f_{2}(u_{2}^{k}(\widetilde{x}))) = f_{2}''(u_{2}^{k}(\widetilde{x}))|\nabla u_{2}^{k}(\widetilde{x})|^{2} + f_{2}'(u_{2}^{k}(\widetilde{x}))\Delta_{x}u_{2}^{k}(\widetilde{x}) \ge 0.$$

Consequently, by assumption (C2), it holds for all $1 \le k \le N$,

$$\frac{1}{\tau}(u_2^k(\widetilde{x}) - u_2^{k-1}(\widetilde{x})) \ge q_1(u_1^k(\widetilde{x})) - q_2(u_2^k(\widetilde{x}))q_3(u_3^k(\widetilde{x})) \ge -C_2C_3u_2^k(\widetilde{x}).$$

This yields

$$u_2^k(\widetilde{x}) \ge (1 + C_2 C_3 \tau)^{-1} u_2^{k-1}(\widetilde{x}) \ge \dots \ge (1 + C_2 C_3 \tau)^{-k} u_2^{in}(\widetilde{x}) \ge (1 + C_2 C_3 T/N)^{-N} \mu > 0,$$
(7.2.6)

where we used assumption (C3) in the last step. Thus,

$$\inf_{x \in \Omega} u_2^k(x) \ge (1 + C_2 C_3 T/N)^{-N} \mu > 0. \tag{7.2.7}$$

Since for $N \to \infty$ the term $(1 + C_2C_3T/N)^{-N}\mu$ converges to $\exp(-C_2C_3T) > 0$, the lower bound in (7.2.7) is positive uniformly in $\tau = T/N > 0$. Exactly the same arguments are also valid for u_1^k and u_3^k , leading to similar positive lower bounds uniform in $\tau > 0$.

Now, due to this strict positivity of u_i^k uniform in $\tau > 0$ for $1 \le k \le N$, assumptions (H1)-(H3) in [51, Theorem 2.2] are satisfied. In fact, assumption (H1) is clear, the lower boundedness of $f_i(u_i)/u_i$ follows from assumption (C1), and for the boundedness of $q_1(u_1^k)/u_1^k$ and $q_2(u_2^k)q_3(u_3^k)/u_i^k$, i = 2, 3 in (H2) we can use the boundedness of $q_i(u_i^k)$ due to (C2) together with the strict positivity of the u_i^k . For assumption (H3), that is proving that $f: \mathbb{R}^3_+ \to \mathbb{R}^3_+$ is a homeomorphism, we can use [51, Proposition 5.2]. Thus by applying [51, Theorem 2.2], we get the existence of a strong solution $(u^k)_{1 \le k \le N-1} \in L^{\infty}(\Omega; \mathbb{R}^3)$. However, we notice that the L^{∞} -bound provided by this theorem is not uniform with respect to the time step $\tau > 0$.

Step 2: uniform (in τ) a priori estimates. Our strategy of proving that $u_i^k \in L^{\infty}(\Omega)$ uniformly in τ is similar to the proof of the strict positivity of the u_i^k . Define $\widetilde{x} \in \overline{\Omega}$ such that $u_i^k(\widetilde{x}) = \sup_{x \in \overline{\Omega}} u_i^k(x)$. If $\widetilde{x} \in \partial \Omega$, we can use again a strong Hopf lemma to obtain that $\nabla_x u_2^k \cdot n(x) > 0$, in contradiction to our assumption of homogeneous Neumann boundary conditions. If $\widetilde{x} \in \Omega$, it holds that $\nabla_x u_i^k(\widetilde{x}) = 0$ and $\Delta_x u_i^k(\widetilde{x}) \leq 0$. Moreover, it holds that

$$\Delta_x(f_1(u_1^k(\widetilde{x}))) = f''(u_1^k(\widetilde{x}))|\nabla u_1^k(\widetilde{x})|^2 + f'(u_1^k(\widetilde{x}))\Delta_x u_1^k(\widetilde{x}) \le 0.$$

Thus, due to assumption (C2), it holds for all $1 \le k \le N$ that

$$\frac{1}{\tau}(u_1^k(\widetilde{x}) - u_1^{k-1}(\widetilde{x})) \le C_2,$$

yielding recursively

$$u_1^k(\widetilde{x}) \le C_2 \tau + u_1^{k-1}(\widetilde{x}) \le \dots \le C_2 N \tau + u_1^{in}(\widetilde{x}).$$

Since $\tau = T/N$, this yields the uniform in $\tau > 0$ upper bound

$$||u_1^k||_{L^{\infty}(\Omega)} \le C_2 T + ||u_1^{in}||_{L^{\infty}(\Omega)},$$
 (7.2.8)

Same arguments are also valid for u_2^k and u_3^k .

Next, we derive a uniform estimate for the gradient $\nabla_x u_i^k(x)$. Multiplying (7.2.5) by $\phi := u_i^k$ and integrating in Ω yields

$$\tau^{-1} \int_{\Omega} (u_i^k - u_i^{k-1}) u_i^k \, dx + \int_{\Omega} f_i'(u_i^k) |\nabla u_i^k|^2 \, dx = \int_{\Omega} Q_i(u^k) u_i^k \, dx.$$

By using assumption (C1) and (C2), we get

$$\int_{\Omega} (u_i^k)^2 dx + C_1 \tau \int_{\Omega} |\nabla u_i^k|^2 dx \le C_2 \tau \int_{\Omega} u_i^k dx + \int_{\Omega} u_i^{k-1} u_i^k dx.$$

Using Young's inequality and summing over k = 1, ..., m yields

$$\int_{\Omega} (u_i^m)^2 dx + 2C_1 \tau \sum_{k=1}^m \int_{\Omega} |\nabla u_i^k|^2 dx \le 2C_2 \tau \sum_{k=1}^m \int_{\Omega} u_i^k dx + \int_{\Omega} (u_i^{in})^2 dx.$$
 (7.2.9)

This yields the bound

$$\|\nabla_x u_i^k\|_{L^2(\Omega)} \le C_T.$$

Now we define the piecewise constant functions in time $u^{(\tau)}(x,t) := u^k(x)$ for x in Ω and $t \in ((k-1)\tau, k\tau], \ k=1,\ldots,N$, and at t=0 set $u^{(\tau)}(\cdot,t)=u^{in}(x)$. Moreover, we define the shift operator $\sigma_\tau u^{(\tau)}(x,t):=u^{k-1}(x)$ for x in Ω and $t\in ((k-1)\tau, k\tau], \ k=1,\ldots,N$.

Identifying the sum over k = 1, ..., m multiplied by the time step τ as the time integral over (0, T) in (7.2.9), we get that

$$\|\nabla_x u_i^{(\tau)}\|_{L^2(0,T;L^2(\Omega))} \le C_T$$
 (uniformly in τ).

Summarizing, we have shown that

$$||u_i^{(\tau)}||_{L^{\infty}(0,T;L^{\infty}(\Omega))} + ||u_i^{(\tau)}||_{L^2(0,T;H^1(\Omega))} \le C_T \quad \text{(uniformly in } \tau\text{)}.$$
 (7.2.10)

Finally, we need to derive a uniform estimate for the discrete time derivative. To this end, let $\phi \in L^2(0,T;H^1(\Omega;\mathbb{R}^3))$. Then

$$\tau^{-1} \left| \int_{\tau}^{T} (u^{(\tau)} - \sigma_{\tau} u^{(\tau)}) \phi \, dx dt \right| \leq \|f'(u^{(\tau)}) \nabla u^{(\tau)}\|_{L^{2}(\Omega_{T})} \|\nabla \phi\|_{L^{2}(\Omega_{T})} + \widetilde{C} \|\phi\|_{L^{1}(\Omega_{T})}$$
$$\leq C \|\phi\|_{L^{2}(0,T;H^{1}(\Omega))}.$$

Thus,

$$\tau^{-1} \| u^{(\tau)} - \sigma_{\tau} u^{(\tau)} \|_{L^{2}(0,T;H^{-1}(\Omega))} \le C_{T}.$$
 (7.2.11)

Step 3: passing to the limit $\tau \to 0$. Due to the uniform estimates (7.2.10) and (7.2.11), we can use the Aubin-Lions lemma [40, Theorem 3] to derive that $u_i^{(\tau)} \to u_i$ strongly in $L^2(\Omega_T)$. Since $u_i^{(\tau)}$ is uniformly bounded in $L^{\infty}(\Omega_T)$ and f' and Q are continuous, also

 $f'(u^{(\tau)})$ and $Q(u^{(\tau)})$ are uniformly bounded in $L^{\infty}(\Omega_T)$, moreover, we get the convergence $u_i(\tau) \to u_i$ strongly in $L^s(\Omega)$ for all $s < \infty$ and a.e. in Ω_T (up to subsequences). Thus we have

$$u^{(\tau)} \to u$$
 strongly in $L^s(0, T; L^s(\Omega)),$ (7.2.12)

$$u^{(\tau)} \rightharpoonup u \quad \text{weakly in } L^2(0, T; H^1(\Omega)), \tag{7.2.13}$$

$$f'(u^{(\tau)})\nabla u^{(\tau)} \rightharpoonup X$$
 weakly in $L^2(0, T, L^2(\Omega)),$ (7.2.14)

$$\tau^{-1}(u^{(\tau)} - \sigma_{\tau}u^{(\tau)}) \rightharpoonup \partial_t u \quad \text{weakly in } L^2(0, T; H^{-1}(\Omega)). \tag{7.2.15}$$

Due to (7.2.12) and (7.2.13), it holds that

$$f'(u^{(\tau)})\nabla u^{(\tau)} \rightharpoonup f'(u)\nabla u$$
 weakly in $L^p(\Omega_T)$ for all $p < 2$,

and consequently $X = f'(u)\nabla u$. Similarly as in the proof of [41, pp. 1208-1209], we derive from (7.2.12) and (7.2.11) that

$$\int_0^T \int_\Omega \tau^{-1}(\sigma_\tau u^{(\tau)} - u^{(\tau)} \cdot \phi) \, dx dt \to \int_0^T \int_\Omega u \cdot \partial_t \phi \, dx dt - \int_\Omega u^0 \cdot \phi(0, x) \, dx.$$

This and the convergence results in (7.2.12)-(7.2.15) are sufficient to pass to the limit $\tau \to 0$, yielding (7.2.3).

Remark 7.2. In fact, assuming enough regularity for the data, the solutions are even classical solutions. This can be shown by a bootstrap argument.

7.2.3. Fast-reaction limit to a cross-diffusion system

Our next goal is to study the rigorous limit of system (7.2.1) to the cross-diffusion system (7.2.2). To this end, we impose the following set of assumptions for this subsection.

Assumptions for subsection 7.2.3

(B1) There exists a constant $C_1 > 0$, such that (for large x) it holds that

$$0 \le f_i(x) \le C_1 x^2 \log(x)$$
 for all $x \ge 0$, $i = 1, 2, 3$.

(B2) There exist $\alpha > 0$ and $C_2 > 0$, such that

$$q_i(x) \ge C_2 x^{\alpha}$$
 for all $x \ge 0$, $i = 1, 2, 3$.

(B3) For some s > 1 and $1 \le p, p' \le \infty$ with 1/p + 1/p' = 1, there exist positive constants $C_3, C_4, C_5 > 0$ such that (for large x)

$$q_1^s(x) \le C_3 x f_1(x)$$
 for all $x \ge 0$,
 $q_2^{sp}(x) \le C_4 x f_2(x)$ for all $x \ge 0$,
 $q_2^{sp'}(x) \le C_5 x f_3(x)$ for all $x \ge 0$.

(B4) The functions $f_i: \mathbb{R}_+ \to \mathbb{R}_+$ and $q_i: \mathbb{R}_+^* \to \mathbb{R}_+$ are in C^1 with $f_i'(x) > 0$ and $q_i'(x) > 0$ for all $x \ge 0$, i = 1, 2, 3.

(B5) There exists a constant $C_6 > 0$, such that

$$q_i(x) \le C_6 x \left(f_i f_i' q_i' \right)(x)$$
 for all $x \ge 0$, $i = 1, 2, 3$.

(B6) For $a: \mathbb{R}^2_+ \to \mathbb{R}^2_+$ with $a(u_2, u_3) = (a_2(u_2, u_3), a_3(u_2, u_3))$ it holds that

$$a_i: \begin{cases} \mathbb{R}_+^2 \to \mathbb{R}_+ \\ (u_2, u_3) \mapsto u_i + q_1^{-1} \Big(q_2(u_2) q_3(q_3) \Big) / u_i \end{cases}$$

are continuous for i = 2, 3.

Fast-reaction limit

Theorem 7.3. Let assumptions (B1) - (B6) hold. Let $\Omega \subseteq \mathbb{R}^N$ be a bounded regular open set of \mathbb{R}^N , and let for any $\varepsilon > 0$, $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}$ denote a weak solution of the reaction-diffusion system (7.2.1) with initial data $u_i^{\varepsilon}(0, x) = u_i^{in}(x) \in L^{\infty}$ for all $x \in \Omega$, i = 1, 2, 3.

Then the following holds: If $\varepsilon \to 0$, there exists a subsequence of $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}$ (which we still denote by $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}$), which converges to u_1, u_2, u_3 in $L^1_{loc}([0, \infty); L^1(\Omega))$.

Moreover, this limit is a weak solution of the cross-diffusion system (7.2.2) belonging to $L^1_{loc}([0,\infty);L^1(\Omega))$.

A priori estimates

Defining the entropy

$$H[u^{\varepsilon}(t)] = \sum_{i=1}^{3} \int_{\Omega} h_i(u_i^{\varepsilon}) dx, \text{ with } h_i(u_i^{\varepsilon}) = \int_{c}^{u_i^{\varepsilon}} \log(q_i(z)) dz, \tag{7.2.16}$$

the entropy dissipation along the flow of system (7.2.1) has the following form

$$\begin{split} D_{\varepsilon}[u^{\varepsilon}(t)] &= -\frac{d}{dt} H[u^{\varepsilon}(t)] \\ &= -\sum_{i=1}^{3} \int_{\Omega} \log q_{i}(u_{i}^{\varepsilon}) \, \partial_{t} u_{i}^{\varepsilon} \, dx \\ &= \sum_{i=1}^{3} \int_{\Omega} \left(\frac{f_{i}' q_{i}'}{q_{i}} \right) (u_{i}^{\varepsilon}) \, |\nabla_{x} u_{i}^{\varepsilon}|^{2} \, dx \\ &+ \frac{1}{\varepsilon} \int_{\Omega} \left(\log(q_{1}(u_{1}^{\varepsilon})) - \log(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})) \right) \left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right) dx \\ &\geq 0. \end{split}$$

Integrating this in time leads to the standard entropy-entropy-dissipation inequality, where the right-hand side is uniform in ε :

$$\sup_{t \in [0,T]} H[u^{\varepsilon}(t)] + \int_0^T D_{\varepsilon}[u^{\varepsilon}(s)]ds \le H[u^{in}]. \tag{7.2.17}$$

Lemma 7.4 (Entropy estimates). The following estimates hold for u_i^{ε} with i = 1, 2, 3:

$$\sum_{i=1}^{3} \int_{0}^{T} \int_{\Omega} \left(\frac{f_{i}' q_{i}'}{q_{i}} \right) (u_{i}^{\varepsilon}) |\nabla_{x} u_{i}^{\varepsilon}|^{2} dx dt \leq C_{T}, \qquad (7.2.18)$$

$$\int_{\Omega} \left(\log(q_1(u_1^{\varepsilon})) - \log(q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon})) \right) \left(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon}) \right) dx dt \le \varepsilon C_T, \tag{7.2.19}$$

$$\sup_{t \in [0,T]} \int_{\Omega} u_i^{\varepsilon} \log(u_i^{\varepsilon}) \, dx \le C_T. \tag{7.2.20}$$

Proof. Estimates (7.2.18) and (7.2.19) follow from the entropy-entropy-dissipation inequality (7.2.17).

Estimate (7.2.20) can be proved in the following way: Thanks to (B2) we know that

$$h_i(x) = \int_c^x \log(q_i(z)) dz \ge \alpha C \int_c^x \log(z) dz.$$

This together with (7.2.17) (and the conservations of $\int_{\Omega} (u_1^{\varepsilon} + u_i^{\varepsilon})$) yields

$$\sup_{t \in [0,T]} \sum_{i=1}^{3} \int_{\Omega} u_i^{\varepsilon} |\log(u_i^{\varepsilon})| \, dx \le C \sup_{t \in [0,T]} H[u^{\varepsilon}(t)] \le C_T.$$

Now we present the following duality lemma from [29], which is a variant of [121] introduced by M. Pierre and D. Schmitt.

176

Lemma 7.5 (Duality lemma). If for $u \ge 0$ and $A := A(t, x) \ge 0$ it holds that

$$\begin{cases} \partial_t u - \Delta_x(Au) \le 0 & in \ \Omega, \\ n(x) \cdot \nabla_x(Au) = 0 & on \ \partial\Omega, \\ u(0,x) = u^{in}(x) \in L^{\infty}(\Omega), \end{cases}$$

then

$$\int_0^T \int_{\Omega} Au^2 \, dx dt \le C \left(1 + \int_0^T \int_{\Omega} A \, dx dt \right),$$

where C depends only on $||u^{in}||_{L^{\infty}(\Omega)}$.

Proof. We integrate $\partial_t u - \Delta_x(Au) \leq 0$ against a test function $w \geq 0$, which is defined as a positive solution of the dual problem

$$\begin{cases} \partial_t w + A\Delta_x w = -Au, \\ \nabla_x w \cdot n(x) = 0 \text{ on } \partial\Omega, \\ w(T, \cdot) = 0. \end{cases}$$
 (7.2.21)

Note that w is indeed nonnegative because u is also assumed to be nonnegative. This leads to

$$\int_{0}^{T} \int_{\Omega} \left(w \partial_{t} u - w \Delta_{x} \left(A u \right) \right) dx dt \leq 0.$$

Integration by parts yields

$$-\int_0^T \int_{\Omega} u \partial_t w \, dx dt - \int_{\Omega} (wu)(0) \, dx - \int_0^T \int_{\Omega} Au \Delta_x w \, dx dt \le 0$$

and using the dual problem (7.2.21) leads to

$$\int_0^T \int_{\Omega} Au^2 \, dx dt \le \int_{\Omega} (wu)(0) \, dx.$$

Using Hölder's inequality yields

$$\int_{0}^{T} \int_{\Omega} Au^{2} dx dt \le ||u(0)||_{L^{\infty}(\Omega)} \int_{\Omega} |w(0)| dx.$$
 (7.2.22)

Next we multiply the dual problem (7.2.21) by $\Delta_x w$ and integrate, which yields

$$-\int_0^T \partial_t \int_{\Omega} \frac{|\nabla_x w|^2}{2} \, dx dt + \int_0^T \int_{\Omega} A \left(\Delta_x w\right)^2 \, dx dt = -\int_0^T \int_{\Omega} A u \Delta_x w \, dx dt.$$

Using Young's inequality leads to

$$\frac{1}{2} \int_{\Omega} |\nabla_x w(0)|^2 \ dx + \int_0^T \int_{\Omega} A \left(\Delta_x w \right)^2 \ dx dt \le \frac{1}{2} \int_0^T \int_{\Omega} A \left(\Delta_x w \right)^2 \ dx dt + \frac{1}{2} \int_0^T \int_{\Omega} A u^2 \ dx dt.$$

Thus

$$\int_{\Omega} |\nabla_x w(0)|^2 dx + \int_0^T \int_{\Omega} A (\Delta_x w)^2 dx dt \le \int_0^T \int_{\Omega} A u^2 dx dt.$$
 (7.2.23)

Now we can write

$$w(0) = -\int_0^T \partial_t w \, dt = \int_0^T \left(A \Delta_x w + A u \right) dt.$$

Thus

$$\int_{\Omega} |w(0)| \, dx \le \int_{0}^{T} \int_{\Omega} \left(|A\Delta_{x}w| + |Au| \right) dx dt.$$

Using Young's inequality for both terms on the right-hand side leads to

$$\int_{\Omega} |w(0)| \ dx \leq \frac{\varepsilon}{2} \int_{0}^{T} \int_{\Omega} Au^{2} dx dt + \frac{\varepsilon}{2} \int_{0}^{T} \int_{\Omega} A(\Delta_{x}w)^{2} dx dt + \frac{1}{\varepsilon} \int_{0}^{T} \int_{\Omega} A dx dt,$$

and thanks to (7.2.23) we get that

$$\int_{\Omega} |w(0)| \ dx \le \frac{1}{\varepsilon} \int_{0}^{T} \int_{\Omega} A \, dx dt + \varepsilon \int_{0}^{T} \int_{\Omega} A u^{2} \, dx dt. \tag{7.2.24}$$

Now we use (7.2.22) and (7.2.24) to obtain

$$\int_0^T \int_{\Omega} Au^2 \, dx dt \le \|u(0)\|_{L^{\infty}(\Omega)} \int_{\Omega} |w(0)| \, dx$$

$$\le \|u(0)\|_{L^{\infty}(\Omega)} \left(\frac{1}{\varepsilon} \int_0^T \int_{\Omega} A \, dx dt + \varepsilon \int_0^T \int_{\Omega} Au^2 \, dx dt\right),$$

which proves the duality lemma by choosing $\varepsilon > 0$ small enough.

Lemma 7.6 (Duality estimate). It holds that

$$\int_0^T \int_{\Omega} \left(f_1(u_1^{\varepsilon}) + f_i(u_i^{\varepsilon}) \right) \left(u_1^{\varepsilon} + u_i^{\varepsilon} \right) dx dt \le C_T, \quad i = 2, 3.$$

Proof. It holds that

$$\partial_t \left(u_1^{\varepsilon} + u_i^{\varepsilon} \right) - \Delta_x \left(\left(\frac{f_1(u_1^{\varepsilon}) + f_i(u_i^{\varepsilon})}{u_1^{\varepsilon} + u_i^{\varepsilon}} \left(u_1^{\varepsilon} + u_i^{\varepsilon} \right) \right) \right) = 0, \quad i = 2, 3.$$
 (7.2.25)

Due to (B1) and (7.2.20) we know that

$$\int_0^T \int_\Omega \frac{f_1(u_1^\varepsilon) + f_i(u_i^\varepsilon)}{u_1^\varepsilon + u_i^\varepsilon} \, dx dt \le C \int_0^T \int_\Omega u_1^\varepsilon \log(u_1^\varepsilon) \, dx dt + \int_\Omega u_i^\varepsilon \log(u_i^\varepsilon) \, dx dt \le C_T.$$

Thus we can use Lemma 7.5 for equation (7.2.25) and get the desired estimate.

Lemma 7.7. Recalling that $u_i^{in}(x) \in L^{\infty}(\Omega)$ for i = 1, 2, 3, we get the following estimates:

$$\begin{aligned} \left\| (q_1(u_1^{\varepsilon}))^{1/2} - \left(q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right)^{1/2} \right\|_{L^2(\Omega_T)} &\leq \sqrt{\varepsilon} C_T, \\ \left\| q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right\|_{L^1(\Omega_T)} &\leq \sqrt{\varepsilon} C_T. \end{aligned}$$

Proof. The elementary inequality

$$4\left|\sqrt{a} - \sqrt{b}\right|^2 \le (a - b)(\log a - \log b), \quad a, b \ge 0,$$

together with Lemma 7.4 yields

$$4 \left\| (q_{1}(u_{1}^{\varepsilon}))^{1/2} - \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right)^{1/2} \right\|_{L^{2}(\Omega_{T})}^{2} =$$

$$= 4 \int_{0}^{T} \int_{\Omega} \left| (q_{1}(u_{1}^{\varepsilon}))^{1/2} - \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right)^{1/2} \right|^{2} dx dt$$

$$\leq \int_{0}^{T} \int_{\Omega} \left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right) \left(\log \left(q_{1}(u_{1}^{\varepsilon}) \right) - \log \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right) \right) dx dt$$

$$< \varepsilon C_{T},$$

which shows the first estimate. By using the Cauchy-Schwarz inequality, we get that

$$\begin{split} &\|q_{1}(u_{1}^{\varepsilon})-q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\|_{L^{1}(\Omega_{T})} = \\ &= \int_{0}^{T} \int_{\Omega} \left((q_{1}(u_{1}^{\varepsilon}))^{1/2} - \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right)^{1/2} \right) \left((q_{1}(u_{1}^{\varepsilon}))^{1/2} + \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right)^{1/2} \right) dxdt \\ &\leq \left(\int_{O_{T}} \left| q_{1}(u_{1}^{\varepsilon})^{1/2} - \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right)^{1/2} \right|^{2} \right)^{1/2} \left(\int_{O_{T}} \left| q_{1}(u_{1}^{\varepsilon})^{1/2} + \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right)^{1/2} \right|^{2} \right)^{1/2} \end{split}$$

For the first factor we can use the first part of this lemma:

$$\int_0^T \int_{\Omega} \left| (q_1(u_1^{\varepsilon}))^{1/2} - \left(q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right)^{1/2} \right|^2 dx dt \le \sqrt{\varepsilon} C_T,$$

and the second factor is bounded due to Young's inequality, (B3) and Lemma 7.6:

$$\int_{0}^{T} \int_{\Omega} \left| (q_{1}(u_{1}^{\varepsilon}))^{1/2} + \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) \right)^{1/2} \right|^{2} dx dt \leq
2 \int_{0}^{T} \int_{\Omega} q_{1}(u_{1}^{\varepsilon}) dx dt + 2 \int_{0}^{T} \int_{\Omega} q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon}) dx dt
\leq C + C \int_{0}^{T} \int_{\Omega} \left(u_{1}^{\varepsilon} f_{1}(u_{1}^{\varepsilon}) + u_{2}^{\varepsilon} f_{2}(u_{2}^{\varepsilon}) + u_{3}^{\varepsilon} f_{3}(u_{3}^{\varepsilon}) \right) dx dt \leq C_{T}.$$

This finishes the proof of the second estimate.

Lemma 7.8. It holds for i = 1, 2, 3 that

$$\|\nabla_x u_i^{\varepsilon}\|_{L^1(\Omega_T)} \le C_T,$$

which shows that u_i^{ε} is bounded in $L^1(0,T;W^{1,1}(\Omega))$.

Proof. It holds by Cauchy-Schwarz that

$$\int_0^T \int_\Omega |\nabla_x u_i^\varepsilon| \, dx dt \leq \left(\int_0^T \int_\Omega \left(\frac{f_i' q_i'}{q_i} \right) (u_i^\varepsilon) \, |\nabla_x u_i^\varepsilon|^2 \, dx dt \right)^{1/2} \left(\int_0^T \int_\Omega \left(\frac{q_i}{f_i' q_i'} \right) (u_i^\varepsilon) \, dx dt \right)^{1/2} dx dt$$

Now it holds that

$$\int_0^T \int_{\Omega} \left(\frac{f_i' q_i'}{q_i} \right) (u_i^{\varepsilon}) |\nabla_x u_i^{\varepsilon}|^2 dx dt \le C_T$$

due to Lemma 7.4, and thanks to (B5) and Lemma 7.6 we get that

$$\int_0^T \int_\Omega \left(\frac{q_i}{f_i'q_i'}\right)(u_i^\varepsilon) \, dx dt \le C \int u_i^\varepsilon f_i(u_i^\varepsilon) \, dx dt \le C_T,$$

which finishes the proof.

Strong compactness for u_i^{ε}

The difficulty in proving strong compactness for u_i^{ε} in system (7.2.1) relies in the fact that we cannot use the Aubin-Lions lemma [130] for u_i^{ε} directly, due to blow-up in the time dervatives $\partial_t u_i^{\varepsilon}$ for $\varepsilon \to 0$. However, we are still able to prove strong compactness for the u_i^{ε} with the help of the following lemmata.

Lemma 7.9. The sequences $(u_1^{\varepsilon} + u_i^{\varepsilon})$, i = 2, 3 are relatively compact in $L^1(\Omega_T)$.

Proof. For the terms $u_1^{\varepsilon} + u_i^{\varepsilon}$, i = 2, 3, in system (7.2.1), we see that the right-hand side does not blow-up for $\varepsilon \to 0$, thus we apply the Aubin-Lions lemma to these terms and get strong compactness for them.

The terms $u_1^{\varepsilon} + u_i^{\varepsilon}$, i = 2, 3, are bounded in $L^1(0, T; W^{1,1}(\Omega))$ due to Lemma 7.8. For the time derivatives $\partial_t(u_1^{\varepsilon} + u_i^{\varepsilon})$, i = 2, 3, we have that

$$\partial_t (u_1^{\varepsilon} + u_i^{\varepsilon}) = \Delta_x (f_1(u_1^{\varepsilon}) + f_i(u_i^{\varepsilon})).$$

Now it holds by Lemma 7.6, (B1) and Lemma 7.4 that

$$\int_{0}^{T} \int_{\Omega} f_{i}(u_{i}^{\varepsilon}) dx dt \leq \left(\int_{0}^{T} \int_{\Omega} f_{i}(u_{i}^{\varepsilon}) u_{i}^{\varepsilon} dx dt \right)^{1/2} \left(\int_{0}^{T} \int_{\Omega} \frac{f_{i}(u_{i}^{\varepsilon})}{u_{i}^{\varepsilon}} dx dt \right)^{1/2} \\
\leq C \left(\int_{0}^{T} \int_{\Omega} f_{i}(u_{i}^{\varepsilon}) u_{i}^{\varepsilon} dx dt \right)^{1/2} \left(\int_{0}^{T} \int_{\Omega} u_{i}^{\varepsilon} \log(u_{i}^{\varepsilon}) dx dt \right)^{1/2} \\
\leq C_{T},$$

which implies that $\partial_t(u_1^{\varepsilon} + u_i^{\varepsilon})$ is bounded in $L^1(0, T; W^{-2,1}(\Omega))$. Thus by the standard Aubin-Lions lemma [130] we get that $(u_1^{\varepsilon} + u_i^{\varepsilon})$ is relatively compact in $L^1(\Omega_T)$.

Lemma 7.10. The function

$$\varphi: \left\{ \begin{array}{l} \mathbb{R}_{+}^{2} \to \mathbb{R}_{+}^{2} \\ (u_{2}, u_{3}) \mapsto \left(u_{2} + q_{1}^{-1} \left(q_{2}(u_{2}) q_{3}(u_{3}) \right), u_{3} + q_{1}^{-1} \left(q_{2}(u_{2}) q_{3}(u_{3}) \right) \right) \end{array} \right.$$

is a homeomorphism on \mathbb{R}^2_+ .

Proof. The function φ can be written as

$$\varphi: \begin{cases} \mathbb{R}_{+}^{2} \to \mathbb{R}_{+}^{2} \\ (u_{2}, u_{3}) \mapsto (a_{2}(u_{2}, u_{3})u_{2}, a_{3}(u_{2}, u_{3})u_{3}), \end{cases}$$

where for i = 2, 3 we have $a_i : \mathbb{R}^2_+ \to \mathbb{R}_+$ with

$$a_i(u_2, u_3) = 1 + \frac{q_1^{-1}(q_2(u_2)q_3(u_3))}{u_i}.$$

By (B6) it holds that the functions a_i are continuous on \mathbb{R}^2_+ . Clearly they are lower bounded by 1. Moreover, on $\mathbb{R}_+ \times \mathbb{R}_+$ it holds thanks to (B4) that φ is strictly increasing (in the sense that each component is strictly increasing w.r.t. each component), and on $\mathbb{R}^*_+ \times \mathbb{R}^*_+$ it holds that φ is C^1 . The determinant of the Jacobian of φ is strictly positive on $\mathbb{R}^*_+ \times \mathbb{R}^*_+$:

$$\det D(\varphi) = 1 + (q_1^{-1})'(q_2(u_2)q_3(u_3)) \cdot (q_2'(u_2)q_3(u_3) + q_2(u_2)q_3'(u_3)) > 0 \quad \forall u_i > 0.$$

Thus we can apply [51, Proposition 5.1] and get that φ is a homeomorphism on \mathbb{R}^2_+ . \square

Lemma 7.11. If (u_i^{ε}) , i = 1, 2, 3 are sequences in \mathbb{R}_+ satisfying for $\varepsilon \to 0$ that

$$\begin{array}{lll} (u_1^\varepsilon + u_2^\varepsilon) \to B & a.e. \ in \ \Omega_T, \\ (u_1^\varepsilon + u_3^\varepsilon) \to C & a.e. \ in \ \Omega_T, \end{array}$$

and if additionally

$$\left(u_1^{\varepsilon}-q_1^{-1}(q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon}))\right)\to 0$$
 a.e. in Ω_T ,

then it holds that

$$\begin{array}{lll} u_1^\varepsilon \to u_1 & a.e. \ in \ \Omega_T, \\ u_2^\varepsilon \to u_2 & a.e. \ in \ \Omega_T, \\ u_3^\varepsilon \to u_3 & a.e. \ in \ \Omega_T, \end{array}$$

with

$$u_2 := \psi_2(B, C), \quad u_3 := \psi_3(B, C), \quad u_1 := q_1^{-1} (q_2(\psi_2(B, C)) q_3(\psi_3(B, C))),$$
 (7.2.27)

where $\psi = (\psi_2, \psi_2) := \varphi^{-1}$, and φ denotes the homeomorphism introduced in Lemma 7.10.

Proof. Due to the assumptions, it holds that

$$u_2^{\varepsilon} + q_1^{-1}(q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon})) \to B$$
 a.e. in Ω_T ,
 $u_3^{\varepsilon} + q_1^{-1}(q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon})) \to C$ a.e. in Ω_T .

Moreover, since φ is bijective with the inverse $\psi := \varphi^{-1}$, it holds that

$$(u_2^{\varepsilon}, u_3^{\varepsilon}) = \psi \left(u_2^{\varepsilon} + q_1^{-1}(q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon})), \quad u_3^{\varepsilon} + q_1^{-1}(q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon})) \right) \text{ for all } \varepsilon > 0,$$

and since ψ is continuous, it holds a.e. in Ω_T that

$$\begin{aligned} (u_2, u_3) &= \lim_{\varepsilon \to 0} (u_2^{\varepsilon}, u_3^{\varepsilon}) \\ &= \lim_{\varepsilon \to 0} \left(\psi_2 \left(u_2^{\varepsilon} + q_1^{-1}(q_2 q_3), u_3^{\varepsilon} + q_1^{-1}(q_2 q_3) \right), \psi_3 \left(u_2^{\varepsilon} + q_1^{-1}(q_2 q_3), u_3^{\varepsilon} + q_1^{-1}(q_2 q_3) \right) \right) \\ &= \left(\psi_2 (B, C), \psi_3 (B, C) \right), \end{aligned}$$

with B,C defined in (7.2.26). This shows that $u_2^{\varepsilon} \to u_2$ a.e. in Ω_T , and $u_3^{\varepsilon} \to u_3$ a.e. in Ω_T with $u_2 = \psi_2(B,C)$ and $u_3 = \psi_3(B,C)$. Finally, since $u_1^{\varepsilon} - q_1^{-1}(q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon})) \to 0$ a.e. in Ω_T , we get that $u_1^{\varepsilon} \to q_1^{-1}(q_2(\psi_2(B,C))q_3(\psi_3(B,C)))$ a.e. in Ω_T , which finishes the proof.

Theorem 7.12. For i = 1, 2, 3 it holds that

$$u_i^{\varepsilon} \to u_i \text{ strongly in } L^1(\Omega_T) \text{ and a.e.}.$$

Proof. Due to Lemma 7.9 we have (for a subsequence) that $u_1^{\varepsilon} + u_2^{\varepsilon} \to B$ a.e. in Ω_T and $u_1^{\varepsilon} + u_3^{\varepsilon} \to C$ a.e. in Ω_T , and due to Lemma 7.7 it holds (for a subsequence) that $\left(u_1^{\varepsilon} - q_1^{-1}(q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon}))\right) \to 0$ a.e. in Ω_T . Thus we can apply Lemma 7.11, and get that $u_i^{\varepsilon} \to u_i$ a.e. in Ω_T .

Thanks to Lemma 7.4 we have for i = 1, 2, 3 that

$$||u_i^{\varepsilon}||_{L(\log L)(\Omega_T)} \le C_T,$$

thus by [105, Lemma 1.3] we get that

$$u_i^{\varepsilon} \to u_i$$
 strongly in $L^1(\Omega_T)$.

Uniform integrability

Lemma 7.13. For $\varepsilon \to 0$ it holds that

$$q_1(u_1^{\varepsilon}) \to q_1(u_1)$$
 strongly in $L^1(\Omega_T)$,
 $q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon}) \to q_2(u_2)q_3(u_3)$ strongly in $L^1(\Omega_T)$.

Proof. Due to Theorem 7.12 we know that $u_i^{\varepsilon} \to u_i$ a.e. in Ω_T . Since q_i is continuous, it holds that

$$q_1(u_1^\varepsilon) \to q_1(u_1) \quad \text{a.e. in} \quad \Omega_T,$$

$$q_2(u_2^\varepsilon)q_3(u_3^\varepsilon) \to q_2(u_2)q_3(u_3) \quad \text{a.e. in} \quad \Omega_T.$$

Thanks to (B3) and Lemma 7.6 we have for some s > 1:

$$||q_1(u_1^{\varepsilon})||_{L^s(\Omega_T)} \le C_T,$$

 $||q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon})||_{L^s(\Omega_T)} \le C_T.$

Thus $q_1(u_1^{\varepsilon})$ and $q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon})$ are uniformly integrable, and applying [105, Lemma 1.3] finishes the proof.

Lemma 7.14. It holds that $f_i(u_i^{\varepsilon}) \to f_i(u_i)$ strongly in $L^1(\Omega_T)$ for i = 1, 2, 3.

Proof. First we prove that $f_i(u_i^{\varepsilon})$ is uniformly integrable, where we recall that $f_i : \mathbb{R}_+ \to \mathbb{R}_+$ with $f_i' > 0$. In the case that that f_i is bounded, this is clearly true. In the case that f_i is unbounded, f_i^{-1} is also unbounded, and we proceed in the following way. Defining the function

$$\Phi_i: \begin{cases} \mathbb{R}_+ \to \mathbb{R}_+ \\ x \mapsto f_i^{-1}(x)x. \end{cases}$$

for i=1,2,3, it holds that Φ_i is strictly increasing and that $\Phi_i(x)/x \to \infty$ for $x \to \infty$. Finally, thanks to Lemma 7.6 we have for all $\varepsilon > 0$ and for i=1,2,3 that

$$\int_0^T \int_\Omega \Phi_i\left(|f_i(u_i^\varepsilon)|\right) \, dx dt = \int_0^T \int_\Omega u_i^\varepsilon f_i(u_i^\varepsilon) \, dx dt \le C_T.$$

Thus $(f_i(u_i^{\varepsilon}))$ is uniformly integrable for i = 1, 2, 3 also in this case. Since $f_i(u_i^{\varepsilon}) \to f_i(u_i)$ a.e. in Ω_T , we can apply [105, Lemma 1.3] to get the desired result.

Passing to the limit

The estimates derived in the last subsections are sufficient to pass to the limit $\varepsilon \to 0$ in the weak formulation of the reaction-diffusion system (7.2.1). Thus we are now able to prove Theorem 7.3.

Proof of Theorem 7.3. The only step which remains to prove is to pass to the limit $\varepsilon \to 0$ for the terms $u_1^{\varepsilon} + u_i^{\varepsilon}$, i = 2, 3 in the weak formulations of the reaction-diffusion system (7.2.1). First we write down the equation satisfied for $u_1^{\varepsilon} + u_i^{\varepsilon}$, i = 2, 3:

$$\begin{cases} \partial_t (u_1^{\varepsilon} + u_i^{\varepsilon}) - \Delta_x \left(f_1(u_1^{\varepsilon}) + f_i(u_i^{\varepsilon}) \right) = 0, \\ (u_1^{\varepsilon} + u_i^{\varepsilon})(0, x) = \left(u_1^{in} + u_i^{in} \right)(x), \\ n(x) \cdot \nabla_x (u_1^{\varepsilon} + u_i^{\varepsilon})(t, x) = 0. \end{cases}$$

The weak form can be written in the following way: For any test function $\psi = \psi(t, x)$ in the set

$$\{\psi \in C^2([0,\infty) \times \overline{\Omega}) : \psi(t) = 0 \text{ for } t \ge T \text{ and } n(x) \cdot \nabla_x \psi(t,x) = 0\},$$

we have that

$$0 = \int_{\Omega_T} (u_1^{\varepsilon} + u_i^{\varepsilon})(t, x) \partial_t \psi(t, x) \, dx dt + \int_{\Omega} (u_1^{in} + u_i^{in})(x) \psi(0, x) \, dx$$
$$+ \int_{\Omega_T} \left(f_1(u_1^{\varepsilon}) + f_i(u_i^{\varepsilon}) \right) \Delta_x \psi(t, x) \, dx dt.$$

Now, by using that

$$u_i^{\varepsilon} \to u_i \text{ strongly in } L^1(\Omega_T), \quad i = 1, 2, 3,$$

 $f_i(u_i^{\varepsilon}) \to f_i(u_i) \text{ strongly in } L^1(\Omega_T), \quad i = 1, 2, 3,$

$$(7.2.28)$$

we can pass to the limit $\varepsilon \to 0$ and get that

$$0 = \int_{\Omega_T} (u_1 + u_i)(t, x) \partial_t \psi(t, x) \, dx dt + \int_{\Omega} (u_1^{in} + u_i^{in})(x) \psi(0, x) \, dx + \int_{\Omega_T} (f_1(u_1) + f_i(u_i)) \Delta_x \psi(t, x) \, dx dt.$$
(7.2.29)

Next we observe that thanks to Lemma 7.13 (second inequality) and Lemma 7.7, we get $q_1(u_1) - q_2(u_2)q_3(u_3)$.

This finishes the proof.
$$\Box$$

Remark 7.15 (Initial layers). The last proof shows that the initial data needs to be well-prepared, i.e.

$$q_2(u_2^{in})q_3(u_3^{in}) = q_1(u_1^{in})$$
 a.e. in Ω .

If this condition is not satisfied, there is an initial layer appearing. But due to the fact that

$$||q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon})q_3(u_3^{\varepsilon})||_{L^1(\Omega_T)} \le \sqrt{\varepsilon}C,$$

this initial layer vanishes after $o_{\varepsilon \to 0}(1)$ time.

7.2.4. Transformed system

First we recall the definition of the homeomorphism $\varphi(u) = (\varphi_2(u), \varphi_3(u))$ for $u = (u_2, u_3)$ defined in Lemma 7.10:

$$\varphi: \begin{cases} \mathbb{R}_{+}^{2} \to \mathbb{R}_{+}^{2}, \\ (u_{2}, u_{3}) \mapsto (w_{2}, w_{3}), \end{cases}$$
 (7.2.30)

where
$$w_2 := u_2 + q_1^{-1}(q_2(u_2)q_3(u_3))$$
 and $w_3 := u_3 + q_1^{-1}(q_2(u_2)q_3(u_3))$, and its inverse
$$\psi := \varphi^{-1}, \quad \psi(w) = (\psi_2(w), \psi_3(w)). \tag{7.2.31}$$

Thus we can formally rewrite the limiting system in terms of the new variables (w_2, w_3) in the following way:

$$\begin{cases} \partial_t w_2 - \Delta_x \left[\left(\frac{f_1(q_1^{-1}(q_2(\psi_2(w))q_3(\psi_3(w)))) + f_2(\psi_2(w))}{w_2} \right) w_2 \right] = 0, \\ \partial_t w_3 - \Delta_x \left[\left(\frac{f_1(q_1^{-1}(q_2(\psi_2(w))q_3(\psi_3(w)))) + f_3(\psi_3(w))}{w_3} \right) w_3 \right] = 0. \end{cases}$$

The entropy can be expressed in the new variables $w = (w_2, w_3)$ in the following way

$$H[w(t)] = \int_{\Omega} \int_{c}^{q_{1}^{-1}(q_{2}(\psi_{2}(w))q_{3}(\psi_{3}(w)))} \log(q_{1}(z)) dz dx + \int_{\Omega} \int_{c}^{\psi_{2}(w)} \log(q_{2}(z)) dz dx + \int_{\Omega} \int_{c}^{\psi_{3}(w)} \log(q_{3}(z)) dz dx.$$

7.2.5. Robustness of the method

Robustness of the method for quadratic diffusivities

Now we study the microscopic model with quadratic diffusivities $f_i(u_i^{\varepsilon}) = (u_i^{\varepsilon})^2$, i = 1, 2, 3, where we allow a small pertubation of cross-diffusion terms in u_i^{ε} , u_i^{ε} for $\delta > 0$, $\gamma > 0$:

$$\begin{cases}
\partial_{t}u_{1}^{\varepsilon} - \Delta_{x}\left(u_{1}^{\varepsilon}\left(u_{1}^{\varepsilon} + \delta u_{2}^{\varepsilon}\right)\right) = -\frac{1}{\varepsilon}\left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right), \\
\partial_{t}u_{2}^{\varepsilon} - \Delta_{x}\left(u_{2}^{\varepsilon}\left(u_{2}^{\varepsilon} + \gamma u_{1}^{\varepsilon}\right)\right) = +\frac{1}{\varepsilon}\left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right), \\
\partial_{t}u_{3}^{\varepsilon} - \Delta_{x}\left(\left(u_{3}^{\varepsilon}\right)^{2}\right) = +\frac{1}{\varepsilon}\left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right).
\end{cases} (7.2.32)$$

The following lemma shows that our method of obtaining entropies for the limiting cross-diffusion systems is robust under small perurbation of cross-diffusion terms.

Lemma 7.16. Let $0 < \delta \le 4$ and $0 < \gamma \le 4$. Assume that there exist positive constants $C_1, C_2 > 0$, such that

$$C_1 \frac{q_2'(u_2)}{q_2(u_2)} \le \frac{q_1'(u_1)}{q_1(u_1)} \le C_2 \frac{q_2'(u_2)}{q_2(u_2)}.$$
 (7.2.33)

Moreover, assume that

$$\delta - \frac{\gamma}{2C_1} \ge 0, \qquad \gamma - \frac{\delta C_2}{2} \ge 0. \tag{7.2.34}$$

Then the entropy is decreasing along the flow of the pertubed system (7.2.32), i.e.

$$\forall \varepsilon > 0: \quad \frac{d}{dt} H[u^{\varepsilon}(t)] \le 0.$$

Proof. It holds that

$$\begin{split} \frac{d}{dt}H[u^{\varepsilon}(t)] &= \int_{\Omega} \log \left(q_{1}(u_{1}^{\varepsilon})\right) \partial_{t}u_{1}^{\varepsilon} \, dx + \int_{\Omega} \log \left(q_{2}(u_{2}^{\varepsilon})\right) \partial_{t}u_{2}^{\varepsilon} \, dx + \int_{\Omega} \log \left(q_{3}(u_{3}^{\varepsilon})\right) \partial_{t}u_{3}^{\varepsilon} \, dx \\ &= -\frac{1}{\varepsilon} \int_{\Omega} \left(\log \left(q_{1}(u_{1}^{\varepsilon})\right) - \log \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right) \right) \left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right) dx \\ &- 2 \int_{\Omega} \frac{q'_{1}(u_{1}^{\varepsilon})}{q_{1}(u_{1}^{\varepsilon})} u_{1}^{\varepsilon} |\nabla_{x}(u_{1}^{\varepsilon})|^{2} \, dx - 2 \int_{\Omega} \frac{q'_{2}(u_{2}^{\varepsilon})}{q_{2}(u_{2}^{\varepsilon})} u_{2}^{\varepsilon} |\nabla_{x}(u_{2}^{\varepsilon})|^{2} \, dx \\ &- 2 \int_{\Omega} \frac{q'_{3}(u_{3}^{\varepsilon})}{q_{3}(u_{3}^{\varepsilon})} u_{3}^{\varepsilon} |\nabla_{x}(u_{3}^{\varepsilon})|^{2} \, dx - \delta \int_{\Omega} \frac{q'_{1}(u_{1}^{\varepsilon})}{q_{1}(u_{1}^{\varepsilon})} u_{2}^{\varepsilon} |\nabla_{x}(u_{1}^{\varepsilon})|^{2} \, dx \\ &- \gamma \int_{\Omega} \frac{q'_{2}(u_{2}^{\varepsilon})}{q_{2}(u_{2}^{\varepsilon})} u_{1}^{\varepsilon} |\nabla_{x}(u_{2}^{\varepsilon})|^{2} \, dx - \delta \int_{\Omega} \frac{q'_{1}(u_{1}^{\varepsilon})}{q_{1}(u_{1}^{\varepsilon})} u_{1}^{\varepsilon} |\nabla_{x}(u_{1}^{\varepsilon}) \cdot \nabla_{x}(u_{2}^{\varepsilon}) \, dx \\ &- \gamma \int_{\Omega} \frac{q'_{2}(u_{2}^{\varepsilon})}{q_{2}(u_{2}^{\varepsilon})} u_{2}^{\varepsilon} |\nabla_{x}(u_{1}^{\varepsilon}) \cdot \nabla_{x}(u_{2}^{\varepsilon}) \, dx. \end{split}$$

Using Young's inequality

$$\int_{\Omega} \frac{q_i'(u_i^{\varepsilon})}{q_i(u_i^{\varepsilon})} u_i^{\varepsilon} \nabla_x(u_i^{\varepsilon}) \cdot \nabla_x(u_j^{\varepsilon}) \, dx \leq \frac{1}{2} \int_{\Omega} \frac{q_i'(u_i^{\varepsilon})}{q_i(u_i^{\varepsilon})} u_i^{\varepsilon} |\nabla_x(u_i^{\varepsilon})|^2 \, dx + \frac{1}{2} \int_{\Omega} \frac{q_i'(u_i^{\varepsilon})}{q_i(u_i^{\varepsilon})} u_i^{\varepsilon} |\nabla_x(u_j^{\varepsilon})|^2 \, dx$$
we get that

$$\begin{split} \frac{d}{dt} H[u^{\varepsilon}(t)] &\leq -\frac{1}{\varepsilon} \int_{\Omega} \Big(\log \left(q_1(u_1^{\varepsilon}) \right) - \log \left(q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \right) \Big) \Big(q_1(u_1^{\varepsilon}) - q_2(u_2^{\varepsilon}) q_3(u_3^{\varepsilon}) \Big) \, dx \\ &- \left(2 - \frac{\delta}{2} \right) \int_{\Omega} \frac{q_1'(u_1^{\varepsilon})}{q_1(u_1^{\varepsilon})} u_1^{\varepsilon} |\nabla_x(u_1^{\varepsilon})|^2 \, dx - \left(2 - \frac{\gamma}{2} \right) \int_{\Omega} \frac{q_2'(u_2^{\varepsilon})}{q_2(u_2^{\varepsilon})} u_2^{\varepsilon} |\nabla_x(u_2^{\varepsilon})|^2 \, dx \\ &- 2 \int_{\Omega} \frac{q_3'(u_3^{\varepsilon})}{q_3(u_3^{\varepsilon})} u_3^{\varepsilon} |\nabla_x(u_3^{\varepsilon})|^2 \, dx - \left(\delta - \frac{\gamma}{2C_1} \right) \int_{\Omega} \frac{q_1'(u_1^{\varepsilon})}{q_1(u_1^{\varepsilon})} u_2^{\varepsilon} |\nabla_x(u_1^{\varepsilon})|^2 \, dx \\ &- \left(\gamma - \frac{\delta C_2}{2} \right) \int_{\Omega} \frac{q_2'(u_2^{\varepsilon})}{q_2(u_2^{\varepsilon})} u_1^{\varepsilon} |\nabla_x(u_2^{\varepsilon})|^2 \, dx \\ &< 0. \end{split}$$

Robustness of the method for cubic diffusivities

Now we study the microscopic model with cubic diffusivities $f_i(u_i^{\varepsilon}) = (u_i^{\varepsilon})^3$, i = 1, 2, 3, where we allow a small pertubation of cross-diffusion terms in u_1^{ε} , u_2^{ε} for $\delta > 0, \gamma > 0$:

$$\begin{cases}
\partial_{t}u_{1}^{\varepsilon} - \Delta_{x}\left(u_{1}^{\varepsilon}\left(\left(u_{1}^{\varepsilon}\right)^{2} + \delta\left(u_{2}^{\varepsilon}\right)^{2}\right)\right) = -\frac{1}{\varepsilon}\left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right), \\
\partial_{t}u_{2}^{\varepsilon} - \Delta_{x}\left(u_{2}^{\varepsilon}\left(\left(u_{2}^{\varepsilon}\right)^{2} + \gamma\left(u_{1}^{\varepsilon}\right)^{2}\right)\right) = +\frac{1}{\varepsilon}\left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right), \\
\partial_{t}u_{3}^{\varepsilon} - \Delta_{x}\left(\left(u_{3}^{\varepsilon}\right)^{3}\right) = +\frac{1}{\varepsilon}\left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right).
\end{cases} (7.2.35)$$

The following lemma shows that our method of obtaining entropies for the limiting cross-diffusion system is robust under small perurbation of cross-diffusion terms also for this model.

Lemma 7.17. Let $\delta > 0$, $\gamma > 0$ and assume that there exist two positive constants $C_1, C_2 > 0$ such that assumption (7.2.33) holds. Moreover, assume that

$$3 - \delta - \frac{\gamma}{C_1} \ge 0,$$
 $3 - \gamma - \delta C_2 \ge 0.$ (7.2.36)

Then the entropy is decreasing along the flow of the pertubed system (7.2.35), i.e.

$$\forall \varepsilon > 0: \quad \frac{d}{dt} H[u^{\varepsilon}(t)] \le 0.$$

Proof. It holds that

$$\begin{split} \frac{d}{dt} H[u^{\varepsilon}(t)] &= \int_{\Omega} \log \left(q_{1}(u_{1}^{\varepsilon})\right) \partial_{t} u_{1}^{\varepsilon} \, dx + \int_{\Omega} \log \left(q_{2}(u_{2}^{\varepsilon})\right) \partial_{t} u_{2}^{\varepsilon} \, dx + \int_{\Omega} \log \left(q_{3}(u_{3}^{\varepsilon})\right) \partial_{t} u_{3}^{\varepsilon} \, dx \\ &= -\frac{1}{\varepsilon} \int_{\Omega} \left(\log \left(q_{1}(u_{1}^{\varepsilon})\right) - \log \left(q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right) \right) \left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon})q_{3}(u_{3}^{\varepsilon})\right) dx \\ &- 3 \int_{\Omega} \frac{q_{3}'(u_{3}^{\varepsilon})}{q_{3}(u_{3}^{\varepsilon})} \left(u_{3}^{\varepsilon}\right)^{2} |\nabla u_{3}^{\varepsilon}|^{2} \, dx \\ &- 3 \int_{\Omega} \frac{q_{1}'(u_{1}^{\varepsilon})}{q_{1}(u_{1}^{\varepsilon})} \left(u_{1}^{\varepsilon}\right)^{2} |\nabla_{x}(u_{1}^{\varepsilon})|^{2} \, dx - 3 \int_{\Omega} \frac{q_{2}'(u_{2}^{\varepsilon})}{q_{2}(u_{2}^{\varepsilon})} \left(u_{2}^{\varepsilon}\right)^{2} |\nabla_{x}(u_{2}^{\varepsilon})|^{2} \, dx \\ &- \delta \int_{\Omega} \frac{q_{1}'(u_{1}^{\varepsilon})}{q_{1}(u_{1}^{\varepsilon})} \left(u_{2}^{\varepsilon}\right)^{2} |\nabla_{x}(u_{1}^{\varepsilon})|^{2} \, dx \\ &- \gamma \int_{\Omega} \frac{q_{2}'(u_{2}^{\varepsilon})}{q_{2}(u_{2}^{\varepsilon})} \left(u_{1}^{\varepsilon}\right)^{2} |\nabla_{x}(u_{2}^{\varepsilon})|^{2} \, dx - 2\delta \int_{\Omega} \frac{q_{1}'(u_{1}^{\varepsilon})}{q_{1}(u_{1}^{\varepsilon})} u_{1}^{\varepsilon} u_{2}^{\varepsilon} \nabla_{x}(u_{1}^{\varepsilon}) \cdot \nabla_{x}(u_{2}^{\varepsilon}) \, dx \\ &- 2\gamma \int_{\Omega} \frac{q_{2}'(u_{2}^{\varepsilon})}{q_{2}(u_{2}^{\varepsilon})} u_{1}^{\varepsilon} u_{2}^{\varepsilon} \nabla_{x}(u_{1}^{\varepsilon}) \cdot \nabla_{x}(u_{2}^{\varepsilon}) \, dx. \end{split}$$

Using Young's inequality in the following way

$$\int_{\Omega} \frac{q_i'(u_i^{\varepsilon})}{q_i(u_i^{\varepsilon})} u_i^{\varepsilon} u_j^{\varepsilon} \nabla_x(u_i^{\varepsilon}) \cdot \nabla_x(u_j^{\varepsilon}) dx
\leq \frac{1}{2} \int_{\Omega} \frac{q_i'(u_i^{\varepsilon})}{q_i(u_i^{\varepsilon})} (u_i^{\varepsilon})^2 |\nabla_x(u_i^{\varepsilon})|^2 dx + \frac{1}{2} \int_{\Omega} \frac{q_i'(u_i^{\varepsilon})}{q_i(u_i^{\varepsilon})} (u_j^{\varepsilon})^2 |\nabla_x(u_j^{\varepsilon})|^2 dx$$

leads to

$$\frac{d}{dt}H[u^{\varepsilon}(t)] \leq -\frac{1}{\varepsilon} \int_{\Omega} \left(\log \left(q_{1}(u_{1}^{\varepsilon}) \right) - \log \left(q_{2}(u_{2}^{\varepsilon}) q_{3}(u_{3}^{\varepsilon}) \right) \right) \left(q_{1}(u_{1}^{\varepsilon}) - q_{2}(u_{2}^{\varepsilon}) q_{3}(u_{3}^{\varepsilon}) \right) dx
- 3 \int_{\Omega} \frac{q_{3}'(u_{3}^{\varepsilon})}{q_{3}(u_{3}^{\varepsilon})} u_{3}^{\varepsilon} |\nabla_{x}(u_{3}^{\varepsilon})|^{2} dx
- \left(3 - \delta - \frac{\gamma}{C_{1}} \right) \int_{\Omega} \frac{q_{1}'(u_{1}^{\varepsilon})}{q_{1}(u_{1}^{\varepsilon})} (u_{1}^{\varepsilon})^{2} |\nabla_{x}(u_{1}^{\varepsilon})|^{2} dx
- \left(3 - \gamma - \delta C_{2} \right) \int_{\Omega} \frac{q_{2}'(u_{2}^{\varepsilon})}{q_{2}(u_{2}^{\varepsilon})} (u_{2}^{\varepsilon})^{2} |\nabla_{x}(u_{2}^{\varepsilon})|^{2} dx
- \delta \int_{\Omega} \frac{q_{1}'(u_{1}^{\varepsilon})}{q_{1}(u_{1}^{\varepsilon})} (u_{2}^{\varepsilon})^{2} |\nabla_{x}(u_{1}^{\varepsilon})|^{2} dx - \gamma \int_{\Omega} \frac{q_{2}'(u_{2}^{\varepsilon})}{q_{2}(u_{2}^{\varepsilon})} (u_{1}^{\varepsilon})^{2} |\nabla_{x}(u_{2}^{\varepsilon})|^{2} dx
< 0$$

7.3. Rigorous limit from four species with one fast and one slow reaction

In this section we prove the rigorous limit of a reaction-diffusion system with one fast reaction and one slow reaction to a limiting cross-diffusion system when the fast-reaction rate $1/\varepsilon$ tends to infinity.

7.3.1. Model for four species with one fast and one slow reaction

Chemical system considered:

Fast reaction (rate $1/\varepsilon$):

$$A_1 \quad \rightleftharpoons \quad A_2 + A_3,$$

Slow reaction (rate 1):

$$A_1 + A_3 \quad \rightleftharpoons \quad A_4.$$

Corresponding system of PDEs for the concentrations $u_i^{\varepsilon} := u_i^{\varepsilon}(t, x) \ge 0$ of species A_i for i = 1, 2, 3, 4 with diffusivities $d_i > 0$:

$$\begin{cases}
\partial_{t}u_{1}^{\varepsilon} - d_{1} \Delta_{x}u_{1}^{\varepsilon} = \frac{1}{\varepsilon} \left(u_{2}^{\varepsilon} u_{3}^{\varepsilon} - u_{1}^{\varepsilon} \right) + \left(u_{4}^{\varepsilon} - u_{1}^{\varepsilon} u_{3}^{\varepsilon} \right) \\
\partial_{t}u_{2}^{\varepsilon} - d_{2} \Delta_{x}u_{2}^{\varepsilon} = -\frac{1}{\varepsilon} \left(u_{2}^{\varepsilon} u_{3}^{\varepsilon} - u_{1}^{\varepsilon} \right) \\
\partial_{t}u_{3}^{\varepsilon} - d_{3} \Delta_{x}u_{3}^{\varepsilon} = -\frac{1}{\varepsilon} \left(u_{2}^{\varepsilon} u_{3}^{\varepsilon} - u_{1}^{\varepsilon} \right) + \left(u_{4}^{\varepsilon} - u_{1}^{\varepsilon} u_{3}^{\varepsilon} \right) \\
\partial_{t}u_{4}^{\varepsilon} - d_{4} \Delta_{x}u_{4}^{\varepsilon} = -\left(u_{4}^{\varepsilon} - u_{1}^{\varepsilon} u_{3}^{\varepsilon} \right)
\end{cases} (7.3.1)$$

together with initial data

$$u_i^{\varepsilon}(0,x) = u_i^{in}(x) \ge 0, \quad i = 1, 2, 3, 4, \quad x \in \Omega,$$

and homogeneous Neumann boundary conditions on the boundary $\partial\Omega$ of a bounded smooth open set Ω of \mathbb{R}^N

$$n(x) \cdot \nabla_x u_i^{\varepsilon}(t, x) = 0, \quad i = 1, 2, 3, 4, \quad x \in \partial\Omega.$$

Formal limit: We are interested in the rigorous limit $\varepsilon \to 0$, with u_i satisfying the following reaction-cross-diffusion system:

$$\begin{cases}
 u_2 u_3 - u_1 = 0, \\
 \partial_t (u_1 + u_2) - \Delta_x (d_1 u_1 + d_2 u_2) = u_4 - u_1 u_3, \\
 \partial_t (u_1 + u_3) - \Delta_x (d_1 u_1 + d_3 u_3) = 2 (u_4 - u_1 u_3). \\
 \partial_t u_4 - d_4 \Delta_x u_4 = -(u_4 - u_1 u_3).
\end{cases}$$
(7.3.2)

7.3.2. Existence of solutions

Theorem 7.18. For any $\varepsilon > 0$, there exists a weak solution $u^{\varepsilon} \in L^{2}_{loc}([0, \infty); L^{2}(\Omega; \mathbb{R}^{4}))$ to system (7.3.1).

Proof. This follows from [49, Theorem 4.1].

7.3.3. Fast-reaction limit to a cross-diffusion system

In this section we will prove the rigorous limit of the reaction-diffusion system (7.3.1) to the limiting reaction-cross-diffusion system (7.3.2). The idea of the proof is, that by duality, we get an $L^2(\log L)^2$ estimate, which directly implies the uniform integrability of the nonlinearities. By compactness properties (Aubin-Lions lemma), we get $L^{4/3}(\Omega_T)$ -compactness of (u_i^{ε}) . Thus we get, up to a subsequence, convergence of the right-hand-side in $L^1(\Omega_T)$.

Theorem 7.19. Let $\Omega \subseteq \mathbb{R}^N$ be a bounded regular open set of \mathbb{R}^N , and let for any $\varepsilon > 0$, $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}, u_4^{\varepsilon}$ denote a weak solution of the reaction-diffusion system (7.3.1) with initial data $u^{in} \log(u^{in}) \in L^2(\Omega)$.

Then the following holds: If $\varepsilon \to 0$, there exists a subsequence of $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}, u_4^{\varepsilon}$ (which we still denote by $u_1^{\varepsilon}, u_2^{\varepsilon}, u_3^{\varepsilon}, u_4^{\varepsilon}$), which converges to u_1, u_2, u_3, u_4 in $L^1_{loc}([0, \infty); L^1(\Omega))$.

Moreover, this limit is a weak solution of the reaction-cross-diffusion system (7.3.2) belonging to $L^2_{loc}([0,\infty);L^2(\Omega))$.

7.3.4. A priori estimates

Using the entropy

$$H[u^{\varepsilon}(t)] = \sum_{i=1}^{4} \int_{\Omega} \left(u_i^{\varepsilon} (\log u_i^{\varepsilon} - 1) + 1 \right) dx, \tag{7.3.3}$$

the entropy dissipation

$$D_{\varepsilon}[u^{\varepsilon}(t)] = -\frac{d}{dt}H[u^{\varepsilon}(t)]$$

along the flow of system (7.3.1) has the following form

$$D_{\varepsilon}[u^{\varepsilon}(t)] = \sum_{i=1}^{4} d_{i} \int_{\Omega} \frac{|\nabla_{x} u_{i}^{\varepsilon}|^{2}}{u_{i}^{\varepsilon}} dx + \frac{1}{\varepsilon} \int_{\Omega} (u_{2}^{\varepsilon} u_{3}^{\varepsilon} - u_{1}^{\varepsilon}) (\log(u_{2}^{\varepsilon} u_{3}^{\varepsilon}) - \log u_{1}^{\varepsilon}) dx + \int_{\Omega} (u_{4}^{\varepsilon} - u_{1}^{\varepsilon} u_{3}^{\varepsilon}) (\log u_{4}^{\varepsilon} - \log(u_{1}^{\varepsilon} u_{3}^{\varepsilon})) dx.$$

Integrating this in time leads to the standard entropy-entropy-dissipation inequality, where the right-hand side is uniform in ε :

$$\sup_{t \in [0,T]} H[u^{\varepsilon}(t)] + \int_0^T D_{\varepsilon}[u^{\varepsilon}(s)] ds \le H[u^{in}]. \tag{7.3.4}$$

Lemma 7.20 (Entropy estimates). Recalling that $u^{in} \log(u^{in}) \in L^2(\Omega)$, the following estimates hold for u_i^{ε} with i = 1, 2, 3, 4:

$$||u_i^{\varepsilon}||_{L^{\infty}(0,T;L\log L)} + ||(u_i^{\varepsilon})^{1/2}||_{L^{2}(0,T;H^{1}(\Omega))} \leq C_T,$$

$$\int_0^T \int_{\Omega} (u_4^{\varepsilon} - u_1^{\varepsilon} u_3^{\varepsilon})(\log u_4^{\varepsilon} - \log(u_1^{\varepsilon} u_3^{\varepsilon})) \leq C_T,$$

$$\int_0^T \int_{\Omega} (u_1^{\varepsilon} - u_2^{\varepsilon} u_3^{\varepsilon})(\log u_1^{\varepsilon} - \log(u_2^{\varepsilon} u_3^{\varepsilon})) \leq \varepsilon C_T.$$

Proof. This follows directly from the entropy-entropy-dissipation inequality (7.3.4) by using the fact that $4|\nabla_x(u_i^\varepsilon)^{1/2}|^2 = |\nabla_x u_i^\varepsilon|^2/u_i^\varepsilon$.

The next lemma can be found in [49, Theorem 8.1].

Lemma 7.21 (Duality lemma). If h is a strong (i.e. $C^2([0,T] \times \overline{\Omega})$) nonnegative solution to the parabolic problem

$$\begin{cases} \partial_t h - \Delta_x (Ah) \le 0, \\ \nabla_x (Ah) \cdot n(x) = 0 \quad on \quad \partial \Omega, \\ h(0, \cdot) = h^{in}(\cdot) \in L^2(\Omega), \end{cases}$$
 (7.3.5)

where A is bounded from below and from above by some positive constants $C_1, C_2 > 0$ in the following way

$$0 < C_1 \le A(x,t) \le C_2$$

then

$$||h||_{L^2(\Omega_T)} \le C_T ||h(0)||_{L^2(\Omega)}.$$

Proof. We integrate $\partial_t h - \Delta_x (Ah) \leq 0$ against a test function $w \geq 0$ which is a defined as a positive solution of the dual problem

$$\begin{cases}
-\left(\partial_t w + A\Delta_x w\right) = \Theta \in C_0^{\infty}(\Omega_T), & \Theta \ge 0, \\
n(x) \cdot \nabla_x w(t, x) = 0 & \text{on } \partial\Omega, \\
w(T, \cdot) = 0.
\end{cases}$$
(7.3.6)

This leads to

$$\int_0^T \int_{\Omega} (w \partial_t h - w \Delta_x(Ah)) \, dx dt \le 0.$$

Integrating by parts yields

$$-\int_0^T \int_{\Omega} h \partial_t w \, dx dt - \int_{\Omega} (wh)(0) \, dx - \int_0^T \int_{\Omega} Ah \Delta_x w \, dx dt \le 0,$$

and using the dual problem (7.3.6) leads to

$$\int_0^T \int_{\Omega} h\Theta \, dx dt \le \int_{\Omega} (hw)(0) \, dx.$$

Next we multiply the dual problem (7.3.6) by $\Delta_x w$ and integrate, which yields

$$-\frac{1}{2}\int_0^T \partial_t \int_{\Omega} |\nabla_x w(t)|^2 dx dt + \int_0^T \int_{\Omega} A(\Delta_x w)^2 dx dt = -\int_0^T \int_{\Omega} \Theta \Delta_x w dx.$$

Using Young's inequality leads to

$$\frac{1}{2} \int_{\Omega} |\nabla_x w(0)|^2 dx + \int_0^T \int_{\Omega} A(\Delta_x w)^2 dx dt
\leq \frac{C_1}{2} \int_0^T \int_{\Omega} (\Delta_x w)^2 dx dt + C(C_1) \int_0^T \int_{\Omega} \Theta^2 dx dt,$$

where we recall that $A \geq C_1 > 0$. Thus

$$\int_{0}^{T} \int_{\Omega} (\Delta_{x} w)^{2} dx dt \leq \widetilde{C} \int_{0}^{T} \int_{\Omega} \Theta^{2} dx dt.$$
 (7.3.7)

Let us now estimate w(0) in $L^2(\Omega)$. For this we write

$$w(0) = -\int_0^T \partial_t w \, dt.$$

Thus by using the dual problem we get that

$$\int_{\Omega} (w(0))^2 dx = -\int_0^T \int_{\Omega} (\partial_t w) w(0) dx dt = \int_0^T \int_{\Omega} w(0) \Theta dx dt + \int_0^T \int_{\Omega} w(0) A \Delta_x w dx dt.$$

Using Young's inequality, we get that

$$\int_{\Omega} w(0)^2 dx \le C \int_{0}^{T} \int_{\Omega} \Theta^2 dx dt.$$

Putting all this estimates together leads to

$$\int_{0}^{T} \int_{\Omega} h\Theta \, dx dt \le \int_{\Omega} h(0)w(0) \, dx$$

$$\le \|w(0)\|_{L^{2}(\Omega)} \|h(0)\|_{L^{2}(\Omega)}$$

$$\le C \|h(0)\|_{L^{2}(\Omega)} \int_{0}^{T} \int_{\Omega} \Theta^{2} \, dx dt.$$

Thus by duality we get that

$$||h||_{L^2(\Omega_T)} \le C||h(0)||_{L^2(\Omega)},$$

which finishes the proof.

Lemma 7.22 (Duality estimate). For i = 1, 2, 3, 4 it holds that

$$||u_i^{\varepsilon}||_{L^2(\log L)^2(\Omega_T)} \le C_T.$$

191

Proof. Set

$$\begin{split} h_{\varepsilon} &:= \sum_{i=1}^{4} (u_{i}^{\varepsilon} \ln u_{i}^{\varepsilon} - u_{i}^{\varepsilon} + 1), \\ w_{\varepsilon} &:= \sum_{i=1}^{4} d_{i} \left(u_{i}^{\varepsilon} \ln u_{i}^{\varepsilon} - u_{i}^{\varepsilon} + 1 \right), \\ j_{\varepsilon} &:= \sum_{i=1}^{4} d_{i} \frac{|\nabla_{x} u_{i}^{\varepsilon}|^{2}}{u_{i}^{\varepsilon}} + \frac{1}{\varepsilon} \left(u_{2}^{\varepsilon} u_{3}^{\varepsilon} - u_{1}^{\varepsilon} \right) \left(\ln(u_{2}^{\varepsilon} u_{3}^{\varepsilon}) - \ln u_{1}^{\varepsilon} \right) \\ &+ \left(u_{4}^{\varepsilon} - u_{1}^{\varepsilon} u_{3}^{\varepsilon} \right) \left(\ln u_{4}^{\varepsilon} - \ln(u_{1}^{\varepsilon} u_{3}^{\varepsilon}) \right), \end{split}$$

then system (7.3.1) exhibits the following entropy structure

$$\partial_t h_{\varepsilon} - \Delta_x w_{\varepsilon} = -j_{\varepsilon}.$$

This implies that the entropy density h_{ε} is a strong (i.e. $C^2([0,T]\times\overline{\Omega})$) nonnegative solution to the parabolic problem

$$\begin{cases} \partial_t h_{\varepsilon} - \Delta_x \left(A_{\varepsilon} h_{\varepsilon} \right) \leq 0, & \text{with } A_{\varepsilon} = w_{\varepsilon} / h_{\varepsilon}, \\ \nabla_x (A_{\varepsilon} h_{\varepsilon}) \cdot n = 0 & \text{on } \partial \Omega, \\ h_{\varepsilon}(0, \cdot) = h^{in}(\cdot) \in L^2(\Omega), \end{cases}$$

$$(7.3.8)$$

where $h^{in} > 0$ is independent of ε , and A_{ε} is bounded from below and from above in the following way

$$0 < \min_{1 \le i \le 4} d_i \le A_{\varepsilon} \le \max_{1 \le i \le 4} d_i.$$

Thus we can apply Lemma 7.21 to h_{ε} and get the desired result.

Lemma 7.23. Recalling that $u^{in} \log (u^{in}) \in L^2(\Omega)$, we get the following estimates:

$$\begin{aligned} \|(u_{1}^{\varepsilon})^{1/2} - (u_{2}^{\varepsilon}u_{3}^{\varepsilon})^{1/2}\|_{L^{2}(\Omega_{T})} & \leq \sqrt{\varepsilon}C_{T}, \\ \|u_{1}^{\varepsilon} - u_{2}^{\varepsilon}u_{3}^{\varepsilon}\|_{L^{1}(\Omega_{T})} & \leq \sqrt{\varepsilon}C_{T}, \\ \|(u_{4}^{\varepsilon})^{1/2} - (u_{1}^{\varepsilon}u_{3}^{\varepsilon})^{1/2}\|_{L^{2}(\Omega_{T})} & \leq C_{T}, \\ \|u_{4}^{\varepsilon} - u_{1}^{\varepsilon}u_{3}^{\varepsilon}\|_{L^{1}(\Omega_{T})} & \leq C_{T}. \end{aligned}$$

Proof. The elementary inequality

$$4\left|\sqrt{a} - \sqrt{b}\right|^2 \le (a - b)(\log a - \log b), \quad a, b \ge 0,$$

together with Lemma 7.22 yields

$$4\|(u_1^{\varepsilon})^{1/2} - (u_2^{\varepsilon}u_3^{\varepsilon})^{1/2}\|_{L^2(\Omega_T)}^2 \le 4\int_0^T \int_{\Omega} \left|(u_1^{\varepsilon})^{1/2} - (u_2^{\varepsilon}u_3^{\varepsilon})^{1/2}\right|^2 dxdt$$

$$\leq \int_0^T \int_{\Omega} (u_1^{\varepsilon} - u_2^{\varepsilon} u_3^{\varepsilon}) (\log(u_1^{\varepsilon}) - \log(u_2^{\varepsilon} u_3^{\varepsilon})) \, dx dt$$

$$\leq \varepsilon C_T,$$

which shows the first estimate. By using the Cauchy-Schwarz inequality, we get that

$$\begin{aligned} &\|u_1^{\varepsilon} - u_2^{\varepsilon} u_3^{\varepsilon}\|_{L^1(\Omega_T)} = \int_0^T \int_{\Omega} \left((u_1^{\varepsilon})^{1/2} - (u_2^{\varepsilon} u_3^{\varepsilon})^{1/2} \right) \left((u_1^{\varepsilon})^{1/2} + (u_2^{\varepsilon} u_3^{\varepsilon})^{1/2} \right) \, dx dt \\ &\leq \left(\int_0^T \int_{\Omega} \left((u_1^{\varepsilon})^{1/2} - (u_2^{\varepsilon} u_3^{\varepsilon})^{1/2} \right)^2 \, dx dt \right)^{1/2} \left(\int_0^T \int_{\Omega} \left((u_1^{\varepsilon})^{1/2} + (u_2^{\varepsilon} u_3^{\varepsilon})^{1/2} \right)^2 \, dx dt \right)^{1/2}. \end{aligned}$$

The boundedness of the last factor by Lemma 7.22 and the previous estimate lead to

$$||u_1^{\varepsilon} - u_2^{\varepsilon} u_3^{\varepsilon}||_{L^1(\Omega_T)} \le \sqrt{\varepsilon} C_T.$$

Similarly,

$$4\|(u_4^{\varepsilon})^{1/2} - (u_1^{\varepsilon}u_3^{\varepsilon})^{1/2}\|_{L^2(\Omega_T)}^2 \le 4\int_0^T \int_{\Omega} \left| (u_4^{\varepsilon})^{1/2} - (u_1^{\varepsilon}u_3^{\varepsilon})^{1/2} \right|^2 dxdt$$

$$\le \int_0^T \int_{\Omega} (u_4^{\varepsilon} - u_1^{\varepsilon}u_3^{\varepsilon}) (\log(u_4^{\varepsilon}) - \log(u_1^{\varepsilon}u_3^{\varepsilon})) dxdt$$

$$\le C_T,$$

and again by using the Cauchy-Schwarz inequality, Lemma 7.22 and the previous estimate, we get that

$$||u_4^{\varepsilon} - u_1^{\varepsilon} u_3^{\varepsilon}||_{L^1(\Omega_T)} \le C_T.$$

Lemma 7.24. It holds that

$$\|\nabla_x u_i^{\varepsilon}\|_{L^{4/3}(\Omega_T)} \le C_T.$$

Proof. Using the fact that $\nabla_x u_i^{\varepsilon} = 2\sqrt{u_i^{\varepsilon}}(\nabla_x \sqrt{u_i^{\varepsilon}})$ with $\sqrt{u_i^{\varepsilon}}$ bounded in $L^4(\Omega \times [0,T])$ due to the duality estimate in Lemma 7.22 and $\nabla_x \sqrt{u_i^{\varepsilon}}$ bounded in $L^2(\Omega \times [0,T])$ due to the entropy estimate in Lemma 7.20, we get by using Hölder's inequality with p = 3, q = 3/2 and 1/p + 1/q = 1, that the product $\nabla_x u_i^{\varepsilon}$ is bounded in $L^{4/3}(\Omega \times [0,T])$:

$$\begin{split} \int_0^T \int_{\Omega} |\nabla_x u_i^{\varepsilon}|^{4/3} \, dx dt &= 2^{4/3} \int_0^T \int_{\Omega} |\sqrt{u_i^{\varepsilon}}|^{4/3} |\nabla_x \sqrt{u_i^{\varepsilon}}|^{4/3} \, dx dt \\ &\leq 2^{4/3} \left(\int_0^T \int_{\Omega} |\sqrt{u_i^{\varepsilon}}|^4 \, dx dt \right)^{1/3} \left(\int_0^T \int_{\Omega} |\nabla_x \sqrt{u_i^{\varepsilon}}|^2 \, dx dt \right)^{2/3} \\ &< C_T. \end{split}$$

7.3.5. Strong compactness

The difficulty in proving strong compactness for the species u_i^{ε} in system (7.3.1) relies in the fact that we cannot use the Aubin-Lions lemma [130] for u_i^{ε} directly, due to blow-up in the time dervatives $\partial_t u_i^{\varepsilon}$ for $\varepsilon \to 0$. However, we are still able to prove strong compactness for the u_i^{ε} with the help of the following lemmata.

Lemma 7.25. The sequences $(u_1^{\varepsilon} + u_2^{\varepsilon})$ and $(u_1^{\varepsilon} + u_3^{\varepsilon})$ are relatively compact in $L^{4/3}(\Omega_T)$.

Proof. For the sums $u_1^{\varepsilon} + u_2^{\varepsilon}$ and $u_1^{\varepsilon} + u_3^{\varepsilon}$ in system (7.3.1), we see that the right-hand side does not blow-up for $\varepsilon \to 0$, thus we apply the Aubin-Lions lemma to these terms and get strong compactness for them. The terms $u_1^{\varepsilon} + u_2^{\varepsilon}$ and $u_1^{\varepsilon} + u_3^{\varepsilon}$ are bounded in $L^{4/3}(0,T;W^{1,4/3}(\Omega))$ due to Lemma 7.24. For the time derivative $\partial_t(u_1^{\varepsilon} + u_2^{\varepsilon})$, we have that

$$\partial_t (u_1^\varepsilon + u_2^\varepsilon) = d_1 \Delta_x u_1^\varepsilon + d_2 \Delta_x u_2^\varepsilon + (u_4^\varepsilon - u_1^\varepsilon u_3^\varepsilon),$$

thus $\partial_t (u_1^{\varepsilon} + u_2^{\varepsilon})$ is bounded in

$$L^{2}(0,T;W^{-2,2}(\Omega)) \cap L^{1}(0,T;L^{1}(\Omega)) \subseteq L^{1}(0,T;W^{-2,1}(\Omega))$$

due to the duality estimate in Lemma 7.22, and the same holds for $\partial_t(u_1^{\varepsilon} + u_3^{\varepsilon})$.

Summarizing, for $f^{\varepsilon} = u_1^{\varepsilon} + u_2^{\varepsilon}$ and $f^{\varepsilon} = u_1^{\varepsilon} + u_3^{\varepsilon}$ respectively, the following estimates hold: (f^{ε}) is bounded in $L^{4/3}(0,T;W^{1,4/3}(\Omega))$, and $(\partial_t f^{\varepsilon})$ is bounded in $L^1(0,T;W^{-2,1}(\Omega))$, and thus by the standard Aubin-Lions lemma [130] we get that (f^{ε}) is relatively compact in $L^{4/3}(\Omega_T)$.

Lemma 7.26. The function

$$\varphi: \begin{cases} \mathbb{R}_+^2 \to \mathbb{R}_+^2 \\ (x,y) \mapsto ((1+y)x, (1+x)y) \end{cases}$$

is a homeomorphism on $(\mathbb{R}_+)^2$.

Proof. The function φ can be written as

$$\varphi: \begin{cases} \mathbb{R}_+^2 \to \mathbb{R}_+^2 \\ (x,y) \mapsto (a(x,y)x, b(x,y)y), \end{cases}$$

where $a: \mathbb{R}^2_+ \to \mathbb{R}_+$, a(x,y) = 1 + y and $b: \mathbb{R}^2_+ \to \mathbb{R}_+$, b(x,y) = 1 + x. Thus we see that a and b are continuous and lower bounded by 1. Moreover, on $\mathbb{R}_+ \times \mathbb{R}_+$ it holds that φ is strictly increasing (in the sense that each component is strictly increasing w.r.t. each component), and on $\mathbb{R}^*_+ \times \mathbb{R}^*_+$ it holds that φ is \mathcal{C}^1 . The determinant of the Jacobian of φ is strictly positive on $\mathbb{R}^*_+ \times \mathbb{R}^*_+$:

$$\det D(\varphi) = 1 + x + y > 0 \quad \forall x > 0, \forall y > 0.$$

Thus we can apply [51, Proposition 5.1] and get that φ is a homeomorphism on \mathbb{R}^2_+ . \square

Lemma 7.27. If $(a_n), (b_n)$ and (c_n) with $n \in \mathbb{N}$ are sequences in \mathbb{R}_+ satisfying for $n \to \infty$ that

$$(a_n + b_n) \to B$$
 a.e. in Ω_T ,
 $(a_n + c_n) \to C$ a.e. in Ω_T , (7.3.9)

and additionally

$$(a_n - b_n c_n) \to 0$$
 a.e. in Ω_T ,

then it holds that

$$\begin{array}{llll} (a_n) \rightarrow a & a.e. \ in \ \Omega_T, \\ (b_n) \rightarrow b & a.e. \ in \ \Omega_T, \\ (c_n) \rightarrow c & a.e. \ in \ \Omega_T, \end{array}$$

with

$$b := \psi_2(B, C), \quad c := \psi_3(B, C), \quad a := bc,$$
 (7.3.10)

where $\psi = (\psi_2, \psi_3) := \varphi^{-1}$, and φ denotes the homeomorphism introduced in Lemma 7.26

$$\varphi: \begin{cases} \mathbb{R}_{+}^{2} \to \mathbb{R}_{+}^{2} \\ (x,y) \mapsto (x(1+y), y(1+x)). \end{cases}$$
 (7.3.11)

Proof. Due to the assumptions, we can write $a_n = b_n c_n + \varepsilon_n$ with $\varepsilon_n \to 0$ for $n \to \infty$, and get that $b_n c_n + b_n + \varepsilon_n \to B$ and $b_n c_n + c_n + \varepsilon_n \to C$. Thus it holds that

$$b_n(1+c_n) \to B$$
 a.e. in Ω_T ,
 $c_n(1+b_n) \to C$ a.e. in Ω_T . (7.3.12)

Since φ is bijective with the inverse $\psi := \varphi^{-1}$, it holds that

$$(b_n, c_n) = \psi(b_n(1+c_n), c_n(1+b_n))$$
 for all $n \in \mathbb{N}$,

and since ψ is continuous, it holds a.e. in Ω_T that

$$(b,c) = \lim_{n \to \infty} (b_n, c_n) = \lim_{n \to \infty} \psi \left(b_n(1 + c_n), c_n(1 + b_n) \right) = \left(\psi_2 \left(B, C \right), \psi_3 \left(B, C \right) \right),$$

with B, C defined in (7.3.9). This shows that $b_n \to b$ a.e. in Ω_T , and $c_n \to c$ a.e. in Ω_T with b and c defined in (7.3.10). Finally, since $a_n - b_n c_n \to 0$ a.e. in Ω_T and $b_n c_n \to bc$ a.e. in Ω_T , we get that $a_n \to bc$ a.e. in Ω_T , which finishes the proof.

Theorem 7.28. For i = 1, 2, 3, 4 it holds that

$$u_i^{\varepsilon} \to u_i \text{ strongly in } L^2(\Omega_T).$$

Proof. Due to Lemma 7.25 we have that $u_1^{\varepsilon} + u_2^{\varepsilon} \to B$ a.e. in Ω_T and $u_1^{\varepsilon} + u_3^{\varepsilon} \to C$ a.e. in Ω_T , and due to Lemma 7.23 it holds that $(u_1^{\varepsilon} - u_2^{\varepsilon} u_3^{\varepsilon}) \to 0$ a.e. in Ω_T . Thus we can apply Theorem 7.27, and get that $u_i^{\varepsilon} \to u_i$ a.e. in Ω_T .

Thus we have for i = 1, 2, 3, 4 that

$$u_i^{\varepsilon} \to u_i$$
 a.e. in Ω_T ,
 $\|u_i^{\varepsilon}\|_{L^2(\log L)^2(\Omega_T)} \le C_T$,

thus by [105, Lemma 1.3] we get that

$$u_i^{\varepsilon} \to u_i$$
 strongly in $L^2(\Omega_T)$,

which finishes the proof.

Corollary 7.29. For $\varepsilon \to 0$ it holds that $\left(u_i^{\varepsilon} u_j^{\varepsilon}\right) \to u_i u_j$ strongly in $L^1(\Omega_T)$ for all $i, j \in \{1, 2, 3, 4\}$.

Proof. Define $g^{\varepsilon} := u_i^{\varepsilon} u_j^{\varepsilon}$ and $g := u_i u_j$ for i, j in $\{1, 2, 3, 4\}$. Then it holds that $g^{\varepsilon} \to g$ a.e. in Ω_T , and that $\|g^{\varepsilon}\|_{L^1(\log L)} \leq C_T$ due to the fact that

$$u_i^\varepsilon u_j^\varepsilon \log(u_i^\varepsilon u_j^\varepsilon) = u_i^\varepsilon \left(u_j^\varepsilon \log(u_j^\varepsilon)\right) + u_j^\varepsilon \left(u_i^\varepsilon \log(u_i^\varepsilon)\right) \ \text{is bounded in } L^1(\Omega_T).$$

Thus [105, Lemma 1.3] implies that $g^{\varepsilon} \to g$ strongly in $L^1(\Omega_T)$, and thus $\left(u_i^{\varepsilon} u_j^{\varepsilon}\right) \to u_i u_j$ strongly in $L^1(\Omega_T)$ for all $i, j \in \{1, 2, 3, 4\}$.

7.3.6. Passing to the limit

The estimates derived in the last subsections are sufficient to pass to the limit $\varepsilon \to 0$ in the weak formulation of the reaction-diffusion system (7.3.1). Thus we are now able to prove Theorem 7.19.

Proof of Theorem 7.19. The only step which remains to prove is to pass to the limit $\varepsilon \to 0$ for the terms $u_1^{\varepsilon} + u_2^{\varepsilon}$, $u_1^{\varepsilon} + u_2^{\varepsilon}$, u_2^{ε} and u_4^{ε} in the weak formulations of the reaction-diffusion system (7.3.1). First we write down the equation satisfied for $u_1^{\varepsilon} + u_2^{\varepsilon}$:

$$\begin{cases} \partial_t (u_1^{\varepsilon} + u_2^{\varepsilon}) - d_1 \, \Delta_x u_1^{\varepsilon} - d_2 \, \Delta_x u_1^{\varepsilon} = (u_4^{\varepsilon} - u_1^{\varepsilon} \, u_3^{\varepsilon}), \\ (u_1^{\varepsilon} + u_2^{\varepsilon})(0, x) = u_1^{in} + u_2^{in}, \\ n(x) \cdot \nabla_x (u_1^{\varepsilon} + u_2^{\varepsilon})(t, x) = 0. \end{cases}$$

The weak form can be written in the following way: For any testfunction $\psi = \psi(t, x)$ in the set

$$\big\{\psi\in C^2([0,\infty)\times\overline{\Omega}): \psi(t)=0 \text{ for } t\geq T \text{ and } n(x)\cdot\nabla_x\psi(t,x)=0\big\},$$

we have that

$$\begin{split} -\int_{\Omega_T} (u_4^\varepsilon - u_1^\varepsilon u_3^\varepsilon) \psi(t,x) \, dx dt &= \int_{\Omega_T} (u_1^\varepsilon + u_2^\varepsilon) \partial_t \psi(t,x) \, dx dt + \int_{\Omega} (u_1^{in} + u_2^{in})(x) \partial_t \psi(0,x) \, dx \\ &+ \int_{\Omega_T} (d_1 u_1^\varepsilon + d_2 u_2^\varepsilon) \Delta_x \psi(t,x) \, dx dt. \end{split}$$

Now, by using the following results obtained above, namely that

$$u_i^{\varepsilon} \to u_i \text{ strongly in } L^2(\Omega_T), \quad i = 1, 2, 3, 4,$$

 $u_i^{\varepsilon} u_j^{\varepsilon} \to u_i u_j \text{ strongly in } L^1(\Omega_T), \quad i, j \in \{1, 2, 3, 4\},$ (7.3.13)

we can pass to the limit $\varepsilon \to 0$ and get that

$$-\int_{\Omega_T} (u_4 - u_1 u_3) \psi(t, x) \, dx dt = \int_{\Omega_T} (u_1 + u_2) \partial_t \psi(t, x) \, dx dt + \int_{\Omega} (u_1^{in} + u_2^{in})(x) \partial_t \psi(0, x) \, dx + \int_{\Omega_T} (d_1 u_1 + d_2 u_2) \Delta_x \psi(t, x) \, dx dt.$$

The procedure for $u_1^{\varepsilon} + u_3^{\varepsilon}$ is exactly the same, thus

$$-2\int_{\Omega_{T}} (u_{4}^{\varepsilon} - u_{1}^{\varepsilon} u_{3}^{\varepsilon}) \psi(t, x) dx dt = \int_{\Omega_{T}} (u_{1}^{\varepsilon} + u_{3}^{\varepsilon}) \partial_{t} \psi(t, x) dx dt + \int_{\Omega} (u_{1}^{in} + u_{3}^{in})(x) \partial_{t} \psi(0, x) dx + \int_{\Omega_{T}} (d_{1} u_{1}^{\varepsilon} + d_{3} u_{3}^{\varepsilon}) \Delta_{x} \psi(t, x) dx dt.$$

converges for $\varepsilon \to 0$ to the limiting system

$$-2\int_{\Omega_{T}} (u_{4} - u_{1}u_{3})\psi(t, x)dxdt = \int_{\Omega_{T}} (u_{1} + u_{3})\partial_{t}\psi(t, x)dxdt + \int_{\Omega} (u_{1}^{in} + u_{3}^{in})(x)\partial_{t}\psi(0, x)dx$$
$$+ \int_{\Omega_{T}} (d_{1}u_{1} + d_{3}u_{3})\Delta_{x}\psi(t, x)dxdt.$$

For u_4^{ε} , we obtain by the same arguments that

$$\begin{split} \int_{\Omega_T} (u_4^\varepsilon - u_1^\varepsilon u_3^\varepsilon) \psi(t,x) \, dx dt &= \int_{\Omega_T} u_4^\varepsilon \partial_t \psi(t,x) \, dx dt + \int_{\Omega} u_4^{in}(x) \partial_t \psi(0,x) \, dx \\ &+ \int_{\Omega_T} d_4 u_4^\varepsilon \Delta_x \psi(t,x) \, dx dt. \end{split}$$

converges for $\varepsilon \to 0$ to

$$\int_{\Omega_T} (u_4 - u_1 u_3) \psi(t, x) \, dx dt = \int_{\Omega_T} u_4 \partial_t \psi(t, x) \, dx dt + \int_{\Omega} u_4^{in}(x) \partial_t \psi(0, x) \, dx$$
$$+ \int_{\Omega_T} d_4 u_4 \Delta_x \psi(t, x) \, dx dt.$$

Using Lemma 7.23 and Theorem 7.28, we get $u_2u_3 - u_1 = 0$.

This shows that the reaction-diffusion system (7.3.1) converges weakly to the reaction-cross-diffusion system (7.3.2), and since the duality estimate in Lemma 7.22 is uniform in ε , we get that the solution of the limiting cross-diffusion system (7.3.2) is bounded in $L^2_{loc}([0,\infty);L^2(\Omega))$.

Remark 7.30 (Initial layers). The last proof shows that the initial data needs to be well-prepared, i.e.

$$u_2^{in}u_3^{in}=u_1^{in}$$
 a.e. in Ω .

If this condition is not satisfied, there is an initial layer appearing. But due to the fact that

$$||u_2^{\varepsilon}u_3^{\varepsilon} - u_1^{\varepsilon}||_{L^2(0,T;L^2(\Omega))} \le \sqrt{\varepsilon}C,$$

this initial layer is vanishing after $o_{\varepsilon \to 0}(1)$ -time.

7.3.7. Global strong solutions to the reaction-diffusion system

Theorem 7.31. Let Ω be a bounded and regular $(C^{2+\alpha} \text{ or } C^{\infty})$ domain of \mathbb{R}^N for $N \leq 5$ and $d_i > 0$, $i = 1, \ldots, 4$. If the nonnegative initial data u_i^{in} , $i = 1, \ldots, 4$ are smooth $(C^2(\overline{\Omega}))$ and compatible with the Neumann boundary conditions, then for each fixed $\varepsilon > 0$ the solution of (7.3.1) is strong and unique.

Proof. We fix $\varepsilon > 0$ for the entire proof. The idea now is to use the bootstrap argument of [48, Proof of Theorem 3.1]. Due to the duality estimate of Lemma 7.22, we know that u_i^{ε} is bounded in $L^2([0,T] \times \Omega)$ for $i = 1, \ldots, 4$. Since

$$\partial_t u_2^{\varepsilon} - d_2 \Delta_x u_2^{\varepsilon} = \frac{1}{\varepsilon} u_1^{\varepsilon} - \frac{1}{\varepsilon} u_2^{\varepsilon} u_3^{\varepsilon} \le \frac{1}{\varepsilon} u_1^{\varepsilon} \text{ is bounded in } L^2([0,T] \times \Omega),$$

we can use the property of the heat kernel in Lemma 7.33 and get that

$$u_2^{\varepsilon} \in L_{loc}^{q_0}([0,\infty) \times \overline{\Omega}) \text{ for } \frac{1}{q_0} > \frac{N-2}{2(N+2)}.$$

The same argument holds for u_3^{ε} , namely since

$$\partial_t u_3^{\varepsilon} - d_3 \Delta_x u_3^{\varepsilon} \le \frac{1}{\varepsilon} u_1^{\varepsilon} + u_4^{\varepsilon}$$
 is bounded in $L^2([0,T] \times \Omega)$,

we get by the property of the heat kernel in Lemma 7.33 that

$$u_3^{\varepsilon} \in L_{loc}^{q_0}([0,\infty) \times \overline{\Omega}) \text{ for } \frac{1}{q_0} > \frac{N-2}{2(N+2)}.$$

Thus by Young's inequality with p = q = 2 and 1/p + 1/q = 1, we conclude that

$$u_2^{\varepsilon}u_3^{\varepsilon} \in L_{loc}^{p_0}([0,\infty) \times \overline{\Omega}) \text{ for } \frac{1}{p_0} > \frac{N-2}{N+2}.$$

Again by Young's inequality with $p = (q_0 + 2)/q_0$, $q = (q_0 + 2)/2$, where $\frac{1}{q_0} > \frac{N-2}{2(N+2)}$ and 1/p + 1/q = 1, we conclude that

$$u_1^{\varepsilon}u_3^{\varepsilon}\in L^{p_1}_{loc}([0,\infty)\times\overline{\Omega}) \text{ for } \frac{1}{p_1}>\frac{N}{N+2}.$$

Now we consider the reaction-diffusion equation satisfied for u_4^{ε} :

$$\partial_t u_4^{\varepsilon} - d_4 \Delta_x u_4^{\varepsilon} \leq \frac{1}{\varepsilon} u_1^{\varepsilon} u_3^{\varepsilon} \text{ bounded in } L^{p_1}([0,T] \times \Omega) \text{ for } \frac{1}{p_1} > \frac{N}{N+2},$$

and again by the property of the heat kernel, we get that

$$u_4^{\varepsilon} \in L_{loc}^{q_1}([0,\infty) \times \overline{\Omega}) \text{ for } \frac{1}{q_1} > \frac{N-2}{N+2},$$

and using the same arguments for u_1^{ε} leads to

$$u_1^{\varepsilon} \in L_{loc}^{q_1}([0,\infty) \times \overline{\Omega}) \text{ for } \frac{1}{q_1} > \frac{N-2}{N+2}.$$

For the remaining part of the proof we suppose that N=5, since the case N<5 can be handled with the same arguments and possibly less steps of bootstrapping. For N=5, we have showed so far that

$$u_2^\varepsilon, u_3^\varepsilon \in L^{q_0}_{loc}([0,\infty) \times \overline{\Omega}) \text{ for } \frac{1}{q_0} > \frac{3}{14}, \quad u_1^\varepsilon, u_4^\varepsilon \in L^{q_1}_{loc}([0,\infty) \times \overline{\Omega}) \text{ for } \frac{1}{q_1} > \frac{3}{7}.$$

Since

$$\partial_t u_i^{\varepsilon} - d_i \Delta_x u_i^{\varepsilon}$$
 is bounded in $L^{7/3}([0,T] \times \Omega)$ for $i=2,3,$

we get by the property of the heat kernel that $u_i \in L^{7-\delta}_{loc}([0,\infty),\overline{\Omega})$, for i=2,3, and a small $\delta > 0$. Thus by using Young's inequality with p=4, q=4/3 and 1/p+1/q=1, we get that

$$u_1^\varepsilon u_3^\varepsilon \in L^{7/4-\delta}([0,T]\times\Omega), \quad u_2^\varepsilon u_3^\varepsilon \in L^{7/4-\delta}([0,T]\times\Omega).$$

Thus

$$\partial_t u_i^{\varepsilon} - d_i \Delta_x u_i^{\varepsilon}$$
 is bounded in $L^{7/4-\delta}([0,T] \times \Omega)$, for $i = 1,4$,

and by the property of the heat kernel we get that

$$u_i^{\varepsilon} \in L_{loc}^{7/2-\delta}([0,\infty) \times \overline{\Omega}), \text{ for } i = 1,4,$$

showing that

$$\partial_t u_i^{\varepsilon} - d_i \Delta_x u_i^{\varepsilon}$$
 is bounded in $L^{7/2-\delta}([0,T] \times \Omega)$, for $i=2,3,$

and by the heat kernel we get that

$$u_i^{\varepsilon} \in L_{loc}^{q_2}([0,\infty) \times \overline{\Omega}), \quad q_2 \in [1,+\infty), \text{ for } i = 2,3.$$

Using Young's inequality leads to

$$u_1^{\varepsilon}u_3^{\varepsilon} \in L^{r_2}([0,T] \times \Omega) \text{ for } \frac{1}{r_2} > \frac{2}{7},$$

and together with the fact that

$$\partial_t u_i^{\varepsilon} - d_i \Delta_x u_i^{\varepsilon}$$
 is bounded in $L^{7/2-\delta}([0,T] \times \Omega)$, for $i = 1,4$

and the property of the kernel leads to

$$u_i^{\varepsilon} \in L_{loc}^{q_2}([0,\infty) \times \overline{\Omega}), \quad q_2 \in [1,+\infty) \text{ for } i = 1,4.$$

Finally, using the property of the heat kernel leads to

$$u_i^{\varepsilon} \in L_{loc}^{\infty}([0,\infty) \times \overline{\Omega}), \quad i = 1, 2, 3, 4.$$

Smoothness $(C^2([0,\infty]\times\overline{\Omega}))$ and uniqueness can be proved by using methods in [101].

Remark 7.32. The previous theorem only provides uniqueness for a given ε , and this does not imply any information about uniqueness of solutions to the limiting cross-diffusion system (7.3.2). However, uniqueness of weak solutions to (7.3.2) may be obtained under additional assumptions by using the technique of Gajewski [68].

7.3.8. Transformed system

Introducing the transformation φ , with $\varphi(u) = (\varphi_2(u), \varphi_3(u)), u = (u_2, u_3)$

$$\varphi: \begin{cases} (\mathbb{R}_{+})^{2} \to (\mathbb{R}_{+})^{2} \\ (u_{2}, u_{3}) \mapsto (w_{2}, w_{3}), \quad w = (w_{2}, w_{3}), \quad w_{2} := u_{2}(1 + u_{3}), \quad w_{3} := u_{3}(1 + u_{2}), \end{cases}$$

$$(7.3.14)$$

and its inverse

$$\psi := \varphi^{-1}, \quad \psi(w) = (\psi_2(w), \psi_3(w)), \tag{7.3.15}$$

we can rewrite the limiting system formally in terms of the new variables (w_2, w_3, u_4) as

$$\begin{cases} \partial_t w_2 - \Delta_x \left(\frac{d_1 \psi_3(w) + d_2}{\psi_3(w) + 1} w_2 \right) = \left(u_4 - \psi_2^2 \psi_3(w) \right), \\ \partial_t w_3 - \Delta_x \left(\frac{d_1 \psi_2(w) + d_3}{\psi_2(w) + 1} w_3 \right) = 2 \left(u_4 - \psi_2^2 \psi_3(w) \right), \\ \partial_t u_4 - d_4 \Delta_x u_4 = - \left(u_4 - \psi_2^2 \psi_3(w) \right). \end{cases}$$

The entropy can be expressed in the new variables $w = (w_2, w_3, u_4)$ in the following way

$$H[w_{2}(t), w_{3}(t), u_{4}(t)] = \int_{\Omega} (\psi_{2}(w)\psi_{3}(w)(\log(\psi_{2}(w)\psi_{3}(w)) - 1) + 1) dx$$

$$+ \int_{\Omega} (\psi_{2}(w)(\log(\psi_{2}(w)) - 1) + 1) dx$$

$$+ \int_{\Omega} (\psi_{3}(w)(\log(\psi_{3}(w)) - 1) + 1) dx$$

$$+ \int_{\Omega} (u_{4}(\log(u_{4}) - 1) + 1) dx.$$

7.4. Auxiliary result: regularizing effect of the heat kernel

The following result [48, p.5] concerning the regularizing effect of the heat kernel has been used in this chapter.

Lemma 7.33. If $f \in L^p([0,T] \times \Omega)$, then the solution u := u(t,x) of

$$\begin{cases} \partial_t u - \Delta_x u = f, & \text{for } t \in [0, T] \text{ and } x \in \Omega, \\ n \cdot \nabla_x u = 0, & \text{for } t \in [0, T] \text{ and } x \in \partial\Omega, \\ u(0, \cdot) \in L^{\infty}(\Omega), \end{cases}$$

lies in $L^q([0,T]\times\Omega)$ for all $q\in[1,+\infty]$ satisfying

$$\frac{1}{p} + \frac{N}{N+2} - 1 < \frac{1}{q} \quad \Rightarrow \quad \frac{1}{q} > \frac{1}{p} - \frac{2}{N+2}.$$

7. From reaction diffusion to cross diffusion in the fast-reaction limit

Bibliography

- [1] R. Alexandre, F. Hérau, and Wei-Xi Li. Global hypoelliptic and symbolic estimates for the linearized Boltzmann operator without angular cutoff. arXiv:1212.4632v2, 2014.
- [2] H. Amann. Dynamic theory of quasilinear parabolic systems. III. Global existence. *Math. Z.*, 202(2):219–250, 1989.
- [3] R. Anguelov and H. M. Tenkam. Lyapunov functional for a class of multi-species models with cross diffusion. *Biomath Communications*, 1(1), 2014.
- [4] C. Baranger and C. Mouhot. Explicit spectral gap estimates for the linearized Boltzmann and Landau operators with hard potentials. *Rev. Mat. Iberoamericana*, 21(3):819–841, 2005.
- [5] M. Bisi, F. Conforto, and L. Desvillettes. Quasi-steady-state approximation for reaction-diffusion equations. *Bull. Inst. Math. Acad. Sin. (N.S.)*, 2(4):823–850, 2007.
- [6] A. V. Bobylëv. The method of the Fourier transform in the theory of the Boltzmann equation for Maxwell molecules. *Dokl. Akad. Nauk SSSR*, 225(6):1041–1044, 1975.
- [7] A. V. Bobylëv. The theory of the nonlinear spatially uniform Boltzmann equation for Maxwell molecules. In *Mathematical physics reviews, Vol. 7*, volume 7 of *Soviet Sci. Rev. Sect. C Math. Phys. Rev.*, pages 111–233. Harwood Academic Publ., Chur, 1988.
- [8] A.V. Bobylev. Moment inequalities for the boltzmann equation and applications to spatially homogeneous problems. *Journal of Statistical Physics*, 88(5-6):1183–1214, 1997.
- [9] A.V. Bobylev, I.M. Gamba, and V.A. Panferov. Moment Inequalities and High-Energy Tails for Boltzmann Equations with Inelastic Interactions. *Journal of Statistical Physics*, 116(5-6):1651–1682, 2004.
- [10] D. Bothe. Nonlinear Evolutions in Banach Spaces. Existence and Qualitative Theory with Applications to Reaction-Diffusion Systems, 1999. Habilitation thesis.
- [11] D. Bothe. Instantaneous limits of reversible chemical reactions in presence of macroscopic convection. J. Differential Equations, 193(1):27–48, 2003.
- [12] D. Bothe. On the Maxwell-Stefan approach to multicomponent diffusion. In *Parabolic problems*, volume 80 of *Progr. Nonlinear Differential Equations Appl.*, pages 81–93. Birkhäuser/Springer Basel AG, Basel, 2011.

- [13] D. Bothe and D. Hilhorst. A reaction-diffusion system with fast reversible reaction. J. Math. Anal. Appl., 286(1):125–135, 2003.
- [14] D. Bothe and M. Pierre. Quasi-steady-state approximation for a reaction-diffusion system with fast intermediate. *J. Math. Anal. Appl.*, 368(1):120–132, 2010.
- [15] D. Bothe and M. Pierre. The instantaneous limit for reaction-diffusion systems with a fast irreversible reaction. *Discrete Contin. Dyn. Syst. Ser. S*, 5(1):49–59, 2012.
- [16] D. Bothe, M. Pierre, and G. Rolland. Cross-diffusion limit for a reaction-diffusion system with fast reversible reaction. Comm. Partial Differential Equations, 37(11):1940–1966, 2012.
- [17] D. Bothe and G. Rolland. Global existence for a class of reaction-diffusion systems with mass action kinetics and concentration-dependent diffusivities. *ArXiv e-prints*, February 2014.
- [18] L. Boudin, B. Grec, and V. Pavan. The Maxwell-Stefan diffusion limit for a kinetic model of mixtures with general cross sections. Working paper or preprint, April 2016.
- [19] L. Boudin, B. Grec, M. Pavić, and F. Salvarani. Diffusion asymptotics of a kinetic model for gaseous mixtures. *Kinetic and Related Models*, 6(1):137–157, 2013.
- [20] L. Boudin, B. Grec, and F. Salvarani. A mathematical and numerical analysis of the Maxwell-Stefan diffusion equations. *Discrete Contin. Dyn. Syst. Ser. B*, 17(5):1427– 1440, 2012.
- [21] L. Boudin, B. Grec, and F. Salvarani. The Maxwell-Stefan Diffusion Limit for a Kinetic Model of Mixtures. *Acta Applicandae Mathematicae*, 136(1):79–90, 2015.
- [22] L. Boudin and F. Salvarani. Compactness of linearized kinetic operators. working paper or preprint, November 2015.
- [23] J.-F. Bourgat, L. Desvillettes, P. Le Tallec, and B. Perthame. Microreversible collisions for polyatomic gases and Boltzmann's theorem. European J. Mech. B Fluids, 13(2):237–254, 1994.
- [24] H. Brezis. Functional analysis, Sobolev spaces and partial differential equations. Universitext. Springer, New York, 2011.
- [25] M. Briant. From the Boltzmann equation to the incompressible Navier–Stokes equations on the torus: A quantitative error estimate. *J. Differential Equations*, 259(11):6072–6141, 2015.
- [26] M. Briant. Perturbative theory for the Boltzmann equation in bounded domains with different boundary conditions. *ArXiv e-prints*, July 2015.
- [27] M. Briant. Stability of global equilibrium for the multi-species Boltzmann equation in L^{∞} settings. ArXiv e-prints, March 2016.

- [28] M. Briant and E. S. Daus. The Boltzmann equation for a multi-species mixture close to global equilibrium. *Arch. Ration. Mech. Anal.*, 222(3):1367–1443, 2016.
- [29] J. A. Cañizo, L. Desvillettes, and K. Fellner. Improved duality estimates and applications to reaction-diffusion equations. Comm. Partial Differential Equations, 39(6):1185–1204, 2014.
- [30] T. Carleman. Problèmes mathématiques dans la théorie cinétique des gaz. Publ. Sci. Inst. Mittag-Leffler. 2. Almqvist & Wiksells Boktryckeri Ab, Uppsala, 1957.
- [31] J. A. Carrillo, L. Desvillettes, and K. Fellner. Rigorous derivation of a nonlinear diffusion equation as fast-reaction limit of a continuous coagulation-fragmentation model with diffusion. *Comm. Partial Differential Equations*, 34(10-12):1338–1351, 2009.
- [32] C. Cercignani. Mathematical methods in kinetic theory. Plenum Press, New York, 1969.
- [33] C. Cercignani. The Boltzmann equation and its applications, volume 67 of Applied Mathematical Sciences. Springer-Verlag, New York, 1988.
- [34] C. Cercignani, R. Illner, and M. Pulvirenti. The mathematical theory of dilute gases, volume 106 of Applied Mathematical Sciences. Springer-Verlag, New York, 1994.
- [35] C. S. W. Chang, J. de Boer, G. W. Ford, and G. E. Uhlenbeck. *Studies in statistical mechanics*. Number v. 5 in Studies in statistical mechanics. North-Holland, 1970.
- [36] F. Charles and L. Desvillettes. Small mass ratio limit of Boltzmann equations in the context of the study of evolution of dust particles in a rarefied atmosphere. *J. Stat. Phys.*, 137(3):539–567, 2009.
- [37] L. Chen and A. Jüngel. Analysis of a multidimensional parabolic population model with strong cross-diffusion. SIAM J. Math. Anal., 36(1):301–322 (electronic), 2004.
- [38] L. Chen and A. Jüngel. Analysis of a parabolic cross-diffusion population model without self-diffusion. *J. Differential Equations*, 224(1):39–59, 2006.
- [39] X. Chen, E. S. Daus, and A. Jüngel. Global existence analysis of cross-diffusion population systems for multiple species. *ArXiv e-prints*, August 2016.
- [40] X. Chen, A. Jüngel, and J.-G. Liu. A note on Aubin-Lions-Dubinskiĭ lemmas. *Acta Appl. Math.*, 133:33–43, 2014.
- [41] X. Chen and J.-G. Liu. Analysis of polymeric flow models and related compactness theorems in weighted spaces. SIAM J. Math. Anal., 45(3):1179–1215, 2013.
- [42] D. S. Clark. Short Proof of a Discrete Gronwall Inequality. *Discrete Appl. Math.*, 16(3):279–281, March 1987.

- [43] F. Conforto and L. Desvillettes. About the Quasi Steady State Approximation for a Reaction Diffusion System Describing a Chain of Irreversible Chemical Reactions. *Journal of Physics: Conference Series*, 482(1):012008, 2014.
- [44] F. Conforto and L. Desvillettes. Rigorous passage to the limit in a system of reaction-diffusion equations towards a system including cross diffusions. *Commun. Math. Sci.*, 12(3):457–472, 2014.
- [45] F. Conforto, L. Desvillettes, and R. Monaco. Some asymptotic limits of reaction—diffusion systems appearing in reversible chemistry. *Ricerche di Matematica*, pages 1–13, 2016.
- [46] E. S. Daus, A. Jüngel, C. Mouhot, and N. Zamponi. Hypocoercivity for a linearized multispecies Boltzmann system. SIAM J. Math. Anal., 48(1):538–568, 2016.
- [47] S. R. de Groot and P. Mazur. *Nonequilibrium thermodynamics*. Dover Publications, Inc., New York, 1984. Reprint of the 1962 original.
- [48] L. Desvillettes and K. Fellner. Duality and entropy methods for reversible reaction-diffusion equations with degenerate diffusion. *Math. Methods Appl. Sci.*, 38(16):3432–3443, 2015.
- [49] L. Desvillettes, K. Fellner, M. Pierre, and J. Vovelle. Global existence for quadratic systems of reaction-diffusion. *Adv. Nonlinear Stud.*, 7(3):491–511, 2007.
- [50] L. Desvillettes, T. Lepoutre, and A. Moussa. Entropy, duality, and cross diffusion. SIAM J. Math. Anal., 46(1):820–853, 2014.
- [51] L. Desvillettes, T. Lepoutre, A. Moussa, and A. Trescases. On the entropic structure of reaction-cross diffusion systems. *Comm. Partial Differential Equations*, 40(9):1705–1747, 2015.
- [52] L. Desvillettes, R. Monaco, and F. Salvarani. A kinetic model allowing to obtain the energy law of polytropic gases in the presence of chemical reactions. *Eur. J. Mech. B Fluids*, 24(2):219–236, 2005.
- [53] L. Desvillettes and A. Trescases. New results for triangular reaction cross diffusion system. J. Math. Anal. Appl., 430(1):32–59, 2015.
- [54] L. Desvillettes and C. Villani. On the trend to global equilibrium in spatially inhomogeneous entropy-dissipating systems: the linear Fokker-Planck equation. *Comm. Pure Appl. Math.*, 54(1):1–42, 2001.
- [55] L. Desvillettes and C. Villani. On the trend to global equilibrium for spatially inhomogeneous kinetic systems: the Boltzmann equation. *Invent. Math.*, 159(2):245–316, 2005.
- [56] P. Deuring. An initial-boundary-value problem for a certain density-dependent diffusion system. *Mathematische Zeitschrift*, 194(3):375–396, 1987.

- [57] J. Dolbeault, F. Hérau, C. Mouhot, and C. Schmeiser. Work in progress, 2016.
- [58] J. Dolbeault, C. Mouhot, and C. Schmeiser. Hypocoercivity for linear kinetic equations conserving mass. *Trans. Amer. Math. Soc.*, 367(6):3807–3828, 2015.
- [59] J. P. Dougherty. Lectures on Gas Theory. By Ludwig Boltzmann. Translated by Stephen G. Brush. *Journal of Fluid Mechanics*, 20:698–699, 1964.
- [60] M. Dreher. Analysis of a population model with strong cross-diffusion in unbounded domains. *Proc. Roy. Soc. Edinburgh Sect. A*, 138(4):769–786, 2008.
- [61] M. Dreher and A. Jüngel. Compact families of piecewise constant functions in $L^p(0,T;B)$. Nonlinear Anal., 75(6):3072–3077, 2012.
- [62] R. Duan. Hypocoercivity of linear degenerately dissipative kinetic equations. Non-linearity, 24(8):2165–2189, 2011.
- [63] M. Dudyński. Spectral properties of the linearized Boltzmann operator in L^p for $1 \le p \le \infty$. J. Stat. Phys., 153(6):1084–1106, 2013.
- [64] J. B. Duncan and H. L. Toor. An experimental study of three component gas diffusion. *AIChE Journal*, 8:38–41, 1962.
- [65] P. J. Dunlop and L. J. Gosting. Interacting flows in liquid diffusion: Expressions for the solute concentration curves in free diffusion, and their use in interpreting Gouy Diffusiometer data for aqueous three-component systems. J. Am. Chem. Soc., 77:5238–5249, 1955.
- [66] R. Esposito, Y. Guo, C. Kim, and R. Marra. Non-isothermal boundary in the Boltzmann theory and Fourier law. *Comm. Math. Phys.*, 323(1):177–239, 2013.
- [67] L. C. Evans. *Partial differential equations*. Graduate studies in mathematics. American Mathematical Society, Providence (R.I.), 1998.
- [68] H. Gajewski. On the uniqueness of solutions to the drift-diffusion model of semiconductor devices. *Math. Models Methods Appl. Sci.*, 4(1):121–133, 1994.
- [69] G. Galiano, M. L. Garzón, and A. Jüngel. Semi-discretization in time and numerical convergence of solutions of a nonlinear cross-diffusion population model. *Numer. Math.*, 93(4):655–673, 2003.
- [70] I. Gallagher, L. Saint-Raymond, and B. Texier. From Newton to Boltzmann: hard spheres and short-range potentials. Zurich Lectures in Advanced Mathematics. European Mathematical Society (EMS), Zürich, 2013.
- [71] I. M. Gamba, V. Panferov, and C. Villani. Upper Maxwellian bounds for the spatially homogeneous Boltzmann equation. *Arch. Ration. Mech. Anal.*, 194(1):253–282, 2009.
- [72] V. Giovangigli. *Multicomponent flow modeling*. Modeling and Simulation in Science, Engineering and Technology. Birkhäuser Boston, Inc., Boston, MA, 1999.

- [73] F. Golse and F. Poupaud. Un résultat de compacité pour l'équation de Boltzmann avec potentiel mou. Application au problème de demi-espace. C. R. Acad. Sci. Paris Sér. I Math., 303(12):583–586, 1986.
- [74] H. Grad. Principles of the kinetic theory of gases. In *Handbuch der Physik (herausgegeben von S. Flügge)*, Bd. 12, Thermodynamik der Gase, pages 205–294. Springer-Verlag, Berlin, 1958.
- [75] H. Grad. Asymptotic theory of the Boltzmann equation. II. In Rarefied Gas Dynamics (Proc. 3rd Internat. Sympos., Palais de l'UNESCO, Paris, 1962), Vol. I, pages 26–59. Academic Press, New York, 1963.
- [76] H. Grad. Asymptotic equivalence of the Navier-Stokes and nonlinear Boltzmann equations. In *Proc. Sympos. Appl. Math.*, Vol. XVII, pages 154–183. Amer. Math. Soc., Providence, R.I., 1965.
- [77] M. Gualdani and N. Zamponi. Spectral gap and exponential convergence to equilibrium for a multi-species Landau system. *ArXiv e-prints*, February 2016.
- [78] M. P. Gualdani, S. Mischler, and C. Mouhot. Factorization for non-symmetric operators and exponential H-theorem. Preprint 2013.
- [79] R. Guillaume. Global Existence and Fast-Reaction Limit in Reaction-Diffusion Systems with Cross Effects. PhD thesis, Technische Universität, Darmstadt, October 2012.
- [80] Y. Guo. The Vlasov-Poisson-Boltzmann system near Maxwellians. Comm. Pure Appl. Math., 55(9):1104–1135, 2002.
- [81] Y. Guo. Classical solutions to the Boltzmann equation for molecules with an angular cutoff. Arch. Ration. Mech. Anal., 169(4):305–353, 2003.
- [82] Y. Guo. Boltzmann diffusive limit beyond the Navier-Stokes approximation. *Comm. Pure Appl. Math.*, 59(5):626–687, 2006.
- [83] Y. Guo. Decay and continuity of the Boltzmann equation in bounded domains. *Arch. Ration. Mech. Anal.*, 197(3):713–809, 2010.
- [84] K. Gustafson and J. Weidmann. On the essential spectrum. J. Math. Anal. Appl., 25:121–127, 1969.
- [85] F. Hérau. Hypocoercivity and exponential time decay for the linear inhomogeneous relaxation Boltzmann equation. *Asymptot. Anal.*, 46(3-4):349–359, 2006.
- [86] F. Hérau and F. Nier. Isotropic hypoellipticity and trend to equilibrium for the Fokker-Planck equation with a high-degree potential. *Arch. Ration. Mech. Anal.*, 171(2):151–218, 2004.
- [87] D. Hilbert. Begründung der kinetischen Gastheorie. Math. Ann., 72(4):562–577, 1912.

- [88] D. Hilhorst, M. Mimura, and H. Ninomiya. Fast reaction limit of competitiondiffusion systems. In *Handbook of differential equations: evolutionary equations. Vol.* V, Handb. Differ. Equ., pages 105–168. Elsevier/North-Holland, Amsterdam, 2009.
- [89] D. Hilhorst, R. van der Hout, and L. A. Peletier. The fast reaction limit for a reaction-diffusion system. J. Math. Anal. Appl., 199(2):349–373, 1996.
- [90] D. Hilhorst, R. van der Hout, and L. A. Peletier. Nonlinear diffusion in the presence of fast reaction. *Nonlinear Anal.*, 41(5-6, Ser. A: Theory Methods):803–823, 2000.
- [91] H. Hutridurga and F. Salvarani. On the Maxwell-Stefan diffusion limit for a mixture of monatomic gases. *ArXiv e-prints*, September 2015.
- [92] M. Iida, M. Mimura, and H. Ninomiya. Diffusion, cross-diffusion and competitive interaction. J. Math. Biol., 53(4):617–641, 2006.
- [93] H. Izuhara and M. Mimura. Reaction-diffusion system approximation to the cross-diffusion competition system. *Hiroshima Math. J.*, 38(2):315–347, 2008.
- [94] A. Jüngel. The boundedness-by-entropy method for cross-diffusion systems. *Nonlinearity*, 28(6):1963–2001, 2015.
- [95] A. Jüngel. Entropy Methods for Diffusive Partial Differential Equations. Springer Briefs in Mathematics. Springer, 2016.
- [96] A. Jüngel and I. V. Stelzer. Existence analysis of Maxwell-Stefan systems for multi-component mixtures. SIAM J. Math. Anal., 45(4):2421–2440, 2013.
- [97] T. Kato. *Perturbation theory for linear operators*. Springer-Verlag, Berlin-New York, second edition, 1976. Grundlehren der Mathematischen Wissenschaften, Band 132.
- [98] T. Kato. *Perturbation theory for linear operators*. Classics in Mathematics. Springer-Verlag, Berlin, 1995. Reprint of the 1980 edition.
- [99] J. Kim. Smooth solutions to a quasi-linear system of diffusion equations for a certain population model. *Nonlinear Anal.*, 8:1121–1144, 1984.
- [100] M. Klaus. The linear Boltzmann operator-spectral properties and short-wavelength limit. *Helv. Phys. Acta*, 48:99–129, 1975.
- [101] O. A. Ladyženskaja, V. A. Solonnikov, and N. N. Uralceva. Linear and quasilinear equations of parabolic type. Translated from the Russian by S. Smith. Translations of Mathematical Monographs, Vol. 23. American Mathematical Society, Providence, R.I., 1968.
- [102] O. E. Lanford III. Time evolution of large classical systems. In *Dynamical systems*, theory and applications (Recontres, Battelle Res. Inst., Seattle, Wash., 1974), pages 1–111. Lecture Notes in Phys., Vol. 38. Springer, Berlin, 1975.
- [103] D. Le. Cross diffusion systems in n spatial dimensional domains. *Indiana Univ. Math.* J., 51:625–643, 2002.

- [104] D. Le. Global existence for a class of strongly coupled parabolic systems. Ann. Mat. Pura Appl. (4), 185(1):133–154, 2006.
- [105] J.-L. Lions. Quelques méthodes de résolution des problèmes aux limites non linéaires. Dunod; Gauthier-Villars, Paris, 1969.
- [106] Y. Lou, W.-M. Ni, and Y. Wu. On the global existence of a cross-diffusion system. Discrete Contin. Dynam. Systems, 4(2):193–203, 1998.
- [107] J. C. Maxwell. On the Dynamical Theory of Gases. *Philosophical Transactions of the Royal Society of London*, 157:49–88, 1867.
- [108] J. C. Maxwell. On Stresses in Rarified Gases Arising from Inequalities of Temperature. *Philosophical Transactions of the Royal Society of London*, 170:231–256, 1879.
- [109] S. Mischler and B. Wennberg. On the spatially homogeneous Boltzmann equation. Ann. Inst. H. Poincaré Anal. Non Linéaire, 16(4):467–501, 1999.
- [110] C. Mouhot. Mathematical study of some collisional kinetic equations. Theses, École normale supérieure de Lyon ENS Lyon, November 2004.
- [111] C. Mouhot. Explicit coercivity estimates for the linearized Boltzmann and Landau operators. Comm. Partial Differential Equations, 31(7-9):1321–1348, 2006.
- [112] C. Mouhot. Rate of convergence to equilibrium for the spatially homogeneous Boltzmann equation with hard potentials. *Comm. Math. Phys.*, 261(3):629–672, 2006.
- [113] C. Mouhot and L. Neumann. Quantitative perturbative study of convergence to equilibrium for collisional kinetic models in the torus. *Nonlinearity*, 19(4):969–998, 2006.
- [114] C. Mouhot and R. M. Strain. Spectral gap and coercivity estimates for linearized Boltzmann collision operators without angular cutoff. *J. Math. Pures Appl.* (9), 87(5):515–535, 2007.
- [115] H. Murakawa. A relation between cross-diffusion and reaction-diffusion. *Discrete Contin. Dyn. Syst. Ser. S*, 5(1):147–158, 2012.
- [116] L. Neumann and C. Schmeiser. A kinetic reaction model: decay to equilibrium and macroscopic limit. arXiv, (1503.05745v1 [math.AP]).
- [117] L. Onsager and R. M. Fuoss. Irreversible processes in electrolytes. Diffusion, conductance, and viscous flow in arbitrary mixtures of strong electrolytes. *Journal of Physical Chemistry*, 35:2689–2778, 1932.
- [118] Y. P. Pao. Boltzmann collision operator with inverse-power intermolecular potentials. I, II. Comm. Pure Appl. Math., 27:407–428; ibid. 27 (1974), 559–581, 1974.
- [119] R. Pettersson. On weak and strong convergence to equilibrium for solutions to the linear Boltzmann equation. J. Statist. Phys., 72(1-2):355–380, 1993.

- [120] M. Pierre. Global existence in reaction-diffusion systems with control of mass: a survey. *Milan J. Math.*, 78(2):417–455, 2010.
- [121] M. Pierre and D. Schmitt. Blowup in reaction-diffusion systems with dissipation of mass. SIAM J. Math. Anal., 28(2):259–269, 1997.
- [122] L. S. Polak and A. V. Khachoyan. Generalization of Boltzmann's H-theorem for a reacting gas mixture. *Sovjet J. Chem. Phys.*, 2:1474–1485, 1985.
- [123] J. Polewczak. The kinetic theory of simple reacting spheres. I. Global existence result in a dilute-gas case. *J. Statist. Phys.*, 100(1-2):327–362, 2000. Special issue dedicated to George Stell.
- [124] A. J. Povzner. On the Boltzmann equation in the kinetic theory of gases. *Mat. Sb.* (N.S.), 58 (100):65–86, 1962.
- [125] M. Pulvirenti, C. Saffirio, and S. Simonella. On the validity of the Boltzmann equation for short range potentials. *Rev. Math. Phys.*, 26(2):1450001, 64, 2014.
- [126] T. Roubíček. Nonlinear partial differential equations with applications, volume 153 of International Series of Numerical Mathematics. Birkhäuser Verlag, Basel, 2005.
- [127] K. Ryu and I. Ahn. Coexistence states of certain population models with nonlinear diffusions among multi-species. *Dyn. Contin. Discrete Impuls. Syst. Ser. A Math. Anal.*, 12(2):235–246, 2005.
- [128] M. Schechter. On the essential spectrum of an arbitrary operator. I. J. Math. Anal. Appl., 13:205–215, 1966.
- [129] N. Shigesada, K. Kawasaki, and E. Teramoto. Spatial segregation of interacting species. J. Theoret. Biol., 79(1):83–99, 1979.
- [130] J. Simon. Compact sets in the space $L^p(0,T;B)$. Ann. Mat. Pura Appl. (4), 146:65–96, 1987.
- [131] P. Suomela. Invariant measures of time-reversible Markov chains. *J. Appl. Probab.*, 16(1):226–229, 1979.
- [132] R. Temam. Infinite-dimensional dynamical systems in mechanics and physics, volume 68 of Applied Mathematical Sciences. Springer-Verlag, New York, second edition, 1997.
- [133] G. Teschl. Mathematical methods in quantum mechanics, volume 99 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, 2009. With applications to Schrödinger operators.
- [134] S. Ukai. On the existence of global solutions of mixed problem for non-linear Boltzmann equation. *Proc. Japan Acad.*, 50:179–184, 1974.

- [135] S. Ukai and T. Yang. Mathematical theory of the Boltzmann equation. Lecture Notes Series, no. 8, Liu Bie Ju Centre for Mathematical Sciences, City University of Hong Kong, 2006.
- [136] V. K. Vanag and I. R. Epstein. Cross-diffusion and pattern formation in reactiondiffusion systems. Phys. Chem. Chem. Phys., 11:897–912, 2009.
- [137] I. Vidav. Spectra of perturbed semigroups with applications to transport theory. *J. Math. Anal. Appl.*, 30:264–279, 1970.
- [138] C. Villani. A review of mathematical topics in collisional kinetic theory. In *Handbook* of mathematical fluid dynamics, Vol. I, pages 71–305. North-Holland, Amsterdam, 2002.
- [139] C. Villani. Hypocoercivity. Mem. Amer. Math. Soc., 202(950):iv+141, 2009.
- [140] Z. Wen and S. Fu. Global solutions to a class of multi-species reaction-diffusion systems with cross-diffusions arising in population dynamics. *J. Comput. Appl. Math.*, 230(1):34–43, 2009.
- [141] A. Yagi. Global solution to some quasilinear parabolic system in population dynamics. *Nonlinear Anal.*, 21(8):603–630, 1993.
- [142] K. Yosida. Functional analysis. Springer-Verlag, New York-Heidelberg, fourth edition, 1974. Die Grundlehren der mathematischen Wissenschaften, Band 123.
- [143] H. Yu. Global classical solutions of the Boltzmann equation near Maxwellians. *Acta Math. Sci. Ser. B Engl. Ed.*, 26(3):491–501, 2006.
- [144] N. Zamponi and A. Jüngel. Analysis of degenerate cross-diffusion population models with volume filling. *ArXiv e-prints*, February 2015.

Acknowledgements

First of all, I want to thank my supervisor, Prof. Ansgar Jüngel, for his outstanding professional support throughout the entire time of my PhD. I deeply appreciate the freedom he gave me to develop my own ideas and the enormous support he provided me to achieve these goals, as well as his great contributions to our joint projects. Moreover, I wish to thank him, as the Speaker of the Doctoral School "Dissipation and Dispersion in Nonlinear Partial Differential Equations" financed by the Austrian National Science Fund (FWF), for supporting me to work for several months in the United Kingdom (University of Cambridge, University of Oxford) and in France (Paris Université Pierre et Marie Curie, Paris Université Diderot) during the time of my PhD. These research stays have certainly had an immense impact on my mindset, first by improving my mathematical understanding and technical skills, but also by helping me to find new friends in other countries, broadening my horizon, and providing me a deeper understanding of these countries. Thank you very much for making all this possible!

I would like to express my deep gratitude to Prof. Laurent Desvillettes for inviting me to Paris and for working with me on our joint research project on cross diffusion. I thank him for his constant effort in deepening my mathematical understanding and for sharing his brilliant mathematical ideas with me, as well as for the many interesting conversations and mathematical discussions we had. Thank you very much, all this was enormously helpful! I also want to thank Dr Ayman Moussa for both his constant mathematical help and his great hospitality. I thank Dr Laurent Boudin and Dr Bérénice Grec for working with me and for helping me to get along in Paris. Of course, I wish to thank my great collaborator, Dr Marc Briant, for working with me, first via email while he was in the United States, and then also directly in Paris, thank you very much for this great collaboration, it was amazing to work on our project! Many thanks, Nina, for sharing your office with me, spending time there was a lot of fun! Finally, I want to thank all the people I met at the Laboratoire Jacques-Louis Lions at the Université Pierre et Marie Curie, as well as those from the Mathematical Institute at the Université Paris Diderot.

Furthermore, I deeply thank Dr Maria Bruna for inviting me to the University of Oxford, it was a great time for me! Thank you very much for taking care of me and for giving me all the good ideas and helpful advice.

I honestly want to thank Prof. Clément Mouhot for inviting me to the Unversity of Cambridge and for working with me on our joint research project in kinetic theory. Thank you very much for the very interesting mathematical discussions and for sharing some intuition with me, which turned out to be so helpful! Moreover, I thank the members of the Cambridge Kinetic Group for our 11 o'clock tea meetings, and for the punting afternoon at the river Cam, thank you, Helge, for organizing it! Moreover, I want to thank Franca

for hosting me, and Sara for inviting me to participate in her video about her research in Cambridge, that was fun! Thanks, Adam and Harold, for sharing your office with me!

Moreover, I want to thank all the members of my Doctoral School for the great time at the Winter and Summer Schools, as well as for the interesting mathematical discussions in our weekly seminar.

Next, I want to acknowledge all the members of my research group at TU Wien and also all the other members of the Institute for Analysis and Scientific Computing for the nice working environment! I thank Dr Nicola Zamponi and Prof. Xiuqing Chen for our great collaboration, it was very nice to work with you. Moreover, thanks to Oliver, Karli, Lara and Stefan for sharing an office with me. I also want to thank Ms. Khaladj for supporting me in the administrative work.

I thank my friends for all the nice time we spent together and the support they provided me in many situation of my life. Many thanks to Ali for his valuable help.

Finally, I want to thank all the members of my family for their advice and support, which helped me to find my way. In particular thank you, Erika, for taking care of me in many situations of my life. I deeply wish to thank Junjian for his constant support, his optimism, his never ending sense of humour, and his deep understanding.

Curriculum vitae

Personal data

Name Esther Sarah Daus

Date of Birth December 23, 1986 in Homburg/Saar, Germany Address Institute for Analysis and Scientific Computing,

Vienna University of Technology, Wiedner Hauptstr. 8-10, 1040 Wien

Website https://esdaus.wordpress.com

Education

11/2013–current	PhD, Mathematics in Technology, TU Wien, Member of the Doctoral
	School "Dissipation and Dispersion in nonlinear PDEs" (FWF).
02/2012 – 09/2013	MSc, Mathematics in Science and Technology, TU Wien, with distinction.
10/2008 – 02/2012	BSc, Mathematics in Science and Technology, TU Wien, with distinction.
10/2005 – 08/2009	BSc, Hungarology, Uni Wien, with distinction.
09/2001 - 06/2004	C-Diploma, Organ music, DKK Conservatory Wien, with distinction.

Scholarships

2013	Merit-based scholarship, TU Wien.
2012	Merit-based scholarship, TU Wien.
2008	Merit-based scholarship, Uni Wien.

09/2006-06/2007 Erasmus outgoing scholarship, ELTE University Budapest.

Research interests

Nonlinear partial differential equations

Kinetic equations

Entropy methods in dissipative systems

Cross-diffusion systems in thermodynamics, physics, and biology

Publications/Preprints/Working papers

- E. S. Daus, L. Desvillettes, A. Jüngel. Cross-diffusion systems and fast-reaction limit. In preparation, will be submitted soon.
- X. Chen, E. S. Daus, A. Jüngel. Global existence analysis of cross-diffusion population systems for multiple species. Submitted for publication in August 2016.
- M. Briant, E. S. Daus. The Boltzmann equation for a multi-species mixture close to global equilibrium. To appear in Arch. Ration. Mech. Anal., 222(3): 1367-1443, 2016.
- E. S. Daus, A. Jüngel, N. Zamponi, C. Mouhot. *Hypocoercivity for a linearized multispecies Boltzmann system*. SIAM J. Math. Anal., 48(1): 538-568, 2016.

Research stays

03/2016 – 06/2016	Paris Université $6/7$, France, Invited by L. Desvillettes.
10/2015 – 12/2015	Paris Université 6/7, France, Invited by L. Desvillettes.
01/2015 – 03/2015	University of Oxford, UK, Invited by M. Bruna.
06/2014 – 07/2014	University of Cambridge, UK, Invited by C. Mouhot.

Scientific talks

05/2016	Mathematical Topics in Kinetic Theory, Cambridge, UK.
11/2015	Seminar of the Doctoral School LJLL, Paris Université 6, France.
09/2015	Multiscale Transport of Particles, WPI Vienna, Austria.
07/2015	Kinetic Equations and Defect Dynamics, HCB Bonn, Germany.
07/2015	Conference Equadiff 2015, Lyon, France.
05/2015	Mathematical Problems in Kinetic Theory, CHL Rennes, France.
02/2015	SIAM Student Chapter Conference, Oxford, UK.
02/2015	WCMB Group Meeting, Oxford, UK.

Teaching experience

2011-2012	Exercise classes for Computer Science students, TU Wien.
2010-2011	Exercise classes for Computer Science students, TU Wien.